Travaux, Incorporated
Resolution No.:
Adopted:

RESOLUTION: APPROVING A CONTRACT WITH HIGHLAND

GARDENS LLC TO BE THE CONSTRUCTION CONTRACTOR FOR THE REHABILITATION OF HIGHLAND GARDENS PROJECT IN AN AMOUNT

NOT-TO-EXCEED \$8,838,593

WHEREAS, the Housing Authority of the City of Milwaukee ("**HACM**") received an allocation of Low Income Housing Tax Credits from the Wisconsin Housing and Economic Development Authority ("**WHEDA**") for the rehabilitation of 114 rental housing units at Highland Gardens (the "**Project**"); and

WHEREAS, HACM incorporated Highland Gardens LLC (the "**Project Owner**") to be the owner entity of Project; and

WHEREAS, HACM is the developer and the Managing Member of the Project Owner through its wholly owned Limited Liability Company, Highland Gardens Development LLC; and

WHEREAS, WHEDA, as lender to the Project, dictates the Contract Form or Document, currently AIA Document A102-2017 titled Standard Form of Agreement Between Owner and Contractor where the basis of payment is the Cost of the Work Plus a Fee with a Guaranteed Maximum Price to be used as contract document; and

WHEREAS, the President recommends that Travaux, Inc. enter into a contract with the Project Owner; now, therefore, be it

RESOLVED, by Travaux, Inc. that they approve entering into a contract with the Project Owner the to be the construction manager as constructor for the Project; and, be it

FURTHER RESOLVED, that the proper officers of Travaux, Inc. are hereby authorized and directed to properly execute said contract for and on behalf of Travaux, Inc.; and, be it

FURTHER RESOLVED, that if any extensions of the time of performance and change orders are needed, the Chairman or President may approve such request without any further action of Travaux Inc.; and, be it

FURTHER RESOLVED, that all acts previously performed, relative to this matter, by the Directors and the Present of Travaux, Inc., be and hereby are approved, ratified, and confirmed in all respects.

Attachments:
Project Description
Project Proforma
Draft of Construction Management Contract

Project Description

Highland Gardens is an existing senior independent living rental development totaling 114 rental units targeting senior residents over the age of 55 and/or adults with disabilities. The Highland Gardens site is located on the near west side of the City of Milwaukee on the northwest corner of the intersection of North 18th Street and West Juneau Avenue. More specifically, the subject site is located at 1818 West Juneau Avenue, Milwaukee, WI, approximately 0.8 miles west of the recently redeveloped Milwaukee Bucks "Deer District" and approximately 1.5 miles west of Milwaukee's central business district.

The existing four-story buildings were originally constructed in 2004 as a low income public housing development whereby existing eligible tenants receive a federal rental subsidy from the US Department of Housing and Urban Development ("HUD") in the form of Capital and Operations Funding under Section 9 of the United State Housing Act of 1937. The existing subject development is currently operating as LIHTC, Section 8, and Rental Assistance Demonstration ("RAD") public housing conversion senior and/or disabled development whereby tenants are paying 30 percent of their adjusted gross income ("AGI") as rent and must income qualify at or below the 80 percent area median income ("AMI") level.



Project Proforma

See attached.

DRAFT

Highland Gardens 10/26/2023

Low-Income Housing Tax Credit Financial Analysis WHEDA 2023 - 4% with State - General Set Aside

0.01% 99.99% 100.00%

90.00% 10.00% 100.00%

0.01%

100.00%

90.00% 10.00% 100.00%

PROJECT INFORMATION			FINANCING ASSUMPTIONS			SYNDICATION INFORMATION				
Project Name		Highland Gardens	<u></u>			General Partner's Capital Contrib	ution		Percent of Owners	hip
Developer Name		HACM	WHEDA Construction - Tax Exempt	\$	8,083,479	Entry Date:	Year	2024	General Partner	0.
Identity of Interest (USE DROPDOV	VNI)	Yes - Buyer and Seller	Interest Rate	Ψ	6.30%	Entry Bate.	Month	7	Investor	99.
identity of interest (OCL BITOI BOV	V14)	res - Buyer and Seller	Term (Months)		24		Day	1	Total	100.
O'the		NATH	Term (Months)		24		Day	1	10tai =	100.
City		Milwaukee						21.2221		
County (USE DROPDOWN)		Milwaukee	WHEDA Construction - Taxable	\$	-	Investor Combined Tax Rate		21.00%	Cash Flow Distribu	
Set Aside (USE DROPDOWN)		General	Interest Rate		6.90%				General Partner	90.
Tenant Type (USE DROPDOWN)		Elderly	Term (Months)		24	Total Investor Capital		\$12,358,488	Investor	10.
Credit % (USE DROPDOWN)		4% with State				Federal LIHTC Investor Capital	- PNC	8,731,894	Total	100.
			Total WHEDA Construction Loan	\$	10,611,101	State LIHTC Investor Capital - S	Sugar Creek	3,626,594	_	
Number of Units		114	Construction Loan - Max	\$	10,611,101				Income, Loss & Cre	edits
LP Formation			Interest Rate		0.00%	1st Contribution			General Partner	0.
	Year	2024	Term (Months)		24	Closing	Date:	7/1/2024	Investor	99.
	Month	7	,			20.00%	Fed LIHTC	1,746,379	Total	100.
			WHEDA Permanent Mortgage	\$	2,527,622		State LIHTC	725,319	=	
Construction Start			Interest Rate	Ψ	6.60%		Total	\$2,471,698	Sale Proceeds Dist	tribution
Construction Start	Year	2024	Amortization				Iotal	Ψ2,47 1,090	General Partner	90.
		7			35 35	On al Contribution				
	Month	I	Term Start Year			2nd Contribution Completion	Doto	9/1/2025	Investor Total	10. 100.
0		40			2026		Date:		i Olai	100.
Construction Period		13	Start Month		5	65.00%	Fed LIHTC	5,675,731		
							State LIHTC	2,357,286		
Operations			WHEDA Subordinate Debt	\$	-		Total	\$8,033,017		
	Year	2025	Interest Rate		3.00%					
	Month	9	Amortization		35	3rd Contribution				
			Term		17	Perm Conversion	Date:	5/1/2026		
Placed in Service Date			Start Year		2026	13.76%	Fed LIHTC	1,201,672		
	Year	2025	Start Month		5		State LIHTC	499,087		
	Month	9	WHEDA Subdebt? (USE DROPDOWN)		No		Total	\$1,700,759		
			WHEDA HTF	\$	-	4th Contribution				
Starting Month of Depreciation		9	Interest Rate		3.00%	8609's	Date:	5/1/2026		
Starting Month of Lease-Up		9	Amortization		35	1.24%	Fed LIHTC	108,112		
Absorption Rate (Mos.)		5	Term		19	1.2170	State LIHTC	44,902		
Absorption Nate (Mos.)		3	Start Year				Total	\$153,014		
Crodit 0/ 700/ D\/		0.000/			2024		IOlai	φ100,014		
Credit % - 70% PV		9.00%	Start Month		/ / / / / / / / / / / / / / / / / / /	Daile and a One of the				
Credit % - 30% PV		4.00%	WHEDA HTF? (USE DROPDOWN)		Yes	Price per Credit				
QCT or DDA?		Yes				Federal Low Income Housing	\$	0.8375		
Construction Type (USE DROPDO)	NN)	Acquisition / Rehab	GP Equity	\$	100	State Low Income Housing	\$	0.7000		
Building Type (USE DROPDOWN)		Elevator Building								
			Deferred Developer Fee	\$	647,721					
			Interest Rate		0.00%	Developer Fee Pay-In Schedule		1,530,137		
						1st Contribution	25.00%	382,534		
			HACM Seller Note	\$	8,550,000	2nd Contribution	25.00%	382,534		
			Interest Rate		3.86%	3rd Contribution	40.00%	612,055		
			Amortization		50	4th Contribution	10.00%	153,014		
			Term		50			,		
			Start Year		2024					
			Start Month		2024					
			Graft Month							
			CMF	\$	-					
			Interest Rate	Ψ	0.00%					
			Amortization		30					
			Term		30					
			Start Year		2					
			Ctart Manth		2024					

Start Month

2024

STATEMENT OF FORECASTED SOURCES AND USES											PAGE 4
SOURCES OF FUNDS		Construction	Post Construction	Permanent							
WHEDA Permanent Mortgage		-	2,527,622	2,527,622							
Federal LIHTC Investor Capital - PNC State LIHTC Investor Capital - Sugar Creek		1,746,379 725,319	6,985,515 2,901,275	8,731,894 3,626,594							
GP Equity HACM Seller Note		100 8,550,000	- -	100 8,550,000							
Deferred Developer Fee		2,527,622	647,721 (2,527,622)	647,721							
WHEDA Permanent Mortgage Available During Construction WHEDA Construction - Tax Exempt		8,083,479	(8,083,479)	- -							
Total Sources Of Funds		21,632,898	2,451,032	24,083,931							
		0	0	U							
			Post	Total		Residential Building	Personal Property	Land Impr.		Non Depreciable/	Eligible
USES OF FUNDS		Construction	Construction	Permanent		27.5 Yr/SL	5 Yr/200%	15 Yr/150%	Amortization	Amortizable	Basis
Acquisition Land		1,250,000	-	1,250,000						1,250,000	
Purchase of Buildings Other Purchase of Blds & Land		7,300,000		7,300,000		7,300,000				- -	7,300,000
Construction Related Expenses											
New Construction and Rehab Hard / Construction Costs - Residential	¢ 69.627	7 004 674		7,824,671		7,824,671					7,824,671
Hard / Construction Costs - Community Service Facility	\$ 68,637 \$ 307	7,824,671 35,000		35,000		35,000		-	-	- -	35,000
E - Equipment & Furnishings F - Special Construction and Demolition		-		-			-	-	-	-	-
Accessory Buildings (Garage, storage, etc.) Other New Construction / Rehabilitation		- -		-		- -	-	- -	- -	-	-
Site Work Costs Off Site - Site Work	\$ -	-		-		-	-	-	-	-	
On Site - Site Work Other Site Work		-		-		- -	-	-	-	-	-
Contractor Fees General Requirements	5.00%	392,984	_	392,984	5%	392,984		_	_	_	392,984
Contractor Overhead Contractor Profit	2.00% 5.00%	165,053 420,885	-	165,053 420,885	2% 5%	165,053 420,885		-	-	- -	165,053 420,885
Construction Supervision	3.0070	-	-	-	370	420,665	-	-	-		-
Contingency Funds	10.050	000.073		000.0==		000.050					202.5
Construction Contingency Other Contingency	10.00%	883,859 -	-	883,859 -		883,859	-	-	-	-	883,859
Construction Period Expenses											
Local or State Agency - Residential Related Fees and Expenses Construction Loan Origination Fee - WHEDA	1.00%	82,485	-	82,485		-	-	-	-	82,485	
Construction Loan Origination Fee - Non WHEDA Bond Related Expenses		-	-	-		-	-	-	-	- ·	-
Cost of Bond Issuance Other Lender and Financing Related Expenses		-	-	-		-	-	-	-	-	
Bridge Loan Fees and Expenses WHEDA Construction Loan Interest		- 585,493	-	-		456,000	-	-	-	-	- 456 000
Other Construction Loan Interest		-	-	585,493		456,092	-	-	-	129,401 -	456,092
Legal Fees - Miscellaneous Other Construction Period Soft Costs		123,000	-	123,000		123,000	-	-	-	-	123,000
Construction Loan Credit Enhancement/LOC Construction Period Real Estate Taxes		- 21,909		- 21,909		- 21,909	-	-	- -	-	- 21,909
Title and Recording Construction Insurance		35,000 29,685		35,000 29,685		35,000 29,685	-	-	-	-	35,000 29,685
Construction Liability Insurance Temporary Relocation Expenses	\$ 4,000	456,000	-	456,000		456,000	<u>-</u>	-	- -	-	456,000
Permanent Relocation Expenses Other Interim/Construction Costs	1,000	30,000	- 1	30,000		30,000	-	-	-	-	30,000
		30,000		30,000		30,000					30,000
Permanent Financing Expenses Lender and Financing Related Expenses											
Permanent Loan Origination Fee - Non WHEDA Permanent Loan Origination Fee - WHEDA	1.00%	25,276	-	- 25,276			-	-	- -	25,276	
Permanent Loan Credit Enhancement Legal Fees - Real Estate	\$ 20,000	20,000	-	20,000		- -	-	-	-	20,000	
Other Financing Fees and Expenses		15,500	-	15,500		•	-	•	-	15,500	
Architectural and Engineering Expenses Architect's Fee - Design		237,000	-	237,000		237,000	-	-	-	-	237,000
Architect's Fee - Inspection/Supervision Engineering Costs		548,000	-	548,000		- 548,000		- -	- -	-	548,000
Survey Other Architect and Engineering		86,900 40,000	-	86,900 40,000		86,900 40,000	<u>-</u>	-	-	- -	86,900 40,000
		40,000		40,000		40,000			_		40,000
Syndication Fees & Expenses Organizational (Partnership)		30,000	-	30,000		-	-	-	-	30,000	
Tax Opinion Other Syndication Costs		-		-		- -	-	-	- -	-	
Capitalized Reserves											
Operating Reserve Replacement Reserve	6 Months	-	475,708 165,000	475,708 165,000		-	-	-	-	475,708 165,000	
Lease-up Operating Deficit Debt Service Reserve		-		-			-	-	-	- - -	
Capital Needs Reserve Other Reserves		-	-	-		-	- -	-	-	- -	
Escrows		-	-	-		-	-	-	-	-	
Other Capitalized Reserves		-	-	-		-	-	•	-	-	
Reports, Studies and Related Work Appraisal(s)		11,000	-	11,000		11,000	-	-	-	-	11,000
Market Study Capital Needs Assessment Report		12,000 25,000	-	12,000 25,000		12,000 25,000	-	- -	- -	- -	12,000 25,000
Environmental Report		5,000	-	5,000		5,000	-	-	-	-	5,000
Other Soft Costs Tax Credit Application Fee	\$ 2,000	2,000	-	2,000				-	-	2,000	
Tax Credit Allocation Fee Tax Credit Compliance Fee		100,394 6,270	-	100,394 6,270		-	-	- -	- -	100,394 6,270	
Water, Sewer, and Impact Fees Cost Certification / Accounting Fees		32,500	15,000	47,500		47,500	- - -	- -	- -		- 32,500
Rent Up Marketing Expense		30,000	-	30,000		-	-	-	-	30,000	32,300
Mortgage Payoff - N/A for Tax Credit Application Other Miscellaneous Costs		290,000		290,000		290,000	-		-		290,000
Developer Earned Fees and Expenses	_										
Developer's Fee Received Developer Overhead		382,534 -	1,795,324 -	2,177,858 -		2,177,858		-		- -	2,177,858
Consultants Other Developer's Fees		97,500 -	-	97,500 -		97,500 -	- -	-	- -	- -	97,500
·		\$ 24 622 000		\$ 24,002,004		\$ 24.754.000	e	•	•	\$ 2222.024	\$ 24.720.000
Total Uses Of Funds	<u>l</u>	\$ 21,632,898	\$ 2,451,032	\$ 24,083,931	ı L	\$ 21,751,896	\$ -	\$ -	\$ -	\$ 2,332,034	\$ 21,736,896

Highland Gardens
PROJECT ASSUMPTIONS
PAGE 3

RENTAL INCOME

															95%		90%
Apartment Type	Tenancy	Bedroom	Number of	Net SF	Number of	AMI Set	Unit Type	Monthly	Total Monthly	Rent	Monthly	Monthly	Total Monthly	Max HTC	% of Max	Estimated	% of Max
	Type	Size	Bathrooms	Per Unit	Units	Aside %		Net Rent	Net Rent	\$ / SF	Utility	Gross Rent	Gross Rent	Limit	HTC Limit	Market Rent	Market Rent
Apartment	Elderly	1 Bedroom	1	650	29	60%	Voucher	\$ 900	\$ 26,100	1.38	\$ -	\$ 900	\$ 26,100	\$ 1,062	84.75%	\$ 909	99.01%
Apartment	Elderly	1 Bedroom	1	650	46	60%	Voucher	\$ 647	\$ 29,762	1.00	\$ -	\$ 647	\$ 29,762	\$ 1,062	60.92%	\$ 909	71.18%
Apartment	Elderly	1 Bedroom	1	650	31	60%	Low Income	\$ 900	\$ 27,900	1.38	\$ -	\$ 900	\$ 27,900	\$ 1,062	84.75%	\$ 909	99.01%
Apartment	Elderly	2 Bedroom	1	928	5	60%	Voucher	\$ 1,050	\$ 5,250	1.13	\$ -	1,050	\$ 5,250	\$ 1,275	82.35%	\$ 1,092	96.15%
Apartment	Elderly	2 Bedroom	1	928	3	60%	Low Income	\$ 1,050	\$ 3,150	1.13	\$ -	\$ 1,050	\$ 3,150	\$ 1,275	82.35%	\$ 1,092	96.15%
				76 324	114	60.00%			\$ 92 162 00				\$ 92 162 00				_

Income Averaging 60.00%
AHP 0.00%
Applicable Fraction (AMI) 100.00%
Applicable Fraction (Square Footage) 100.00%

		Lo	ow Income Unit	St	atistics
AMI %		# of Units	% of Units		Gross Annual Rental Income
	20%	0	0.00%	\$	-
	30%	0	0.00%	\$	-
	40%	0	0.00%	\$	-
	50%	0	0.00%	\$	-
	60%	114	100.00%	\$	1,105,944
	70%	0	0.00%	\$	-
	80%	0	0.00%	\$	-
Low Income Totals		114	100.00%		1,105,944
Ma	arket	0	0.00%	\$	-
Totals		114	100.00%		1,105,944

RESIDENTIAL INCOME	
Base Net Residential Rent	\$ 1,105,944
Rent Increases	2.00%
Other Income Increases	2.00%
Residential Vacancy Loss	7.00%
Other Income (PUPM)	\$ 4.59
Other Income (annually)	\$ 6,279
Other Income Vacancy Loss - LIHTC	7.00%
EGI	\$ 1,034,368

CASH FLOW DISTRIBUTION		
Asset Management Fee	8,550	3%

	Studio	1 Bedroom	2 Bedroom	3 Bedroom	4 Bedroom
Heating	0	0	0	0	0
Cooling	0	0	0	0	0
Cooking	0	0	0	0	0
Electricity	0	0	0	0	0
Hot Water	0	0	0	0	0
Other	0	0	0	0	0
	Λ	0	0	0	0

Other Income		Explanation
Parking	\$ -	
Laundry and Vending	6,279	Laundry Income
Misc. Income	0	
Other	0	
Total Other Income	\$ 6,279	

ACQUISITION CREDIT		STATE CREDIT ANALYSIS	LIH Credit	
Acquisition price (Building and Land) Less: Land & other non-depreciable costs Less: Other	\$ 8,550,000 (1,250,000)	Annual State Credit Calculated Annual State Credit Reserved	\$ 855,039 855,039	
Adjusted Eligible Basis	\$ 7,300,000	Minimum Credit or Allocation	\$ 855,039	
Qualified Census Tract Adjustment Eligible Basis	\$ 7,300,000	Bonus Credit - Energy Efficiency Bonus Credit - Additional 30% Units	8,550 0	
Applicable Fraction (Low Inc. %) Qualified Basis	100.00% \$ 7,300,000	Allocation with Bonus Reserved Annual Credit	\$ 863,589 \$ 863,561	
Credit Percentage (30% PV)	4.00%	Total State Credit	\$ 5,181,366	
		Investor Percentage	99.99% \$ 5,180,848	
		Price per Credit	0.700	
Potential Annual Credit	\$ 292,000	State Tax Credit Equity Raised	\$ 3,626,594	
CONSTRUCTION / REHAB CREDIT		FEDERAL CREDIT ANALYSIS	Federal Credit	State Fed Credit
Eligible Building Basis Less: Acquisition Price Less: Federal Historic Tax Credit Allocation	\$ 21,736,896 (7,300,000)	Annual LIH Credit Calculated Annual LIH Credit Reserved	\$ 1,042,719 1,042,719	\$ 1,042,719 1,042,719
Adjusted Eligible Basis	\$ 14,436,896	Minimum Credit or Allocation	\$ 1,042,719	\$ 1,042,719
QCT or DDA Boost HFA Boost Allowed Boost	130% 100% 130%	Bonus Credit - Energy Efficiency Bonus Credit - Additional 30% Units	0	
Eligible Basis	\$ 18,767,965	Allocation with Bonus	\$ 1,042,719	
Applicable Fraction (Low Inc. %)	100.00%	Minimum Credit or Allocation	\$ 1,042,719	
Qualified Basis	\$ 18,767,965 4,331,069	Total Tax Credit	\$ 10,427,186	\$ 10,427,186
Federal Credit Percentage	4.00%	Investor Percentage	99.98%	0.01%
Potential Annual Federal Credit	\$ 750,719	Tax Credits for Syndication	\$ 10,425,101	\$ 1,043
State Credit Percentage	3.00%	Price per Credit	0.838	0.838
Potential Annual State Credit	\$ 563,039	Tax Credit Equity Raised	\$ 8,731,021	\$ 873

		Expense	% Effective
	Annual Expense	per Unit 114	Gross Income
t Expenses	Expense	114	lilcome
Convention and Meeting	900.00	8	0.1%
Management Consultants	0.00 300.00	0 3	0.0% 0.0%
Ivertising / Marketing Expense her	0.00	0	0.0%
total Rent Expense	1,200.00	11	0.1%
inistrative Expenses			
fice Salaries	56,000.00	491	5.4%
fice Expenses	3,019.00	26	0.3%
ffice or Model Apartment Rent	0.00	0	0.0%
anagement Fee - Residential Rents anagement Fee - Commercial Rents	51,426.40 0.00	451 0	5.0% 0.0%
anagement Fee - Misc. Income	0.00	0	0.0%
anager / Superintendent Salaries	0.00	0	0.0%
dministrative Rent - Free Unit egal Expenses - Project Only	0.00 2,148.00	0 19	0.0% 0.2%
pokkeeping Fees / Accounting Services	21,330.00	187	2.1%
ax Credit Monitoring Fees	3,420.00	30	0.3%
ad Debt Expense	0.00	0	0.0%
ther Administrative Expenses total Administrative Expense	13,960.00 151,303.40	122 1,327	1.3% 14.6%
·	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	,-	
ties Expense	0.00	0	0.00/
uel Oil ectricity	0.00 68,900.00	0 604	0.0% 6.7%
ater	13,050.00	114	1.3%
as ewer	110,500.00 13,050.00	969 114	10.7% 1.3%
wner Paid Amenities	0.00	0	0.0%
total: Utilities Expense	205,500.00	1,803	19.9%
rating Maintenance Expense			
ayroll	68,432.00	600	6.6%
upplies ontracts	0.00 5,000.00	0 44	0.0% 0.5%
perating and Maintenance Rent Free Unit	0.00	0	0.0%
arbage and Trash Removal ecurity Payroll / Contracts (incl taxes and benefits)	39,996.00 3,496.00	351 31	3.9% 0.3%
ecurity Rent Free Unit	0.00	0	0.0%
eating / Cooling Maintenance ehicle / Maintenance Equipment Operation & Repairs	3,180.00 4,732.00	28 42	0.3% 0.5%
now Removal	0.00	0	0.0%
isc. Operating & Maintenance Expenses total: Operating Maintenance Expense	59,432.00 184,268.00	521 1,616	5.7% 17.8%
otal. Operating maintenance Expense	104,200.00	1,010	17.070
s and Insurance			
al Estate & Property Taxes	21,274.70	187	2.1%
perty and Liability Insurance (Hazard) yroll Taxes - Project Share	41,004.00 0.00	360 0	4.0% 0.0%
lelity Bond Insurance	0.00	0	0.0%
orkmen's Compensation alth Insurance and Other Employee Benefits	0.00 33,334.00	0 292	0.0% 3.2%
sc. Taxes, Licenses, Permits, and Insurance	93,995.90	825	9.1%
otal: Taxes and Insurance	189,608.60	1,663	18.3%
Sorvings Expanse			
Services Expense tary Services	0.00	0	0.0%
etary Purchased Serv	0.00	0	0.0%
ood egistered Nurse Salary	0.00 0.00	0	0.0% 0.0%
ousekeeping Salary	0.00	0	0.0%
ousekeeping Supply ther Housekeeping	0.00 0.00	0 0	0.0% 0.0%
edical Supplies	0.00	0	0.0%
edical Purchased Serv	0.00	0	0.0%
aundry / Linen aundry Salaries	0.00 0.00	0	0.0% 0.0%
nundry Purchased Serv	0.00	0	0.0%
undry Supplies edical Records Salary	0.00	0	0.0% 0.0%
edical Records Salary edical Records Supply	0.00	0	0.0%
ed Records Purchased SERV	0.00	0	0.0%
ecreation / Rehab ctivities Supplies	0.00 0.00	0	0.0% 0.0%
ctivities Purchased Serv	0.00	0	0.0%
ehab Salaries ehab Supplies	0.00 0.00	0	0.0% 0.0%
hab Supplies hab Purchased Serv	0.00	0	0.0%
her Support Serv	0.00	0	0.0%
total: Total Services Expense	0	0	0.0%
nual Replacement Reserves	34,200	300	3.3%
al Operating Expenses	\$ 766,080		
Unit Per Month	\$ 560		
	OPEX OUTS	SIDE OF WHED	A RANGE
otor Accet Management Foo	¢ 0.550		

\$ 8,550

Investor Asset Management Fee

		1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	TOTAL
INCOME	0.000/					* * * * * * * * * * * * * * * * * * *		* 4 0 4 5 4 5 0		* 4 00 = = 00				*	4 400 050	4 450 050	* 40 40= ==4
Gross Potential Income: (Excluding TIF & Other)	2.00% \$, , ,	, ,	. , , ,	, ,		, ,		. , ,	. , , , .	, ,	. , , .	, ,	\$ 1,402,604 \$	1,430,656 \$., ,	\$ 19,125,551
Other Income Source	2.00%	6,279	6,405	6,533	6,663	6,797	6,933	7,071	7,213	7,357	7,504	7,654	7,807	7,963	8,123	8,285	108,587
Vacancy - Rental Income	-7.00%	(77,416)	(78,964)	(80,544)	(82,155)	(83,798)	(85,474)	(87,183)	(88,927)	(90,705)	(92,519)	(94,370)	(96,257)	(98,182)	(100,146)	(102,149)	(1,338,789)
Vacancy - Other Income	-7.00%	(440)	(448)	(457)	(466)	(476)	(485)	(495)	(505)	(515)	(525)	(536)	(547)	(557)	(569)	(580)	(7,601)
Total Vacancy	\$	(77,856) \$	(79,413)	\$ (81,001) \$	(82,621)	\$ (84,273) \$	(85,959)	\$ (87,678)	\$ (89,432)	\$ (91,220) \$	(93,045)	(94,906) \$	(96,804)	\$ (98,740) \$	(100,715) \$	(102,729)	\$ (1,346,390)
Effective Gross Income		1,034,368	1,055,055	1,076,156	1,097,679	1,119,633	1,142,025	1,164,866	1,188,163	1,211,926	1,236,165	1,260,888	1,286,106	1,311,828	1,338,065	1,364,826	17,887,748
EXPENSES																	
Rent Expense	3.00%	1,200	1,236	1,273	1,311	1,351	1,391	1,433	1,476	1,520	1,566	1,613	1,661	1,711	1,762	1,815	22,319
Management Fees	3.00%	51,426	52,969	54,558	56,195	57,881	59,617	61,406	63,248	65,145	67,100	69,113	71,186	73,322	75,521	77,787	956,475
Other Admin Expenses (Less Management Fee	3.00%	99,877	102,873	105,960	109,138	112,412	115,785	119,258	122,836	126,521	130,317	134,226	138,253	142,401	146,673	151,073	1,857,604
Utilities Expense	3.00%	205,500	211,665	218,015	224,555	231,292	238,231	245,378	252,739	260,321	268,131	276,175	284,460	292,994	301,784	310,837	3,822,077
Operating and Maintenance Expense	3.00%	184,268	189,796	195,490	201,355	207,395	213,617	220,026	226,626	233,425	240,428	247,641	255,070	262,722	270,604	278,722	3,427,185
Real Estate and Personal Property Taxes	3.00%	21,275	21,913	22,570	23,247	23,945	24,663	25,403	26,165	26,950	27,759	28,591	29,449	30,333	31,243	32,180	395,686
Other Taxes and Insurance	3.00%	168,334	173,384	178,585	183,943	189,461	195,145	200,999	207,029	213,240	219,638	226,227	233,013	240,004	247,204	254,620	3,130,828
Service Expense	3.00%	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Replacement Reserve	3.00%	34,200	35,226	36,283	37,371	38,492	39,647	40,837	42,062	43,324	44,623	45,962	47,341	48,761	50,224	51,731	636,083
Total Expenses		766,080	789,062	812,734	837,116	862,230	888,097	914,740	942,182	970,447	999,561	1,029,547	1,060,434	1,092,247	1,125,014	1,158,765	14,248,256
Net Operating Income		268,288	265,992	263,422	260,563	257,403	253,929	250,126	245,981	241,479	236,604	231,341	225,672	219,581	213,050	206,061	3,639,492
Debt Service - 1st Mortgage																	
WHEDA Permanent Mortgage		185,337	185,337	185,337	185,337	185,337	185,337	185,337	185,337	185,337	185,337	185,337	185,337	185,337	185,337	185,337	2,780,049
Total Debt Service		185,337	185,337	185,337	185,337	185,337	185,337	185,337	185,337	185,337	185,337	185,337	185,337	185,337	185,337	185,337	2,780,049
DSCR- 1st		1.448	1.435	1.421	1.406	1.389	1.370	1.350	1.327	1.303	1.277	1.248	1.218	1.185	1.150	1.112	
Cash Flow		82,951	80,656	78,085	75,226	72,066	68,592	64,790	60,645	56,143	51,268	46,004	40,335	34,245	27,714	20,725	859,443
Asset Management Fee	8,550	8,550	8,807	9,071	9,343	9,623	9,912	10,209	10,515	10,831	11,156	11,490	11,835	12,190	12,556	12,933	159,021
Cash Flow		74,401	71,849	69,014	65,883	62,443	58,680	54,580	50,129	45,312	40,112	34,514	28,500	22,054	15,158	7,792	700,423
DDF Interest Rate	647,721 <i>0%</i>	74,401 -	71,849 -	69,014 -	65,883 -	62,443 -	58,680 -	54,580 -	50,129 -	45,312 -	40,112 -	34,514 -	20,803	-	<u>-</u>	-	647,721 -
		573,320	501,471	432,456	366,573	304,130	245,450	190,869	140,740	95,428	55,316	20,803	-	-	-	-	

Highland Gardens
Construction Interest

Construction Interest		lr Ir	Construction Loa nterest during Co nterest during Le Fotal Construction	onstruction ease Up	Total 456,092 129,401 585,493	2024 91,647 0 91,647	2025 364,445 64,701 429,146	2026 0 64,701 64,701	E	Construction Lo Budgeted Interes Estimated Interes Excess / (Shortag	st st	585,493 585,493 0													PAGE 8
Sources of Funds	Permanent	Closing Jul 2024 0	Aug 2024 1	Sep 2024 2	Oct 2024 3	Nov 2024 4	Dec 2024 5	Lease Up Jan 2025 6	Lease Up Feb 2025 7	Lease Up Mar 2025 8	Lease Up Apr 2025 9	Lease Up May 2025 10	Lease Up Jun 2025 11	Lease Up Jul 2025 12	Lease Up Aug 2025 13	Lease Up Completion Sep 2025 14	Lease Up Oct 2025 15	Lease Up Nov 2025 16	Lease Up	Lease Up Jan 2026 20	Lease Up Feb 2026 21	Mar 2026 22	Apr 2026 23	Perm Conv 8,609 May 2026 24	Tota
WHEDA Permanent Mortgage WHEDA Subordinate Debt Federal LIHTC Investor Capital - PNC State LIHTC Investor Capital - Sugar Creek	2,527,622 0 8,731,894 3,626,594	- - 1,746,379 725,319	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- -	- - 5,675,731 2,357,286	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	-	2,527,622 - 1,309,784 543,989	2,527,622 - 8,731,894 3,626,594
GP Equity FHLBC - AHP HACM Existing Reserves HACM Seller Note	100 0 0 8,550,000	100 - - 8,550,000	- - -	- - -	- - -	- - -	- - -	- - - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	100 - - 8,550,000
CMF Deferred Developer Fee WHEDA Permanent Mortgage Available During Construction 2,527,622 WHEDA Construction - Tax Exempt WHEDA Construction - Taxable Construction Loan	0 647,721	- 303,155 - -	- - 779,464 - -	- - 1,172,649 - -	- 272,354 906,744 -	- - - 990,908 -	- - - 996,110 -	- - - 806,890 -	- - - 811,127 -	- - - 620,936 -	- - - 624,196 -	- - - 627,473 -	- - - 630,767 -	- - - 634,079 -	- - - 434,251 -	- - - (7,650,483) 0	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- (2,527,622) (432,996) -	-
Total Sources Of Funds	24,083,931	11,324,953	779,464	1,172,649	1,179,098	990,908	996,110	806,890	811,127	620,936	624,196	627,473	630,767	634,079	434,251	382,534	0	0	0	0	0	0	0	1,420,777	23,436,210
WHEDA Permanent Mortgage Available During Construction Balance Interest Rate Interest Eligible Interest	2,527,622 6.60%	303,155 0.55% 1,667 1,667	1,082,619 0.55% 5,954 5,954	2,255,268 0.55% 12,404 12,404	2,527,622 0.55% 13,902 13,902	2,527,622 0.55% 13,902 13,902	2,527,622 0.55% 13,902 13,902	2,527,622 0.55% 13,902 13,902	2,527,622 0.55% 13,902 13,902	2,527,622 0.55% 13,902 13,902	2,527,622 0.55% 13,902 13,902	2,527,622 0.55% 13,902 13,902	2,527,622 0.55% 13,902 13,902	2,527,622 0.55% 13,902 13,902	2,527,622 0.55% 13,902 13,902	2,527,622 0.55% 13,902	0.55% 13,902 -	0.55% 13,902 -	0.55% 13,902 -	0.55% 13,902 -	0.55% 13,902 -	2,527,622 0.55% 13,902	2,527,622 0.55% 13,902	0 0.55% - -	284,162 172,947
WHEDA Construction - Tax Exempt Balance Interest Rate Interest Eligible Interest	8,083,479 6.30%	- 0.53% -	0 0.53% -	0 0.53% -	906,744 0.53% 4,760 4,760	1,897,651 0.53% 9,963 9,963	2,893,761 0.53% 15,192 15,192	3,700,651 0.53% 19,428 19,428	4,511,778 0.53% 23,687 23,687	5,132,714 0.53% 26,947 26,947	5,756,910 0.53% 30,224 30,224	6,384,382 0.53% 33,518 33,518	7,015,149 0.53% 36,830 36,830	7,649,228 0.53% 40,158 40,158	8,083,479 0.53% 42,438 42,438	13,902 432,996 0.53% 2,273	13,902 432,996 0.53% 2,273	13,902 432,996 0.53% 2,273	13,902 432,996 0.53% 2,273	13,902 432,996 0.53% 2,273	13,902 432,996 0.53% 2,273	13,902 432,996 0.53% 2,273	13,902 432,996 0.53% 2,273	0 0.53% 0	301,331 283,145
WHEDA Construction - Taxable Balance Interest Rate	0 6.90%	- 0.58%	- 0 0.58%	- 0 0.58%	- 0 0.58%	- 0 0.58%	0 0.58%	0 0.58%	0 0.58%	0 0.58%	0 0.58%	- 0 0.58%	- 0 0.58%	- 0 0.58%	- 0 0.58%	2,273 0 0,58%	2,273 0 0,58%	2,273 0 0.58%	2,273 0 0,58%	2,273 0 0,58%	2,273 0 0.58%	2,273 0 0.58%	2,273 0 0.58%	0.000 0 0.58%	18,186
Interest Rate Interest Eligible Interest Ineligible Interest Total Construction Interest	0.90%	- - - - 1,667	- - - - 5,954	- - - - - 12,404	18,662	23,865	29,094	33,330	37,589		44,126	- - - - 47,420	50,731	- - - - 54,060	- - - - 56,340	0 - 0	0 - 0	0.38%	0.3878	0 - 0	0 - 0	0 - 0	0.38% 0 - 0 16,175	0.000	0 0 0 585,493
Uses of Funds			8.00%	12.00%	12.00%	10.00%	10.00%	8.00%	8.00%	6.00%	6.00%	6.00%	6.00%	6.00%	2.00%										
Acquisition Land Purchase of Buildings Other Purchase of Blds & Land	1,250,000 7,300,000 -	1,250,000 7,300,000 -	:	: : :	: : :	: :	- - -	- - -	:	:	:	- - -	:	- - -	- -	- - -	: : :	- - -	-	:	:	:	- -	:	1,250,000 7,300,000 - -
Construction Related Expenses New Construction and Rehab Hard / Construction Costs - Residential Hard / Construction Costs - Community Service Facility E - Equipment & Furnishings	7,824,671 35,000 -	- - -	625,974 2,800 -	938,960 4,200 -	938,960 4,200 -	782,467 3,500	782,467 3,500 -	625,974 2,800 -	625,974 2,800 -	469,480 2,100 -	469,480 2,100 -	469,480 2,100 -	469,480 2,100 -	469,480 2,100 -	156,493 700 -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	7,824,671 35,000 -
F - Special Construction and Demolition Accessory Buildings (Garage, storage, etc.) Other New Construction / Rehabilitation Site Work Costs Off Site - Site Work	- - -	-	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -
On Site - Site Work Other Site Work Contractor Fees General Requirements	- - 392,984	- -	- - 31,439	- - 47,158	- - 47,158	- - 39,298	- - 39,298	- - 31,439	- - 31,439	- - 23,579	- - 23,579	- - 23,579	- - 23,579	- - 23,579	- - 7,860	- -	-	- -	- -	-	- -	-	-	-	- - 392,984
Contractor Overhead Contractor Profit Construction Supervision	165,053 420,885 -	- - -	13,204 33,671 -	19,806 50,506 -	19,806 50,506 -	16,505 42,089 -	16,505 42,089 -	13,204 33,671 -	13,204 33,671 -	9,903 25,253 -	9,903 25,253 -	9,903 25,253 -	9,903 25,253 -	9,903 25,253 -	3,301 8,418 -	- - -	- - -	- - -	- - -	-	-	-	- - -	- - -	165,053 420,885 -
Contingency Funds Construction Contingency Other Contingency Construction Period Expenses	883,859 -	:	70,709 -	106,063 -	106,063 -	88,386 -	88,386 -	70,709 -	70,709 -	53,032 -	53,032 -	53,032 -	53,032 -	53,032 -	17,677 -	Ė	Ē	:	Ξ	Ī	Ē	:	Ξ	:	883,859 -
Local or State Agency - Residential Related Fees and Expenses Construction Loan Origination Fee - WHEDA Construction Loan Origination Fee - Non WHEDA Bond Related Expenses Cost of Bond Issuance	82,485 - -	82,485 - -	- -	- -	- -	- -		- -	- -		- -	- -	-	- -	- -	- -	- -	- -	-	-	-	- -		- -	82,485 - -
Other Lender and Financing Related Expenses Bridge Loan Fees and Expenses WHEDA Construction Loan Interest Other Construction Loan Interest Legal Fees - Miscellaneous	585,493 - 123,000	- - - 123,000	- 1,667 - -	- 5,954 - -	- 12,404 - -	- 18,662 - -	- 23,865 - -	- 29,094 - -	- 33,330 - -	- 37,589 - -	- 40,849 - -	- 44,126 - -	- 47,420 - -	- 50,731 - -	- 239,802 - -	- - - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- 585,493 - 123,000
Other Construction Period Soft Costs Construction Loan Credit Enhancement/LOC Construction Period Real Estate Taxes Title and Recording Construction Insurance Construction Liability Insurance	21,909 35,000 29,685	21,909 35,000 29,685	- - -	- - - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- 21,909 35,000 29,685
Temporary Relocation Expenses Permanent Relocation Expenses Other Interim/Construction Costs Permanent Financing Expenses	456,000 - 30,000	456,000 - 30,000	:	-	-	:	-	-	-	:	:	:	-	-	-	- - -	- - -	-	:	- - -	- - -	:	-	:	456,000 - 30,000
Lender and Financing Related Expenses Permanent Loan Origination Fee - Non WHEDA Permanent Loan Origination Fee - WHEDA Permanent Loan Credit Enhancement Legal Fees - Real Estate	- 25,276 - 20,000	- 25,276 - 20,000	: :	- - - -	- - - -	: : :	- - - -	- - - -	- - - -	- - - -	:	- - - -	- - - -	- - - -	- - - -	: : :	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- 25,276 - 20,000
Other Financing Fees and Expenses Architectural and Engineering Expenses Architect's Fee - Design	15,500 237,000	15,500 237,000	-	-	- -	-	- -	-	-		-	-	-	-	-	- -	-	- -	-	-	-	-	-	·	15,500 237,000
Architect's Fee - Inspection/Supervision Engineering Costs Survey Other Architect and Engineering	548,000 86,900 40,000	548,000 86,900 40,000	- - -	- - - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - - -	- - - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	548,000 86,900 40,000
Syndication Fees & Expenses Organizational (Partnership) Tax Opinion Other Syndication Costs	30,000 - -	30,000 - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	30,000 - -
Capitalized Reserves Operating Reserve Replacement Reserve Lease-up Operating Deficit	475,708 165,000 -	- - -	- - -	- - - -	- - -	- - -	- - -	- - -	: : :	- - -	- - -	- - -	: : :	- - -	- - -	- - - -	: : :	- - -		: : :	: : :	- - -	- - -	475,708.30 165,000.00 -	475,708 165,000
Debt Service Reserve Capital Needs Reserve Other Reserves Escrows Other Capitalized Reserves	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -
Reports, Studies and Related Work Appraisal(s) Market Study Capital Needs Assessment Report	11,000 12,000 25,000	11,000 12,000 25,000	- - -	- - -	- - -	- - -	- - -	- - -	- - -	:	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	11,000 12,000 25,000
Other Soft Costs Tax Credit Application Fee Tax Credit Allocation Fee	2,000 100,394	2,000 100,394	- - -	- - -	- - - -	- - -	- - -	- - -	- - -	:	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	· :	2,000 100,394
Tax Credit Compliance Fee Water, Sewer, and Impact Fees Cost Certification / Accounting Fees Rent Up Marketing Expense	6,270 - 47,500 30,000	6,270 - 32,500 30,000	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - 15,000.00 -	6,270 - 47,500 30,000
Mortgage Payoff - N/A for Tax Credit Application Other Miscellaneous Costs	30,000 - 290,000	30,000 - 290,000	- - -	- - -	- - -	:	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	- - -	-	- - -	- - -	290,000
Developer Earned Fees and Expenses Developer's Fee Received Developer's Fee - Deferred Developer Overhead Consultants	2,177,858 - - - 97,500	382,534 - - - 97,500	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - - -	- - -	- - - -	- - - -	- - - -	- - - -	- - - -	382,534 - - -	- - - -	- - - -	- - -	- - - -	- - - -	- - - -	- - -	765,069 - - -	1,530,137 - - - 97,500
Other Developer's Fees Total Uses Of Funds	24,083,931	-	- 779,464	- 1,172,649	- 1,179,098	990,908	996,110	806,890	- 811,127	620,936	- 624,196	627,473	630,767	634,079	- 434,251	382,534	-	-	-	-	-	-	-	- 1,420,777	-

Draft of Construction Management Contract

See attached.



Standard Form of Agreement Between Owner and Contractor where the basis of payment is the Cost of the Work Plus a Fee with a Guaranteed Maximum Price

CONTRACT 23-006

AGREEMENT made effective as of the 9th day of November in the year 2023 (*In words, indicate day, month and year.*)

BETWEEN the Owner:

(Name, legal status, address and other information)

Highland Gardens LLC c/o Housing Authority of the City of Milwaukee 809 North Broadway, 3rd Floor Milwaukee, WI 53202

and the Contractor:

(Name, legal status, address and other information)

Travaux Inc.

c/o Housing Authority of the City of Milwaukee 809 North Broadway, 3rd Floor Milwaukee, WI 53202

for the following Project:

(Name, location and detailed description)

Highland Gardens

Located at 1818 West Juneau Avenue, Milwaukee, WI 53233 Rehabilitation of 114-unit apartment building and associated site improvements

The Architect:

(Name, legal status, address and other information)

Galbraith Carnahan Architects LLC 6404 West North Avenue Wauwatosa, WI 53213

The Contractor's Designated Representative:

(Name, address and other information)

Warren B. Jones Travaux Inc. 401 East Ogden Avenue Milwaukee, WI 53202

The Architect's Designated Representative:

(Name, address and other information)

Joseph Galbraith Calbraith Carnahan Architects LLC 6404 West North Avenue

ADDITIONS AND DELETIONS:

The author of this document has added information needed for its completion. The author may also have revised the text of the original AIA standard form. An *Additions and Deletions Report* that notes added information as well as revisions to the standard form text is available from the author and should be reviewed. A vertical line in the left margin of this document indicates where the author has added necessary information and where the author has added to or deleted from the original AIA text.

This document has important legal consequences. Consultation with an attorney is encouraged with respect to its completion or modification.

The parties should complete A102™–2017, Exhibit A, Insurance and Bonds, contemporaneously with this Agreement. AIA Document A201™–2017, General Conditions of the Contract for Construction, is adopted in this document by reference. Do not use with other general conditions unless this document is modified.

Init.

AIA Document A102 – 2017. Copyright © 1920, 1925, 1951, 1958, 1961, 1963, 1967, 1974, 1978, 1987, 1997, 2007 and 2017. All rights reserved. "The American Institute of Architects," "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are trademarks of The American Institute of Architects. This document was produced at 15:38:28 ET on 10/30/2023 under Order No.2114399858 which expires on 02/01/2024, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail docinfo@aiacontracts.com.

Wauwatosa, WI 53213

The Owner's Designated Representative: (Name, address and other information)

Willie L. Hines, Jr. Highland Gardens LLC c/o Housing Authority of the City of Milwaukee 809 North Broadway, 3rd Floor Milwaukee, WI 53202

The Owner and Contractor agree as follows.

TABLE OF ARTICLES

- 1 THE CONTRACT DOCUMENTS
- 2 THE WORK OF THIS CONTRACT
- 3 RELATIONSHIP OF THE PARTIES
- 4 DATE OF COMMENCEMENT AND SUBSTANTIAL COMPLETION
- 5 CONTRACT SUM
- 6 CHANGES IN THE WORK
- 7 COSTS TO BE REIMBURSED
- 8 COSTS NOT TO BE REIMBURSED
- 9 DISCOUNTS, REBATES AND REFUNDS
- 10 SUBCONTRACTS AND OTHER AGREEMENTS
- 11 ACCOUNTING RECORDS
- 12 PAYMENTS
- 13 DISPUTE RESOLUTION
- 14 TERMINATION OR SUSPENSION
- 15 MISCELLANEOUS PROVISIONS
- 16 ENUMERATION OF CONTRACT DOCUMENTS

ARTICLE 1 THE CONTRACT DOCUMENTS

The Contract Documents consist of this Agreement, Conditions of the Contract (General, Supplementary, and other Conditions), HUD 5370 General Conditions of the Contract for Construction Drawings, Specifications, Addenda issued prior to execution of this Agreement, other documents listed in this Agreement and Modifications issued after execution of this Agreement, all of which form the Contract, and are as fully a part of the Contract as if attached to this Agreement or repeated herein. The Contract represents the entire and integrated agreement between the parties hereto and supersedes prior negotiations, representations, or agreements, either written or oral. If anything in the other Contract Documents, other than a Modification, is inconsistent with this Agreement, this Agreement shall govern. The terms of AIA Document A201TM–2017, General Conditions of the Contract for Construction (sometimes referred to herein as the "General Conditions") and the terms of HUD 5370 General Conditions of the Contract for Construction shall be construed and interpreted to the full extent possible to give effect to all such terms. In the event of any conflict between the terms, the terms of HUD 5370 General Conditions of the Contract for Construction shall govern and control. This Agreement may be amended only by written instrument signed by both the Owner and Contractor. An enumeration of the Contract Documents, other than a Modification, appears in Article 16.

ARTICLE 2 THE WORK OF THIS CONTRACT

§ 2.1 The Contractor shall perform the services described in this Article. The Contractor has identified a representative above authorized to act on behalf of the Contractor with respect to the Project.

§ 2.1.2 General

- § 2.1.2.1 For purposes of Section 8.1.2 of the General Conditions, the date of means of the Work shall mean the date of commencement of the Construction Phase. The Construction Phase shall commence upon the Owner's issuance of a Notice to Proceed.
- § 2.1.2.2 The Contractor shall have the control over and charge of and shall be responsible for construction means, methods, techniques, sequences and procedures, and the safety precautions and programs in connection with the Work. The Contractor shall be responsible for its and its Subcontractor's failure to carry out the Work in accordance with the Contract Documents. The Contractor shall coordinate the sequence of the construction and the responsibilities of the Subcontractors and be responsible for the acts and omissions of the Subcontractors and their agents and employees.
- **§2.1.2.3** The Contractor shall monitor the Work of the Contractors, Subcontractors and suppliers, and coordinate their Work with the activities and responsibilities of the Architect, the Contractor, and the Owner to complete the Project in accordance with the Owner's objectives of cost, time and quality.
- § 2.1.2.4 The Contractor shall maintain a competent staff at the Project Site to coordinate and provide general direction of the Work and facilitate progress of the Contractors, Subcontractors and suppliers on the Project.
- § 2.1.2.5 The Contractor shall establish procedures for coordination among the Architect, Contractors, Subcontractors and suppliers and the Contractor with respect to all aspects of the Project and shall implement such procedures.
- § 2.2.1.6 The Contractor shall determine the adequacy of the Contractors', Subcontractors' and suppliers' personnel and equipment and the availability of materials and supplies to meet the schedule. The Contractor shall take appropriate action when requirements of a contract are not being met.
- § 2.1.2.7 If needed, the Contractor shall provide temporary construction office facility for use by Owner, Architect, and Contractor at the Project site.
- § 2.1.2.8 The Contractor shall manage the mobilization activities with the Subcontractors.
- § 2.1.2.9 The Contractor shall arrange for the delivery, storage, protection, and security of Owner-purchased materials, systems, and equipment that (a) are obtained or acquired through the Contractor or the Contractor's purchasing entity, or (b) are items to be installed by the Contractor.
- § 2.1.2.10 The Contractor shall provide assignment of responsibilities for temporary project facilities and equipment, materials and services for common use of the Contractor and the Subcontractors.

§ 2.2.1 Administration

- § 2.2.1.1 Those portions of the Work that the Contractor does not customarily perform with the Contractor's own personnel shall be performed under subcontracts or by other appropriate agreements with the Contractor. The Owner may designate specific persons from whom, or entities from which, the Contractor shall obtain bids. The Contractor shall obtain bids from Subcontractors and from suppliers of materials or equipment fabricated especially for the Work and shall deliver such bids to the Owner and Architect. The Owner shall then determine, with the advice of the Contractor and the Architect, which bids will be accepted. The Contractor shall not be required to contract with anyone to whom the Contractor has reasonable objection. The Contractor shall be responsible for the proper and complete performance of the Work in accordance with the terms and provisions of the Contract Documents, whether performed by the Contractor or by Subcontractors.
- § 2.2.1.2 Subcontracts or other agreements shall conform to the applicable payment provisions of this Agreement, and shall not be awarded on the basis of cost plus a fee without the prior consent of the Owner. If the Subcontract is awarded on a cost-plus a fee basis, the Contractor shall provide in the Subcontract for the Owner to receive the same audit rights with regard to the Subcontractor as the Owner receives with regard to the Contractor in Article 12.
- § 2.2.1.3 Intentionally omitted..

- § 2.2.1.5 The Contractor shall schedule and conduct meetings at which the Owner, Architect, Contractor and appropriate Subcontractors can discuss such matters as procedures, progress, coordination, scheduling, and status of the Work. The Contractor shall prepare and promptly distribute minutes to the Owner and Architect.
- § 2.2.1.6 Upon the execution of this Agreement the Contractor shall prepare and submit to the Owner and Architect a construction schedule for the Work and submittal schedule in accordance with Section 3.10 of the General Conditions.
- § 2.2.1.7 The Contractor shall record the progress of the Project. On a quarterly basis, or otherwise as agreed to by the Owner, the Contractor shall submit written progress reports to the Owner and Architect, showing percentages of completion and other information required by the Owner. The Contractor shall also keep, and make available to the Owner and Architect, a daily log containing a record for each day of weather, portions of the Work in progress, Work accomplished, number of workers on site, Subcontractors working on site, identification of equipment on site, problems that might affect progress of the work, accidents, injuries, and other information required by the Owner.
- § 2.2.1.8 The Contractor shall develop a system of cost control for the Work, including regular monitoring of actual costs for activities in progress and estimates for uncompleted tasks and proposed changes. The Contractor shall identify variances between actual and estimated costs and report the variances to the Owner and Architect and shall provide this information in its quarterly reports to the Owner and Architect, in accordance with Section 2.2.1.7above.
- § 2.2.1.9 The Contractor shall receive, review and approve all certificates of insurance, lien waivers and similar information and shall forward copies, if requested, to the Architect and the Owner.
- § 2.2.1.10 The Contractor shall develop and implement a system for the preparation, review and processing of change orders, and shall recommend necessary or desirable changes to the Owner.
- § 2.2.1.11 In collaboration with the Architect, the Contractor shall establish and implement procedures for expediting the processing and approval of shop drawings and samples.
- § 2.2.1.12 The Contractor shall work with the Architect to prepare a punch-list of defective or incomplete work, and shall coordinate completion of punch-list items.
- § 2.2.1.13 The Contractor shall secure and transmit to the Owner all required warranties and lien waivers and turn over to the Owner all keys, manuals, record drawings and maintenance stocks. The Contractor shall coordinate the collection of the Contractor and Subcontractor field-annotated record drawings and shall mark up a consolidated print for the Architect's use.
- § 2.2.1.14 The Contractor shall prepare and deliver to the Owner a final accounting for all costs incurred.

ARTICLE 3 RELATIONSHIP OF THE PARTIES

The Contractor accepts the relationship of trust and confidence established with the Owner by this Agreement and covenants with the Owner to cooperate with the Architect and exercise the Contractor's skill and judgment in furthering the interests of the Owner; to furnish efficient construction administration, management services and supervision; to furnish at all times an adequate supply of workers and materials; and to use the Contractor's best efforts to perform the Work in an expeditious and economical manner consistent with the Owner's interests. All parties shall endeavor to promote harmony and cooperation among the Owner, Architect, Contractor and other persons and entities employed by the Owner for the Project.

ARTICLE 4 DATE OF COMMENCEMENT AND SUBSTANTIAL COMPLETION

§ 4.1 The date of commencement of the Work shall be: *(Check one of the following boxes.)*

[]	The date of this Agreement.
[X]	A date set forth in a notice to proceed issued by the Owner.
[]	Established as follows:

If a date of commencement of the Work is not selected, then the date of commencement shall be the date of this Agreement.

§ 4.2 The Contract Time shall be measured from the date of commencement of the Work.

§ 4.3 Substantial Completion

§ 4.3.1 Subject to adjustments of the Contract Time as provided in the Contract Documents, the Contractor shall achieve Substantial Completion of the entire Work:

(Check one of the following boxes and complete the necessary information.)

[]	Not later than () calendar days from the date of commencement of the Work.
[X]	By the following date: December 31, 2025

§ 4.3.2 Subject to adjustments of the Contract Time as provided in the Contract Documents, if portions of the Work are to be completed prior to Substantial Completion of the entire Work, the Contractor shall achieve Substantial Completion of such portions by the following dates:

Portion of Work

Substantial Completion Date

NONE

§ 4.3.3 If the Contractor fails to achieve Substantial Completion as provided in this Section 4.3, liquidated damages, if any, shall be assessed as set forth in Section 5.1.6.

ARTICLE 5 COMPENSATION FOR CONSTRUCTION PHASE SERVICES

§ 5.1 For the Contractor's performance of the Work as described in Article 2, the Owner shall pay the Contractor the Contract Sum in current funds for the Contractor's performance of the Contract. The Contract Sum is the Cost of the Work as defined in Article 7 plus the Contractor's Fee.

§ 5.1.1 The Contractor's Fee:

(State a lump sum, percentage of Cost of the Work, or other provision for determining the Contractor's Fee.)

A lump sum of Four Hundred Forty-One Thousand, Nine Hundred Twenty-Nine and 00/100 Dollars (\$441,929.00), exclusive of Architect and Engineer fees, Owner's FF&E, performance/payment bond, and builder's risk insurance. Owner FF&E if incorporated into the budget as an allowance would be considered a portion of the Cost of Work for purposes of calculating the fee.

§ 5.1.2 Intentionally omitted.

§ 5.1.3 Intentionally omitted.

§ 5.1.4 Rental rates for Contractor-owned equipment shall not exceed eighty-five percent (85 %) of the rental rates listed in the latest edition of the Associated Equipment Distributors "green book" (a/k/a AED Green Book) of nationally averaged rates. Charges for equipment rented from third parties shall be the actual rental rates, but shall in no even exceed local market rates.

§ 5.1.5 Unit prices, if any:

(Identify the item and state the unit price and quantity limitations, if any, to which the unit price will be applicable.)

Item Units and Limitations Price Per Unit (\$0.00)
NONE

Init.

AIA Document A102 – 2017. Copyright © 1920, 1925, 1951, 1958, 1961, 1963, 1967, 1974, 1978, 1987, 1997, 2007 and 2017. All rights reserved. "The American Institute of Architects," "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are trademarks of The American Institute of Architects. This document was produced at 15:38:28 ET on 10/30/2023 under Order No.2114399858 which expires on 02/01/2024, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail docinfo@aiacontracts.com.

§ 5.1.6

(Paragraphs deleted) Intentionally omitted.

§ 5.1.7 Other:

(Insert provisions for bonus, cost savings or other incentives, if any, that might result in a change to the Contract Sum.)

§ 5.2 Guaranteed Maximum Price

§ 5.2.1 The Contract Sum is guaranteed by the Contractor not to exceed Eight Million, Eight Hundred Thirty-Eight Thousand, Five Hundred Ninety-Three and 00/100 Dollars (\$8,838,593.00) Total, subject to additions and deductions by Change Order as provided in the Contract Documents. This maximum sum is referred to in the Contract Documents as the Guaranteed Maximum Price. Costs which would cause the Guaranteed Maximum Price to be exceeded shall be paid by the Contractor without reimbursement by the Owner. Attached as Exhibit A is the Schedule of Values allocating the entire amount of the Guaranteed Maximum Price..

To the extent the actual sum of the Cost of the Work and the Contractor's Fee are less than the final adjusted Guaranteed Maximum Price, such difference ("Savings") shall be shared between Owner and Contractor with Owner receiving 50% of the Savings and Contractor receiving 50% of the Savings. Contractor's portion of the Savings shall be paid at the time final payment is due pursuant to the terms of the Contract Documents.

§ 5.2.2 Alternates

§ 5.2.2.1 Alternates, if any, included in the Guaranteed Maximum Price: (Row deleted)

NONE

§ 5.2.2.2 Subject to the conditions noted below, the following alternates may be accepted by the Owner following execution of this Agreement. Upon acceptance, the Owner shall issue a Modification to this Agreement. (Insert below each alternate and the conditions that must be met for the Owner to accept the alternate.) (Row deleted)

NONE

§ 5.2.3 Allowances, if any, included in the Guaranteed Maximum Price: (*Identify each allowance.*) (Row deleted)

NONE

§ 5.2.4 Assumptions, if any, upon which the Guaranteed Maximum Price is based: (*Identify each assumption.*)

The following items are excluded from the Guaranteed Maximum Price:

Any alterations due to blower door testing results

Exit signs

Any parking lot work

New monument sign

Green roof replacement or modification

Existing elevator renovation

Work in existing elevators or shaft

Refinishing of kitchen cabinets

Init.

AIA Document A102 – 2017. Copyright © 1920, 1925, 1951, 1958, 1961, 1963, 1967, 1974, 1978, 1987, 1997, 2007 and 2017. All rights reserved. "The American Institute of Architects," "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are trademarks of The American Institute of Architects. This document was produced at 15:38:28 ET on 10/30/2023 under Order No.2114399858 which expires on 02/01/2024, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail docinfo@aiacontracts.com.

Painting of residential units
Replacement or painting of any doors
Hollow metal entrance door painting or refinishing
Hazardous materials (it is assumed that there are no hazardous materials)

- § 5.2.5 To the extent that the Contract Documents are anticipated to require further development, the Guaranteed Maximum Price includes the costs attributable to such further development consistent with the Contract Documents and reasonably inferable therefrom. Such further development does not include changes in scope, systems, kinds and quality of materials, finishes or equipment, all of which, if required, shall be incorporated by Change Order.
- § 5.2.6 The Owner shall authorize preparation of revisions to the Contract Documents that incorporate the agreed-upon assumptions contained in Section 5.2.4. The Owner shall promptly furnish such revised Contract Documents to the Contractor. The Contractor shall notify the Owner and Architect of any inconsistencies between the agreed-upon assumptions contained in Section 5.2.4 and the revised Contract Documents.

ARTICLE 6 CHANGES IN THE WORK

- § 6.1 The Owner may, without invalidating the Contract, order changes in the Work within the general scope of the Contract consisting of additions, deletions or other revisions. The Owner shall issue such changes in writing. The Architect may make minor changes in the Work as provided in Article 7 of the General Conditions and any other applicable terms. The Contractor shall be entitled to an equitable adjustment in the Contract Time as a result of changes in the Work.
- § 6.2 Adjustments to subcontracts awarded on the basis of a stipulated sum shall be determined in accordance with Article 7 of A201–2017, as they refer to "cost" and "fee," and not by Articles 5, 7 and 8 of this Agreement. Adjustments to subcontracts awarded with the Owner's prior written consent on the basis of cost plus a fee shall be calculated in accordance with the terms of those subcontracts.
- § 6.3 In calculating adjustments to subcontracts (except those awarded with the Owner's prior consent on the basis of cost plus a fee), the terms "cost" and "fee" as used in Section 7.3.3.3 of the General Conditions and the term "costs" as used in Section 7.3.4 of the General Conditions shall have the meanings assigned to them in the General Conditions and shall not be modified by Sections 5.1 and 5.2, Sections 7.1 through 7.7, and Section 7.8 of this Agreement. Adjustments to subcontracts awarded with the Owner's prior consent on the basis of cost plus a fee shall be calculated in accordance with the terms of those subcontracts.
- § 6.4 If no specific provision is made in Article 5 for adjustment of the Contractor's Fee in the case of changes in the Work, or if the extent of such changes is such, in the aggregate, that application of the adjustment provisions of Article 5 will cause substantial inequity to the Owner or Contractor, the Contractor's Fee shall be equitably adjusted on the same basis that was used to establish the Fee for the original Work, and the Guaranteed Maximum Price shall be adjusted accordingly.

ARTICLE 7 COSTS TO BE REIMBURSED

§ 7.1 Cost of the Work

- § 7.1.1 The term Cost of the Work shall mean costs necessarily incurred by the Contractor in the proper performance of the Work and shall include only the items set forth in this Article 7. Such costs shall be at rates not higher than the standard paid at the place of the Project except with prior consent of the Owner. The Cost of the Work shall include only the items set forth in Article 7.
- § 7.1.2 Where any cost is subject to the Owner's prior approval, the Contractor shall obtain such approval prior to incurring the cost.
- § 7.1.3 Intentionally omitted.
- § 7.2 Labor Costs
- § 7.2.1 Intentionally omitted.

§ 7.2.2 Wages or salaries of the Contractor's supervisory and administrative personnel when stationed at the site and performing Work, with the Owner's prior approval.

§ 7.2.2.1

(Paragraphs deleted) Intentionally omitted.

- § 7.2.3 Wages or salaries of the Contractor's supervisory or administrative personnel engaged at factories, workshops or on the road, in expediting the production or transportation of materials or equipment required for the Work, but only for that portion of their time required for the Work.
- § 7.2.4 Costs paid or incurred by the Contractor for taxes, insurance, contributions, assessments, and benefits and, for personnel not covered by collective bargaining agreements, customary benefits such as sick leave, medical and health benefits, holidays, vacations and pensions, provided such costs are based on wages and salaries included in the Cost of the Work under Sections 7.2.2 through 7.2.3.
- § 7.2.5 Bonuses, profit sharing, incentive compensation and any other discretionary payments paid to anyone hired by the Contractor or paid to any Subcontractor or vendor, with the Owner's prior approval..

§ 7.3 Subcontract Costs

Payments made by the Contractor to Subcontractors in accordance with the requirements of the subcontracts.

§ 7.4 Costs of Materials and Equipment Incorporated in the Completed Construction

- § 7.4.1 Costs, including transportation and storage at the site, of materials and equipment incorporated, or to be incorporated, in the completed construction.
- § 7.4.2 Costs of materials described in the preceding Section 7.4.1 in excess of those actually installed to allow for reasonable waste and spoilage. Unused excess materials, if any, shall become the Owner's property at the completion of the Work or, at the Owner's option, shall be sold by the Contractor. Any amounts realized from such sales shall be credited to the Owner as a deduction from the Cost of the Work.

§ 7.5 Costs of Other Materials and Equipment, Temporary Facilities and Related Items

- § 7.5.1 Costs of transportation, storage, installation, dismantling, maintenance, and removal of materials, supplies, temporary facilities, machinery, equipment and hand tools not customarily owned by construction workers that are provided by the Contractor at the site and fully consumed in the performance of the Work. Costs of materials, supplies, temporary facilities, machinery, equipment, and tools, that are not fully consumed, shall be based on the cost or value of the item at the time it is first used on the Project site less the value of the item when it is no longer used at the Project site. Costs for items not fully consumed by the Contractor shall mean fair market value.
- § 7.5.2 Rental charges for temporary facilities, machinery, equipment, and hand tools not customarily owned by construction workers that are provided by the Contractor at the site, and the costs of transportation, installation, dismantling, minor repairs, and removal. The total rental cost of any Contractor-owned equipment and quantities of equipment shall be subject to Owner's prior approval.
- § 7.5.3 Costs of removal of debris from the site of the Work and its proper and legal disposal.
- § 7.5.4 Costs of document reproductions, facsimile transmissions and long-distance telephone calls, postage and parcel delivery charges, telephone service at the site and reasonable petty cash expenses of the site office.
- § 7.5.5 That portion of the reasonable expenses of the Contractor's supervisory or administrative personnel incurred while traveling in discharge of duties connected with the Work.
- § 7.5.6 Costs of materials and equipment suitably stored off the site at a mutually acceptable location, subject to the Owner's prior approval.

§ 7.6 Miscellaneous Costs

- § 7.6.1 Premiums for that portion of insurance and bonds required by the Contract Documents that can be directly attributed to this Contract. Self-insurance for either full or partial amounts of the coverages required by the Contract Documents, with the Owner's prior approval
- § 7.6.1.1 Costs for self-insurance, for either full or partial amounts of the coverages required by the Contract Documents, with the Owner's prior approval. Self-insurance for either full or partial amounts of the coverages required by the Contract Documents, with the Owner's prior approval.
- § 7.6.1.2 Intentionally omitted.
- § 7.6.2 Sales, use, or similar taxes, imposed by a governmental authority, that are related to the Work and for which the Contractor is liable and actually pays.
- § 7.6.3 Fees and assessments for the building permit, and for other permits, licenses, and inspections, for which the Contractor is required by the Contract Documents to pay.
- § 7.6.4 Fees of laboratories for tests required by the Contract Documents; except those related to defective or nonconforming Work for which reimbursement is excluded under Article 13.4.3 of AIA Document A201–2017 or by other provisions of the Contract Documents, and which do not fall within the scope of Section 7.7.3.
- § 7.6.5 Royalties and license fees paid for the use of a particular design, process, or product, required by the Contract Documents and the cost of defending suits or claims for infringement of patent rights arising from such requirement of the Contract Documents provided Contractor has given Owner notice and opportunity to defend; payments made in accordance with legal judgments against the Contractor resulting from such suits or claims and payments of settlements made with the Owner's consent. However, such costs of legal defenses, judgments and settlements shall not be included in the calculation of the Contractor's Fee or subject to the Guaranteed Maximum Price. However, such royalties, fees and costs are not part of the Cost of Work if Contractor knows that the required design process, or product is, or could be, an infringement of a copyright or patent and the Contractor fails to promptly furnish such information to the Architect.
- § 7.6.5.1 Intentionally omitted.
- § 7.6.6 Costs for electronic equipment and software, directly related to the Work with the Owner's prior approval.
- § 7.6.7 Deposits lost for causes other than the Contractor's negligence or failure to fulfill a specific responsibility in the Contract Documents.
- § 7.6.8 Legal, mediation and arbitration costs, including attorneys' fees, other than those arising from disputes between the Owner and Contractor, reasonably incurred by the Contractor after the execution of this Agreement in the performance of the Work and with the Owner's prior approval, which shall not be unreasonably withheld.
- § 7.6.9 Subject to the Owner's prior written approval, expenses incurred in accordance with the Contractor's standard written personnel policy for relocation and temporary living allowances of the Contractor's personnel required for the Work.
- § 7.6.10 Costs of additional services or Work required in connection with repairs, removal or replacements necessitated by defects in materials and workmanship with the Owner's prior written approval, which shall not be unreasonably withheld.
- § 7.6.11 The cost of performing warranty work with the Owner's prior written approval, which shall not be unreasonably withheld.

§ 7.7 Other Costs and Emergencies

§ 7.7.1 Other costs incurred in the performance of the Work if, and to the extent, approved in advance in writing by the Owner.

- § 7.7.2 Costs incurred in taking action to prevent threatened damage, injury, or loss, in case of an emergency affecting the safety of persons and property, as provided in Article 10.4 of AIA Document A201–2017.
- § 7.7.3 Costs of repairing or correcting damaged or nonconforming Work executed by the Contractor, Subcontractors, or suppliers, provided that such damaged or nonconforming Work was not caused by the negligence of, or failure to fulfill a specific responsibility by, the Contractor, and only to the extent that the cost of repair or correction is not recovered by the Contractor from insurance, sureties, Subcontractors, suppliers, or others.

§ 7.8 Related Party Transactions

- § 7.8.1 For purposes of this Section 7.8, the term "related party" shall mean (1) a parent, subsidiary, affiliate, or other entity having common ownership or management with the Contractor; (2) any entity in which any stockholder in, or management employee of, the Contractor holds an equity interest in excess of ten percent in the aggregate; or (3) any entity which has the right to control the business or affairs of the Contractor. The term "Related Party" includes any member of the immediate family of any person identified above.
- § 7.8.2 If any of the costs to be reimbursed arise from a transaction between the Contractor and a related party, the Contractor shall notify the Owner of the specific nature of the contemplated transaction, including the identity of the related party and the anticipated cost to be incurred, before any such transaction is consummated or cost incurred. If the Owner, after such notification, authorizes the proposed transaction, then the cost incurred shall be included as a cost to be reimbursed, and the Contractor shall procure the Work, equipment, goods, or service, from the related party, as a Subcontractor, according to the terms of Sections 2.3.2.1, 2.3.2.2 and 2.3.2.3. If the Owner fails to authorize the transaction in writing, the Contractor shall procure the Work, equipment, goods, or service from some person or entity other than a related party according to the terms of Sections 2.3.2.1, 2.3.2.2 and 2.3.2.3.

ARTICLE 8 COSTS NOT TO BE REIMBURSED

- **§ 8.1** The Cost of the Work shall not include the items listed below:
 - Salaries and other compensation of the Contractor's personnel stationed at the Contractor's principal office or offices other than the site office, except as specifically provided in Section 7.2, or as may be provided in Article 15;
 - .2 Expenses of the Contractor's principal office and offices other than the site office;
 - .3 Contractor and Subcontractors overhead and general expenses, except as may be expressly included in Sections 7.1 to 7.7;
 - .4 The Contractor's and Subcontractor's capital expenses, including interest on the Contractor's and Subcontractor's capital employed for the Work;
 - .5 Except as provided in Section 7.7.3 of this Agreement, costs due to the acts or omissions of the Contractor or Subcontractors, or failure of the Contractor, Subcontractors and suppliers or anyone directly or indirectly employed by any of them or for whose acts any of them may be liable to fulfill an obligation set forth in the Contract Documents;
 - **.6** Any cost not specifically and expressly described in Sections 6.1 to 6.7;
 - .7 Costs, other than costs included in Change Orders approved by the Owner, that would cause the Guaranteed Maximum Price to be exceeded.
 - **.8** Intentionally omitted.
 - .9 Rental costs of machinery and equipment, except as specifically provided herein;
 - .10 Costs of removing condemned or rejected materials, resulting from the acts or omissions of the Contractor or Subcontractor;
 - .11 Costs in excess of the normal costs for permits and licenses to the extent such costs resulted from Contractor's or Subcontractor's failure to timely obtain permits and licenses required to be obtained by Contractor or Subcontractor;
 - .12 Intentionally deleted;
 - .13 Costs incurred as a result of any inspection or test which reveals nonconforming or defective work not performed in accordance with the Contract Documents;
 - .14 The cost of fines and penalties, including interest thereon, assessed against Contractor or Subcontractor by any federal, state or local government or quasigovernmental authorities;
 - Any cost which would cause the Guaranteed Maximum Price to be exceeded, in which case the Contractor shall pay all such costs and indemnify and hold harmless Owner therefrom;
 - .16 The cost of bonus and incentive plans for Contractor's executives.

ARTICLE 9 DISCOUNTS, REBATES AND REFUNDS

§ 9.1 Cash discounts obtained on payments made by the Contractor shall accrue to the Owner if (1) before making the payment, the Contractor included them in an Application for Payment and received payment from the Owner, or (2) the Owner has deposited funds with the Contractor with which to make payments; otherwise, cash discounts shall accrue to the Contractor. Trade discounts, rebates, refunds, and amounts received from sales of surplus materials and equipment shall accrue to the Owner, and the Contractor shall make provisions so that they can be obtained.

§ 9.2 Amounts that accrue to the Owner in accordance with the provisions of Section 9.1 shall be credited to the Owner as a deduction from the Cost of the Work.

ARTICLE 10 SUBCONTRACTS AND OTHER AGREEMENTS

§ 10.1 Those portions of the Work that the Contractor does not customarily perform with the Contractor's own personnel shall be performed under subcontracts or other appropriate agreements with the Contractor. The Owner may designate specific persons from whom, or entities from which, the Contractor shall obtain bids. The Contractor shall obtain bids from Subcontractors, and from suppliers of materials or equipment fabricated especially for the Work, who are qualified to perform that portion of the Work in accordance with the requirements of the Contract Documents. The Contractor shall deliver such bids to the Architect and Owner with an indication as to which bids the Contractor intends to accept. The Owner then has the right to review the Contractor's list of proposed subcontractors and suppliers in consultation with the Architect and, subject to Section 10.1.1, to object to any subcontractor or supplier. Any advice of the Architect, or approval or objection by the Owner, shall not relieve the Contractor of its responsibility to perform the Work in accordance with the Contract Documents. The Contractor shall not be required to contract with anyone to whom the Contractor has reasonable objection.

§ 10.1.1 When a specific subcontractor or supplier (1) is recommended to the Owner by the Contractor; (2) is qualified to perform that portion of the Work; and (3) has submitted a bid that conforms to the requirements of the Contract Documents without reservations or exceptions, but the Owner requires that another bid be accepted, then the Contractor may require that a Change Order be issued to adjust the Guaranteed Maximum Price by the difference between the bid of the person or entity recommended to the Owner by the Contractor and the amount of the subcontract or other agreement actually signed with the person or entity designated by the Owner.

§ 10.2 Subcontracts or other agreements shall conform to the applicable payment provisions of this Agreement, and shall not be awarded on the basis of cost plus a fee without the Owner's prior written approval. If a subcontract is awarded on the basis of cost plus a fee, the Contractor shall provide in the subcontract for the Owner to receive the same audit rights with regard to the Subcontractor as the Owner receives with regard to the Contractor in Article 11.

(Paragraphs deleted)

ARTICLE 11 ACCOUNTING RECORDS; RIGHT TO AUDIT

11.1 Contractor agrees to make, keep and maintain in accordance with the Owner's financial reporting requirements and generally accepted accounting principles and practices consistently applied, complete books, records, invoices and records of payments relating to the Work while it is being performed and for a period of seven (7) years following completion of the Work, or such longer time as required based on the Owner's funding requirements. Contractor shall further comply with any and all HUD reporting requirements.

11.2 Owner and Owner's lender shall have the right to examine and/or audit, either directly or through its authorized representatives or agents, during business hours and for a reasonable period of time, all nonfinancial records, correspondence, instructions, specifications, plans, drawings, receipts, manuals and memoranda insofar as they are pertinent to any question relating to an audit. This right of inspection shall not apply to trade secrets or other proprietary information properly designated or asserted as such, or to any documents protected in good faith by Attorney-Client or Attorney Work Product privileges provided such claim of Attorney-Client or Attorney Work Product privilege is not made for purposes of thwarting the intent of this Section.

11.3 Owner and Owner's lender shall be granted access to all of Contractor's internal audit information relating to this Agreement, including reports of corrective actions taken as a result of such audit, but access shall not be granted to (i) general financial records not related to this Agreement except as may be solely necessary to verify the audit; or (ii) documents or other items protected in good faith by Attorney-Client or Attorney Work Product privileges provided

such claim of Attorney-Client or Attorney Work Product privilege is not made for purposes of thwarting the intent of this Section.

11.4 Regardless of the time when the audit is conducted, in the seven year (or longer) audit period, if the audit determines overcharges by Contractor, Contractor shall repay Owner the amount of said overcharges.

ARTICLE 12 PAYMENTS

§ 12.1 Progress Payments

- § 12.1.1 Based upon Applications for Payment submitted to the Architect by the Contractor, and Certificates for Payment issued by the Architect, the Owner shall make progress payments on account of the Contract Sum, to the Contractor, as provided below and elsewhere in the Contract Documents.
- § 12.1.2 The period covered by each Application for Payment shall be one calendar month ending on the last day of the month.
- § 12.1.3 Payments are due and payable twenty-five (25) days from the date the Construction Manager's Pay Application is received by the Owner.
- § 12.1.4 Payments may be withheld from Construction Manager in accordance with Section 9.5.1 of the General Conditions.
- § 12.1.5 In exchange for final payment for all services and expenses under this Agreement, Construction Manager shall convey full and final lien waivers for its work under this Agreement and any of its consultants or subcontractors.
- § 12.1.5.1 Intentionally omitted.
- § 12.1.5.2 Intentionally omitted.
- § 12.1.5.3 Intentionally omitted.
- § 12.1.6 Provided that Owner has made payments of the amounts due to Construction Manager under this Agreement, Construction Manager shall keep the Project Site free from any liens by Construction Manager or any person or entity retained by Construction Manager, or their respective assigns. If, because of any act or omission of Construction Manager or anyone claiming through or under Construction Manager, any mechanic's or other lien, encumbrance or order is filed against Owner or the Project Site, Construction Manager shall, at Construction Manager's own cost and expense, cause the same to be canceled, bonded-over and discharged of record promptly after receiving notice thereof, and shall indemnify and hold Owner harmless from and against all costs, expenses, claims, losses or damages, including reasonable attorneys' fees, resulting therefrom or by reason thereof. No Uniform Commercial Code (UCC) or similar state statutory filings shall be made by Construction Manager, except upon express written consent of Owner
- § 12.1.7 Subject to other provisions of the Contract Documents, the amount of each progress payment shall be computed as follows:
 - .1 Take that portion of the Guaranteed Maximum Price properly allocable to completed Work as determined by multiplying the percentage of completion of each portion of the Work by the share of the Guaranteed Maximum Price allocated to that portion of the Work in the schedule of values. Pending final determination of cost to the Owner of changes in the Work, amounts not in dispute shall be included as provided in Section 7.3.9 of the General Conditions. Contractor shall provide accounting documentation as proof of costs incurred for that portion of the Work that the Contractor self-performs and for all material/equipment expenses. For subcontracts that were competitively bid, the Guaranteed Maximum Price shall be determined by the percentage of completion of each portion of the Work by the share of the Guaranteed Maximum Price allocated to that portion of the Work. For subcontracts that were not competitively bid, accounting documentation shall be provided as proof of costs incurred for that portion of the Work;

- Add that portion of the Guaranteed Maximum Price properly allocable to materials and equipment delivered and suitably stored at the site for subsequent incorporation in the Work, or if approved in advance by the Owner, suitably stored off the site at a location agreed upon in writing;
- .3 Add the Contractor's Fee, less retainage of five percent (5%) until project completion. The Contractor's Fee shall be computed upon the Cost of the Work at the rate stated in Section 5.1.1 or, if the Contractor's Fee is stated as a fixed sum in that Section, shall be an amount that bears the same ratio to that fixed-sum fee as the Cost of the Work bears to a reasonable estimate of the probable Cost of the Work upon its completion;
- 4 Subtract retainage equal to the amount in the agreement between the Contractor and Subcontractor;
- .5 Subtract the aggregate of previous payments made by the Owner;
- .6 Subtract the shortfall, if any, indicated by the Contractor in the documentation required by Section 12.1.4 to substantiate prior Applications for Payment, or resulting from errors subsequently discovered by the Owner's auditors in such documentation; and
- .7 Subtract amounts, if any, for which the Architect has withheld or nullified a Certificate for Payment as provided in Section 9.5 of the General Conditions.

§ 12.1.7.1 The amount of each progress payment shall first include:

- That portion of the Guaranteed Maximum Price properly allocable to completed Work as determined by multiplying the percentage of completion of each portion of the Work by the share of the Guaranteed Maximum Price allocated to that portion of the Work in the most recent schedule of values;
- .2 That portion of the Guaranteed Maximum Price properly allocable to materials and equipment delivered and suitably stored at the site for subsequent incorporation in the completed construction or, if approved in writing in advance by the Owner, suitably stored off the site at a location agreed upon in writing;
- .3 That portion of Construction Change Directives that the Architect determines, in the Architect's professional judgment, to be reasonably justified; and
- .4 The Contractor's Fee, computed upon the Cost of the Work described in the preceding Sections 12.1.7.1.1 and 12.1.7.1.2 at the rate stated in Section 5.1.1 or, if the Contractor's Fee is stated as a fixed sum in that Section, an amount that bears the same ratio to that fixed-sum fee as the Cost of the Work included in Sections 12.1.7.1.1 and 12.1.7.1.2 bears to a reasonable estimate of the probable Cost of the Work upon its completion.

§ 12.1.7.2 The amount of each progress payment shall then be reduced by:

- .1 The aggregate of any amounts previously paid by the Owner;
- .2 The amount, if any, for Work that remains uncorrected and for which the Architect has previously withheld a Certificate for Payment as provided in Article 9 of AIA Document A201–2017;
- .3 Any amount for which the Contractor does not intend to pay a Subcontractor or material supplier, unless the Work has been performed by others the Contractor intends to pay;
- .4 For Work performed or defects discovered since the last payment application, any amount for which the Architect may withhold payment, or nullify a Certificate of Payment in whole or in part, as provided in Article 9 of AIA Document A201–2017;
- .5 The shortfall, if any, indicated by the Contractor in the documentation required by Section 12.1.4 to substantiate prior Applications for Payment, or resulting from errors subsequently discovered by the Owner's auditors in such documentation; and
- **.6** Retainage withheld pursuant to Section 12.1.8.

§ 12.1.8 Except with the Owner's prior approval, payments to Subcontractors shall be subject to retention of not less than five percent (5%) up to the time the Work is fifty percent (50%) complete and then zero percent (0%) thereafter which amount shall be reduced consistent with the reduction in Section 12.1.7.4. The Owner and Contractor shall agree upon a mutually acceptable procedure for review and approval of payments and retention for Subcontracts and, when necessary, shall adjust the percentage of retainage held on Subcontracts to a mutually agreeable amount.

(Paragraphs deleted)

§ 12.1.9 Except with the Owner's prior approval, the Contractor shall not make advance payments to suppliers for materials or equipment which have not been delivered and stored at the site.

docinfo@aiacontracts.com.

User Notes: (729100899)

- § 12.1.10 In taking action on the Contractor's Applications for Payment, the Architect shall be entitled to rely on the accuracy and completeness of the information furnished by the Contractor and shall not be deemed to represent that the Architect has made a detailed examination, audit or arithmetic verification of the documentation submitted in accordance with Section 7.1.4 or other supporting data; that the Architect has made exhaustive or continuous on-site inspections; or that the Architect has made examinations to ascertain how or for what purposes the Contractor has used amounts previously paid on account of the Contract. Such examinations, audits and verifications, if required by the Owner, will be performed by the Owner's auditors acting in the sole interest of the Owner.
- **§ 12.1.11** No payment under this Agreement, either final or progress, shall release the Contractor from any obligation under the Contract Documents.
- § 12.1.12 The Owner reserves the right to withhold, on account of subsequently discovered evidence, subsequent observations or otherwise, the whole or any part of any payment which would otherwise be due under this Agreement, to such extent as (a) the Owner may have any claim or offset against the Contractor and/or (b) the Owner, in its reasonable opinion, has a loss or shall consider necessary to protect itself from loss because of any of the items enumerated in Section 9.5.1 of the General Conditions.

§ 12.2 Final Payment

- § 12.2.1 Final payment, constituting the entire unpaid balance of the Contract Sum, shall be made by the Owner to the Contractor when
 - .1 the Contractor has fully performed the Contract, except for the Contractor's responsibility to correct Work as provided in Article 12.2 of AIA Document A201–2017, and to satisfy other requirements, if any, which extend beyond final payment;
 - .2 the Contractor has submitted a final accounting for the Cost of the Work and a final Application for Payment, which documents have been reviewed by Owner's representative; and
 - .3 a final Certificate for Payment has been issued by the Architect.

The Owner's final payment to the Contractor shall be made no later than 30 days after the issuance of the Architect's final Certificate for Payment, or as follows:

- § 12.2.2 The amount of the final payment shall be calculated as follows:
 - .1 Take the sum of the Cost of the Work substantiated by the Contractor's final accounting and the Contractor's Fee, but not more than the Guaranteed Maximum Price.
 - 2 Subtract amounts, if any, for which the Architect withholds, in whole or in part, a final Certificate for Payment as provided in Section 9.5 of the General Conditions or other provisions of the Contract Documents.
 - .3 Subtract the aggregate of previous payments made by the Owner.
- § 12.2.2.1 Intentionally omitted.
- § 12.2.2.2 Intentionally omitted.
- § 12.2.2.3 The Owner's auditors will review and report in writing on the Contractor's final accounting within 30 days after delivery of the final accounting to the Architect by the Contractor. Based upon such Cost of the Work as the Owner's auditors report to be substantiated by the Contractor's final accounting, and provided the other conditions of Section 7.2.1 have been met, the Architect will, within seven days after receipt of the written report of the Owner's auditors, either issue to the Owner a final Certificate for Payment with a copy to the Contractor, or notify the Contractor and Owner in writing of the Architect's reasons for withholding a certificate as provided in Section 9.5.1 of the General Conditions. The time periods stated in this Section supersede those stated in Section 9.4.1 of the General Conditions. The Architect is not responsible for verifying the accuracy of the Contractor's final accounting.
- § 12.2.3 Intentionally omitted.

§ 12.2.4 If, subsequent to final payment, and at the Owner's request, the Contractor incurs costs, described in Section Article 7 and not excluded by Article 8, to correct defective or nonconforming Work, the Owner shall reimburse the Contractor for such costs, and the Contractor's Fee applicable thereto, on the same basis as if such costs had been incurred prior to final payment, but not in excess of the Guaranteed Maximum Price. If adjustments to the Contract Sum are provided for in Section 5.1.7, the amount of those adjustments shall be recalculated and appropriate credit given to the Owner in determining the net amount to be paid by the Owner to the Contractor.

§ 12.3 Interest

(Paragraphs deleted)

Amounts unpaid thirty (30) days from the Owner's receipt of the Construction Manager's approved Pay Application shall bear interest at twelve percent (12%) per annum

ARTICLE 13 DISPUTE RESOLUTION

§ 13.1

(Paragraphs deleted)

Any Claim between the Owner and Contractor shall be resolved in accordance with the provisions set forth in this Article 9 and Article 15 of the General Conditions

§ 13.2

(Paragraphs deleted) Intentionally Omitted.

ARTICLE 14 TERMINATION OR SUSPENSION

§ 14.1 Termination

§ 14.1.1 The Contract may be terminated by the Owner or the Contractor as provided in Article 14 of AIA Document A201–2017...

(Paragraphs deleted)

§ 14.1.1.2 If the Owner terminates the Contract for convenience, the amount payable to the Contractor pursuant to Section 14.4.3 of the General Conditions shall not exceed the amount the Contractor would otherwise have received pursuant to Sections 10.1.2 and 10.1.3 of this Agreement.

§ 14.1.1.3 If the Contractor terminates the Contract the amount payable to the Contractor under Section 14.1.3 of the General Conditions shall not exceed the amount the Contractor would otherwise have received under Sections 10.1.2 and 10.1.3 above.

§ 14.2 Suspension

The Work may be suspended by the Owner as provided in Section 14.3 of the General Conditions; in such case, the Guaranteed Maximum Price and Contract Time shall be increased as provided in Section 14.3.2 of the General Conditions, except that the term "Contract Sum" in that Section shall be understood to mean the Cost of the Work and the term "profit" shall be understood to mean the Contractor's Fee as described in Sections 5.1 and 5.3.5 of this Agreement.

ARTICLE 15 MISCELLANEOUS PROVISIONS

§ 15.1 Where reference is made in this Agreement to a provision of AIA Document A201–2017 or another Contract Document, the reference refers to that provision as amended or supplemented by other provisions of the Contract Documents.

§ 15.2 The Owner's representative:

(Paragraphs deleted) The Owner's Representative shall have express authority to bind the Owner with respect to all matters requiring the Owner's approval or authorization. This representative shall have the authority to make decisions on behalf of the Owner concerning estimates and schedules, construction budgets, and changes in the Work, and shall render such decisions promptly and furnish information expeditiously, so as to avoid unreasonable delay in the services or Work of the Contractor. Except as otherwise provided in Section 4.2.1 of the General Conditions, the Architect does not have such authority.

§ 15.3

(Paragraphs deleted) Intentionally omitted.

§ 15.4 Neither the Owner's nor the Contractor's representative shall be changed without ten days' prior notice to the other party.

§ 15.5 Insurance and Bonds

§ 15.5.1 The Owner and the Contractor shall purchase and maintain insurance as set forth in Article 11 of the General Conditions.

§ 15.5.2 Intentionally omitted.

§ 15.6

(Paragraphs deleted) Intentionally omitted.

§ 15.7 Other provisions:

- § 15.1.7.1 The Owner shall furnish tests, inspections and reports required by law and as otherwise agreed to by the parties, such as structural, mechanical, and chemical tests, tests for air and water pollution, and tests for hazardous materials. The Contractor shall, if requested by Owner, assist Owner in furnishing such tests, inspections and reports as the Owner may deem necessary.
- § 15.1.7.2 During the Construction Phase, the Owner shall furnish information or services reasonably required of the Owner by the Contract Documents with reasonable promptness. The Owner shall also furnish any other information or services under the Owner's control and relevant to the Contractor's performance of the Work with reasonable promptness after receiving the Contractor's written request for such information or services.
- § 15.1.7.3 The Contractor, when such services are reasonably required by the scope of the Project and are requested and approved by Owner, shall furnish services of geotechnical engineers, which may include but are not limited to test borings, test pits, determinations of soil bearing values, percolation tests, evaluations of hazardous materials, seismic evaluation, ground corrosion tests and resistivity tests, including necessary operations for anticipating subsoil conditions, with written reports and appropriate recommendations.
- § 15.1.7.4 The Owner shall furnish information or services reasonably required of the Owner by the Contract Documents with reasonable promptness. The Owner shall also furnish any other information or services under the Owner's control and relevant to the Contractor's performance of the Work with reasonable promptness after receiving the Contractor's written request for such information or services.
- § 15.1.7.5 Legal Requirements. The Owner shall furnish all legal, insurance and accounting services, including auditing services, that may be reasonably necessary at any time for the Project to meet the Owner's needs and interests.
- § 15.1.7.6 Architect. The Owner shall retain an Architect to provide services, duties and responsibilities, including any additional services requested by the Contractor that are necessary for the Construction Phase services under this Agreement. The Owner shall provide the Contractor a copy of the executed agreement between the Owner and the Architect, and any further modifications to the agreement.

Governing Law

User Notes:

§ 15.1.7.7 The Contract shall be governed by the law of the State of Wisconsin.

§ 15.1.7.8 Assignment

The Owner and Contractor, respectively, bind themselves, their agents, successors, assigns and legal representatives to this Agreement. Neither the Owner nor the Contractor shall assign this Agreement without the written consent of the other, except that the Owner may assign this Agreement to a lender providing financing for the Project if the lender agrees to assume the Owner's rights and obligations under this Agreement. Except as provided in Section 13.2.1 of the General Conditions, neither party to the Contract shall assign the Contract as a whole without written consent of the

other. If either party attempts to make such an assignment without such consent, that party shall nevertheless remain legally responsible for all obligations under the Contract.

SCOPE OF THE AGREEMENT

§ 15.1.7.9 This Agreement represents the entire and integrated agreement between the Owner and the Contractor and supersedes all prior negotiations, representations or agreements, either written or oral. This Agreement may be amended only by written instrument signed by both Owner and Contractor.

ARTICLE 16 ENUMERATION OF CONTRACT DOCUMENTS

§ 16.1 This Agreement is comprised of the following documents:

- .1 AIA Document A102TM–2017, Standard Form of Agreement Between Owner and Contractor where the basis of payment is the Cost of the Work Plus a Fee with a Guaranteed Maximum Price
- .2 Intentionally omitted
- 3 AIA Document A201TM–2017, General Conditions of the Contract for Construction, as modified
- 4 AIA Document E203™–2013, Building Information Modeling and Digital Data Exhibit, dated as indicated below:

(Insert the date of the E203-2013 incorporated into this Agreement.)

NOT USED

.5 Intentionally omitted

(Row deleted)

(Paragraph deleted)

.6 Intentionally omitted

(Row deleted)

(Paragraph deleted)

.7 Intentionally omitted

(Row deleted)

Portions of Addenda relating to bidding or proposal requirements are not part of the Contract Documents unless the bidding or proposal requirements are also enumerated in this Article 16.

.8 Other Exhibits:

(Check all boxes that apply.)

(Paragraphs deleted)

[X] Supplementary and other Conditions of the Contract:

Document	Title	Date	Pages
HUD-5370	General Conditions for	01/2014	19
	Construction Contracts –	exp11/30/2023	
	Public Housing Programs		

.9 Other documents, if any, listed below:

(List here any additional documents that are intended to form part of the Contract Documents. AIA Document A201–2017 provides that the advertisement or invitation to bid, Instructions to Bidders, sample forms, the Contractor's bid or proposal, portions of Addenda relating to bidding or proposal requirements, and other information furnished by the Owner in anticipation of receiving bids or proposals, are not part of the Contract Documents unless enumerated in this Agreement. Any such documents should be listed here only if intended to be part of the Contract Documents.)

Init.

AIA Document A102 – 2017. Copyright © 1920, 1925, 1951, 1958, 1961, 1963, 1967, 1974, 1978, 1987, 1997, 2007 and 2017. All rights reserved. "The American Institute of Architects," "American Institute of Architects," "Alla," the AIA Logo, and "AIA Contract Documents" are trademarks of The American Institute of Architects. This document was produced at 15:38:28 ET on 10/30/2023 under Order No.2114399858 which expires on 02/01/2024, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail

docinfo@aiacontracts.com.

User Notes: (729100899)

- A Specifications Project Manual Table of Contents, dated xx/xx/xxxx
- B Drawings Cover Sheet and Index (General and Elevators), dated xx/xx/xxxx
- C Scope Narrative, dated xx/xx/xxxx
- D Itemized Statement of the Guaranteed Maximum Price, dated xx/xx/xxxx
- E Construction Schedule, dated xx/xx/xxxx



This Agreement entered into as of the day and year first written above.

HIGHLAND GARDENS LLC

A Wisconsin Limited Liability Company

By: Highland Gardens Development LLC A Wisconsin Limited Liability Company, its Manager

By: Housing Authority of the City of Milwaukee A Wisconsin Public Body Corporate and Politic,

its Manager

OWNER (Signature)

Willie L. Hines, Jr.

Secretary-Executive Director

(Printed name and title)

TRAVAUX INC.

CONTRACTOR (Signature)

Willie L. Hines, Jr.

President

(Printed name and title)

Additions and Deletions Report for

AIA® Document A102® – 2017

This Additions and Deletions Report, as defined on page 1 of the associated document, reproduces below all text the author has added to the standard form AIA document in order to complete it, as well as any text the author may have added to or deleted from the original AIA text. Added text is shown underlined. Deleted text is indicated with a horizontal line through the original AIA text.

Note: This Additions and Deletions Report is provided for information purposes only and is not incorporated into or constitute any part of the associated AIA document. This Additions and Deletions Report and its associated document were generated simultaneously by AIA software at 15:38:28 ET on 10/30/2023.

PAGE 1

CONTRACT 23-006

AGREEMENT made <u>effective</u> as of the <u>9th</u> day of <u>November</u> in the year <u>2023</u>

...

Highland Gardens LLC c/o Housing Authority of the City of Milwaukee 809 North Broadway, 3rd Floor Milwaukee, WI 53202

...

Travaux Inc.

c/o Housing Authority of the City of Milwaukee 809 North Broadway, 3rd Floor Milwaukee, WI 53202

...

Highland Gardens

Located at 1818 West Juneau Avenue, Milwaukee, WI 53233
Rehabilitation of 114-unit apartment building and associated site improvements

...

(Name, legal status, address and other information)

Galbraith Carnahan Architects LLC 6404 West North Avenue

Wauwatosa, WI 53213

The Contractor's Designated Representative:

(Name, address and other information)

Warren B. Jones
Travaux Inc.
401 East Ogden Avenue
Milwaukee, WI 53202

The Architect's Designated Representative:

(Name, address and other information)

Additions and Deletions Report for AIA Document A102 – 2017. Copyright © 1920, 1925, 1951, 1958, 1961, 1963, 1967, 1974, 1978, 1987, 1997, 2007 and 2017. All rights reserved. "The American Institute of Architects," "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are trademarks of The American Institute of Architects. This document was produced at 15:38:28 ET on 10/30/2023 under Order No.2114399858 which expires on 02/01/2024, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail docinfo@aiacontracts.com.

Joseph Galbraith
Calbraith Carnahan Architects LLC
6404 West North Avenue
Wauwatosa, WI 53213

The Owner's Designated Representative: (Name, address and other information)

Willie L. Hines, Jr.

Highland Gardens LLC
c/o Housing Authority of the City of Milwaukee
809 North Broadway, 3rd Floor
Milwaukee, WI 53202

<u>The Owner and Contractor agree as follows.</u> **PAGE 3**

16 ENUMERATION OF CONTRACT DOCUMENTS

EXHIBIT A INSURANCE AND BONDS

...

The Contract Documents consist of this Agreement, Conditions of the Contract (General, Supplementary, and other Conditions), HUD 5370 General Conditions of the Contract for Construction Drawings, Specifications, Addenda issued prior to execution of this Agreement, other documents listed in this Agreement and Modifications issued after execution of this Agreement, all of which form the Contract, and are as fully a part of the Contract as if attached to this Agreement or repeated herein. The Contract represents the entire and integrated agreement between the parties hereto and supersedes prior negotiations, representations, or agreements, either written or oral. If anything in the other Contract Documents, other than a Modification, is inconsistent with this Agreement, this Agreement shall govern. The terms of AIA Document A201TM–2017, General Conditions of the Contract for Construction (sometimes referred to herein as the "General Conditions") and the terms of HUD 5370 General Conditions of the Contract for Construction shall be construed and interpreted to the full extent possible to give effect to all such terms. In the event of any conflict between the terms, the terms of HUD 5370 General Conditions of the Contract for Construction shall govern and control. This Agreement may be amended only by written instrument signed by both the Owner and Contractor. An enumeration of the Contract Documents, other than a Modification, appears in Article 16.

...

The Contractor shall fully execute the Work described in the Contract Documents, except as specifically indicated in the Contract Documents to be the responsibility of others. § 2.1 The Contractor shall perform the services described in this Article. The Contractor has identified a representative above authorized to act on behalf of the Contractor with respect to the Project.

§ 2.1.2 General

§ 2.1.2.1 For purposes of Section 8.1.2 of the General Conditions, the date of means of the Work shall mean the date of commencement of the Construction Phase. The Construction Phase shall commence upon the Owner's issuance of a Notice to Proceed.

§ 2.1.2.2 The Contractor shall have the control over and charge of and shall be responsible for construction means, methods, techniques, sequences and procedures, and the safety precautions and programs in connection with the Work. The Contractor shall be responsible for its and its Subcontractor's failure to carry out the Work in accordance with the Contract Documents. The Contractor shall coordinate the sequence of the construction and the responsibilities of the Subcontractors and be responsible for the acts and omissions of the Subcontractors and their agents and employees.

Additions and Deletions Report for AIA Document A102 – 2017. Copyright © 1920, 1925, 1951, 1958, 1961, 1963, 1967, 1974, 1978, 1987, 1997, 2007 and 2017>. All rights reserved. "The American Institute of Architects," "American Institute of Architects," "AllA," the AIA Logo, and "AIA Contract Documents" are trademarks of The American Institute of Architects. This draft was produced at 15:38:28 ET on 10/30/2023 under Order No.2114399858 which expires on 02/01/2024, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail docinfo@aiacontracts.com.

- §2.1.2.3 The Contractor shall monitor the Work of the Contractors, Subcontractors and suppliers, and coordinate their Work with the activities and responsibilities of the Architect, the Contractor, and the Owner to complete the Project in accordance with the Owner's objectives of cost, time and quality.
- § 2.1.2.4 The Contractor shall maintain a competent staff at the Project Site to coordinate and provide general direction of the Work and facilitate progress of the Contractors, Subcontractors and suppliers on the Project.
- § 2.1.2.5 The Contractor shall establish procedures for coordination among the Architect, Contractors, Subcontractors and suppliers and the Contractor with respect to all aspects of the Project and shall implement such procedures.
- § 2.2.1.6 The Contractor shall determine the adequacy of the Contractors', Subcontractors' and suppliers' personnel and equipment and the availability of materials and supplies to meet the schedule. The Contractor shall take appropriate action when requirements of a contract are not being met.
- § 2.1.2.7 If needed, the Contractor shall provide temporary construction office facility for use by Owner, Architect, and Contractor at the Project site.
- § 2.1.2.8 The Contractor shall manage the mobilization activities with the Subcontractors.
- § 2.1.2.9 The Contractor shall arrange for the delivery, storage, protection, and security of Owner-purchased materials, systems, and equipment that (a) are obtained or acquired through the Contractor or the Contractor's purchasing entity, or (b) are items to be installed by the Contractor.
- § 2.1.2.10 The Contractor shall provide assignment of responsibilities for temporary project facilities and equipment, materials and services for common use of the Contractor and the Subcontractors.

§ 2.2.1 Administration

- § 2.2.1.1 Those portions of the Work that the Contractor does not customarily perform with the Contractor's own personnel shall be performed under subcontracts or by other appropriate agreements with the Contractor. The Owner may designate specific persons from whom, or entities from which, the Contractor shall obtain bids. The Contractor shall obtain bids from Subcontractors and from suppliers of materials or equipment fabricated especially for the Work and shall deliver such bids to the Owner and Architect. The Owner shall then determine, with the advice of the Contractor and the Architect, which bids will be accepted. The Contractor shall not be required to contract with anyone to whom the Contractor has reasonable objection. The Contractor shall be responsible for the proper and complete performance of the Work in accordance with the terms and provisions of the Contract Documents, whether performed by the Contractor or by Subcontractors.
- § 2.2.1.2 Subcontracts or other agreements shall conform to the applicable payment provisions of this Agreement, and shall not be awarded on the basis of cost plus a fee without the prior consent of the Owner. If the Subcontract is awarded on a cost-plus a fee basis, the Contractor shall provide in the Subcontract for the Owner to receive the same audit rights with regard to the Subcontractor as the Owner receives with regard to the Contractor in Article 12.
- § 2.2.1.3 Intentionally omitted...
- § 2.2.1.5 The Contractor shall schedule and conduct meetings at which the Owner, Architect, Contractor and appropriate Subcontractors can discuss such matters as procedures, progress, coordination, scheduling, and status of the Work. The Contractor shall prepare and promptly distribute minutes to the Owner and Architect.
- § 2.2.1.6 Upon the execution of this Agreement the Contractor shall prepare and submit to the Owner and Architect a construction schedule for the Work and submittal schedule in accordance with Section 3.10 of the General Conditions.
- § 2.2.1.7 The Contractor shall record the progress of the Project. On a quarterly basis, or otherwise as agreed to by the Owner, the Contractor shall submit written progress reports to the Owner and Architect, showing percentages of completion and other information required by the Owner. The Contractor shall also keep, and make available to the Owner and Architect, a daily log containing a record for each day of weather, portions of the Work in progress, Work

accomplished, number of workers on site, Subcontractors working on site, identification of equipment on site, problems that might affect progress of the work, accidents, injuries, and other information required by the Owner.

- § 2.2.1.8 The Contractor shall develop a system of cost control for the Work, including regular monitoring of actual costs for activities in progress and estimates for uncompleted tasks and proposed changes. The Contractor shall identify variances between actual and estimated costs and report the variances to the Owner and Architect and shall provide this information in its quarterly reports to the Owner and Architect, in accordance with Section 2.2.1.7above.
- § 2.2.1.9 The Contractor shall receive, review and approve all certificates of insurance, lien waivers and similar information and shall forward copies, if requested, to the Architect and the Owner.
- § 2.2.1.10 The Contractor shall develop and implement a system for the preparation, review and processing of change orders, and shall recommend necessary or desirable changes to the Owner.
- § 2.2.1.11 In collaboration with the Architect, the Contractor shall establish and implement procedures for expediting the processing and approval of shop drawings and samples.
- § 2.2.1.12 The Contractor shall work with the Architect to prepare a punch-list of defective or incomplete work, and shall coordinate completion of punch-list items.
- § 2.2.1.13 The Contractor shall secure and transmit to the Owner all required warranties and lien waivers and turn over to the Owner all keys, manuals, record drawings and maintenance stocks. The Contractor shall coordinate the collection of the Contractor and Subcontractor field-annotated record drawings and shall mark up a consolidated print for the Architect's use.
- § 2.2.1.14 The Contractor shall prepare and deliver to the Owner a final accounting for all costs incurred. PAGE 5

The Contractor accepts the relationship of trust and confidence established with the Owner by this Agreement and covenants with the Owner to cooperate with the Architect and exercise the Contractor's skill and judgment in furthering the interests of the Owner; to furnish efficient business administration construction administration, management services and supervision; to furnish at all times an adequate supply of workers and materials; and to use the Contractor's best efforts to perform the Work in an expeditious and economical manner consistent with the Owner's interests. The Owner agrees to furnish and approve, in a timely manner, information required by the Contractor and to make payments to the Contractor in accordance with the requirements of the Contract Documents. All parties shall endeavor to promote harmony and cooperation among the Owner, Architect, Contractor and other persons and entities employed by the Owner for the Project.

[X] A date set forth in a notice to proceed issued by the Owner.

PAGE 6

[X] By the following date: <u>December 31, 2025</u>

NONE

User Notes:

ARTICLE 5 CONTRACT SUM

ARTICLE 5 COMPENSATION FOR CONSTRUCTION PHASE SERVICES

§ 5.1 The For the Contractor's performance of the Work as described in Article 2, the Owner shall pay the Contractor the Contract Sum in current funds for the Contractor's performance of the Contract. The Contract Sum is the Cost of the Work as defined in Article 7 plus the Contractor's Fee.

A lump sum of Four Hundred Forty-One Thousand, Nine Hundred Twenty-Nine and 00/100 Dollars (\$441,929.00), exclusive of Architect and Engineer fees, Owner's FF&E, performance/payment bond, and builder's risk insurance. Owner FF&E if incorporated into the budget as an allowance would be considered a portion of the Cost of Work for purposes of calculating the fee.

- § 5.1.2 The method of adjustment of the Contractor's Fee for changes in the Work: Intentionally omitted.
- § 5.1.3 Limitations, if any, on a Subcontractor's overhead and profit for increases in the cost of its portion of the Work:

Intentionally omitted.

§ 5.1.4 Rental rates for Contractor-owned equipment shall not exceed percent (-%) of the standard rental rate paid at the place of the Project-eighty-five percent (85 %) of the rental rates listed in the latest edition of the Associated Equipment Distributors "green book" (a/k/a AED Green Book) of nationally averaged rates. Charges for equipment rented from third parties shall be the actual rental rates, but shall in no even exceed local market rates.

NONE

§ 5.1.6 Liquidated damages, if any:

(Insert terms and conditions for liquidated damages, if any.)

Intentionally omitted.

PAGE 7

§ 5.2.1 The Contract Sum is guaranteed by the Contractor not to exceed (\$___), Eight Million, Eight Hundred Thirty-Eight Thousand, Five Hundred Ninety-Three and 00/100 Dollars (\$8,838,593.00) Total, subject to additions and deductions by Change Order as provided in the Contract Documents. This maximum sum is referred to in the Contract Documents as the Guaranteed Maximum Price. Costs which would cause the Guaranteed Maximum Price to be exceeded shall be paid by the Contractor without reimbursement by the Owner. Attached as Exhibit A is the Schedule of Values allocating the entire amount of the Guaranteed Maximum Price..

To the extent the actual sum of the Cost of the Work and the Contractor's Fee are less than the final adjusted Guaranteed Maximum Price, such difference ("Savings") shall be shared between Owner and Contractor with Owner receiving 50% of the Savings and Contractor receiving 50% of the Savings. Contractor's portion of the Savings shall be paid at the time final payment is due pursuant to the terms of the Contract Documents.

ltem	Price	
NONE		
Item	Price	Conditions for Acceptance
NONE		

Additions and Deletions Report for AIA Document A102 – 2017. Copyright © 1920, 1925, 1951, 1958, 1961, 1963, 1967, 1974, 1978, 1987, 1997, 2007 and 2017>. All rights reserved. "The American Institute of Architects," "AMERICAN Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are trademarks of The American Institute of Architects. This draft was produced at 15:38:28 ET on 10/30/2023 under Order No.2114399858 which expires on 02/01/2024, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail docinfo@aiacontracts.com. **User Notes:**

Item Price

NONE

..

The following items are excluded from the Guaranteed Maximum Price:

Any alterations due to blower door testing results

Exit signs

Any parking lot work

New monument sign

Green roof replacement or modification

Existing elevator renovation

Work in existing elevators or shaft

Refinishing of kitchen cabinets

Painting of residential units

Replacement or painting of any doors

Hollow metal entrance door painting or refinishing

Hazardous materials (it is assumed that there are no hazardous materials)

PAGE 8

§ 6.1 Adjustments to the Guaranteed Maximum Price on account of changes in the Work may be determined by any of the methods listed in Article 7 of AIA Document A201TM 2017, General Conditions of the Contract for Construction. The Owner may, without invalidating the Contract, order changes in the Work within the general scope of the Contract consisting of additions, deletions or other revisions. The Owner shall issue such changes in writing. The Architect may make minor changes in the Work as provided in Article 7 of the General Conditions and any other applicable terms. The Contractor shall be entitled to an equitable adjustment in the Contract Time as a result of changes in the Work.

...

§ 6.3 In calculating adjustments to the Guaranteed Maximum Price, the terms "cost" and "costs" as used in Article 7 of AIA Document A201 2017 shall mean the Cost of the Work as defined in Article 7 of this Agreement and the term "fee" shall mean the Contractor's Fee as defined in Section 5.1.1 of this Agreement.subcontracts (except those awarded with the Owner's prior consent on the basis of cost plus a fee), the terms "cost" and "fee" as used in Section 7.3.3.3 of the General Conditions and the term "costs" as used in Section 7.3.4 of the General Conditions shall have the meanings assigned to them in the General Conditions and shall not be modified by Sections 5.1 and 5.2, Sections 7.1 through 7.7, and Section 7.8 of this Agreement. Adjustments to subcontracts awarded with the Owner's prior consent on the basis of cost plus a fee shall be calculated in accordance with the terms of those subcontracts.

...

- § 7.1.1 The term Cost of the Work shall mean costs necessarily incurred by the Contractor in the proper performance of the Work. Work and shall include only the items set forth in this Article 7. Such costs shall be at rates not higher than the standard paid at the place of the Project except with prior consent of the Owner. The Cost of the Work shall include only the items set forth in this Article 7. Article 7.
- § 7.1.2 Where, pursuant to the Contract Documents, Where any cost is subject to the Owner's prior approval, the Contractor shall obtain such approval in writing prior to incurring the cost.
- § 7.1.3 Costs shall be at rates not higher than the standard paid at the place of the Project, except with prior approval of the Owner. Intentionally omitted.

•••

- § 7.2.1 Wages or salaries of construction workers directly employed by the Contractor to perform the construction of the Work at the site or, with the Owner's prior approval, at off site workshops. Intentionally omitted.

 PAGE 9
- § 7.2.2.1 Wages or salaries of the Contractor's supervisory and administrative personnel when performing Work and stationed at a location other than the site, but only for that portion of time required for the Work, and limited to the personnel and activities listed below:

(Identify the personnel, type of activity and, if applicable, any agreed upon percentage of time to be devoted to the Work.)

Intentionally omitted.

- § 7.2.3 Wages or salaries of the Contractor's supervisory or administrative personnel engaged at factories, workshops or while traveling, on the road, in expediting the production or transportation of materials or equipment required for the Work, but only for that portion of their time required for the Work.
- § 7.2.4 Costs paid or incurred by the Contractor, as required by law or collective bargaining agreements, Contractor for taxes, insurance, contributions, assessments, and benefits and, for personnel not covered by collective bargaining agreements, customary benefits such as sick leave, medical and health benefits, holidays, vacations and pensions, provided such costs are based on wages and salaries included in the Cost of the Work under Sections 7.2.1-7.2.2 through 7.2.3.
- § 7.2.5 If agreed rates for labor costs, in lieu of actual costs, are provided in this Agreement, the rates shall remain unchanged throughout the duration of this Agreement, unless the parties execute a Modification. Bonuses, profit sharing, incentive compensation and any other discretionary payments paid to anyone hired by the Contractor or paid to any Subcontractor or vendor, with the Owner's prior approval..

...

Payments made by the Contractor to Subcontractors in accordance with the requirements of the subcontracts and this Agreement.subcontracts.

...

§ 7.5.2 Rental charges for temporary facilities, machinery, equipment, and hand tools not customarily owned by construction workers that are provided by the Contractor at the site, and the costs of transportation, installation, dismantling, minor repairs, and removal of such temporary facilities, machinery, equipment, and hand tools. Rates and quantities of equipment owned by the Contractor, or a related party as defined in Section 7.8, shall be subject to the Owner's prior approval. The total rental cost of any such equipment may not exceed the purchase price of any comparable item.removal. The total rental cost of any Contractor-owned equipment and quantities of equipment shall be subject to Owner's prior approval.

...

- § 7.5.4 Costs of the Contractor's site office, including general office equipment and supplies.document reproductions, facsimile transmissions and long-distance telephone calls, postage and parcel delivery charges, telephone service at the site and reasonable petty cash expenses of the site office.
- § 7.5.5 Costs of materials and equipment suitably stored off the site at a mutually acceptable location, subject to the Owner's prior approval. That portion of the reasonable expenses of the Contractor's supervisory or administrative personnel incurred while traveling in discharge of duties connected with the Work.
- § 7.5.6 Costs of materials and equipment suitably stored off the site at a mutually acceptable location, subject to the Owner's prior approval.

PAGE 10

- § 7.6.1 Premiums for that portion of insurance and bonds required by the Contract Documents that can be directly attributed to this Contract. Self-insurance for either full or partial amounts of the coverages required by the Contract Documents, with the Owner's prior approval
- § 7.6.1.1 Costs for self-insurance, for either full or partial amounts of the coverages required by the Contract Documents, with the Owner's prior approval. Self-insurance for either full or partial amounts of the coverages required by the Contract Documents, with the Owner's prior approval.
- § 7.6.1.2 Costs for insurance through a captive insurer owned or controlled by the Contractor, with the Owner's prior approval. Intentionally omitted.
- § 7.6.2 Sales, use, or similar taxes, imposed by a governmental authority, that are related to the Work and for which the Contractor is liable.liable and actually pays.

- § 7.6.4 Fees of laboratories for tests required by the Contract Documents; except those related to defective or nonconforming Work for which reimbursement is excluded under Article 13-13.4.3 of AIA Document A201-2017 or by other provisions of the Contract Documents, and which do not fall within the scope of Section 7.7.3.
- § 7.6.5 Royalties and license fees paid for the use of a particular design, process, or product, required by the Contract Documents and the cost of defending suits or claims for infringement of patent rights arising from such requirement of the Contract Documents provided Contractor has given Owner notice and opportunity to defend; payments made in accordance with legal judgments against the Contractor resulting from such suits or claims and payments of settlements made with the Owner's consent. However, such costs of legal defenses, judgments and settlements shall not be included in the calculation of the Contractor's Fee or subject to the Guaranteed Maximum Price. However, such royalties, fees and costs are not part of the Cost of Work if Contractor knows that the required design process, or product is, or could be, an infringement of a copyright or patent and the Contractor fails to promptly furnish such information to the Architect.
- § 7.6.5.1 The cost of defending suits or claims for infringement of patent rights arising from requirements of the Contract Documents, payments made in accordance with legal judgments against the Contractor resulting from such suits or claims, and payments of settlements made with the Owner's consent, unless the Contractor had reason to believe that the required design, process or product was an infringement of a copyright or a patent, and the Contractor failed to promptly furnish such information to the Architect as required by Article 3 of AIA Document A201 2017. The costs of legal defenses, judgments, and settlements, shall not be included in the Cost of the Work used to calculate the Contractor's Fee or subject to the Guaranteed Maximum Price. Intentionally omitted.
- § 7.6.6 Costs for communications services, electronic equipment, electronic equipment and software, directly related to the Work and located at the site, with the Owner's prior approval.
- § 7.6.7 Costs of document reproductions and delivery charges. Deposits lost for causes other than the Contractor's negligence or failure to fulfill a specific responsibility in the Contract Documents.
- § 7.6.8 Deposits lost for causes other than the Contractor's negligence or failure to fulfill a specific responsibility in the Contract Documents. Legal, mediation and arbitration costs, including attorneys' fees, other than those arising from disputes between the Owner and Contractor, reasonably incurred by the Contractor after the execution of this Agreement in the performance of the Work and with the Owner's prior approval, which shall not be unreasonably withheld.
- § 7.6.9 Legal, mediation and arbitration costs, including attorneys' fees, other than those arising from disputes between the Owner and Contractor, reasonably incurred by the Contractor after the execution of this Agreement in the performance of the Work and with the Owner's prior approval, which shall not be unreasonably withheld. Subject to the Owner's prior written approval, expenses incurred in accordance with the Contractor's standard written personnel policy for relocation and temporary living allowances of the Contractor's personnel required for the Work.

- § 7.6.10 Expenses incurred in accordance with the Contractor's standard written personnel policy for relocation and temporary living allowances of the Contractor's personnel required for the Work, with the Owner's prior approval. Costs of additional services or Work required in connection with repairs, removal or replacements necessitated by defects in materials and workmanship with the Owner's prior written approval, which shall not be unreasonably withheld.
- § 7.6.11 That portion of the reasonable expenses of the Contractor's supervisory or administrative personnel incurred while traveling in discharge of duties connected with the Work. The cost of performing warranty work with the Owner's prior written approval, which shall not be unreasonably withheld.

..

- § 7.7.1 Other costs incurred in the performance of the Work, with the Owner's prior approval. Work if, and to the extent, approved in advance in writing by the Owner.
- § 7.7.2 Costs incurred in taking action to prevent threatened damage, injury, or loss, in case of an emergency affecting the safety of persons and property, as provided in Article 10-10.4 of AIA Document A201-2017.

 PAGE 11
- § 7.8.1 For purposes of this Section 7.8, the term "related party" shall mean (1) a parent, subsidiary, affiliate, or other entity having common ownership of, or sharing common management with, or management with the Contractor; (2) any entity in which any stockholder in, or management employee of, the Contractor holds an equity interest in excess of ten percent in the aggregate; or (3) any entity which has the right to control the business or affairs of the Contractor; or (4) any person, or Contractor. The term "Related Party" includes any member of the immediate family of any person, who has the right to control the business or affairs of the Contractor. person identified above.
- § 7.8.2 If any of the costs to be reimbursed arise from a transaction between the Contractor and a related party, the Contractor shall notify the Owner of the specific nature of the contemplated transaction, including the identity of the related party and the anticipated cost to be incurred, before any such transaction is consummated or cost incurred. If the Owner, after such notification, authorizes the proposed transaction in writing, transaction, then the cost incurred shall be included as a cost to be reimbursed, and the Contractor shall procure the Work, equipment, goods, or service, from the related party, as a Subcontractor, according to the terms of Article 10. Sections 2.3.2.1, 2.3.2.2 and 2.3.2.3. If the Owner fails to authorize the transaction in writing, the Contractor shall procure the Work, equipment, goods, or service from some person or entity other than a related party according to the terms of Article 10. Sections 2.3.2.1, 2.3.2.2 and 2.3.2.3.

..

- .2 Bonuses, profit sharing, incentive compensation, and any other discretionary payments, paid to anyone hired by the Contractor or paid to any Subcontractor or vendor, unless the Owner has provided prior approval;
- 3 Expenses of the Contractor's principal office and offices other than the site office;
- 4 Overhead 3Contractor and Subcontractors overhead and general expenses, except as may be expressly included in Article 7; Sections 7.1 to 7.7;
- 5 _____.4 The Contractor's <u>and Subcontractor's</u> capital expenses, including interest on the Contractor's <u>and Subcontractor's</u> capital employed for the Work;
- .5 Except as provided in Section 7.7.3 of this Agreement, costs due to the negligence of, or failure to fulfill a specific responsibility of the Contract by, the Contractor, Subcontractors, and suppliers, acts or omissions of the Contractor or Subcontractors, or failure of the Contractor, Subcontractors and suppliers or anyone directly or indirectly employed by any of them or for whose acts any of them may be liable; liable to fulfill an obligation set forth in the Contract Documents;
- .7 ____.6 Any cost not specifically and expressly described in Article 7; and Sections 6.1 to 6.7;
- .7 Costs, other than costs included in Change Orders approved by the Owner, that would cause the Guaranteed Maximum Price to be exceeded.
- 8 Intentionally omitted.
- .9 Rental costs of machinery and equipment, except as specifically provided herein;
- .10 Costs of removing condemned or rejected materials, resulting from the acts or omissions of the

- Contractor or Subcontractor;
- .11 Costs in excess of the normal costs for permits and licenses to the extent such costs resulted from Contractor's or Subcontractor's failure to timely obtain permits and licenses required to be obtained by Contractor or Subcontractor;
- .12 Intentionally deleted;
- Costs incurred as a result of any inspection or test which reveals nonconforming or defective work not performed in accordance with the Contract Documents;
- The cost of fines and penalties, including interest thereon, assessed against Contractor or Subcontractor by any federal, state or local government or quasigovernmental authorities;
- .15 Any cost which would cause the Guaranteed Maximum Price to be exceeded, in which case the Contractor shall pay all such costs and indemnify and hold harmless Owner therefrom;
- The cost of bonus and incentive plans for Contractor's executives.

PAGE 12

§ 9.1 Cash discounts obtained on payments made by the Contractor shall accrue to the Owner if (1) before making the payment, the Contractor included the amount to be paid, less such discount, them in an Application for Payment and received payment from the Owner, or (2) the Owner has deposited funds with the Contractor with which to make payments; otherwise, cash discounts shall accrue to the Contractor. Trade discounts, rebates, refunds, and amounts received from sales of surplus materials and equipment shall accrue to the Owner, and the Contractor shall make provisions so that they can be obtained.

ARTICLE 11 ACCOUNTING RECORDS

The Contractor shall keep full and detailed records and accounts related to the Cost of the Work, and exercise such controls, as may be necessary for proper financial management under this Contract and to substantiate all costs incurred. The accounting and control systems shall be satisfactory to the Owner. The Owner and the Owner's auditors shall, during regular business hours and upon reasonable notice, be afforded access to, and shall be permitted to audit and copy, the Contractor's records and accounts, including complete documentation supporting accounting entries, books, job cost reports, correspondence, instructions, drawings, receipts, subcontracts, Subcontractor's proposals, Subcontractor's invoices, purchase orders, vouchers, memoranda, and other data relating to this Contract. The Contractor shall preserve these records for a period of three years after final payment, or for such longer period as may be required by law.

ARTICLE 11 ACCOUNTING RECORDS; RIGHT TO AUDIT

- 11.1 Contractor agrees to make, keep and maintain in accordance with the Owner's financial reporting requirements and generally accepted accounting principles and practices consistently applied, complete books, records, invoices and records of payments relating to the Work while it is being performed and for a period of seven (7) years following completion of the Work, or such longer time as required based on the Owner's funding requirements. Contractor shall further comply with any and all HUD reporting requirements.
- 11.2 Owner and Owner's lender shall have the right to examine and/or audit, either directly or through its authorized representatives or agents, during business hours and for a reasonable period of time, all nonfinancial records, correspondence, instructions, specifications, plans, drawings, receipts, manuals and memoranda insofar as they are pertinent to any question relating to an audit. This right of inspection shall not apply to trade secrets or other proprietary information properly designated or asserted as such, or to any documents protected in good faith by Attorney-Client or Attorney Work Product privileges provided such claim of Attorney-Client or Attorney Work Product privilege is not made for purposes of thwarting the intent of this Section.
- 11.3 Owner and Owner's lender shall be granted access to all of Contractor's internal audit information relating to this Agreement, including reports of corrective actions taken as a result of such audit, but access shall not be granted to (i) general financial records not related to this Agreement except as may be solely necessary to verify the audit; or (ii) documents or other items protected in good faith by Attorney-Client or Attorney Work Product privileges provided such claim of Attorney-Client or Attorney Work Product privilege is not made for purposes of thwarting the intent of this Section.

Additions and Deletions Report for AIA Document A102 – 2017. Copyright © 1920, 1925, 1951, 1958, 1961, 1963, 1967, 1974, 1978, 1987, 1997, 2007 and 2017>. All rights reserved. "The American Institute of Architects," "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are trademarks of The American Institute of Architects. This draft was produced at 15:38:28 ET on 10/30/2023 under Order No.2114399858 which expires on 02/01/2024, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail docinfo@aiacontracts.com. **User Notes:**

11.4 Regardless of the time when the audit is conducted, in the seven year (or longer) audit period, if the audit determines overcharges by Contractor, Contractor shall repay Owner the amount of said overcharges.

PAGE 13

§ 12.1.2 The period covered by each Application for Payment shall be one calendar month ending on the last day of the month, or as follows:

month.

- § 12.1.3 Provided that an Application for Payment is received by the Architect not later than the day of a month, the Owner shall make payment of the amount certified to the Contractor not later than the day of the month. If an Application for Payment is received by the Architect after the application date fixed above, payment of the amount certified shall be made by the Owner not later than () days after the Architect receives the Application for Payment. (Federal, state or local laws may require payment within a certain period of time.) Payments are due and payable twenty-five (25) days from the date the Construction Manager's Pay Application is received by the Owner.
- § 12.1.4 With each Application for Payment, the Contractor shall submit payrolls, petty cash accounts, receipted invoices or invoices with check vouchers attached, and any other evidence required by the Owner or Architect to demonstrate that payments already made by the Contractor on account of the Cost of the Work equal or exceed progress payments already received by the Contractor plus payrolls for the period covered by the present Application for Payment, less that portion of the progress payments attributable to the Contractor's Fee. Payments may be withheld from Construction Manager in accordance with Section 9.5.1 of the General Conditions.
- § 12.1.5 Each Application for Payment shall be based on the most recent schedule of values submitted by the Contractor in accordance with the Contract Documents. The schedule of values shall allocate the entire Guaranteed Maximum Price among: (1) the various portions of the Work; (2) any contingency for costs that are included in the Guaranteed Maximum Price but not otherwise allocated to another line item or included in a Change Order; and (3) the Contractor's Fee. In exchange for final payment for all services and expenses under this Agreement, Construction Manager shall convey full and final lien waivers for its work under this Agreement and any of its consultants or subcontractors.
- § 12.1.5.1 The schedule of values shall be prepared in such form and supported by such data to substantiate its accuracy as the Architect may require. The schedule of values shall be used as a basis for reviewing the Contractor's Applications for Payment. Intentionally omitted.
- § 12.1.5.2 The allocation of the Guaranteed Maximum Price under this Section 12.1.5 shall not constitute a separate guaranteed maximum price for the Cost of the Work of each individual line item in the schedule of values. Intentionally omitted.
- § 12.1.5.3 When the Contractor allocates costs from a contingency to another line item in the schedule of values, the Contractor shall submit supporting documentation to the Architect. Intentionally omitted.
- § 12.1.6 Applications for Payment shall show the percentage of completion of each portion of the Work as of the end of the period covered by the Application for Payment. The percentage of completion shall be the lesser of (1) the percentage of that portion of the Work which has actually been completed; or (2) the percentage obtained by dividing (a) the expense that has actually been incurred by the Contractor on account of that portion of the Work and for which the Contractor has made payment or intends to make payment prior to the next Application for Payment, by (b) the share of the Guaranteed Maximum Price allocated to that portion of the Work in the schedule of values. Provided that Owner has made payments of the amounts due to Construction Manager under this Agreement, Construction Manager shall keep the Project Site free from any liens by Construction Manager or any person or entity retained by Construction Manager, or their respective assigns. If, because of any act or omission of Construction Manager or anyone claiming through or under Construction Manager, any mechanic's or other lien, encumbrance or order is filed against Owner or the Project Site, Construction Manager shall, at Construction Manager's own cost and expense, cause the same to be canceled, bonded-over and discharged of record promptly after receiving notice thereof, and shall indemnify and hold Owner harmless from and against all costs, expenses, claims, losses or damages, including

reasonable attorneys' fees, resulting therefrom or by reason thereof. No Uniform Commercial Code (UCC) or similar state statutory filings shall be made by Construction Manager, except upon express written consent of Owner.

- § 12.1.7 In accordance with AIA Document A201 2017 and subject Subject to other provisions of the Contract Documents, the amount of each progress payment shall be computed as follows:
 - Take that portion of the Guaranteed Maximum Price properly allocable to completed Work as determined by multiplying the percentage of completion of each portion of the Work by the share of the Guaranteed Maximum Price allocated to that portion of the Work in the schedule of values. Pending final determination of cost to the Owner of changes in the Work, amounts not in dispute shall be included as provided in Section 7.3.9 of the General Conditions. Contractor shall provide accounting documentation as proof of costs incurred for that portion of the Work that the Contractor self-performs and for all material/equipment expenses. For subcontracts that were competitively bid, the Guaranteed Maximum Price shall be determined by the percentage of completion of each portion of the Work by the share of the Guaranteed Maximum Price allocated to that portion of the Work. For subcontracts that were not competitively bid, accounting documentation shall be provided as proof of costs incurred for that portion of the Work;
 - Add that portion of the Guaranteed Maximum Price properly allocable to materials and equipment delivered and suitably stored at the site for subsequent incorporation in the Work, or if approved in advance by the Owner, suitably stored off the site at a location agreed upon in writing;
 - .3 Add the Contractor's Fee, less retainage of five percent (5%) until project completion. The

 Contractor's Fee shall be computed upon the Cost of the Work at the rate stated in Section 5.1.1 or, if
 the Contractor's Fee is stated as a fixed sum in that Section, shall be an amount that bears the same ratio
 to that fixed-sum fee as the Cost of the Work bears to a reasonable estimate of the probable Cost of the
 Work upon its completion;
 - 4 Subtract retainage equal to the amount in the agreement between the Contractor and Subcontractor;
 - .5 Subtract the aggregate of previous payments made by the Owner;
 - Subtract the shortfall, if any, indicated by the Contractor in the documentation required by Section 12.1.4 to substantiate prior Applications for Payment, or resulting from errors subsequently discovered by the Owner's auditors in such documentation; and
 - .7 Subtract amounts, if any, for which the Architect has withheld or nullified a Certificate for Payment as provided in Section 9.5 of the General Conditions.

PAGE 14

§ 12.1.8 Retainage Except with the Owner's prior approval, payments to Subcontractors shall be subject to retention of not less than five percent (5%) up to the time the Work is fifty percent (50%) complete and then zero percent (0%) thereafter which amount shall be reduced consistent with the reduction in Section 12.1.7.4. The Owner and Contractor shall agree upon a mutually acceptable procedure for review and approval of payments and retention for Subcontracts and, when necessary, shall adjust the percentage of retainage held on Subcontracts to a mutually agreeable amount.

§ 12.1.8.1 For each progress payment made prior to Substantial Completion of the Work, the Owner may withhold the following amount, as retainage, from the payment otherwise due:

(Insert a percentage or amount to be withheld as retainage from each Application for Payment. The amount of retainage may be limited by governing law.)

§ 12.1.8.1.1 The following items are not subject to retainage:

(Insert any items not subject to the withholding of retainage, such as general conditions, insurance, etc.)

§ 12.1.8.2 Reduction or limitation of retainage, if any, shall be as follows:

(If the retainage established in Section 12.1.8.1 is to be modified prior to Substantial Completion of the entire Work, insert provisions for such modification.)

§ 12.1.8.3 Except as set forth in this Section 12.1.8.3, upon Substantial Completion of the Work, the Contractor may submit an Application for Payment that includes the retainage withheld from prior Applications for Payment pursuant to this Section 12.1.8. The Application for Payment submitted at Substantial Completion shall not include retainage as

(Insert any other conditions for release of retainage, such as upon completion of the Owner's audit and reconciliation, upon Substantial Completion.)

- § 12.1.9 If final completion of the Work is materially delayed through no fault of the Contractor, the Owner shall pay the Contractor any additional amounts in accordance with Article 9 of AIA Document A201 2017. Except with the Owner's prior approval, the Contractor shall not make advance payments to suppliers for materials or equipment which have not been delivered and stored at the site.
- § 12.1.10 Except with the Owner's prior written approval, the Contractor shall not make advance payments to suppliers for materials or equipment which have not been delivered and suitably stored at the site. In taking action on the Contractor's Applications for Payment, the Architect shall be entitled to rely on the accuracy and completeness of the information furnished by the Contractor and shall not be deemed to represent that the Architect has made a detailed examination, audit or arithmetic verification of the documentation submitted in accordance with Section 7.1.4 or other supporting data; that the Architect has made exhaustive or continuous on-site inspections; or that the Architect has made examinations to ascertain how or for what purposes the Contractor has used amounts previously paid on account of the Contract. Such examinations, audits and verifications, if required by the Owner, will be performed by the Owner's auditors acting in the sole interest of the Owner.
- § 12.1.11 The Owner and the Contractor shall agree upon a mutually acceptable procedure for review and approval of payments to Subcontractors, and the percentage of retainage held on Subcontracts, and the Contractor shall execute subcontracts in accordance with those agreements. No payment under this Agreement, either final or progress, shall release the Contractor from any obligation under the Contract Documents.
- § 12.1.12 In taking action on the Contractor's Applications for Payment the Architect shall be entitled to rely on the accuracy and completeness of the information furnished by the Contractor, and such action shall not be deemed to be a representation that (1) the Architect has made a detailed examination, audit, or arithmetic verification, of the documentation submitted in accordance with Section 12.1.4 or other supporting data; (2) that the Architect has made exhaustive or continuous on site inspections; or (3) that the Architect has made examinations to ascertain how or for what purposes the Contractor has used amounts previously paid on account of the Contract. Such examinations, audits, and verifications, if required by the Owner, will be performed by the Owner's auditors acting in the sole interest of the Owner. The Owner reserves the right to withhold, on account of subsequently discovered evidence, subsequent observations or otherwise, the whole or any part of any payment which would otherwise be due under this Agreement, to such extent as (a) the Owner may have any claim or offset against the Contractor and/or (b) the Owner, in its reasonable opinion, has a loss or shall consider necessary to protect itself from loss because of any of the items enumerated in Section 9.5.1 of the General Conditions.

PAGE 15

- .1 the Contractor has fully performed the Contract, except for the Contractor's responsibility to correct Work as provided in Article 12-12.2 of AIA Document A201-2017, and to satisfy other requirements, if any, which extend beyond final payment;
- the Contractor has submitted a final accounting for the Cost of the Work and a final Application for Payment; Payment, which documents have been reviewed by Owner's representative; and
- .3 a final Certificate for Payment has been issued by the Architect in accordance with Section 12.2.2.the Architect.

The Owner's final payment to the Contractor shall be made no later than 30 days after the issuance of the Architect's final Certificate for Payment, or as follows:

- § 12.2.2 Within 30 days of the Owner's receipt of the Contractor's final accounting for the Cost of the Work, the Owner shall conduct an audit of the Cost of the Work or notify the Architect that it will not conduct an audit. The amount of the final payment shall be calculated as follows:
 - Take the sum of the Cost of the Work substantiated by the Contractor's final accounting and the Contractor's Fee, but not more than the Guaranteed Maximum Price.
 - Subtract amounts, if any, for which the Architect withholds, in whole or in part, a final Certificate for Payment as provided in Section 9.5 of the General Conditions or other provisions of the Contract Documents.
 - Subtract the aggregate of previous payments made by the Owner.
- § 12.2.2.1 If the Owner conducts an audit of the Cost of the Work, the Owner shall, within 10 days after completion of the audit, submit a written report based upon the auditors' findings to the Architect. Intentionally omitted.
- § 12.2.2.2 Within seven days after receipt of the written report described in Section 12.2.2.1, or receipt of notice that the Owner will not conduct an audit, and provided that the other conditions of Section 12.2.1 have been met, the Architect will either issue to the Owner a final Certificate for Payment with a copy to the Contractor, or notify the Contractor and Owner in writing of the Architect's reasons for withholding a certificate as provided in Article 9 of AIA Document A201 2017. The time periods stated in this Section 12.2.2 supersede those stated in Article 9 of AIA Document A201 2017. The Architect is not responsible for verifying the accuracy of the Contractor's final accounting. Intentionally omitted.
- § 12.2.2.3 If the Owner's auditors' report concludes that the Cost of the Work, as-The Owner's auditors will review and report in writing on the Contractor's final accounting within 30 days after delivery of the final accounting to the Architect by the Contractor. Based upon such Cost of the Work as the Owner's auditors report to be substantiated by the Contractor's final accounting, is less than claimed by the Contractor, the Contractor shall be entitled to request mediation of the disputed amount without seeking an initial decision pursuant to Article 15 of AIA Document A201 2017. A request for mediation shall be made by the Contractor within 30 days after the Contractor's receipt of a copy of the Architect's final Certificate for Payment. Failure to request mediation within this 30-day period shall result in the substantiated amount reported by the Owner's auditors becoming binding on the Contractor. Pending a final resolution of the disputed amount, the Owner shall pay the Contractor the amount certified in the Architect's final Certificate for Payment and provided the other conditions of Section 7.2.1 have been met, the Architect will, within seven days after receipt of the written report of the Owner's auditors, either issue to the Owner a final Certificate for Payment with a copy to the Contractor, or notify the Contractor and Owner in writing of the Architect's reasons for withholding a certificate as provided in Section 9.5.1 of the General Conditions. The time periods stated in this Section supersede those stated in Section 9.4.1 of the General Conditions. The Architect is not responsible for verifying the accuracy of the Contractor's final accounting.
- § 12.2.3 The Owner's final payment to the Contractor shall be made no later than 30 days after the issuance of the Architect's final Certificate for Payment, or as follows:

Intentionally omitted.

§ 12.2.4 If, subsequent to final payment, and at the Owner's request, the Contractor incurs costs, described in Section Article 7 and not excluded by Article 8, to correct defective or nonconforming Work, the Owner shall reimburse the Contractor for such costs, and the Contractor's Fee applicable thereto, on the same basis as if such costs had been incurred prior to final payment, but not in excess of the Guaranteed Maximum Price. If adjustments to the Contract Sum are provided for in Section 5.1.7, the amount of those adjustments shall be recalculated, taking into account any reimbursements made pursuant to this Section 12.2.4-recalculated and appropriate credit given to the Owner in determining the net amount to be paid by the Owner to the Contractor.

PAGE 16

Payments due and unpaid under the Contract shall bear interest from the date payment is due at the rate stated below, or in the absence thereof, at the legal rate prevailing from time to time at the place where the Project is located. (Insert rate of interest agreed upon, if any.)

Application shall bear interest at twelve percent (12%) per annum

§ 13.1 Initial Decision Maker

The Architect will serve as Initial Decision Maker pursuant to Article 15 of AIA Document A201 2017, unless the parties appoint below another individual, not a party to the Agreement, to serve as Initial Decision Maker. (If the parties mutually agree, insert the name, address and other contact information of the Initial Decision Maker, if other than the Architect.)

Any Claim between the Owner and Contractor shall be resolved in accordance with the provisions set forth in this Article 9 and Article 15 of the General Conditions

§ 13.2 Binding Dispute Resolution

For any Claim subject to, but not resolved by mediation pursuant to Article 15 of AIA Document A201 2017, the method of binding dispute resolution shall be as follows: (Check the appropriate box.)

[-]	Arbitration pursuant to Section 15 of AIA Document A201 2017
[-]-	Litigation in a court of competent jurisdiction
	Other (Specify)

If the Owner and Contractor do not select a method of binding dispute resolution, or do not subsequently agree in writing to a binding dispute resolution method other than litigation, Claims will be resolved by litigation in a court of competent jurisdiction.Intentionally Omitted.

§ 14.1.1 The Contract may be terminated by the Owner or the Contractor as provided in Article 14 of AIA Document A201 2017.A201-2017...

§ 14.1.2 Termination by the Owner for Cause

§ 14.1.2.1 If the Owner terminates the Contract for cause as provided in Article 14 of AIA Document A201 2017, the amount, if any, to be paid to the Contractor under Article 14 of AIA Document A201 2017 shall not cause the Guaranteed Maximum Price to be exceeded, nor shall it exceed an amount calculated as follows:

- Take the Cost of the Work incurred by the Contractor to the date of termination;
- .2 Add the Contractor's Fee, computed upon the Cost of the Work to the date of termination at the rate stated in Section 5.1.1 or, if the Contractor's Fee is stated as a fixed sum in that Section, an amount that bears the same ratio to that fixed sum Fee as the Cost of the Work at the time of termination bears to a reasonable estimate of the probable Cost of the Work upon its completion;
- Subtract the aggregate of previous payments made by the Owner; and
- .4 Subtract the costs and damages incurred, or to be incurred, by the Owner under Article 14 of AIA Document A201-2017.

§ 14.1.2.2 The Owner shall also pay the Contractor fair compensation, either by purchase or rental at the election of the Owner, for any equipment owned by the Contractor that the Owner elects to retain and that is not otherwise included in the Cost of the Work under Section 14.1.2.1.1. To the extent that the Owner elects to take legal assignment of subcontracts and purchase orders (including rental agreements), the Contractor shall, as a condition of receiving the payments referred to in this Article 14, execute and deliver all such papers and take all such steps, including the legal assignment of such subcontracts and other contractual rights of the Contractor, as the Owner may require for the purpose of fully vesting in the Owner the rights and benefits of the Contractor under such subcontracts or purchase orders.

§ 14.1.3 Termination by the Owner for Convenience

If the Owner terminates the Contract for convenience in accordance with Article 14 of AIA Document A201 2017, then the Owner shall pay the Contractor a termination fee as follows:

(Insert the amount of or method for determining the fee, if any, payable to the Contractor following a termination for the Owner's convenience.)

§ 14.1.1.2 If the Owner terminates the Contract for convenience, the amount payable to the Contractor pursuant to Section 14.4.3 of the General Conditions shall not exceed the amount the Contractor would otherwise have received pursuant to Sections 10.1.2 and 10.1.3 of this Agreement.

§ 14.1.1.3 If the Contractor terminates the Contract the amount payable to the Contractor under Section 14.1.3 of the General Conditions shall not exceed the amount the Contractor would otherwise have received under Sections 10.1.2 and 10.1.3 above.

The Work may be suspended by the Owner as provided in Article 14 of AIA Document A201 2017; Section 14.3 of the General Conditions; in such case, the Guaranteed Maximum Price and Contract Time shall be increased as provided in Article 14 of AIA Document A201 2017, except that the term Section 14.3.2 of the General Conditions, except that the term "Contract Sum" in that Section shall be understood to mean the Cost of the Work and the term "profit" shall be understood to mean the Contractor's Fee as described in Article 5 and Section 6.4 Sections 5.1 and 5.3.5 of this Agreement.

(Name, address, email address and other information)

The Owner's Representative shall have express authority to bind the Owner with respect to all matters requiring the Owner's approval or authorization. This representative shall have the authority to make decisions on behalf of the Owner concerning estimates and schedules, construction budgets, and changes in the Work, and shall render such decisions promptly and furnish information expeditiously, so as to avoid unreasonable delay in the services or Work of the Contractor. Except as otherwise provided in Section 4.2.1 of the General Conditions, the Architect does not have such authority.

§ 15.3 The Contractor's representative:

(Name, address, email address and other information)

Intentionally omitted.

PAGE 17

§ 15.5.1 The Owner and the Contractor shall purchase and maintain insurance as set forth in AIA Document A102TM 2017, Standard Form of Agreement Between Owner and Contractor where the basis of payment is the Cost of the Work Plus a Fee with a Guaranteed Maximum Price, Exhibit A, Insurance and Bonds, and elsewhere in the Contract Documents. Article 11 of the General Conditions.

- § 15.5.2 The Contractor shall provide bonds as set forth in AIA Document A102TM 2017 Exhibit A, and elsewhere in the Contract Documents. Intentionally omitted.
- § 15.6 Notice in electronic format, pursuant to Article 1 of AIA Document A201–2017, may be given in accordance with AIA Document E203TM–2013, Building Information Modeling and Digital Data Exhibit, if completed, or as otherwise set forth below:

(If other than in accordance with AIA Document E203 2013, insert requirements for delivering notice in electronic format such as name, title, and email address of the recipient and whether and how the system will be required to generate a read receipt for the transmission.)

Intentionally omitted.

...

- § 15.1.7.1 The Owner shall furnish tests, inspections and reports required by law and as otherwise agreed to by the parties, such as structural, mechanical, and chemical tests, tests for air and water pollution, and tests for hazardous materials. The Contractor shall, if requested by Owner, assist Owner in furnishing such tests, inspections and reports as the Owner may deem necessary.
- § 15.1.7.2 During the Construction Phase, the Owner shall furnish information or services reasonably required of the Owner by the Contract Documents with reasonable promptness. The Owner shall also furnish any other information or services under the Owner's control and relevant to the Contractor's performance of the Work with reasonable promptness after receiving the Contractor's written request for such information or services.
- § 15.1.7.3 The Contractor, when such services are reasonably required by the scope of the Project and are requested and approved by Owner, shall furnish services of geotechnical engineers, which may include but are not limited to test borings, test pits, determinations of soil bearing values, percolation tests, evaluations of hazardous materials, seismic evaluation, ground corrosion tests and resistivity tests, including necessary operations for anticipating subsoil conditions, with written reports and appropriate recommendations.
- § 15.1.7.4 The Owner shall furnish information or services reasonably required of the Owner by the Contract

 Documents with reasonable promptness. The Owner shall also furnish any other information or services under the

 Owner's control and relevant to the Contractor's performance of the Work with reasonable promptness after receiving
 the Contractor's written request for such information or services.
- § 15.1.7.5 Legal Requirements. The Owner shall furnish all legal, insurance and accounting services, including auditing services, that may be reasonably necessary at any time for the Project to meet the Owner's needs and interests.
- § 15.1.7.6 Architect. The Owner shall retain an Architect to provide services, duties and responsibilities, including any additional services requested by the Contractor that are necessary for the Construction Phase services under this Agreement. The Owner shall provide the Contractor a copy of the executed agreement between the Owner and the Architect, and any further modifications to the agreement.

Governing Law

User Notes:

§ 15.1.7.7 The Contract shall be governed by the law of the State of Wisconsin.

§ 15.1.7.8 Assignment

The Owner and Contractor, respectively, bind themselves, their agents, successors, assigns and legal representatives to this Agreement. Neither the Owner nor the Contractor shall assign this Agreement without the written consent of the other, except that the Owner may assign this Agreement to a lender providing financing for the Project if the lender agrees to assume the Owner's rights and obligations under this Agreement. Except as provided in Section 13.2.1 of the General Conditions, neither party to the Contract shall assign the Contract as a whole without written consent of the other. If either party attempts to make such an assignment without such consent, that party shall nevertheless remain legally responsible for all obligations under the Contract.

SCOPE OF THE AGREEMENT

§ 15.1.7.9 This Agreement represents the entire and integrated agreement between the Owner and the Contractor and supersedes all prior negotiations, representations or agreements, either written or oral. This Agreement may be amended only by written instrument signed by both Owner and Contractor.

PAGE 18

.1 .2 .3	AIA Document A102 TM —2017, Stand the basis of payment is the Cost of the AIA Document A102 TM —2017, Exhibit AIA Document A201 TM —2017, Generomodified	e Work Plus a Fee with a Gu oit A, Insurance and Bonds <u>Ir</u>	aranteed Maximum tentionally omitted	Price		
	.5 Drawings NOT USED					
	.5 Intentionally omitted Number	Title	Date			
.6	- Specifications					
	.6 Intentionally omitted Section	Title	Date	Pages		
.7—	-Addenda, if any:					
	.7 Intentionally omitted Number	Date	Pages			
	[] AIA Document E204TM 201	17, Sustainable Projects Exhi 2017 incorporated into this 2		ted below:		
	The Sustainability Plan:					
	Title	Date	Pages			

<u>HUD-5370</u> <u>General Conditions for O1/2014</u> <u>19</u> <u>Construction Contracts – exp11/30/2023</u>

Supplementary and other Conditions of the Contract:

Public Housing Programs

Additions and Deletions Report for AIA Document A102 – 2017. Copyright © 1920, 1925, 1951, 1958, 1961, 1963, 1967, 1974, 1978, 1987, 1997, 2007 and 2017>. All rights reserved. "The American Institute of Architects," "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are trademarks of The American Institute of Architects. This draft was produced at 15:38:28 ET on 10/30/2023 under Order No.2114399858 which expires on 02/01/2024, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail docinfo@aiacontracts.com.

[<u>X</u>]

- A Specifications Project Manual Table of Contents, dated xx/xx/xxxx
- B Drawings Cover Sheet and Index (General and Elevators), dated xx/xx/xxxx
- C Scope Narrative, dated xx/xx/xxxx
- D Itemized Statement of the Guaranteed Maximum Price, dated xx/xx/xxxx
- E Construction Schedule, dated xx/xx/xxxx



PAGE 20

HIGHLAND GARDENS LLC

TRAVAUX INC.

A Wisconsin Limited Liability Company

By: Highland Gardens Development LLC A Wisconsin Limited Liability Company, its Manager

By: Housing Authority of the City of Milwaukee A Wisconsin Public Body Corporate and Politic, its Manager

...

Willie L. Hines, Jr.
Secretary-Executive Director

Willie L. Hines, Jr.

President

Certification of Document's Authenticity

AIA® Document D401™ - 2003

I, , hereby certify, to the best of my knowledge, information and belief, that I c simultaneously with its associated Additions and Deletions Report and this certification.	
under Order No. 2114399858 from AIA Contract Documents software and that	in preparing the attached final
document I made no changes to the original text of AIA® Document A102 TM –	
Between Owner and Contractor where the basis of payment is the Cost of the Waximum Price, other than those additions and deletions shown in the associated	
	or reasonable and Bereine Repers
(Signed)	
(Title)	
(D. 44. 4)	
(Dated)	

General Conditions of the Contract for Construction

for the following PROJECT:

(Name and location or address)

CONTRACT 23-006

Highland Gardens
Located at 1818 West Juneau Avenue, Milwaukee, WI 53233
Rehabilitation of 114-unit apartment building and associated site improvements

THE OWNER:

(Name, legal status and address)

Highland Gardens LLC c/o Housing Authority of the City of Milwaukee 809 North Broadway, 3rd Floor Milwaukee, WI 53202

THE ARCHITECT:

(Name, legal status and address)

Galbraith Carnahan Architects LLC 6404 West North Avenue Wauwatosa, WI 53213

TABLE OF ARTICLES

- 1 GENERAL PROVISIONS
- 2 OWNER
- 3 CONTRACTOR
- 4 ARCHITECT
- 5 SUBCONTRACTORS
- 6 CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS
- 7 CHANGES IN THE WORK
- 8 TIME

User Notes:

- 9 PAYMENTS AND COMPLETION
- 10 PROTECTION OF PERSONS AND PROPERTY
- 11 INSURANCE AND BONDS
- 12 UNCOVERING AND CORRECTION OF WORK

ADDITIONS AND DELETIONS:

The author of this document has added information needed for its completion. The author may also have revised the text of the original AIA standard form. An Additions and Deletions Report that notes added information as well as revisions to the standard form text is available from the author and should be reviewed. A vertical line in the left margin of this document indicates where the author has added necessary information and where the author has added to or deleted from the original AIA text.

This document has important legal consequences. Consultation with an attorney is encouraged with respect to its completion or modification.

For guidance in modifying this document to include supplementary conditions, see AIA Document A503™, Guide for Supplementary Conditions.

- 13 MISCELLANEOUS PROVISIONS
- 14 TERMINATION OR SUSPENSION OF THE CONTRACT
- 15 CLAIMS AND DISPUTES



INDEX Architect's Authority to Reject Work (Topics and numbers in bold are Section headings.) 1.1.7, 1.5 Acceptance of Nonconforming Work 9.6.6, 9.9.3, 12.3 Acceptance of Work 9.6.6, 9.8.2, 9.9.3, 9.10.1, 9.10.3, 12.3 13.4.2, 15.2 Access to Work **3.16**, 6.2.1, 12.1 **Accident Prevention** 10 Acts and Omissions 3.2, 3.3.2, 3.12.8, 3.18, 4.2.3, 8.3.1, 9.5.1, 10.2.5, 4.2.11, 4.2.12 10.2.8, 13.3.2, 14.1, 1, 15.2 Addenda 4.2.10 1.1.1 Additional Costs, Claims for 3.7.4, 3.7.5, 10.3.2, 15.1.5 **Additional Inspections and Testing** 9.4.2, 9.8.3, 12.2.1, **13.4** Additional Time, Claims for 3.2.4, 3.7.4, 3.7.5, 3.10.2, 8.3.2, 15.1.6 **Administration of the Contract** 3.1.3, **4.2**, 9.4, 9.5 Advertisement or Invitation to Bid 1.1.1 Aesthetic Effect Asbestos 4.2.13 10.3.1 Attorneys' Fees **Allowances Applications for Payment** 4.2.5, 7.3.9, 9.2, **9.3**, 9.4, 9.5.1, 9.5.4, 9.6.3, 9.7, 9.10 6.1.1, 6.1.2 2.1.1, 2.3.1, 2.5, 3.1.3, 3.10.2, 3.12.8, 3.12.9, 3.12.10.1, 4.2.7, 9.3.2, 13.4.1 5.2 Arbitration 8.3.1, 15.3.2, **15.4** 1.1 ARCHITECT 1.1.1 Architect, Definition of 4.1.1 Architect, Extent of Authority 2.5, 3.12.7, 4.1.2, 4.2, 5.2, 6.3, 7.1.2, 7.3.4, 7.4, 9.2, Bonds, Lien 9.3.1, 9.4, 9.5, 9.6.3, 9.8, 9.10.1, 9.10.3, 12.1, 12.2.1, 13.4.1, 13.4.2, 14.2.2, 14.2.4, 15.1.4, 15.2.1 Architect, Limitations of Authority and Responsibility 2.1.1, 3.12.4, 3.12.8, 3.12.10, 4.1.2, 4.2.1, 4.2.2, 4.2.3, 4.2.6, 4.2.7, 4.2.10, 4.2.12, 4.2.13, 5.2.1, 7.4, 9.4.2, 1.8 9.5.4, 9.6.4, 15.1.4, 15.2 **Building Permit** Architect's Additional Services and Expenses 3.7.1 2.5, 12.2.1, 13.4.2, 13.4.3, 14.2.4 Capitalization Architect's Administration of the Contract

3.5, 4.2.6, 12.1.2, 12.2.1 Architect's Copyright Architect's Decisions 3.7.4, 4.2.6, 4.2.7, 4.2.11, 4.2.12, 4.2.13, 4.2.14, 6.3, 7.3.4, 7.3.9, 8.1.3, 8.3.1, 9.2, 9.4.1, 9.5, 9.8.4, 9.9.1, Architect's Inspections 3.7.4, 4.2.2, 4.2.9, 9.4.2, 9.8.3, 9.9.2, 9.10.1, 13.4 Architect's Instructions 3.2.4, 3.3.1, 4.2.6, 4.2.7, 13.4.2 Architect's Interpretations Architect's Project Representative Architect's Relationship with Contractor 1.1.2, 1.5, 2.3.3, 3.1.3, 3.2.2, 3.2.3, 3.2.4, 3.3.1, 3.4.2, 3.5, 3.7.4, 3.7.5, 3.9.2, 3.9.3, 3.10, 3.11, 3.12, 3.16, 3.18, 4.1.2, 4.2, 5.2, 6.2.2, 7, 8.3.1, 9.2, 9.3, 9.4, 9.5, 9.7, 9.8, 9.9, 10.2.6, 10.3, 11.3, 12, 13.3.2, 13.4, 15.2 Architect's Relationship with Subcontractors 1.1.2, 4.2.3, 4.2.4, 4.2.6, 9.6.3, 9.6.4, 11.3 Architect's Representations 9.4.2, 9.5.1, 9.10.1 Architect's Site Visits 3.7.4, 4.2.2, 4.2.9, 9.4.2, 9.5.1, 9.9.2, 9.10.1, 13.4 3.18.1, 9.6.8, 9.10.2, 10.3.3 Award of Separate Contracts Award of Subcontracts and Other Contracts for Portions of the Work **Basic Definitions Bidding Requirements** Binding Dispute Resolution 8.3.1, 9.7, 11.5, 13.1, 15.1.2, 15.1.3, 15.2.1, 15.2.5, 15.2.6.1, 15.3.1, 15.3.2, 15.3.3, 15.4.1 7.3.4.4, 9.6.8, 9.10.2, 9.10.3 Bonds, Performance, and Payment 7.3.4.4, 9.6.7, 9.10.3, **11.1.2**, 11.1.3, **11.5 Building Information Models Use and Reliance** Certificate of Substantial Completion 9.8.3, 9.8.4, 9.8.5

Init.

3.1.3, 3.7.4, 15.2, 9.4.1, 9.5

Architect's Approvals 2.5, 3.1.3, 3.5, 3.10.2, 4.2.7

AIA Document A201 – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017. All rights reserved. "The American Institute of Architects," "American Institute of Architects," "AlA," the AIA Logo, and "AIA Contract Documents" are trademarks of The American Institute of Architects. This document was produced at 15:38:55 ET on 10/30/2023 under Order No.2114399858 which expires on 02/01/2024, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail docinfo@aiacontracts.com.

Certificates for Payment

4.2.1, 4.2.5, 4.2.9, 9.3.3, **9.4**, 9.5, 9.6.1, 9.6.6, 9.7, 9.10.1, 9.10.3, 14.1.1.3, 14.2.4, 15.1.4

Certificates of Inspection, Testing or Approval 13.4.4

Certificates of Insurance

9.10.2

Change Orders

1.1.1, 3.4.2, 3.7.4, 3.8.2.3, 3.11, 3.12.8, 4.2.8, 5.2.3, 7.1.2, 7.1.3, **7.2**, 7.3.2, 7.3.7, 7.3.9, 7.3.10, 8.3.1, 9.3.1.1, 9.10.3, 10.3.2, 11.2, 11.5, 12.1.2

Change Orders, Definition of

7.2.1

CHANGES IN THE WORK

2.2.2, 3.11, 4.2.8, **7**, 7.2.1, 7.3.1, 7.4, 8.3.1, 9.3.1.1, 11.5

Claims, Definition of

15.1.1

Claims, Notice of

1.6.2, 15.1.3

CLAIMS AND DISPUTES

3.2.4, 6.1.1, 6.3, 7.3.9, 9.3.3, 9.10.4, 10.3.3, **15**, 15.4 Claims and Timely Assertion of Claims

Claims for Additional Cost

3.2.4, 3.3.1, 3.7.4, 7.3.9, 9.5.2, 10.2.5, 10.3.2, **15.1.5**

Claims for Additional Time

3.2.4, 3.3.1, 3.7.4, 6.1.1, 8.3.2, 9.5.2, 10.3.2, **15.1.6**

Concealed or Unknown Conditions, Claims for 3.7.4

Claims for Damages

3.2.4, 3.18, 8.3.3, 9.5.1, 9.6.7, 10.2.5, 10.3.3, 11.3, 11.3.2, 14.2.4, 15.1.7

Claims Subject to Arbitration

15.4.1

Cleaning Up

3.15, 6.3

Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.2, **15.1.5**

Commencement of the Work, Definition of 8.1.2

Communications

3.9.1, 4.2.4

Completion, Conditions Relating to

3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 14.1.2, 15.1.2

COMPLETION, PAYMENTS AND

9

Completion, Substantial

3.10.1, 4.2.9, 8.1.1, 8.1.3, 8.2.3, 9.4.2, 9.8, 9.9.1,

9.10.3, 12.2, 15.1.2

Compliance with Laws

2.3.2, 3.2.3, 3.6, 3.7, 3.12.10, 3.13, 9.6.4, 10.2.2, 13.1, 13.3, 13.4.1, 13.4.2, 13.5, 14.1.1, 14.2.1.3, 15.2.8, 15.4.2, 15.4.2

15.4.2, 15.4.3

Concealed or Unknown Conditions

3.7.4, 4.2.8, 8.3.1, 10.3

Conditions of the Contract

1.1.1, 6.1.1, 6.1.4

Consent, Written

3.4.2, 3.14.2, 4.1.2, 9.8.5, 9.9.1, 9.10.2, 9.10.3, 13.2,

15.4.4.2

Consolidation or Joinder

15.4.4

CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS

1.1.4.6

Construction Change Directive, Definition of

7.3.1

Construction Change Directives

1.1.1, 3.4.2, 3.11, 3.12.8, 4.2.8, 7.1.1, 7.1.2, 7.1.3, **7.3**,

9.3.1.1

Construction Schedules, Contractor's

3.10, 3.11, 3.12.1, 3.12.2, 6.1.3, 15.1.6.2

Contingent Assignment of Subcontracts

5.4, 14.2.2.2

Continuing Contract Performance

15.1.4

Contract, Definition of

1.1.2

CONTRACT, TERMINATION OR

SUSPENSION OF THE

5.4.1.1, 5.4.2, 11.5, **14**

Contract Administration

3.1.3, 4, 9.4, 9.5

Contract Award and Execution, Conditions Relating

to

3.7.1, 3.10, 5.2, 6.1

Contract Documents, Copies Furnished and Use of 1.5.2, 2.3.6, 5.3

Contract Documents, Definition of

1.1.1

Contract Sum

2.2.2, 2.2.4, 3.7.4, 3.7.5, 3.8, 3.10.2, 5.2.3, 7.3, 7.4, **9.1**, 9.2, 9.4.2, 9.5.1.4, 9.6.7, 9.7, 10.3.2, 11.5, 12.1.2,

12.3, 14.2.4, 14.3.2, 15.1.4.2, **15.1.5, 15.2.5**

Contract Sum, Definition of

9.1

Contract Time

1.1.4, 2.2.1, 2.2.2, 3.7.4, 3.7.5, 3.10.2, 5.2.3, 6.1.5, 7.2.1.3, 7.3.1, 7.3.5, 7.3.6, 7, 7, 7.3.10, 7.4, 8.1.1,

8.2.1, 8.2.3, 8.3.1, 9.5.1, 9.7, 10.3.2, 12.1.1, 12.1.2,

14.3.2, 15.1.4.2, 15.1.6.1, 15.2.5

Contract Time, Definition of

8.1.1

CONTRACTOR

3

Contractor, Definition of

3.1, 6.1.2

Contractor's Construction and Submittal Schedules

Schedules

3.10, 3.12.1, 3.12.2, 4.2.3, 6.1.3, 15.1.6.2

Init.

AlA Document A201 – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017. All rights reserved. "The American Institute of Architects," "AlA," the AlA Logo, and "AlA Contract Documents" are trademarks of The American Institute of Architects. This document was produced at 15:38:55 ET on 10/30/2023 under Order No.2114399858 which expires on 02/01/2024, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AlA Contract Documents® Terms of Service. To report copyright violations, e-mail docinfo@aiacontracts.com.

User Notes: (1162898282)

Contractor's Employees Damage to Construction of Owner or Separate 2.2.4, 3.3.2, 3.4.3, 3.8.1, 3.9, 3.18.2, 4.2.3, 4.2.6, 10.2, Contractors 10.3, 11.3, 14.1, 14.2.1.1 3.14.2, 6.2.4, 10.2.1.2, 10.2.5, 10.4, 12.2.4 Contractor's Liability Insurance Damage to the Work 11.1 3.14.2, 9.9.1, 10.2.1.2, 10.2.5, 10.4, 12.2.4 Contractor's Relationship with Separate Contractors Damages, Claims for 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.3.2, and Owner's Forces 3.12.5, 3.14.2, 4.2.4, 6, 11.3, 12.2.4 11.3, 14.2.4, 15.1.7 Contractor's Relationship with Subcontractors Damages for Delay 1.2.2, 2.2.4, 3.3.2, 3.18.1, 3.18.2, 4.2.4, 5, 9.6.2, 9.6.7, 6.2.3, 8.3.3, 9.5.1.6, 9.7, 10.3.2, 14.3.2 9.10.2, 11.2, 11.3, 11.4 Date of Commencement of the Work, Definition of Contractor's Relationship with the Architect 1.1.2, 1.5, 2.3.3, 3.1.3, 3.2.2, 3.2.3, 3.2.4, 3.3.1, 3.4.2, Date of Substantial Completion, Definition of 3.5.1, 3.7.4, 3.10, 3.11, 3.12, 3.16, 3.18, 4.2, 5.2, 6.2.2, 8.1.3 Day, Definition of 7, 8.3.1, 9.2, 9.3, 9.4, 9.5, 9.7, 9.8, 9.9, 10.2.6, 10.3, 11.3, 12, 13.4, 15.1.3, 15.2.1 8.1.4 Contractor's Representations Decisions of the Architect 3.7.4, 4.2.6, 4.2.7, 4.2.11, 4.2.12, 4.2.13, 6.3, 7.3.4,3.2.1, 3.2.2, 3.5, 3.12.6, 6.2.2, 8.2.1, 9.3.3, 9.8.2 Contractor's Responsibility for Those Performing the 7.3.9, 8.1.3, 8.3.1, 9.2, 9.4, 9.5.1, 9.8.4, 9.9.1, 13.4.2, Work 14.2.2, 14.2.4, 15.1, 15.2 3.3.2, 3.18, 5.3, 6.1.3, 6.2, 9.5.1, 10.2.8 **Decisions to Withhold Certification** Contractor's Review of Contract Documents 9.4.1, **9.5**, 9.7, 14.1.1.3 Defective or Nonconforming Work, Acceptance, Rejection and Correction of Contractor's Right to Stop the Work 2.2.2, 9.7 2.5, 3.5, 4.2.6, 6.2.3, 9.5.1, 9.5.3, 9.6.6, 9.8.2, 9.9.3, Contractor's Right to Terminate the Contract 9.10.4, 12.2.1 14.1 **Definitions** Contractor's Submittals 1.1, 2.1.1, 3.1.1, 3.5, 3.12.1, 3.12.2, 3.12.3, 4.1.1, 5.1, 3.10, 3.11, 3.12, 4.2.7, 5.2.1, 5.2.3, 9.2, 9.3, 9.8.2, 6.1.2, 7.2.1, 7.3.1, 8.1, 9.1, 9.8.1, 15.1.1 9.8.3, 9.9.1, 9.10.2, 9.10.3 **Delays and Extensions of Time** Contractor's Superintendent **3.2**, **3.7.4**, 5.2.3, 7.2.1, 7.3.1, **7.4**, **8.3**, 9.5.1, **9.7**, 3.9, 10.2.6 10.3.2, **10.4**, 14.3.2, **15.1.6**, 15.2.5 Contractor's Supervision and Construction **Digital Data Use and Transmission Procedures** 1.7 1.2.2, 3.3, 3.4, 3.12.10, 4.2.2, 4.2.7, 6.1.3, 6.2.4, 7.1.3, Disputes 7.3.4, 7.3.6, 8.2, 10, 12, 14, 15.1.4 6.3, 7.3.9, 15.1, 15.2 Coordination and Correlation **Documents and Samples at the Site** 1.2, 3.2.1, 3.3.1, 3.10, 3.12.6, 6.1.3, 6.2.1 3.11 Copies Furnished of Drawings and Specifications Drawings, Definition of 1.5, 2.3.6, 3.11 1.1.5 Copyrights Drawings and Specifications, Use and Ownership of 1.5, 3.17 Correction of Work Effective Date of Insurance 2.5, 3.7.3, 9.4.2, 9.8.2, 9.8.3, 9.9.1, 12.1.2, 12.2, 12.3, 8.2.2 15.1.3.1, 15.1.3.2, 15.2.1 **Emergencies Correlation and Intent of the Contract Documents 10.4**, 14.1.1.2, **15.1.5** 1.2 Employees, Contractor's Cost, Definition of 3.3.2, 3.4.3, 3.8.1, 3.9, 3.18.2, 4.2.3, 4.2.6, 10.2, 7.3.4 10.3.3, 11.3, 14.1, 14.2.1.1 Costs Equipment, Labor, or Materials 2.5, 3.2.4, 3.7.3, 3.8.2, 3.15.2, 5.4.2, 6.1.1, 6.2.3, 1.1.3, 1.1.6, 3.4, 3.5, 3.8.2, 3.8.3, 3.12, 3.13, 3.15.1, 7.3.3.3, 7.3.4, 7.3.8, 7.3.9, 9.10.2, 10.3.2, 10.3.6, 11.2, 4.2.6, 4.2.7, 5.2.1, 6.2.1, 7.3.4, 9.3.2, 9.3.3, 9.5.1.3, 12.1.2, 12.2.1, 12.2.4, 13.4, 14 9.10.2, 10.2.1, 10.2.4, 14.2.1.1, 14.2.1.2 **Cutting and Patching** Execution and Progress of the Work **3.14**, 6.2.5 1.1.3, 1.2.1, 1.2.2, 2.3.4, 2.3.6, 3.1, 3.3.1, 3.4.1, 3.7.1, 3.10.1, 3.12, 3.14, 4.2, 6.2.2, 7.1.3, 7.3.6, 8.2, 9.5.1, 9.9.1, 10.2, 10.3, 12.1, 12.2, 14.2, 14.3.1, 15.1.4

Init.

AlA Document A201 – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017. All rights reserved. "The American Institute of Architects," "American Institute of Architects," "AlA," the AlA Logo, and "AlA Contract Documents" are trademarks of The American Institute of Architects. This document was produced at 15:38:55 ET on 10/30/2023 under Order No.2114399858 which expires on 02/01/2024, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AlA Contract Documents® Terms of Service. To report copyright violations, e-mail docinfo@aiacontracts.com.

(1162898282)

User Notes:

Extensions of Time Insurance, Stored Materials 3.2.4, 3.7.4, 5.2.3, 7.2.1, 7.3, 7.4, 9.5.1, 9.7, 10.3.2, 9.3.2 INSURANCE AND BONDS 10.4, 14.3, 15.1.6, **15.2.5 Failure of Payment** 9.5.1.3, 9.7, 9.10.2, 13.5, 14.1.1.3, 14.2.1.2 Insurance Companies, Consent to Partial Occupancy Faulty Work (See Defective or Nonconforming Work) Insured loss, Adjustment and Settlement of **Final Completion and Final Payment** 4.2.1, 4.2.9, 9.8.2, **9.10**, 12.3, 14.2.4, 14.4.3 Intent of the Contract Documents Financial Arrangements, Owner's 1.2.1, 4.2.7, 4.2.12, 4.2.13 2.2.1, 13.2.2, 14.1.1.4 **Interest GENERAL PROVISIONS** 13.5 Interpretation 1.1.8, 1.2.3, **1.4**, 4.1.1, 5.1, 6.1.2, 15.1.1 **Governing Law** 13.1 Interpretations, Written Guarantees (See Warranty) 4.2.11, 4.2.12 **Hazardous Materials and Substances** Judgment on Final Award 10.2.4, **10.3** 15.4.2 Identification of Subcontractors and Suppliers Labor and Materials, Equipment 5.2.1 1.1.3, 1.1.6, **3.4**, 3.5, 3.8.2, 3.8.3, 3.12, 3.13, 3.15.1, Indemnification 5.2.1, 6.2.1, 7.3.4, 9.3.2, 9.3.3, 9.5.1.3, 9.10.2, 10.2.1, 3.17, **3.18**, 9.6.8, 9.10.2, 10.3.3, 11.3 10.2.4, 14.2.1.1, 14.2.1.2 Information and Services Required of the Owner Labor Disputes 2.1.2, **2.2**, 2.3, 3.2.2, 3.12.10.1, 6.1.3, 6.1.4, 6.2.5, 8.3.1 9.6.1, 9.9.2, 9.10.3, 10.3.3, 11.2, 13.4.1, 13.4.2, Laws and Regulations 1.5, 2.3.2, 3.2.3, 3.2.4, 3.6, 3.7, 3.12.10, 3.13, 9.6.4, 14.1.1.4, 14.1.4, 15.1.4 **Initial Decision** 9.9.1, 10.2.2, 13.1, 13.3.1, 13.4.2, 13.5, 14, 15.2.8, 15.2 15.4 Initial Decision Maker, Definition of Liens 1.1.8 2.1.2, 9.3.1, 9.3.3, 9.6.8, 9.10.2, 9.10.4, 15.2.8 Initial Decision Maker, Decisions Limitations, Statutes of 14.2.4, 15.1.4.2, 15.2.1, 15.2.2, 15.2.3, 15.2.4, 15.2.5 12.2.5, 15.1.2, 15.4.1.1 Initial Decision Maker, Extent of Authority Limitations of Liability 14.2.4, 15.1.4.2, 15.2.1, 15.2.2, 15.2.3, 15.2.4, 15.2.5 3.2.2, 3.5, 3.12.10, 3.12.10.1, 3.17, 3.18.1, 4.2.6, Injury or Damage to Person or Property 4.2.7, 6.2.2, 9.4.2, 9.6.4, 9.6.7, 9.6.8, 10.2.5, 10.3.3, 10.2.8, 10.4 11.3, 12.2.5, 13.3.1 Inspections Limitations of Time 3.1.3, 3.3.3, 3.7.1, 4.2.2, 4.2.6, 4.2.9, 9.4.2, 9.8.3, 2.1.2, 2.2, 2.5, 3.2.2, 3.10, 3.11, 3.12.5, 3.15.1, 4.2.7, 9.9.2, 9.10.1, 12.2.1, 13.4 5.2, 5.3, 5.4.1, 6.2.4, 7.3, 7.4, 8.2, 9.2, 9.3.1, 9.3.3, Instructions to Bidders 9.4.1, 9.5, 9.6, 9.7, 9.8, 9.9, 9.10, 12.2, 13.4, 14, 15, 1.1.1 15.1.2, 15.1.3, 15.1.5 Instructions to the Contractor Materials, Hazardous 3.2.4, 3.3.1, 3.8.1, 5.2.1, 7, 8.2.2, 12, 13.4.2 10.2.4, 10.3 Instruments of Service, Definition of Materials, Labor, Equipment and 1.1.7 1.1.3, 1.1.6, 3.4.1, 3.5, 3.8.2, 3.8.3, 3.12, 3.13, 3.15.1, 5.2.1, 6.2.1, 7.3.4, 9.3.2, 9.3.3, 9.5.1.3, 9.10.2, 6.1.1, 7.3.4, 8.2.2, 9.3.2, 9.8.4, 9.9.1, 9.10.2, 10.2.5, 11 10.2.1.2, 10.2.4, 14.2.1.1, 14.2.1.2 Insurance, Notice of Cancellation or Expiration Means, Methods, Techniques, Sequences and 11.1.4, 11.2.3 Procedures of Construction Insurance, Contractor's Liability 3.3.1, 3.12.10, 4.2.2, 4.2.7, 9.4.2 11.1 Mechanic's Lien Insurance, Effective Date of 2.1.2, 9.3.1, 9.3.3, 9.6.8, 9.10.2, 9.10.4, 15.2.88.2.2, 14.4.2 Mediation Insurance, Owner's Liability 8.3.1, 15.1.3.2, 15.2.1, 15.2.5, 15.2.6, **15.3**, 15.4.1, 15.4.1.1 **Insurance, Property** Minor Changes in the Work 1.1.1, 3.4.2, 3.12.8, 4.2.8, 7.1, **7.4 10.2.5**, 11.2, 11.4, 11.5

Init.

AIA Document A201 – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017. All rights reserved. "The American Institute of Architects," "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are trademarks of The American Institute of Architects. This document was produced at 15:38:55 ET on 10/30/2023 under Order No.2114399858 which expires on 02/01/2024, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail docinfo@aiacontracts.com.

MISCELLANEOUS PROVISIONS

13

Modifications, Definition of

Modifications to the Contract

1.1.1, 1.1.2, 2.5, 3.11, 4.1.2, 4.2.1, 5.2.3, 7, 8.3.1, 9.7, 10.3.2

Mutual Responsibility

Nonconforming Work, Acceptance of

9.6.6, 9.9.3, 12.3

Nonconforming Work, Rejection and Correction of 2.4, 2.5, 3.5, 4.2.6, 6.2.4, 9.5.1, 9.8.2, 9.9.3, 9.10.4, 12.2

Notice

1.6, 1.6.1, 1.6.2, 2.1.2, 2.2.2., 2.2.3, 2.2.4, 2.5, 3.2.4, 3.3.1, 3.7.4, 3.7.5, 3.9.2, 3.12.9, 3.12.10, 5.2.1, 7.4, 8.2.2 9.6.8, 9.7, 9.10.1, 10.2.8, 10.3.2, 11.5, 12.2.2.1, 13.4.1, 13.4.2, 14.1, 14.2.2, 14.4.2, 15.1.3, 15.1.5, 15.1.6, 15.4.1

Notice of Cancellation or Expiration of Insurance 11.1.4, 11.2.3

Notice of Claims

1.6.2, 2.1.2, 3.7.4, 9.6.8, 10.2.8, **15.1.3**, 15.1.5, 15.1.6, 15.2.8, 15.3.2, 15.4.1

Notice of Testing and Inspections

13.4.1, 13.4.2

Observations, Contractor's

3.2, 3.7.4

Occupancy

2.3.1, 9.6.6, 9.8

Orders, Written

1.1.1, 2.4, 3.9.2, 7, 8.2.2, 11.5, 12.1, 12.2.2.1, 13.4.2, 14.3.1

OWNER

2

Owner, Definition of

2.1.1

Owner, Evidence of Financial Arrangements

2.2, 13.2.2, 14.1.1.4

Owner, Information and Services Required of the 2.1.2, **2.2**, 2.3, 3.2.2, 3.12.10, 6.1.3, 6.1.4, 6.2.5, 9.3.2, 9.6.1, 9.6.4, 9.9.2, 9.10.3, 10.3.3, 11.2, 13.4.1, 13.4.2, 14.1.1.4, 14.1.4, 15.1.4

Owner's Authority

1.5, 2.1.1, 2.3.32.4, 2.5, 3.4.2, 3.8.1, 3.12.10, 3.14.2, 4.1.2, 4.2.4, 4.2.9, 5.2.1, 5.2.4, 5.4.1, 6.1, 6.3, 7.2.1, 7.3.1, 8.2.2, 8.3.1, 9.3.2, 9.5.1, 9.6.4, 9.9.1, 9.10.2, 10.3.2, 11.4, 11.5, 12.2.2, 12.3, 13.2.2, 14.3, 14.4, 15.2.7

Owner's Insurance

User Notes:

Owner's Relationship with Subcontractors 1.1.2, 5.2, 5.3, 5.4, 9.6.4, 9.10.2, 14.2.2

Owner's Right to Carry Out the Work **2.5**, 14.2.2

Owner's Right to Clean Up

Owner's Right to Perform Construction and to **Award Separate Contracts**

6.1

Owner's Right to Stop the Work

Owner's Right to Suspend the Work

Owner's Right to Terminate the Contract

14.2, 14.4

Ownership and Use of Drawings, Specifications

and Other Instruments of Service

1.1.1, 1.1.6, 1.1.7, **1.5**, 2.3.6, 3.2.2, 3.11, 3.17, 4.2.12,

Partial Occupancy or Use

9.6.6, **9.9**

Patching, Cutting and

3.14, 6.2.5 Patents

3.17

Payment, Applications for

4.2.5, 7.3.9, 9.2, **9.3**, 9.4, 9.5, 9.6.3, 9.7, 9.8.5, 9.10.1,

14.2.3, 14.2.4, 14.4.3

Payment, Certificates for

4.2.5, 4.2.9, 9.3.3, **9.4**, 9.5, 9.6.1, 9.6.6, 9.7, 9.10.1,

9.10.3, 14.1.1.3, 14.2.4

Payment, Failure of 9.5.1.3, **9.7**, 9.10.2, 13.5, 14.1.1.3, 14.2.1.2

Payment, Final

4.2.1, 4.2.9, **9.10**, 12.3, 14.2.4, 14.4.3

Payment Bond, Performance Bond and

7.3.4.4, 9.6.7, 9.10.3, **11.1.2**

Payments, Progress

9.3, **9.6**, 9.8.5, 9.10.3, 14.2.3, 15.1.4

PAYMENTS AND COMPLETION

Payments to Subcontractors

5.4.2, 9.5.1.3, 9.6.2, 9.6.3, 9.6.4, 9.6.7, 14.2.1.2

PCB

10.3.1

Performance Bond and Payment Bond

7.3.4.4, 9.6.7, 9.10.3, **11.1.2**

Permits, Fees, Notices and Compliance with Laws

2.3.1, **3.7**, 3.13, 7.3.4.4, 10.2.2

PERSONS AND PROPERTY, PROTECTION OF

Polychlorinated Biphenyl

10.3.1

Product Data, Definition of

3.12.2

Product Data and Samples, Shop Drawings

3.11, **3.12**, 4.2.7

Progress and Completion

4.2.2, **8.2**, 9.8, 9.9.1, 14.1.4, 15.1.4

Progress Payments

9.3, **9.6**, 9.8.5, 9.10.3, 14.2.3, 15.1.4

Init.

AIA Document A201 – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017. All rights reserved. "The American Institute of Architects," "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are trademarks of The American Institute of Architects. This document was produced at 15:38:55 ET on 10/30/2023 under Order No.2114399858 which expires on 02/01/2024, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail docinfo@aiacontracts.com.

(1162898282)

Project, Definition of Separate Contracts and Contractors 1.1.4, 3.12.5, 3.14.2, 4.2.4, 4.2.7, 6, 8.3.1, 12.1.2 1.1.4 Separate Contractors, Definition of Project Representatives 4.2.10 6.1.1 **Property Insurance** Shop Drawings, Definition of 10.2.5, **11.2** 3.12.1 **Proposal Requirements Shop Drawings, Product Data and Samples** 3.11, 3.12, 4.2.7 PROTECTION OF PERSONS AND PROPERTY Site, Use of 10 **3.13**, 6.1.1, 6.2.1 Site Inspections Regulations and Laws 3.2.2, 3.3.3, 3.7.1, 3.7.4, 4.2, 9.9.2, 9.4.2, 9.10.1, 13.4 1.5, 2.3.2, 3.2.3, 3.6, 3.7, 3.12.10, 3.13, 9.6.4, 9.9.1, 10.2.2, 13.1, 13.3, 13.4.1, 13.4.2, 13.5, 14, 15.2.8, 15.4 Site Visits, Architect's 3.7.4, 4.2.2, 4.2.9, 9.4.2, 9.5.1, 9.9.2, 9.10.1, 13.4 Rejection of Work 4.2.6, 12.2.1 Special Inspections and Testing Releases and Waivers of Liens 4.2.6, 12.2.1, 13.4 9.3.1, 9.10.2 Specifications, Definition of Representations 1.1.6 3.2.1, 3.5, 3.12.6, 8.2.1, 9.3.3, 9.4.2, 9.5.1, 9.10.1**Specifications** Representatives 1.1.1, **1.1.6**, 1.2.2, 1.5, 3.12.10, 3.17, 4.2.14 2.1.1, 3.1.1, 3.9, 4.1.1, 4.2.10, 13.2.1 Statute of Limitations Responsibility for Those Performing the Work 15.1.2, 15.4.1.1 3.3.2, 3.18, 4.2.2, 4.2.3, 5.3, 6.1.3, 6.2, 6.3, 9.5.1, 10 Stopping the Work 2.2.2, 2.4, 9.7, 10.3, 14.1 9.3.1, 9.6.2, 9.8.5, 9.9.1, 9.10.2, 9.10.3 Stored Materials **Review of Contract Documents and Field** 6.2.1, 9.3.2, 10.2.1.2, 10.2.4 **Conditions by Contractor** Subcontractor, Definition of **3.2**, 3.12.7, 6.1.3 5.1.1 Review of Contractor's Submittals by Owner and **SUBCONTRACTORS** Architect 3.10.1, 3.10.2, 3.11, 3.12, 4.2, 5.2, 6.1.3, 9.2, 9.8.2 Subcontractors, Work by Review of Shop Drawings, Product Data and Samples 1.2.2, 3.3.2, 3.12.1, 3.18, 4.2.3, 5.2.3, 5.3, 5.4, 9.3.1.2, by Contractor 3.12 **Subcontractual Relations Rights and Remedies 5.3**, 5.4, 9.3.1.2, 9.6, 9.10, 10.2.1, 14.1, 14.2.1 1.1.2, 2.4, 2.5, 3.5, 3.7.4, 3.15.2, 4.2.6, 5.3, 5.4, 6.1, Submittals 6.3, 7.3.1, 8.3, 9.5.1, 9.7, 10.2.5, 10.3, 12.2.1, 12.2.2, 3.10, 3.11, 3.12, 4.2.7, 5.2.1, 5.2.3, 7.3.4, 9.2, 9.3, 9.8, 12.2.4, 13.3, 14, 15.4 9.9.1, 9.10.2, 9.10.3 Royalties, Patents and Copyrights Submittal Schedule 3.17 3.10.2, 3.12.5, 4.2.7 Rules and Notices for Arbitration Subrogation, Waivers of 15.4.1 6.1.1, **11.3** Safety of Persons and Property Substances, Hazardous 10.2, 10.4 10.3 **Safety Precautions and Programs Substantial Completion** 3.3.1, 4.2.2, 4.2.7, 5.3, **10.1**, 10.2, 10.4 4.2.9, 8.1.1, 8.1.3, 8.2.3, 9.4.2, **9.8**, 9.9.1, 9.10.3, 12.2, Samples, Definition of 15.1.2 3.12.3 Substantial Completion, Definition of Samples, Shop Drawings, Product Data and 9.8.1 Substitution of Subcontractors 3.11, **3.12**, 4.2.7 Samples at the Site, Documents and 5.2.3, 5.2.4 3.11 Substitution of Architect Schedule of Values 2.3.3 **9.2**, 9.3.1 Substitutions of Materials

Init.

Schedules, Construction

3.10, 3.12.1, 3.12.2, 6.1.3, 15.1.6.2

AIA Document A201 – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017. All rights reserved. "The American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are trademarks of The American Institute of Architects. This document was produced at 15:38:55 ET on 10/30/2023 under Order No.2114399858 which expires on 02/01/2024, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail docinfo@aiacontracts.com.

5.1.2

3.4.2, 3.5, 7.3.8

Sub-subcontractor, Definition of

Subsurface Conditions

3.7.4

Successors and Assigns

13.2

Superintendent

3.9, 10.2.6

Supervision and Construction Procedures

1.2.2, **3.3**, 3.4, 3.12.10, 4.2.2, 4.2.7, 6.1.3, 6.2.4, 7.1.3,

7.3.4, 8.2, 8.3.1, 9.4.2, 10, 12, 14, 15.1.4

Suppliers

1.5, 3.12.1, 4.2.4, 4.2.6, 5.2.1, 9.3, 9.4.2, 9.5.4, 9.6,

9.10.5, 14.2.1

Surety

5.4.1.2, 9.6.8, 9.8.5, 9.10.2, 9.10.3, 11.1.2, 14.2.2,

15.2.7

Surety, Consent of

9.8.5, 9.10.2, 9.10.3

Surveys

1.1.7, 2.3.4

Suspension by the Owner for Convenience

14.3

Suspension of the Work

3.7.5, 5.4.2, 14.3

Suspension or Termination of the Contract

5.4.1.1, 14

Taxes

3.6, 3.8.2.1, 7.3.4.4

Termination by the Contractor

14.1, 15.1.7

Termination by the Owner for Cause

5.4.1.1, **14.2,** 15.1.7

Termination by the Owner for Convenience

14.4

Termination of the Architect

233

Termination of the Contractor Employment

14.2.2

TERMINATION OR SUSPENSION OF THE CONTRACT

14

Tests and Inspections

3.1.3, 3.3.3, 3.7.1, 4.2.2, 4.2.6, 4.2.9, 9.4.2, 9.8.3,

9.9.2, 9.10.1, 10.3.2, 12.2.1, 13.4

TIME

8

Time, Delays and Extensions of

3.2.4, 3.7.4, 5.2.3, 7.2.1, 7.3.1, 7.4, **8.3**, 9.5.1, 9.7,

10.3.2, 10.4, 14.3.2, 15.1.6, 15.2.5

Time Limits

2.1.2, 2.2, 2.5, 3.2.2, 3.10, 3.11, 3.12.5, 3.15.1, 4.2,

5.2, 5.3, 5.4, 6.2.4, 7.3, 7.4, 8.2, 9.2, 9.3.1, 9.3.3, 9.4.1,

9.5, 9.6, 9.7, 9.8, 9.9, 9.10, 12.2, 13.4, 14, 15.1.2,

15.1.3, 15.4

Time Limits on Claims

3.7.4, 10.2.8, 15.1.2, 15.1.3

Title to Work

9.3.2, 9.3.3

UNCOVERING AND CORRECTION OF WORK

12

Uncovering of Work

12.1

Unforeseen Conditions, Concealed or Unknown

3.7.4, 8.3.1, 10.3

Unit Prices

7.3.3.2, 9.1.2

Use of Documents

1.1.1, 1.5, 2.3.6, 3.12.6, 5.3

Use of Site

3.13, 6.1.1, 6.2.1

Values, Schedule of

9.2, 9.3.1

Waiver of Claims by the Architect

13.3.2

Waiver of Claims by the Contractor

9.10.5, 13.3.2, 15.1.7

Waiver of Claims by the Owner

9.9.3, 9.10.3, 9.10.4, 12.2.2.1, 13.3.2, 14.2.4, **15.1.7**

Waiver of Consequential Damages

14.2.4, 15.1.7

Waiver of Liens

9.3, 9.10.2, 9.10.4

Waivers of Subrogation

6.1.1, **11.3**

Warranty

3.5, 4.2.9, 9.3.3, 9.8.4, 9.9.1, 9.10.2, 9.10.4, 12.2.2,

15.1.2

Weather Delays

8.3, 15.1.6.2

Work, Definition of

1.1.3

Written Consent

1.5.2, 3.4.2, 3.7.4, 3.12.8, 3.14.2, 4.1.2, 9.3.2, 9.10.3,

13.2, 13.3.2, 15.4.4.2

Written Interpretations

4.2.11, 4.2.12

Written Orders

1.1.1, 2.4, 3.9, 7, 8.2.2, 12.1, 12.2, 13.4.2, 14.3.1

ARTICLE 1 GENERAL PROVISIONS

§ 1.1 Basic Definitions

§ 1.1.1 The Contract Documents

The Contract Documents are enumerated in the Agreement between the Owner and Contractor (hereinafter the Agreement) and consist of the Agreement, Conditions of the Contract (General, Supplementary and other Conditions), Drawings, Specifications, Addenda issued prior to execution of the Contract, other documents listed in the Agreement, and Modifications issued after execution of the Contract. A Modification is (1) a written amendment to the Contract signed by both parties, (2) a Change Order, (3) a Construction Change Directive, (4) a written order for a minor change in the Work issued by the Architect, or (5) Field Change (as defined herein). Unless specifically enumerated in the Agreement, the Contract Documents do not include the advertisement or invitation to bid, Instructions to Bidders, sample forms, other information furnished by the Owner in anticipation of receiving bids or proposals, the Contractor's bid or proposal, or portions of Addenda relating to bidding or proposal requirements.

§ 1.1.2 The Contract

The Contract Documents form the Contract for Construction. The Contract represents the entire and integrated agreement between the parties hereto and supersedes prior negotiations, representations, or agreements, either written or oral. The Contract may be amended or modified only by a Modification. The Contract Documents shall not be construed to create a contractual relationship of any kind (1) between the Contractor and the Architect or the Architect's consultants, (2) between the Owner and a Subcontractor or a Sub-subcontractor, (3) between the Owner and the Architect or the Architect's consultants, or (4) between any persons or entities other than the Owner and the Contractor. The Architect shall, however, be entitled to performance and enforcement of obligations under the Contract intended to facilitate performance of the Architect's duties and the Owner shall be entitled to performance and enforcement of obligations under the agreement between Contractor and Subcontractors intended to complete the Work.

§ 1.1.3 The Work

The term "Work" means the construction and services required by the Contract Documents, whether completed or partially completed, and includes all other labor, materials, equipment, and services provided or to be provided by the Contractor to fulfill the Contractor's obligations. The Work may constitute the whole or a part of the Project.

§ 1.1.4 The Project

The Project is the total construction of which the Work performed under the Contract Documents may be the whole or a part and which may include construction by the Owner and by Separate Contractors.

§ 1.1.5 The Drawings

The Drawings are the graphic and pictorial portions of the Contract Documents showing the design, location and dimensions of the Work, generally including plans, elevations, sections, details, schedules, diagrams, and any building information modeling (BIM).

§ 1.1.6 The Specifications

The Specifications are that portion of the Contract Documents consisting of the written requirements for materials, equipment, systems, standards and workmanship for the Work, and performance of related services. The Specifications shall also include performance specifications necessary for Contractor to complete any design-assist portion of the Work.

§ 1.1.7 Instruments of Service

Instruments of Service are representations, in any medium of expression now known or later developed, of the tangible and intangible creative work performed by the Architect and the Architect's consultants under their respective professional services agreements, in addition to any tangible and intangible creative work performed by Contractor related to any design-assist portion of the Work. Instruments of Service may include, without limitation, studies, surveys, models, sketches, drawings, specifications, and other similar materials.

(Paragraphs deleted)

§ 1.1.8 I INTENTIONALLY OMITTED

§ 1.2 Correlation and Intent of the Contract Documents

- § 1.2.1 The Contract Documents shall include all items necessary for the proper execution and completion of the Work by the Contractor. The Contract Documents are complementary, and what is required by one shall be as binding as if required by all; performance by the Contractor shall be required only to the extent consistent with the Contract Documents and reasonably inferable from them as being necessary to produce the indicated results. In the event of any inconsistency among the Contract Documents, at Owner's discretion, the better quality or more stringent requirement shall apply.
- § 1.2.1.1 The invalidity of any provision of the Contract Documents shall not invalidate the Contract or its remaining provisions. If it is determined that any provision of the Contract Documents violates any law, or is otherwise invalid or unenforceable, then that provision shall be revised to the extent necessary to make that provision legal and enforceable. In such case the Contract Documents shall be construed, to the fullest extent permitted by law, to give effect to the parties' intentions and purposes in executing the Contract.
- § 1.2.2 Organization of the Specifications into divisions, sections and articles, and arrangement of Drawings shall not control the Contractor in dividing the Work among Subcontractors or in establishing the extent of Work to be performed by any trade.
- § 1.2.3 Unless otherwise stated in the Contract Documents, words that have well-known technical or construction industry meanings are used in the Contract Documents in accordance with such recognized meanings.
- § 1.2.4 The general character and scope of Work is indicated by the Drawings. Where a portion of the Work is fully drawn and the remainder is merely indicated, the portion fully drawn shall apply to all similar parts of the Work to the extent it can be reasonably inferred.
- § 1.2.5 Lists of "work included" and "work excluded" are not intended to enumerate each and every item of work or appurtenants required. All work indicated shall be supplied except items specially noted as "by others", "by Owner", "not in contract", "existing", or similarly noted. Words and abbreviations which have well-known technical or trade meanings are used in the Contract Documents in accordance with such recognized meanings and consistent with the Stipulated Sum and the detail provided therein.
- § 1.2.6 Where items of material, equipment and labor are referred to in the singular, such item or items shall be provided in the number necessary for the proper completion of the Work.
- § 1.2.7 Unless otherwise specifically indicated, all references to "days" refers to calendar days The term "calendar day" shall mean the period from midnight of one day to the following midnight of the next day. The term "working day" shall mean the calendar days Mondays through Fridays, excluding holidays recognized by the City of Milwaukee, Wisconsin."

§ 1.3 Capitalization

Terms capitalized in these General Conditions include those that are (1) specifically defined, (2) the titles of numbered articles, or (3) the titles of other documents published by the American Institute of Architects.

§ 1.4 Interpretation

In the interest of brevity the Contract Documents frequently omit modifying words such as "all" and "any" and articles such as "the" and "an," but the fact that a modifier or an article is absent from one statement and appears in another is not intended to affect the interpretation of either statement.

§ 1.5 Ownership and Use of Drawings, Specifications, and Other Instruments of Service

- § 1.5.1 The Owner shall be deemed the owner of the Instruments of Service, including the Drawings and Specifications. The Contractor, Subcontractors, Sub-subcontractors, and material or equipment suppliers shall not own or claim a copyright in the Instruments of Service. Submittal or distribution to meet official regulatory requirements or for other purposes in connection with this Project is not to be construed as publication in derogation of the Architect's or Architect's consultants' reserved rights.
- § 1.5.2 The Contractor, Subcontractors, Sub-subcontractors, and material or equipment suppliers are authorized to use and reproduce the Instruments of Service provided to them, solely and exclusively for execution of the Work. All

copies made under this authorization shall bear the copyright notice, if any, shown on the Instruments of Service. The Contractor, Subcontractors, Sub-subcontractors, and material or equipment suppliers may not use the Instruments of Service on other projects or for additions to this Project outside the scope of the Work without the specific written consent of the Owner, and, where applicable, Architect and the Architect's consultants.

§ 1.5.3 Contractor shall indemnify, hold harmless and defend Owner, Architect, Architect's Consultants, against any claim, loss, damage or injury resulting from Contractor's or Subcontractor's: (1) unauthorized use of the Instruments of Service; or (2) use of the Instruments of Service on any other project. This duty shall include the reasonable attorneys' fees incurred by Owner in enforcing the terms set forth in this section.

§ 1.6 Notice

- § 1.6.1 Except as otherwise provided in Section 1.6.2, where the Contract Documents require one party to notify or give notice to the other party, such notice shall be provided in writing to the designated representative of the party to whom the notice is addressed and shall be deemed to have been duly served if delivered in person, by mail, by courier, or by electronic transmission if a method for electronic transmission is set forth in the Agreement.
- § 1.6.2 Notice of Claims as provided in Section 15.1.3 shall be provided in writing and shall be deemed to have been duly served only if delivered to the designated representative of the party to whom the notice is addressed by certified or registered mail, or by courier providing proof of delivery.

§ 1.7 Digital Data Use and Transmission

The parties shall agree upon protocols governing the transmission and use of Instruments of Service or any other information or documentation in digital form. Unless otherwise indicated in the Agreement or herein, the parties will use AIA Document E203TM–2013, Building Information Modeling and Digital Data Exhibit, to establish the protocols for the development, use, transmission, and exchange of digital data.

(Paragraphs deleted)

§ 1.8 Building Information Models Use

Contractor shall provide Drawings and BIM as necessary to meet permitting and design coordination requirements for the Work.

ARTICLE 2 OWNER

§ 2.1 General

- § 2.1.1 The Owner is the entity identified as such in the Agreement and is referred to throughout the Contract Documents as if singular in number. The Owner shall designate in writing those representatives who shall have express authority to bind the Owner with respect to all matters requiring the Owner's approval or authorization. Except as otherwise provided in Section 4.2.1, the Architect does not have such authority. The term "Owner" means the Owner or the Owner's authorized representative.
- § 2.1.2 The Owner shall furnish to the Contractor, within fifteen (15) days after receipt of a written request, information necessary and relevant for the Contractor to evaluate, give notice of, or enforce mechanic's lien rights. Such information shall include a correct statement of the record legal title to the property on which the Project is located, usually referred to as the site, and the Owner's interest therein.

§ 2.2 Evidence of the Owner's Financial Arrangements

- § 2.2.1 Prior to commencement of the Work, the Contractor may request in writing that the Owner provide reasonable evidence that the Owner has made financial arrangements to fulfill the Owner's obligations under the Contract.
- § 2.2.2 Thereafter, the Contractor may only request such reasonable evidence if (1) the Owner fails to make payments to the Contractor as the Contract Documents require; (2) the Contractor identifies in writing a reasonable concern regarding the Owner's ability to make payment when due; or (3) a change in the Work materially changes the Contract Sum. If the Owner fails to provide such evidence, as required, within fourteen (14) days of the Contractor's request, the Contractor may immediately stop the Work and, in that event, shall notify the Owner that the Work has stopped. However, if the request is made because a change in the Work materially changes the Contract Sum under (3) above, the Contractor may immediately stop only that portion of the Work affected by the change until reasonable evidence is provided. If the Work is stopped under this Section 2.2.2, the Contract Time shall be extended appropriately and the

Contract Sum shall be increased by the amount of the Contractor's reasonable costs of shutdown, delay and start-up, plus interest as provided in the Contract Documents.

§ 2.2.3 INTENTIONALLY OMITTED

§ 2.2.4 Where the Owner has designated information furnished under this Section 2.2 as "confidential," the Contractor shall keep the information confidential and shall not disclose it to any other person. However, the Contractor may disclose "confidential" information, after seven (7) days' notice to the Owner, where disclosure is required by law, including a subpoena or other form of compulsory legal process issued by a court or governmental entity, or by court or arbitrator(s) order. The Contractor may also disclose "confidential" information to its employees, consultants, sureties, Subcontractors and their employees, Sub-subcontractors, and others who need to know the content of such information solely and exclusively for the Project and who agree to maintain the confidentiality of such information.

§ 2.3 Information and Services Required of the Owner

- § 2.3.1 Except for permits and fees that are the responsibility of the Contractor under the Contract Documents, including those required under Section 3.7.1, the Owner shall secure and pay for necessary approvals, easements, assessments and charges required for construction, use or occupancy of permanent structures or for permanent changes in existing facilities. Owner shall furnish and pay for water and utilities utilized by the Contractor, Subcontractors, Sub-subcontractors and Architect in connection with their performance of the Work.
- § 2.3.2 The Owner shall retain an architect lawfully licensed to practice architecture, or an entity lawfully practicing architecture, in the jurisdiction where the Project is located. That person or entity is identified as the Architect in the Agreement and is referred to throughout the Contract Documents as if singular in number.
- § 2.3.3 If the employment of the Architect terminates, the Owner shall employ a successor to whom the Contractor has no reasonable objection and whose status under the Contract Documents shall be that of the Architect.
- § 2.3.4 Except as set forth in the Contract Documents, the Owner shall furnish surveys describing physical characteristics, legal limitations and utility locations for the site of the Project, and a legal description of the site. The Contractor shall be entitled to rely on the accuracy of information furnished by the Owner but shall exercise proper precautions relating to the safe performance of the Work. Contractor shall notify Owner of any discrepancies or information which is lacking in the information provided by the Owner of which Contractor is aware or reasonably should be aware.
- § 2.3.5 The Owner shall furnish information or services required of the Owner by the Contract Documents with reasonable promptness. The Owner shall also furnish any other information or services under the Owner's control that are necessary for completion of the Work with reasonable promptness after receiving the Contractor's written request for such information or services.
- § 2.3.6 Unless otherwise provided in the Contract Documents, the Owner shall furnish to the Contractor one copy of the Contract Documents for purposes of making reproductions pursuant to Section 1.5.2.

§ 2.4 Owner's Right to Stop the Work

If the Contractor: (a) fails to correct Work that is not in accordance with the requirements of the Contract Documents as required by Section 12.2 after fifteen (15) days prior written notice; or (b) repeatedly fails to carry out Work in accordance with the Contract Documents and, after fifteen (15) days prior written notice, and in the case under either (a) or (b) the Contractor does not undertake efforts to cure the non-conforming Work,, the Owner may issue a written order to the Contractor to stop the Work, or any portion thereof, until the cause for such order has been eliminated; however, the right of the Owner to stop the Work shall not give rise to a duty on the part of the Owner to exercise this right for the benefit of the Contractor or any other person or entity, except to the extent required by Section 6.1.3.

§ 2.5 Owner's Right to Carry Out the Work

If the Contractor defaults or neglects to carry out the Work in accordance with the Contract Documents and fails within a fifteen-day period after receipt of written notice from the Owner to commence and continue correction of such default or neglect with diligence and promptness, the Owner may, without prejudice to other remedies the Owner may have, correct such deficiencies. In such case an appropriate Change Order shall be issued deducting from payments then or thereafter due the Contractor the reasonable cost of correcting such deficiencies, including Owner's

expenses and compensation for the Architect's additional services and expenses made necessary by such default, neglect, or failure. If payments then or thereafter due the Contractor are not sufficient to cover such amounts, the Contractor shall pay the difference to the Owner.

ARTICLE 3 CONTRACTOR

§ 3.1 General

- § 3.1.1 The Contractor is the person or entity identified as such in the Agreement and is referred to throughout the Contract Documents as if singular in number. The Contractor and all applicable Subcontractors and suppliers shall be lawfully licensed and registered, if required in the jurisdiction where the Project is located. The Contractor shall designate in writing a representative who shall have express authority to bind the Contractor with respect to all matters under this Contract. The term "Contractor" means the Contractor or the Contractor's authorized representative.
- § 3.1.2 The Contractor shall perform the Work in accordance with the Contract Documents and Submittals approved pursuant to Section 3.12 herein.
- § 3.1.3 The Contractor shall not be relieved of its obligations to perform the Work in accordance with the Contract Documents either by activities or duties of the Architect in the Architect's administration of the Contract, or by tests, inspections or approvals required or performed by persons or entities other than the Contractor.

§ 3.2 Review of Contract Documents and Field Conditions by Contractor

- § 3.2.1 Execution of the Contract by the Contractor is a representation that the Contractor has visited the site, become generally familiar with local conditions under which the Work is to be performed, and correlated personal observations with requirements of the Contract Documents. Notwithstanding such observations, the Contractor shall nevertheless be entitled to rely upon any Owner provided information about the Project, Work or site, including, but not limited to, surveys, subsurface tests and data, the identification and location of hazardous substances and the condition and characteristics of structures on the site, unless Contractor has specific knowledge that the Owner provided information contrary to existing conditions. If that occurs, then Contractor shall ask for clarification from Owner or Architect. Whenever possible, the Contractor and all of its necessary Subcontractors, as appropriate, shall endeavor to verify existing field conditions that are visible (i.e., accessible through normal and customary methods of observation and investigation). Where existing field conditions are not visible (e.g., below grade) or concealed within construction assemblies, and existing items are indicated on the drawings, such items shall be considered as being shown schematically only, if so represented or designated as such by Architect.
- § 3.2.2 Because the Contract Documents are complementary, the Contractor shall, before starting each portion of the Work, carefully study and compare the various Contract Documents relative to that portion of the Work, as well as the information furnished by the Owner pursuant to Section 2.3.4, shall take field measurements of any existing conditions related to that portion of the Work, and shall observe any conditions at the site affecting it. These obligations are for the purpose of facilitating coordination and construction by the Contractor and are not for the purpose of discovering errors, omissions, or inconsistencies in the Contract Documents. The Contractor shall promptly report to the Architect and Owner any errors, inconsistencies or omissions discovered, including design errors or omissions, by or made known to the Contractor as a request for information in such form as the Architect may require. The Contractor shall halt all affected work until such discrepancies have been corrected or until directed otherwise by Owner or Architect. The direct out-of-pocket costs related to this Work stoppage shall be reimbursed to the Contractor and, if the Work stoppage results in a delay in the Project Schedule, the Completion Date shall be extended by Change Order. It is recognized that the Contractor's review is made in the Contractor's capacity as a contractor and not as a licensed design professional, unless otherwise specifically provided in the Contract Documents.
- § 3.2.3 Except as specifically set forth in the Contract Documents, the Contractor is not required to ascertain that the Contract Documents are in accordance with applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, but the Contractor shall promptly report to the Architect and Owner any nonconformity discovered by or made known to the Contractor as a request for information in such form as the Architect may require.
- § 3.2.4 If the Contractor believes that additional cost or time is involved because of clarifications or instructions the Architect issues in response to the Contractor's notices or requests for information pursuant to Sections 3.2.2 or 3.2.3, or the Architect's failure to respond to such requests for information or notices, the Contractor shall make Claims as

User Notes:

provided in Article 15. If the Contractor fails to perform the obligations of Sections 3.2.2 or 3.2.3, the Contractor shall pay such costs and damages to the Owner, as would have been avoided if the Contractor had performed such obligations. If the Contractor performs those obligations, the Contractor shall not be liable to the Owner or Architect for damages resulting from errors, inconsistencies or omissions in the Contract Documents, for differences between field measurements or conditions and the Contract Documents, or for nonconformities of the Contract Documents to applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities, unless the Contractor recognized such error, inconsistency, omission or difference and failed to report it to the Architect. If the Contractor performs any construction activity involving an error, inconsistency, omission or difference in the Contract Documents that Contractor knowingly recognized or should have reasonably recognized without such notice to the Architect and Owner, the costs for correction will be allocated among the responsible parties.

§ 3.2.5 The Contractor shall provide reasonably necessary estimating and coordination services to evaluate the value engineering and cost savings options. Contractor shall not be entitled to an increase in the Contract Price for such services. For those options the Owner approves as a lump sum adjustment, the Contract Price shall be reduced by the full amount of direct costs associated with the change through the Change Order process.

§ 3.3 Supervision and Construction Procedures

- § 3.3.1 The Contractor shall supervise and direct the Work, using the Contractor's best skill and attention. The Contractor shall be solely responsible for, and have control over, construction means, methods, techniques, sequences, and procedures, and for coordinating all portions of the Work under the Contract, unless the Contract Documents give other specific instructions concerning these matters. If the Contract Documents give specific instructions concerning construction means, methods, techniques, sequences, or procedures, the Contractor shall evaluate the jobsite safety thereof and, except as stated below, shall be fully and solely responsible for the jobsite safety of such means, methods, techniques, sequences, or procedures may not be safe, the Contractor determines that such means, methods, techniques, sequences or procedures may not be safe, the Contractor shall give timely written notice to the Owner and Architect, and shall not proceed with that portion of the Work without further written instructions from the Architect. If the Contractor is then instructed to proceed with the required means, methods, techniques, sequences, or procedures without acceptance of changes proposed by the Contractor, the Owner shall be solely responsible for any loss or damage arising from those Owner-required means, methods, techniques, sequences, or procedures, unless Contractor is negligent.
- § 3.3.2 The Contractor shall be responsible to the Owner for acts and omissions of the Contractor's employees, Subcontractors and their agents and employees, and other persons or entities performing portions of the Work for, or on behalf of, the Contractor or any of its Subcontractors.
- § 3.3.3 The Contractor shall be responsible for inspection of portions of Work already performed to determine that such portions are in proper condition to receive subsequent Work.

§ 3.4 Labor and Materials

- § 3.4.1 Unless otherwise provided in the Contract Documents, the Contractor shall provide and pay for labor, materials, equipment, tools, construction equipment and machinery, heat, transportation, and other facilities and services necessary for proper execution and completion of the Work, whether temporary or permanent and whether or not incorporated or to be incorporated in the Work. The Owner will not be liable to assume any responsibility for the damage or wear to the Contractor's tools, materials and/or equipment except to the extent covered by any applicable insurance or to the extent that the Owner and its agents, employees or contractors are responsible for the damage or wear.
- § 3.4.2 Except in the case of minor changes in the Work authorized by the Architect in accordance with Sections 3.12.8 or 7.4, the Contractor may make substitutions of specified materials or equipment only with the consent and approval of the Owner, after evaluation and recommendation by the Architect and in accordance with a Change Order or Construction Change Directive.
- § 3.4.2.1 Where a definite material or method is specified, it is not the intention to discriminate against any "approved equal" product by another manufacturer. Rather, it is the intention to set a definite standard. Open competition is expected. The Contractor shall submit to the Architect the substitutions it proposes together with samples, complete evidence of quality and any credits that may accrue to the Owner for allowing the various substitutions. The Architect shall investigate all proposed substitutions, consult with the Owner for its review and approval, and render final decisions as is necessary to avoid any delay in the Work.

User Notes:

- § 3.4.2.2 The Contractor shall ascertain that items offered as equals to specified items will fit the physical limits of space shown on the Drawings, and leave ample clearance for proper installation, operation and servicing of the item and all adjacent items.
- § 3.4.2.3 Materials and equipment proposed as substitutes for specified items may be rejected by the Architect and Owner.
- § 3.4.2.4 If during the performance of Work any materials or equipment specified in the Contract Documents become unavailable because of government restrictions or because of other market conditions (which are not the result of Contractor's delay in order or purchasing), the Owner, Architect and Contractor shall collectively suggest and consider alternatives and substitutes. The Architect shall then issue a final recommendation for consideration and approval by Owner. Contractor will be entitled to an increase in the Contract Sum for actual increased costs incurred directly as a result of an alternative or substitution required because of government restrictions or other market conditions to the extent such cost could not be mitigated or otherwise avoided. Contractor will provide Owner with invoices or such other documents as Owner may reasonably require to verify Contractor's increased costs.
- § 3.4.3 The Contractor shall enforce strict discipline and good order among the Contractor's employees and other persons carrying out the Work. The Contractor shall not permit employment of unfit persons or persons not properly skilled in tasks assigned to them. The Contractor shall promptly remove any persons of whom Owner has a reasonable objection to their fitness for work.
- § 3.4.4 Contractor shall provide to Architect the required information for all materials and equipment which, pursuant to the Contract Documents must be approved by Architect. The Contractor shall provide such information to the Architect according to the submittal schedule included in the Construction Schedule, or, if not specifically included in the Construction Schedule, then timely enough to allow for Architect's review and to otherwise not cause a delay in the Project Schedule.

§ 3.5 Warranty

- § 3.5.1 The Contractor warrants to the Owner and Architect that materials and equipment furnished under the Contract will be of good quality and new unless the Contract Documents require or permit otherwise. The Contractor further warrants that the Work will conform to the requirements of the Contract Documents and will be free from defects, except for those inherent in the quality of the Work the Contract Documents require or permit. Work, materials, or equipment not conforming to these requirements, including substitutions not properly approved and authorized, may be considered defective. The Contractor's warranty excludes remedy for damage or defect caused by abuse, alterations to the Work not executed by the Contractor, improper or insufficient maintenance, improper operation, or normal wear and tear and normal usage. If required by the Architect, the Contractor shall furnish satisfactory evidence as to the kind and quality of materials and equipment.
- § 3.5.2 The Contractor's warranty under this Section is not exclusive, and any other express warranties stated elsewhere, may also be exercised by Owner at its option. In addition, the Contractor shall assign to the Owner all manufacturers' and suppliers' warranties, express or implied, respecting any part of the Work which Contractor or Subcontractors receive not later than at the time Final Payment is made. The assignments, copies of all warranties and all product operation manuals for proper use and maintenance of equipment shall be conveyed to the Owner prior to Final Payment for the Work (i.e., final retainage). Owner may, at its option, release final retainage for that portion of the Work for which all contract close-out requirements have been satisfied.
- § 3.5.3 The Contractor and any applicable Subcontractors agree, as part of its post completion Work and its warranty obligations, to participate in warranty walk-throughs at the Project Site with the Owner and its Architect eleven (11) months after Substantial Completion of each phase of the Work to identify warranty work and any other items which must be corrected to conform with the requirements of the Contract Documents.

§ 3.6 Taxes

User Notes:

The Contractor shall pay sales, consumer, use and similar taxes for the Work provided by the Contractor that are legally enacted when bids are received or negotiations concluded, whether or not yet effective or merely scheduled to go into effect.

§ 3.7 Permits, Fees, Notices and Compliance with Laws

- § 3.7.1 Unless otherwise provided in the Contract Documents, the Contractor shall secure and pay for the building permit as well as for other permits, fees, licenses, and inspections by government agencies necessary for proper execution and completion of the Work that are customarily secured after execution of the Contract and legally required at the time bids are received or negotiations concluded.
- § 3.7.2 The Contractor shall comply with and give notices required by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities applicable to performance of the Work.
- § 3.7.3 If the Contractor performs Work knowing it to be contrary to applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, the Contractor shall assume appropriate responsibility for such Work and shall bear the costs attributable to correction. However, if the Contractor observes that portions of the Contract Documents are at variance therewith and promptly notifies the Architect and Owner in writing, necessary changes shall be accomplished by appropriate Modification. Any Modification shall be in accordance with Article 7. Notwithstanding the above, for those trades requiring licensed tradesmen, Work depicted for those trades shall be installed in a code conforming manner irrespective of references or representations in the documents to the contrary, at no additional cost to the Owner.

§ 3.7.4 Concealed or Unknown Conditions

If the Contractor encounters conditions at the site that are (1) subsurface or otherwise concealed physical conditions that differ materially from those indicated in the Contract Documents or (2) unknown physical conditions of an unusual nature that differ materially from those ordinarily found to exist and generally recognized as inherent in construction activities of the character provided for in the Contract Documents, the Contractor shall promptly provide notice to the Owner and the Architect before conditions are disturbed and in no event later than 21 days after first observance of the conditions. The Architect will promptly investigate such conditions and, if the Architect determines that they differ materially and cause an increase or decrease in the Contractor's cost of, or time required for, performance of any part of the Work, will recommend, for Owner's review and approval or disapproval, an equitable adjustment in the Contract Time. Any adjustment to the Contract Sum recommended by the Architect will be based on the factors set forth in Sections 7.3.4.1 to 7.3.4.5. If the Architect determines that the conditions at the site are not materially different from those indicated in the Contract Documents and that no change in the terms of the Contract is justified, the Architect shall promptly notify the Owner and Contractor of its recommendation in writing, stating the reasons. If either party disputes the Architect's recommendation, that party proceed as provided in Article 15.

§ 3.7.5 If, in the course of the Work, the Contractor encounters human remains or recognizes the existence of burial markers, archaeological sites or wetlands not indicated in the Contract Documents, the Contractor shall immediately suspend any operations that would affect them and shall notify the Owner and Architect. Upon receipt of such notice, the Owner shall promptly take any action necessary to obtain governmental authorization required to resume the operations. The Contractor shall continue to suspend such operations until otherwise instructed by the Owner but shall continue with all other operations that do not affect those remains or features. If requested by Contractor, any adjustment to the Contract Sum or Contract Time will be based on the factors set forth in Sections 7.3.4.1 to 7.3.4.5.

§ 3.8 Allowances

§ 3.8.1 The Contractor shall include in the Contract Sum all allowances stated in the Contract Documents. Items covered by allowances shall be supplied for such amounts and by such persons or entities as the Owner may direct, but the Contractor shall not be required to employ persons or entities to whom the Contractor has reasonable objection.

- § 3.8.2 Unless otherwise provided in the Contract Documents,
 - allowances shall cover the cost to the Contractor of materials and equipment delivered at the site and all required taxes, less applicable trade discounts;
 - .2 Contractor's costs for unloading and handling at the site, labor, installation costs, overhead, profit, and other expenses contemplated for stated allowance amounts shall be included in the Contract Sum but not in the allowances; and
 - .3 whenever costs are more than or less than allowances, the Contract Sum shall be adjusted accordingly by Change Order. The amount of the Change Order shall reflect (1) the difference between actual costs and the allowances under Section 3.8.2.1 and (2) changes in Contractor's costs under Section 3.8.2.2.
- § 3.8.3 Materials and equipment under an allowance shall be selected by the Owner with reasonable promptness.

User Notes:

§ 3.9 Superintendent

- § 3.9.1 The Contractor shall employ a competent superintendent and necessary assistants who shall be at the Project site as necessary for the performance of the Work, and who shall not be replaced without the consent of the Owner. If the superintendent or an assistant leaves the employment of the Contractor, the replacement shall be subject to Owner's approval. The superintendent shall represent the Contractor, and communications given to the superintendent shall be as binding as if given to the Contractor. All communications shall be confirmed in writing.
- § 3.9.2 The Contractor, as soon as practicable after award of the Contract, shall furnish in writing to the Owner the name and qualifications of a proposed superintendent. The Owner may reply within fourteen (14) days to the Contractor in writing, stating whether the Owner (1) has reasonable objection to the proposed superintendent or (2) requires additional time to review. Failure to reply within the 14-day period shall constitute notice of no reasonable objection.
- § 3.9.3 The Contractor shall not employ a proposed superintendent to whom the Owner has made reasonable and timely objection. The Contractor shall not change the superintendent without the Owner's consent, which shall not unreasonably be withheld or delayed.
- § 3.9.4 The Contractor shall have the Superintendent and other competent representatives familiar with the Work attend meetings which may be held at the job site or at such place as Owner designates.

§ 3.10 Contractor's Construction and Submittal Schedules

- § 3.10.1 If the Owner requires, the Contractor, promptly after being awarded the Contract, shall prepare and submit for the Owner's review and Architect's approval a Contractor's construction schedule for the Work. The schedule shall contain detail appropriate for the Project, including (1) the date of commencement of the Work, interim schedule milestone dates, and the date of Substantial Completion; (2) an apportionment of the Work by construction activity; and (3) the time required for completion of each portion of the Work. The schedule shall provide for the orderly progression of the Work to completion and shall not exceed time limits current under the Contract Documents. The schedule shall be revised at appropriate intervals as required by the conditions of the Work and Project, shall be related to the entire Project to the extent required by the Contract Documents, and shall provide for expeditious and practicable execution of the Work.
- § 3.10.2 The Contractor shall prepare a submittal schedule promptly after being awarded the Contract and thereafter as necessary to maintain a current submittal schedule, and shall submit the schedule for the Architect's and Owner's review and approval, which approval shall not be unreasonably delayed or withheld. The submittal schedule shall (1) be coordinated with the Contractor's construction schedule, and (2) allow the Architect and Owner reasonable time to review submittals. If the Contractor fails to submit a submittal schedule, or fails to provide submittals in accordance with the approved submittal schedule, the Contractor shall not be entitled to any increase in Contract Sum or extension of Contract Time based on the time required for review of submittals.
- § 3.10.3 The Contractor shall perform the Work in general accordance with the most recent schedules submitted to the Owner and Architect.

§ 3.11 Documents and Samples at the Site

- § 3.11.1 The Contractor shall maintain at the Project site, one (1) copy of the Contract Documents, including Drawings, Specifications, Addenda, Change Orders, Construction Change Directives, and other Modifications, in good order and marked currently to indicate Field Changes and selections made during construction, and the approved Shop Drawings, Product Data, Samples, and similar required submittals. These shall be in electronic form or paper copy, available to the Architect and Owner, and delivered to the Architect for submittal to the Owner upon completion of the Work as a record of the Work as constructed and prior to final payment (i.e. final retainage). Owner may, at its option, release final retainage for that portion of the Work for which all contract close-out requirements have been satisfied.
- § 3.11.2 At completion of each phase of the Work, record drawings of all mechanical and electrical systems, if any, shall be submitted to the Architect, which indicate all Field Changes. If not submitted, Applications for Payment will not be processed.

User Notes:

§ 3.12 Shop Drawings, Product Data and Samples

- § 3.12.1 Shop Drawings are drawings, diagrams, schedules, and other data specially prepared for the Work by the Contractor or a Subcontractor, Sub-subcontractor, manufacturer, supplier, or distributor to illustrate some portion of the Work.
- § 3.12.2 Product Data are illustrations, standard schedules, performance charts, instructions, brochures, diagrams, and other information furnished by the Contractor to illustrate materials or equipment for some portion of the Work.
- § 3.12.3 Samples are physical examples that illustrate materials, equipment, or workmanship, and establish standards by which the Work will be judged.
- § 3.12.4 Shop Drawings, Product Data, Samples, and similar submittals are not Contract Documents. Their purpose is to demonstrate the way by which the Contractor proposes to conform to the information given and the design concept expressed in the Contract Documents for those portions of the Work for which the Contract Documents require submittals. Review by the Architect is subject to the limitations of Section 4.2.7. Informational submittals upon which the Architect is not expected to take responsive action may be so identified in the Contract Documents. Submittals that are not required by the Contract Documents may be returned by the Architect without action.
- § 3.12.5 The Contractor shall review for compliance with the Contract Documents, approve, and submit to the Architect, Shop Drawings, Product Data, Samples, and similar submittals required by the Contract Documents, in accordance with the submittal schedule approved by the Architect and Owner or, in the absence of an approved submittal schedule, with reasonable promptness and in such sequence as to cause no delay in the Work or in the activities of the Owner or of Separate Contractors.
- § 3.12.6 By submitting Shop Drawings, Product Data, Samples, and similar submittals, the Contractor represents to the Owner and Architect that the Contractor has (1) reviewed and approved them, (2) determined and verified materials, field measurements and field construction criteria related thereto, or will do so, and (3) checked and coordinated the information contained within such submittals with the requirements of the Work and of the Contract Documents.
- § 3.12.7 The Contractor shall perform no portion of the Work for which the Contract Documents require submittal and review of Shop Drawings, Product Data, Samples, or similar submittals until the respective submittal has been approved by the Architect.
- § 3.12.8 The Work shall be in accordance with approved submittals except that the Contractor shall not be relieved of responsibility for deviations from the requirements of the Contract Documents by the Architect's approval of Shop Drawings, Product Data, Samples, or similar submittals, unless the Contractor has specifically informed the Architect in writing of such deviation at the time of submittal and (1) the Architect has given written approval to the specific deviation as a minor change in the Work, or (2) a Change Order or Construction Change Directive has been issued authorizing the deviation. The Contractor shall not be relieved of responsibility for errors or omissions in Shop Drawings, Product Data, Samples, or similar submittals, by the Architect's approval thereof.
- § 3.12.9 The Contractor shall direct specific attention, in writing or on resubmitted Shop Drawings, Product Data, Samples, or similar submittals, to revisions other than those requested by the Architect on previous submittals. In the absence of such written notice, the Architect's approval of a resubmission shall not apply to such revisions.
- § 3.12.10 The Contractor shall not be required to provide professional services that constitute the practice of architecture or engineering unless such services are specifically required by the Contract Documents for a portion of the Work or unless the Contractor needs to provide such services in order to carry out the Contractor's responsibilities for construction means, methods, techniques, sequences, and procedures. The Contractor shall not be required to provide professional services in violation of applicable law.
- § 3.12.10.1 If professional design services or certifications by a design professional related to systems, materials, or equipment are specifically required of the Contractor by the Contract Documents, the Owner and the Architect will specify all performance and design criteria that such services must satisfy. The Contractor shall cause such services or certifications to be provided by a properly licensed design professional, whose signature and seal shall appear on all drawings, calculations, specifications, certifications, Shop Drawings, and other submittals prepared by such

professional. Shop Drawings, and other submittals related to the Work, designed or certified by such professional, if prepared by others, shall bear such professional's written approval when submitted to the Architect. The Owner and the Architect shall be entitled to rely upon the adequacy, accuracy, and completeness of the services, certifications, and approvals performed or provided by such design professionals, provided the Owner and Architect have specified to the Contractor appropriate performance and design criteria that such services must satisfy. The Architect will review and approve, reject, or take other appropriate action on submittals (i.e., "revise and resubmit" or "proceed as revised") in accordance with Section 4.2.7 herein. Except to the extent Contractor is provided Design-Build services or has retained Subcontractors that are performing design-build services (e.g., MEP Subcontractors), the Contractor shall not be responsible for the adequacy of the performance and design criteria specified in the Contract Documents, but shall be responsible to promptly report to the Architect and Owner any errors, inconsistencies or omissions discovered in the performance and design criteria.

- § 3.12.10.2 If the Contract Documents require the Contractor's design professional to certify that the Work has been performed in accordance with the design criteria, the Contractor shall furnish such certifications to the Architect at the time and in the form specified by the Architect.
- § 3.12.11 Contractor shall furnish three (3) complete sets of manuals containing the manufacturers' warranties and instructions necessary for maintenance and operation of each item of equipment and apparatus it furnishes under the Contract Documents, warranty information and any additional data specifically requested under the various sections of the Specifications for each division of the Work to the Architect for subsequent conveyance with all Contractors' manuals to the Owner at or before Final Payment. Each manual shall be arranged in logical order, indexed and suitably bound. Alternatively, the Contractor may provide this information in the as-built BIM for the Work.

§ 3.13 Use of Site

- § 3.13.1 The Contractor shall confine operations at the site to areas permitted by applicable laws, statutes, ordinances, codes, rules and regulations, lawful orders of public authorities, and the Contract Documents and shall not unreasonably encumber the site with materials or equipment.
- § 3.13.2 The Contractor shall store materials on the site where directed and in such a manner as will not damage the area in which they are stored. Material deliveries shall be scheduled so that they are not stored longer than necessary. All items furnished to the site by the Owner shall be stored as directed.
- § 3.13.3 Where the Contractor's operations will affect the Owner's operation and use of existing facilities, the Contractor will arrange its work to minimize its effect on the existing facility, and will coordinate the timing, sequencing and duration of the disruptions with the Owner's representative and proceed with such work only after receiving authority to do so.

§ 3.14 Cutting and Patching

- § 3.14.1 The Contractor shall be responsible for cutting, fitting, or patching required to complete the Work or to make its parts fit together properly. All areas requiring cutting, fitting, and patching shall be restored to the condition existing prior to the cutting, fitting, and patching, unless otherwise required by the Contract Documents.
- § 3.14.2 The Contractor shall not damage or endanger a portion of the Work or fully or partially completed construction of the Owner or Separate Contractors by cutting, patching, or otherwise altering such construction, or by excavation. The Contractor shall not cut or otherwise alter such construction by the Owner or a Separate Contractor except with written consent of the Owner and of such Separate Contractor. Consent shall not be unreasonably withheld. The Contractor shall not unreasonably withhold, from the Owner or a Separate Contractor, its consent to cutting or otherwise altering the Work.

§ 3.15 Cleaning Up

User Notes:

- § 3.15.1 The Contractor shall keep the premises and surrounding area free from accumulation of waste materials or rubbish caused by operations under the Contract. At completion of the Work, the Contractor shall remove waste materials, rubbish, the Contractor's tools, construction equipment, machinery, and surplus materials from and about the Project.
- § 3.15.2 If the Contractor fails to clean up as provided in the Contract Documents, the Owner may do so and the Owner shall be entitled to reimbursement from the Contractor.

(1162898282)

§ 3.16 Access to Work

The Contractor shall provide the Owner and Architect with access to the Work in preparation and progress wherever located.

§ 3.17 Royalties, Patents and Copyrights

The Contractor shall pay all royalties and license fees. The Contractor shall defend suits or claims for infringement of copyrights and patent rights and shall hold the Owner and Architect harmless from loss on account thereof, but shall not be responsible for such defense or loss when a particular design, process, or product of a particular manufacturer or manufacturers is required by the Contract Documents, or where the copyright violations are contained in Drawings, Specifications, or other documents prepared by the Owner or Architect. However, if the Contractor has reason to believe that the required design, process or produce is an infringement of a copyright or patent is discovered by, or made known to, the Contractor, the Contractor shall be responsible for such loss unless such information is promptly furnished to the Architect.

§ 3.18 Indemnification

§ 3.18.1 To the fullest extent permitted by law, the Contractor shall indemnify, defend, and hold harmless the Owner and its board members, officers, directors, agents and employees from and against claims, damages, losses, and expenses, including, but not limited to attorneys' fees, arising out of or resulting from performance of the Work, provided that such claim, damage, loss, or expense is attributable to bodily injury, sickness, disease or death, or to injury to or destruction of tangible property, but only to the extent caused by the negligent acts or omissions of the Contractor, a Subcontractor, anyone directly or indirectly employed by them, or anyone for whose acts they may be liable, regardless of whether or not such claim, damage, loss, or expense is caused in part by a party indemnified hereunder. Such obligation shall not be construed to negate, abridge, or reduce other rights or obligations of indemnity that would otherwise exist as to a party or person described in this Section 3.18.

§ 3.18.2 In claims against any person or entity indemnified under this Section 3.18 by an employee of the Contractor, a Subcontractor, anyone directly or indirectly employed by them, or anyone for whose acts they may be liable, the indemnification obligation under Section 3.18.1 shall not be limited by a limitation on amount or type of damages, compensation, or benefits payable by or for the Contractor or a Subcontractor under workers' compensation acts, disability benefit acts, or other employee benefit acts.

ARTICLE 4 ARCHITECT

§ 4.1 General

§ 4.1.1 The Architect is the person or entity retained by the Owner pursuant to Section 2.3.2 and identified as such in the Agreement.

§ 4.1.2 Duties, responsibilities, and limitations of authority of the Architect as set forth in the Contract Documents shall not be restricted, modified, or extended without written consent of the Owner and written notice of such change is provided to the Contractor.

§ 4.2 Administration of the Contract

§ 4.2.1 The Architect will provide administration of the Contract as described in the Contract Documents and will be an Owner's representative (1) during construction, (2) until the date the Architect issues the final Certificate for Payment, and (3) with the Owner's concurrence, from time to time during the one-year period of correction of Work described in Section 12.2.2. The Architect will have authority to act on behalf of the Owner only to the extent provided in the Contract Documents.

§ 4.2.2 The Architect will visit the site at intervals appropriate to the stage of construction, or as otherwise agreed with the Owner, (1) to become generally familiar with the progress and quality of the portion of the Work completed, (2) to inspect the Work, (3) to determine in general if the Work is being performed in a manner indicating that the Work, when fully completed, will be in accordance with the Contract Documents, and (4) when applicable, to determine whether to issue a Certificate for Payment. However, the Architect will not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. The Architect will not have control over, charge of, or responsibility for the construction means, methods, techniques, sequences or procedures, or for the safety precautions and programs in connection with the Work, since these are solely the Contractor's rights and responsibilities under the Contract Documents, except as provided in Section 3.3.1.

§ 4.2.3 On the basis of the site visits, the Architect will keep the Owner reasonably informed about the progress and quality of the portion of the Work completed, and promptly report to the Owner (1) known deviations from the Contract Documents and from the most recent construction schedule submitted by the Contractor, and (2) defects and deficiencies observed in the Work. The Architect will not be responsible for the Contractor's failure to perform the Work in accordance with the requirements of the Contract Documents, except as set forth herein. The Architect will not have control over or charge of, and will not be responsible for acts or omissions of, the Contractor, Subcontractors, or their agents or employees, or any other persons or entities performing portions of the Work.

§ 4.2.4 Communications

Except as otherwise provided in the Contract Documents, the Owner and Contractor shall include the Architect in all communications that relate to or affect the Architect's services or professional responsibilities. The Owner shall notify the Architect of the substance of any direct communications between the Owner and the Contractor otherwise relating to the Project. Communications by and with the Architect's consultants shall be through the Architect. Communications by and with Subcontractors and suppliers shall be through the Contractor. Communications by and with Separate Contractors shall be through the Owner. The Contract Documents may specify other communication protocols.

- § 4.2.5 Based on the Architect's evaluations of the Contractor's Applications for Payment, the Architect will review and certify the amounts due the Contractor and will issue Certificates for Payment in such amounts.
- § 4.2.6 The Architect has authority to reject Work that does not conform to the Contract Documents. Whenever the Architect considers it necessary or advisable, the Architect will have authority to require inspection or testing of the Work in accordance with Sections 13.4.2 and 13.4.3, whether or not such Work is fabricated, installed or completed. However, neither this authority of the Architect nor a decision made in good faith either to exercise or not to exercise such authority shall give rise to a duty or responsibility of the Architect to the Contractor, Subcontractors, material and equipment suppliers, their agents or employees, or other persons or entities performing portions of the Work.
- § 4.2.7 The Architect will review and approve, or take other appropriate action upon, the Contractor's submittals such as Shop Drawings, Product Data, and Samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Architect's action will be taken in accordance with the submittal schedule approved by the Architect or, in the absence of an approved submittal schedule, with reasonable promptness while allowing sufficient time in the Architect's professional judgment to permit adequate review. Review of such submittals is not conducted for the purpose of determining the accuracy and completeness of other details such as dimensions and quantities, or for substantiating instructions for installation or performance of equipment or systems, all of which remain the responsibility of the Contractor as required by the Contract Documents. The Architect's review of the Contractor's submittals shall not relieve the Contractor of the obligations under Sections 3.3, 3.5, and 3.12. The Architect's review shall not constitute approval of safety precautions or, unless otherwise stated by the Architect, of any construction means, methods, techniques, sequences, or procedures. The Architect's approval of a specific item shall not indicate approval of an assembly of which the item is a component.
- § 4.2.8 The Architect will prepare Change Orders and Construction Change Directives, and may authorize minor changes in the Work as provided in Section 7.4. The Architect will investigate and make determinations and recommendations regarding concealed and unknown conditions as provided in Section 3.7.4.
- § 4.2.9 The Architect will conduct inspections to determine the date or dates of Substantial Completion and the date of final completion; issue Certificates of Substantial Completion pursuant to Section 9.8; receive and forward to the Owner, for the Owner's review and records, written warranties and related documents required by the Contract and assembled by the Contractor pursuant to Section 9.10; and issue a final Certificate for Payment pursuant to Section 9.10.
- § 4.2.10 If the Owner and Architect agree, the Architect will provide one or more Project representatives to assist in carrying out the Architect's responsibilities at the site. The duties, responsibilities and limitations of authority of such Project representatives shall be as set forth in an exhibit to be incorporated in the Contract Documents.

- § 4.2.11 The Architect will interpret and decide matters concerning performance under, and requirements of, the Contract Documents on written request of either the Owner or Contractor. The Architect's response to such requests will be made in writing within any time limits agreed upon or otherwise with reasonable promptness.
- § 4.2.12 Interpretations and decisions of the Architect will be consistent with the intent of, and reasonably inferable from, the Contract Documents and will be in writing or in the form of drawings. When making such interpretations and decisions, the Architect will endeavor to secure faithful performance by both Owner and Contractor, will not show partiality to either and will not be liable for results of interpretations or decisions rendered in good faith.
- § 4.2.13 The Architect's decisions on matters relating to aesthetic effect will be final if consistent with the intent expressed in the Contract Documents.
- § 4.2.14 The Architect will review and respond to requests for information about the Contract Documents. The Architect's response to such requests will be made in writing within any time limits agreed upon or otherwise with reasonable promptness. If appropriate, the Architect will prepare and issue supplemental Drawings and Specifications in response to the requests for information.

ARTICLE 5 SUBCONTRACTORS

§ 5.1 Definitions

- § 5.1.1 A Subcontractor is a person or entity who has a direct contract with the Contractor to perform a portion of the Work at the site. The term "Subcontractor" is referred to throughout the Contract Documents as if singular in number and means a Subcontractor or an authorized representative of the Subcontractor. The term "Subcontractor" does not include a Separate Contractor or the subcontractors of a Separate Contractor.
- § 5.1.2 A Sub-subcontractor is a person or entity who has a direct or indirect contract with a Subcontractor to perform a portion of the Work at the site. The term "Sub-subcontractor" is referred to throughout the Contract Documents as if singular in number and means a Sub-subcontractor or an authorized representative of the Sub-subcontractor.

§ 5.2 Award of Subcontracts and Other Contracts for Portions of the Work

- § 5.2.1 Unless otherwise stated in the Contract Documents or the bidding requirements, the Contractor, as soon as practicable after award of the Contract, shall furnish in writing to the Owner through the Architect the names of persons or entities proposed for each principal portion of the Work, including those who are to furnish materials or equipment fabricated to a special design. Within 10 days of receipt of the information, the Owner or Architect may notify the Contractor in writing whether the Owner or the Architect has reasonable objection to any such proposed person or entity or that the Owner or the Architect requires additional time for review. Failure of the Owner or Architect to provide notice within the 10-day period shall constitute notice of no reasonable objection.
- § 5.2.2 The Contractor shall not contract with a proposed person or entity to whom the Owner or Architect has made reasonable and timely objection. The Contractor shall not be required to contract with anyone to whom the Contractor has made reasonable objection. It shall be considered reasonable for the Contractor to reject a Subcontractor who fails or refuses to sign a form of subcontract reasonably acceptable to the Contractor.
- § 5.2.3 If the Owner or Architect has reasonable objection to a person or entity proposed by the Contractor, the Contractor shall propose another to whom the Owner or Architect has no reasonable objection. If the proposed but rejected Subcontractor was reasonably capable of performing the Work, the Contract Sum and Contract Time shall be increased or decreased by the difference, if any, occasioned by such change, and an appropriate Change Order shall be issued before commencement of the substitute Subcontractor's Work. However, no increase in the Contract Sum or Contract Time shall be allowed for such change unless the Contractor has acted promptly and responsively in submitting names as required.
- § 5.2.4 The Contractor shall not substitute a Subcontractor, person, or entity previously selected if the Owner or Architect makes reasonable objection to such substitution.

§ 5.3 Subcontractual Relations

§ 5.3.1 By appropriate written agreement, the Contractor shall require each Subcontractor, to the extent of the Work to be performed by the Subcontractor, to be bound to the Contractor by terms of the Contract Documents, and to assume toward the Contractor all the obligations and responsibilities, including the responsibility for safety of the

Subcontractor's Work that the Contractor, by these Contract Documents, assumes toward the Owner and Architect. Each subcontract agreement shall preserve and protect the rights of the Owner and Architect under the Contract Documents with respect to the Work to be performed by the Subcontractor so that subcontracting thereof will not prejudice such rights. Where appropriate, the Contractor shall require each Subcontractor to enter into similar agreements with Sub-subcontractors. The Contractor shall make available to each proposed Subcontractor, prior to the execution of the subcontract agreement, copies of the Contract Documents to which the Subcontractor will be bound. Subcontractors will similarly make copies of applicable portions of such documents available to their respective proposed Sub-subcontractors.

§ 5.3.2 Owner is an intended third-party beneficiary of all subcontracts and material supply contracts of whatever tier, with the right to directly enforce, both during and after the construction period, subcontractor and material supplier obligations to meet prevailing standards or workmanship and to comply with the contract documents including but not limited to all applicable express and implied warranties. During the construction period, that right shall only be exercised in cooperation with Contractor, unless Contractor is in default under the Contract and fails to cure the same.

§ 5.4 Contingent Assignment of Subcontracts

- § 5.4.1 Each subcontract agreement for a portion of the Work is assigned by the Contractor to the Owner, provided that
 - assignment is effective only after termination of the Contract by the Owner for cause pursuant to Section 14.2 and only for those subcontract agreements that the Owner accepts by notifying the Subcontractor and Contractor in writing; and
 - .2 assignment is subject to the prior rights of the surety, if any, obligated under bond relating to the Contract.

When the Owner accepts the assignment of a subcontract agreement, the Owner assumes the Contractor's rights under the subcontract.

- § 5.4.2 Upon such assignment, if the Work has been suspended for more than 30 days, the Subcontractor's compensation shall be equitably adjusted for actual increases in the direct costs resulting from the suspension.
- § 5.4.3 Upon such assignment to the Owner under this Section 5.4, the Owner may further assign the subcontract to a successor contractor or other entity.

ARTICLE 6 CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS

§ 6.1 Owner's Right to Perform Construction and to Award Separate Contracts

- § 6.1.1 The term "Separate Contractor(s)" shall mean other contractors retained by the Owner under separate agreements. The Owner reserves the right to perform construction or operations related to the Project with the Owner's own forces, and with Separate Contractors retained in connection with other portions of the Project or other construction or operations on the Project site under Conditions of the Contract identical or substantially similar to those of this Contract, including those provisions related to insurance and waiver of subrogation.
- § 6.1.2 When separate contracts are awarded for different portions of the Project or other construction or operations on the site, the term "Contractor" in the Contract Documents in each case shall mean the Contractor who executes each separate Owner-Contractor Agreement.
- § 6.1.3 The Owner shall provide for coordination of the activities of the Owner's own forces and of each Separate Contractor with the Work of the Contractor, who shall cooperate with them. The Contractor shall participate with other Separate Contractors and the Owner in reviewing their construction schedules. The Contractor shall make any revisions to the construction schedule deemed necessary after a joint review and mutual agreement. The construction schedules shall then constitute the schedules to be used by the Contractor, Separate Contractors, and the Owner until subsequently revised.
- § 6.1.4 Unless otherwise provided in the Contract Documents, when the Owner performs construction or operations related to the Project with the Owner's own forces or with Separate Contractors, the Owner or its Separate Contractors shall be deemed to be subject to the same obligations and to have the same rights that apply to the Contractor has under the Conditions of the Contract, as stated in Article 3, this Article 6, and Articles 10, 11, and 12.

§ 6.2 Mutual Responsibility

- § 6.2.1 The Contractor shall afford the Owner and Separate Contractors reasonable opportunity for introduction and storage of their materials and equipment and performance of their activities, and shall connect and coordinate the Contractor's construction and operations with theirs as required by the Contract Documents.
- § 6.2.2 If part of the Contractor's Work depends for proper execution or results upon construction or operations by the Owner or a Separate Contractor, the Contractor shall, prior to proceeding with that portion of the Work, promptly report to the Architect apparent discrepancies or defects in such other construction or operations that would render it unsuitable for such proper execution. Failure of the Contractor to so report shall constitute an acknowledgment that the Owner's or Separate Contractor's completed or partially completed construction is fit and proper to receive the Contractor's Work, except as to defects not then reasonably discoverable.
- § 6.2.3 The Contractor shall reimburse the Owner for costs the Owner incurs that are payable to a Separate Contractor because of the Contractor's delays, improperly timed activities or defective construction. The Owner shall be responsible to the Contractor for actual and direct costs the Contractor incurs because of a Separate Contractor's delays, improperly timed activities, damage to the Work or defective construction.
- § 6.2.4 The Contractor shall promptly remedy damage that the Contractor wrongfully causes to completed or partially completed construction or to property of the Owner or Separate Contractor as provided in Section 10.2.5, but only to the extent caused by Contractor.
- **§ 6.2.5** The Owner and each Separate Contractor shall have the same responsibilities for cutting and patching as are described for the Contractor in Section 3.14.

§ 6.3 Owner's Right to Clean Up

If a dispute arises among the Contractor, Separate Contractors, and the Owner as to the responsibility under their respective contracts for maintaining the premises and surrounding area free from waste materials and rubbish, the Owner may clean up allocate the cost among those responsible.

ARTICLE 7 CHANGES IN THE WORK

§ 7.1 General

- § 7.1.1 Changes in the Work may be accomplished after execution of the Contract, and without invalidating the Contract, by Change Order, Construction Change Directive or order for a minor change in the Work, subject to the limitations stated in this Article 7 and elsewhere in the Contract Documents.
- § 7.1.2 A Change Order shall be based upon agreement among the Owner, Contractor, and Architect. A Construction Change Directive requires agreement by the Owner and Architect and may or may not be agreed to by the Contractor. An order for a minor change in the Work may be issued by the Architect alone.
- § 7.1.3 Changes in the Work shall be performed under applicable provisions of the Contract Documents. The Contractor shall proceed promptly with changes in the Work, unless otherwise provided in the Change Order, Construction Change Directive, or order for a minor change in the Work.
- § 7.1.4 Except for reasons of emergency or for minor changes in the Work (see Sections 7.4 herein), there will be no Changes in the Work unless first authorized in writing by Owner and Architect pursuant to the terms of this Article 7.

§ 7.2 Change Orders

- § 7.2.1 A Change Order is a written instrument prepared by the Architect and signed by the Owner, Contractor, and Architect stating their agreement upon all of the following:
 - .1 The change in the Work;
 - .2 The amount of the adjustment, if any, in the Contract Sum; and
 - **.3** The extent of the adjustment, if any, in the Contract Time.
- § 7.2.2 Methods used in determining adjustments to the Contract Sum may include those listed in Section 7.3.3.

§ 7.3 Construction Change Directives

- § 7.3.1 A Construction Change Directive is a written order prepared by the Owner and signed by the Owner and Architect, the form of which is attached hereto as "Exhibit 3," directing a change in the Work prior to agreement on adjustment, if any, in the Contract Sum or Contract Time, or both. The Owner may by Construction Change Directive, without invalidating the Contract, order changes in the Work within the general scope of the Contract consisting of additions, deletions, or other revisions, the Contract Sum and Contract Time being adjusted accordingly.
- § 7.3.2 A Construction Change Directive shall be used in the absence of total agreement on the terms of a Change Order.
- § 7.3.3 If the Construction Change Directive provides for an adjustment to the Contract Sum, the adjustment shall be based on one of the following methods:
 - .1 Mutual acceptance of a lump sum properly itemized and supported by sufficient substantiating data to permit evaluation;
 - .2 Unit prices stated in the Contract Documents or subsequently agreed upon;
 - 3 Cost to be determined in a manner agreed upon by the parties and a mutually acceptable fixed or percentage fee; or
 - .4 As provided in Section 7.3.4.
- § 7.3.4 If the Contractor does not respond promptly or disagrees with the method for adjustment in the Contract Sum, the Architect shall determine the adjustment on the basis of reasonable expenditures and savings of those performing the Work attributable to the change, including, in case of an increase in the Contract Sum, an amount for overhead and profit as set forth in the Agreement, or if no such amount is set forth in the Agreement, a reasonable amount, all on a not to exceed basis unless otherwise agreed. In such case, and also under Section 7.3.3.3, the Contractor shall keep and present, an itemized accounting together with appropriate supporting data. Unless otherwise provided in the Contract Documents, costs for the purposes of this Section 7.3.4 shall be limited to the following:
 - Actual costs of labor, including social security, old age and unemployment insurance, fringe benefits required by agreement or custom, and workers' compensation insurance,;
 - .2 Costs of materials, supplies, and equipment, including cost of transportation, whether incorporated or consumed:
 - .3 Rental costs of machinery and equipment, exclusive of hand tools, whether rented from the Contractor or others:
 - .4 Costs of premiums for all bonds and insurance (itemized separately and subject to the provisions of the Agreement), permit fees, and sales, use, or similar taxes, directly related to the change; and
 - .5 Additional costs of supervision and field office personnel directly attributable to the change.
- § 7.3.5 If the Contractor disagrees with the adjustment in the Contract Time, the Contractor may make a Claim in accordance with applicable provisions of Article 15.
- § 7.3.6 Upon receipt of a Construction Change Directive, the Contractor shall promptly proceed with the change in the Work involved and advise the Architect of the Contractor's agreement or disagreement with the method, if any, provided in the Construction Change Directive for determining the proposed adjustment in the Contract Sum or Contract Time.
- § 7.3.7 A Construction Change Directive signed by the Contractor indicates the Contractor's agreement therewith, including adjustment in Contract Sum and Contract Time or the method for determining them. Such agreement shall be effective immediately and shall be recorded as a Change Order.
- § 7.3.8 The amount of credit to be allowed by the Contractor to the Owner for a deletion or change that results in a net decrease in the Contract Sum shall be actual net cost as confirmed by the Architect. When both additions and credits covering related Work or substitutions are involved in a change, the allowance for overhead and profit shall be figured on the basis of net increase, if any, with respect to that change.
- § 7.3.9 The Contractor may request payment for changes in the Work not in dispute in Applications for Payment accompanied by a Change Order indicating the parties' agreement with part or all of such costs. For any portion of such cost that remains in dispute, the Architect will make an interim determination for purposes of monthly certification for payment for those costs and certify for payment the amount that the Architect determines, in the

Architect's professional judgment, to be reasonably justified. The Architect's interim determination of cost shall adjust the Contract Sum on the same basis as a Change Order, subject to the right of either party to disagree and assert a Claim in accordance with Article 15.

§ 7.3.10 When the Owner and Contractor reach an agreement concerning the adjustments in the Contract Sum and Contract Time, or otherwise reach agreement upon the adjustments, such agreement shall be effective immediately and shall be recorded by preparation and execution of an appropriate Change Order. The Contractor shall be responsible for preparation of the Change Order. The Contractor's agreement with a Change Order constitutes a waiver and release by the Contractor and its Subcontractors and suppliers for any claim for delay, cumulative impact, cost of extended general conditions, and any other indirect cost associated with the Owner-directed change in Work.

§ 7.4 Minor Changes in the Work

The Architect has the authority to order minor changes in the Work that are consistent with the intent of the Contract Documents and do not involve an adjustment in the Contract Sum or an extension of the Contract Time. The Architect's order for minor changes shall be in writing signed by the Architect and shall be binding on the Owner and Contractor. The Contractor shall carry out such written orders promptly.

§ 7.5 FIELD CHANGES

- § 7.5.1 Upon the written request of Contractor, Owner may authorize minor changes in the Work which occur in the field ("Field Change(s)") without prior presentation to the Architect, so long as each of the following criteria is satisfied:
 - the proposed Field Change is consistent with the intent of the Contract Documents;
 - the proposed Field Change will not result in an extension of the Contract Time;
 - the proposed Field Change will not result in a cumulative increase in the GMP or the Contract Sum by more than \$5,000.00.
- § 7.5.2 By presenting the proposed Field Change, the Contractor is verifying that each of the criteria listed above is satisfied with respect to the proposed Field Change.
- § 7.5.3 If a Field Change may result in a cumulative increase in the Contract Sum of more than \$5,000.00, the Field Change set forth in Section 7.5.1 above does not apply. Rather, the Contractor must follow the written change order provisions set forth in Sections 7.1 and 7.2 above.
- § 7.5.4 The issuance of a Field Change and the provisions of this Section 7.5 shall not limit the Owner's right of review and approval of any Field Change in the progress payment process or as part of final payment.

ARTICLE 8 TIME

§ 8.1 Definitions

- § 8.1.1 Unless otherwise provided, Contract Time is the period of time, including authorized adjustments, allotted in the Contract Documents for Substantial Completion of the Work.
- § 8.1.2 The date of commencement of the Work is the date established in the Notice to Proceed. If there is no Notice to Proceed, it shall be the date of the Owner-Contractor Agreement or such other date as may be established therein.
- § 8.1.3 The date of Substantial Completion of the Work or designated portion thereof is the date certified by the Architect in accordance with Section 9.8.
- § 8.1.4 The term "day" as used in the Contract Documents shall mean calendar day unless otherwise specifically defined.

§ 8.2 Progress and Completion

§ 8.2.1 Time limits stated in the Contract Documents are of the essence of the Contract. By executing the Agreement, the Contractor confirms that the Contract Time is a reasonable period for performing the Work and the dates established for Substantial Completion and Final Completion are reasonable.

User Notes:

(1162898282)

§ 8.2.2 The Contractor shall not knowingly, except by agreement or instruction of the Owner in writing, prematurely commence operations on the Project site or elsewhere prior to the effective date of insurance required by Article 11 to be furnished by the Contractor and Owner. The date of commencement of the Work shall not be changed by the effective date of such insurance. Unless the date of commencement is established by the Contract Documents or a notice to proceed given by the Owner, the Contractor shall notify the Owner in writing not less than five (5) days or other agreed period before commencing the Work to permit the timely filing of mortgages, mechanic's liens and other security interests.

§ 8.2.3 The Contractor shall proceed expeditiously with adequate forces and shall achieve Substantial Completion within the Contract Time.

§ 8.3 Delays and Extensions of Time

§ 8.3.1 If the Contractor is delayed at any time in the commencement or progress of the Work by (1) an act of negligence of the Owner or Architect, or of an employee of either, or of a Separate Contractor; (2) by changes ordered in the Work; (3) by labor disputes, fire, unusual delay in deliveries, unavoidable casualties or other causes beyond the Contractor's control; (4) by delay authorized by the Owner pending mediation and binding dispute resolution; or (5) by other causes that the Contractor asserts, and the Architect and/or Owner determines may justify delay, then the Contract Time shall be extended by Change Order for such reasonable time as the Owner may allow.

§ 8.3.2 Claims relating to time shall be made in accordance with applicable provisions of Article 15.

§ 8.3.3 Contractor's remedy for delays caused by the Owner or Architect shall be limited solely to (a) extra time; and (b) any applicable "general conditions" expenses allowed by Section 9.7 and Section 14.3 of the General Conditions. With respect to any delays caused by Owner or Architect, if Owner contends that acceleration of the Work is necessary, Contractor shall accelerate the Work as directed by Owner. Owner shall pay Contractor only the Contractor's extra labor cost over the amount for regular time during the period of such overtime, including additional insurance and taxes incurred by the Contractor with respect thereto. Time slips covering said overtime must be submitted to Owner's designated representative for checking and approval.

Contractor's sole remedy in the event of (i) any delay not caused by or attributable to the Owner; or (ii) a delay excusable per Section 8.3.1 above; shall be an equitable extension of the Contract Time and Contractor shall not obtain any monetary damages for the delay or any addition to the Contract Sum or Guaranteed Maximum Price; however, Contractor shall be entitled to use Contractor's construction contingency funds (if any) to cover additional costs incurred as a result of any such delay.

Notwithstanding any implication to the contrary in Section 8.3.1 or elsewhere herein, Contractor shall not be entitled to any time extension for any delay that is the fault of Contractor or any of its Subcontractors or suppliers, or their Subcontractors or suppliers at any level. There shall be no adjustment of compensation for acceleration efforts undertaken by Contractor to meet the Substantial Completion date, as the same may be modified as provided for herein, as a result of any delay that is the fault of Contractor or any of its Subcontractors or suppliers, or their Subcontractors or supplier at any level. However, Contractor shall be entitled to use Contractor's construction contingency funds (if any) to cover additional costs incurred as a result of any such delay.

§ 8.3.4 If there is a delay or anticipated delay of the Construction Schedule because of the actions or omissions of the Contractor or any Subcontractor and the Contractor is unable to produce within fifteen (15) working days after notice by the Owner a recovery schedule for its Work to address such delay or anticipated delay acceptable to the Owner, then the Owner shall have the right to order the Contractor to take such actions as may be necessary, consistent with the same performance of the Work affected thereby, to recapture the time lost by any such delay. Such action shall include increasing staff; increase in shifts or hours worked per day, or performance of work on Saturdays, Sundays or national holidays; use of any available work float in the Project schedule; and changing the sequence of work activities. Owner shall pay Contractor only the Contractor's extra labor cost over the amount for regular time during the period of such overtime, including additional insurance and taxes incurred by the Contractor with respect thereto. Time slips covering said overtime must be submitted to Owner's designated representative for checking and approval.

The Owner, at its option, shall also be entitled to accelerate performance of the Work where there is no delay or anticipated delay. The costs of such acceleration where there is no delay or anticipated delay shall be paid for by the

Owner.

§ 8.3.5 If Contractor intends to claim an extension of time to perform as a result of a delay not caused by Contractors or any of its Subcontractors or suppliers, Contractor must give Owner written notice within seven (7) days after the event giving rise to the claim and follow the procedures in Article 7 for securing a Change Order; otherwise such a claim shall be deemed waived.

ARTICLE 9 PAYMENTS AND COMPLETION

§ 9.1 Contract Sum

§ 9.1.1 The Contract Sum is stated in the Agreement and, including authorized adjustments, is the total amount payable by the Owner to the Contractor for performance of the Work under the Contract Documents.

(Paragraph deleted)

§ 9.2 Schedule of Values

Where the Contract is based on a stipulated sum or Guaranteed Maximum Price, the Contractor shall submit a schedule of values to the Owner and Architect before the first Application for Payment, allocating the entire Contract Sum to the various portions of the Work. The schedule of values shall be prepared in the form, and supported by the data to substantiate its accuracy, required by the Architect. This schedule, unless objected to by the Owner or Architect, shall be used as a basis for reviewing the Contractor's Applications for Payment.

§ 9.3 Applications for Payment

- § 9.3.1 At least ten days before the date established for each progress payment, the Contractor shall submit to the Owner and Architect an itemized Application for Payment prepared in accordance with the schedule of values, if required under Section 9.2, for completed portions of the Work. Such application shall be notarized, if required, and supported by such data substantiating the Contractor's right to payment as the Owner or Architect may require, such as copies of requisitions, and releases and waivers of liens from Subcontractors and material suppliers, and shall reflect retainage if provided for in the Contract Documents.
- § 9.3.1.1 As provided in Section 7.3.9, such applications may include requests for payment on account of changes in the Work that have been properly authorized by Construction Change Directives, or by interim determinations of the Architect, but not yet included in Change Orders.
- § 9.3.1.2 Applications for Payment shall not include requests for payment for portions of the Work for which the Contractor does not intend to pay a Subcontractor or material supplier, unless such Work has been performed by others whom the Contractor intends to pay.
- § 9.3.2 Unless otherwise provided in the Contract Documents, payments shall be made on account of materials and equipment delivered and suitably stored at the site for subsequent incorporation in the Work. If approved in advance by the Owner, payment may similarly be made for materials and equipment suitably stored off the site at a location agreed upon in writing. Payment for materials and equipment stored on or off the site shall be conditioned upon compliance by the Contractor with procedures satisfactory to the Owner to establish the Owner's title to such materials and equipment or otherwise protect the Owner's interest, and shall include the costs of applicable insurance, storage, and transportation to the site for such materials and equipment stored off the site.
- § 9.3.3 The Contractor warrants that title to all Work covered by an Application for Payment will pass to the Owner no later than the time of payment. The Contractor further warrants that upon submittal of an Application for Payment all Work for which Certificates for Payment have been previously issued and payments received from the Owner shall, to the best of the Contractor's knowledge, information, and belief, be free and clear of liens, claims, security interests, or encumbrances, in favor of the Contractor, Subcontractors, material suppliers, or other persons or entities making a claim by reason of having provided labor, materials, and equipment relating to the Work.
- § 9.3.4 With each Application for Payment, the Contractor shall provide to Owner; (a) a partial lien waiver and release from each subcontractor or supplier who was to receive funds from the previous Application for Payment, and (b) the Contractor's own conditional lien waiver for all funds to be received for that Application.

(1162898282)

§ 9.3.5 Provided that Owner shall make payment of amounts due to Contractor, Contractor shall keep the Project site free from any liens by Contractor, its Subcontractors, Architect or engineers retained by Contractor, or their respective assigns. No Uniform Commercial Code (UCC) or similar state statutory filings shall be made by Contractor, Subcontractors, or such Architect or engineers, except upon express prior written consent of Owner. If, because of any act or omission of Contractor or anyone claiming through or under Contractor, any mechanic's or other lien, encumbrance or order is filed against Owner or the Project site, Contractor shall, at Contractor's own cost and expense, cause the same to be canceled and discharged of record or bonded over within thirty (30) days after receiving notice thereof, and shall indemnify and hold Owner harmless from and against all costs, expenses, claims, losses or damages, including reasonable attorneys' fees, resulting therefrom or by reason thereof.

§ 9.4 Certificates for Payment

§ 9.4.1 The Architect will, within seven days after receipt of the Contractor's Application for Payment, either issue to the Owner a Certificate for Payment, with a copy to the Contractor for such amount as the Architect determines is properly due, or notify the Contractor and Owner in writing of the Architect's reasons for withholding certification in part as provided in Section 9.5.1.

§ 9.4.2 The issuance of a Certificate for Payment will constitute a representation by the Architect to the Owner, based on the Architect's evaluation of the Work and the data comprising the Application for Payment, that, to the best of the Architect's knowledge, information, and belief, the Work has progressed to the point indicated, and that the quality of the Work is in accordance with the Contract Documents. The foregoing representations are subject to an evaluation of the Work for conformance with the Contract Documents upon Substantial Completion, to results of subsequent tests and inspections, to correction of minor deviations from the Contract Documents prior to completion, and to specific qualifications expressed by the Architect. The issuance of a Certificate for Payment will further constitute a representation that the Contractor is entitled to payment in the amount certified. However, the issuance of a Certificate for Payment will not be a representation that the Architect has (1) made exhaustive or continuous on-site inspections to check the quality or quantity of the Work; (2) reviewed construction means, methods, techniques, sequences, or procedures; (3) reviewed copies of requisitions received from Subcontractors and material suppliers and other data requested by the Owner to substantiate the Contractor's right to payment; or (4) made examination to ascertain how or for what purpose the Contractor has used money previously paid on account of the Contract Sum.

§ 9.5 Decisions to Withhold Certification

§ 9.5.1 The Architect and/or Owner may withhold a Certificate for Payment in whole or in part, to the extent reasonably necessary to protect the Owner, if in the Owner's or Architect's opinion the representations to the Owner required by Section 9.4.2 cannot be made. If the Architect is unable to certify payment in the amount of the Application, the Architect will notify the Contractor and Owner as provided in Section 9.4.1. If the Contractor and Architect cannot agree on a revised amount, the Architect will promptly issue a Certificate for Payment for the amount for which the Architect is able to make such representations to the Owner and the Owner shall pay Contractor in accordance with Section 9.6.1. The Architect and/or Owner may also withhold a Certificate for Payment or, because of subsequently discovered evidence, may nullify the whole or a part of a Certificate for Payment previously issued, to such extent as may be necessary in the Owner's or Architect's opinion to protect the Owner from loss for which the Contractor is responsible, including loss resulting from acts and omissions described in Section 3.3.2, because of

- .1 defective Work not remedied;
- .2 third party claims not resulting from the Owner's failure or refusal to pay Contractor filed or reasonable evidence indicating probable filing of such claims, unless security acceptable to the Owner is provided by the Contractor;
- failure of the Contractor to make payments properly to Subcontractors or for labor, materials or equipment;
- .4 reasonable evidence that the Work cannot be completed for the unpaid balance of the Contract Sum;
- .5 damage to the Owner or a Separate Contractor and not covered by insurance;
- .6 reasonable evidence that the Work will not be completed within the Contract Time as a result of the Contractor's fault, and that the unpaid balance would not be adequate to cover actual or liquidated damages for the anticipated delay;
- .7 repeated failure to carry out the Work in accordance with the Contract Documents;
- .8 failure of the Contractor to comply with the most current approved Project construction schedule; or
- .9 the existence of any event of material default under the Contract Documents

§ 9.5.2 INTENTIONALLY OMITTED

- § 9.5.3 When the above reasons for withholding certification are removed, certification will be made for amounts previously withheld.
- § 9.5.4 If the Owner or Architect withholds certification for payment under Section 9.5.1.3, the Owner may, at its sole option, issue joint checks to the Contractor and to any Subcontractor or material or equipment suppliers to whom the Contractor failed to make payment for Work properly performed or material or equipment suitably delivered. If the Owner makes payments by joint check, the Owner shall notify the Architect and the Architect shall reflect such payment on the next Certificate for Payment.

§ 9.6 Progress Payments

- § 9.6.1 After the Architect has issued a Certificate for Payment, the Owner shall make payment in the manner and within the time provided in the Contract Documents, and shall so notify the Architect.
- § 9.6.2 The Contractor shall pay each Subcontractor, no later than seven days after receipt of payment from the Owner, the amount to which the Subcontractor is entitled, either by agreement or by reflecting percentages actually retained from payments to the Contractor on account of the Subcontractor's portion of the Work. The Contractor shall, by appropriate agreement with each Subcontractor, require each Subcontractor to make payments to Sub-subcontractors in a similar manner.

§ 9.6.3 INTENTIONALLY OMITTED

- § 9.6.4 The Owner has the right to request written evidence from the Contractor that the Contractor has properly paid Subcontractors and material and equipment suppliers amounts paid by the Owner to the Contractor for subcontracted Work. If the Contractor fails to furnish such evidence within seven days, the Owner shall have the right to contact Subcontractors and material and equipment suppliers to ascertain whether they have been properly paid. Neither the Owner nor Architect shall have an obligation to pay, or to see to the payment of money to, a Subcontractor or material or equipment supplier, except as may otherwise be required by law.
- § 9.6.5 The Contractor's payments to material and equipment suppliers shall be treated in a manner similar to that provided in Sections 9.6.2, 9.6.3 and 9.6.4.
- § 9.6.6 A Certificate for Payment, a progress payment, or partial or entire use or occupancy of the Project by the Owner shall not constitute acceptance of Work not in accordance with the Contract Documents.
- § 9.6.7 Unless the Contractor provides the Owner with a payment bond in the full penal sum of the Contract Sum, payments received by the Contractor for Work properly performed by Subcontractors or provided by material and equipment suppliers shall be held by the Contractor for those Subcontractors or suppliers who performed Work or furnished materials, or both, under contract with the Contractor for which payment was made by the Owner. Nothing contained herein shall require money to be placed in a separate account and not commingled with money of the Contractor, shall create any fiduciary liability or tort liability on the part of the Contractor for breach of trust, or shall entitle any person or entity to an award of punitive damages against the Contractor for breach of the requirements of this provision.
- § 9.6.8 Provided the Owner has fulfilled its payment obligations under the Contract Documents, the Contractor shall defend and indemnify the Owner from all loss, liability, damage or expense, including reasonable attorney's fees and litigation expenses, arising out of any lien claim or other claim for payment by any Subcontractor or supplier of any tier. Upon receipt of notice of a lien claim or other claim for payment, the Owner shall notify the Contractor. If approved by the applicable court, when required, the Contractor may substitute a surety bond for the property against which the lien or other claim for payment has been asserted.

§ 9.7 Failure of Payment

User Notes:

If the Architect does not issue a Certificate for Payment, through no fault of the Contractor, within seven days after receipt of the Contractor's Application for Payment, or if the Owner does not pay the Contractor within seven days after the date established in the Contract Documents, the amount certified by the Architect or awarded by binding dispute resolution, then the Contractor may, upon seven additional days' written notice to the Owner and Architect, stop the Work until payment of the amount owing has been received. The Contract Time shall be extended

appropriately and the Contract Sum shall be increased by the amount of the Contractor's reasonable costs of shutdown, delay and start-up, plus interest as provided for in the Contract Documents.

§ 9.8 Substantial Completion

- § 9.8.1 Substantial Completion is the stage in the progress of the Work when (i) the Work or designated portion thereof is sufficiently complete in accordance with the Contract Documents so that the Owner can occupy or utilize the Work for its intended use, subject only to the completion of Punch list items or other Work that cannot be completed because of weather conditions or because of unavailability of products or materials, none of which may materially affect Owner's intended use and enjoyment of the Project or cause unreasonable disruption of interference of Owner's intended use or otherwise affect Owner's ability to fully utilize the Project for its intended use; (ii) all required permits and approvals, including a certificate of occupancy, have been issued and obtained, unless any such permit, approval or certificate has not been issued through no fault of the Contractor; and (iii) all utilities required for Owner's operation of the Project are in place and operational, including gas, electricity, water, sanitary, sewer, and telephone; (iv) a Certificate of Substantial Completion is issued and signed-off on by Owner; and (v) Contractor has complied with all terms and conditions of the Agreement.
- § 9.8.2 When the Contractor considers that the Work, or a portion thereof which the Owner agrees to accept separately, is substantially complete, the Contractor shall prepare and submit to the Architect a comprehensive list of items to be completed or corrected prior to final payment. The Contractor shall proceed promptly to complete and correct items on the list. Failure to include an item on such list does not alter the responsibility of the Contractor to complete all Work in accordance with the Contract Documents. Observation by the Architect and/or the Owner during the construction shall not be considered as acceptance of any part of the Work, and the conditions found when final inspection is made shall be taken as governing the fitness of the Work and whether or not it conforms to the provisions of the Contract.
- § 9.8.3 Upon receipt of the Contractor's list, the Architect will make an inspection to determine whether the Work or designated portion thereof is substantially complete. If the Architect's inspection discloses any item, whether or not included on the Contractor's list not sufficiently complete in accordance with the Contract Documents so that the Owner can occupy or utilize the Work or designated portion thereof for its intended use, the Contractor shall, before issuance of the Certificate of Substantial Completion, complete or correct such item upon notification by the Architect. In such case, the Contractor shall then submit a request for another inspection by the Architect to determine Substantial Completion.
- § 9.8.4 When the Work or designated portion thereof is substantially complete, the Architect will prepare a Certificate of Substantial Completion that shall establish the date of Substantial Completion; shall establish responsibilities of the Owner and Contractor for security, maintenance, heat, utilities, damage to the Work and insurance; and shall fix the time within which the Contractor shall finish all items on the list accompanying the Certificate. Warranties required by the Contract Documents shall commence on the date of Substantial Completion of the Work or designated portion thereof unless otherwise provided in the Certificate of Substantial Completion.
- § 9.8.5 The Certificate of Substantial Completion shall be submitted to the Owner and Contractor for their written acceptance of responsibilities assigned to them in such Certificate. Upon such acceptance, and consent of surety if any, the Owner shall make payment of retainage applying to such Work or designated portion thereof. Such payment shall be adjusted for Work that is incomplete or not in accordance with the requirements of the Contract Documents.
- § 9.8.6 Contractor shall reach Final Completion of the Work within 60 days of the date of Substantial Completion except for those items which cannot be completed because of weather conditions or because of unavailability of products or materials or that Owner agrees do not have to be completed within the 60-day period.

§ 9.9 Partial Occupancy or Use

§ 9.9.1 The Owner may occupy or use any completed or partially completed portion of the Work at any stage when such portion is designated by separate agreement with the Contractor, provided such occupancy or use is consented to by the insurer as required under Section 11.2.9 and authorized by public authorities having jurisdiction over the Project. Such partial occupancy or use may commence whether or not the portion is substantially complete, provided the Owner and Contractor have accepted in writing the responsibilities assigned to each of them for payments, retainage, if any, security, maintenance, heat, utilities, damage to the Work and insurance, and have agreed in writing concerning the period for correction of the Work and commencement of warranties required by the Contract

Documents. When the Contractor considers a portion substantially complete, the Contractor shall prepare and submit a list to the Architect as provided under Section 9.8.2. Consent of the Contractor to partial occupancy or use shall not be unreasonably withheld. The stage of the progress of the Work shall be determined by written agreement between the Owner and Contractor or, if no agreement is reached, by decision of the Architect.

§ 9.9.2 Immediately prior to such partial occupancy or use, the Owner, Contractor, and Architect shall jointly inspect the area to be occupied or portion of the Work to be used in order to determine and record the condition of the Work.

§ 9.9.3 Unless otherwise agreed upon in writing and signed by Owner, partial occupancy or use of a portion or portions of the Work shall not constitute acceptance of Work not complying with the requirements of the Contract Documents.

§ 9.10 Final Completion and Final Payment

§ 9.10.1 Upon receipt of the Contractor's written notice that the Work is ready for final inspection and acceptance and upon receipt of a final Application for Payment, the Architect will promptly make such inspection, and when the Architect finds the Work acceptable under the Contract Documents and the Contract fully performed, the Architect will promptly issue a final Certificate for Payment stating that to the best of the Architect's knowledge, information and belief, and on the basis of the Architect's on-site visits and inspections, the Work has been completed in accordance with terms and conditions of the Contract Documents and that the entire balance found to be due the Contractor and noted in the final Certificate is due and payable. The Architect's final Certificate for Payment will constitute a further representation that conditions listed in Section 9.10.2 as precedent to the Contractor's being entitled to final payment have been fulfilled.

§ 9.10.2 Neither final payment nor any remaining retained percentage shall become due until the Contractor submits to the Architect (1) an affidavit that payrolls, bills for materials and equipment, and other indebtedness connected with the Work for which the Owner or the Owner's property might be responsible or encumbered (less amounts withheld by Owner) have been paid or otherwise satisfied, (2) a certificate evidencing that insurance required by the Contract Documents to remain in force after final payment is currently in effect and will not be cancelled or allowed to expire until at least 30 days' prior written notice has been given to the Owner, (3) a written statement that the Contractor knows of no substantial reason that the insurance will not be renewable to cover the period required by the Contract Documents, (4) consent of surety, if any, to final payment, (5) documentation of any special warranties, such as manufacturers' warranties or specific Subcontractor warranties, (6) a marked-up version of all Construction Documents showing the record condition of the Work; (7) a Final Lien Waiver and Release from Contractor and from all subcontractors and suppliers who provided any labor, materials or Work for the Project; and (8) if required by the Owner, other data establishing payment or satisfaction of obligations, such as receipts and releases and waivers of liens, claims, security interests, or encumbrances arising out of the Contract, to the extent and in such form as may be designated by the Owner. If a Subcontractor refuses to furnish a release or waiver required by the Owner, the Contractor may furnish a bond satisfactory to the Owner to indemnify the Owner against such lien, claim, security interest, or encumbrance. If such lien, claim, security interest, or encumbrance remains unsatisfied after payments are made, the Contractor shall refund to the Owner all money that the Owner may be compelled to pay in discharging such lien, claim, security interest, or encumbrance, including all costs and reasonable attorneys' fees, unless Contractor is proceeding diligently and in good faith to contest such lien, claim, security interest, or encumbrance and has furnished to Owner reasonable security, such as a bond, title insurance, or letter of credit to reasonably protect Owner against such lien, claim, security interest, or encumbrance.

§ 9.10.3 If, after Substantial Completion of the Work, final completion thereof is materially delayed through no fault of the Contractor or by issuance of Change Orders affecting final completion, and the Architect so confirms, the Owner shall, upon application by the Contractor and certification by the Architect, and without terminating the Contract, make payment of the balance due for that portion of the Work fully completed and accepted. If the remaining balance for Work not fully completed or corrected is less than retainage stipulated in the Contract Documents, and if bonds have been furnished, the written consent of surety to payment of the balance due for that portion of the Work fully completed and accepted shall be submitted by the Contractor to the Architect prior to certification of such payment. Such payment shall be made under terms and conditions governing final payment, except that it shall not constitute a waiver of Claims.

- § 9.10.4 The making of final payment shall constitute a waiver of Claims by the Owner except those arising from
 - .1 liens, Claims, security interests, or encumbrances arising out of the Contract and unsettled;
 - .2 failure of the Work to comply with the requirements of the Contract Documents; or

- .3 terms of special warranties required by the Contract Documents.
- § 9.10.5 Acceptance of final payment by the Contractor, a Subcontractor, or a material or equipment supplier, shall constitute a waiver of claims by that payee except those previously made in writing and identified by that payee as unsettled at the time of final Application for Payment.

ARTICLE 10 PROTECTION OF PERSONS AND PROPERTY

§ 10.1 Safety Precautions and Programs

The Contractor shall be responsible for initiating, maintaining, and supervising all safety precautions and programs in connection with the performance of the Contract.

§ 10.2 Safety of Persons and Property

- § 10.2.1 The Contractor shall take reasonable precautions for safety of, and shall provide reasonable protection to prevent damage, injury, or loss to
 - .1 employees on the Work and other persons who may be affected thereby;
 - .2 the Work and materials and equipment to be incorporated therein, whether in storage on or off the site, under care, custody, or control of the Contractor, a Subcontractor, or a Sub-subcontractor; and
 - .3 other property at the site or adjacent thereto, such as trees, shrubs, lawns, walks, pavements, roadways, structures, and utilities not designated for removal, relocation, or replacement in the course of construction.
- § 10.2.2 The Contractor shall comply with, and give notices required by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities, bearing on safety of persons or property or their protection from damage, injury, or loss.
- § 10.2.3 The Contractor shall implement, erect, and maintain, as required by existing conditions and performance of the Contract, reasonable safeguards for safety and protection, including posting danger signs and other warnings against hazards; promulgating safety regulations; and notifying the owners and users of adjacent sites and utilities of the safeguards.
- § 10.2.4 When use or storage of explosives or other hazardous materials or equipment, or unusual methods are necessary for execution of the Work, the Contractor shall exercise utmost care and carry on such activities under supervision of properly qualified personnel.
- § 10.2.5 The Contractor shall promptly remedy damage and loss (other than damage or loss insured under property insurance required by the Contract Documents) to property referred to in Sections 10.2.1.2 and 10.2.1.3 to the extent caused by the Contractor, a Sub-subcontractor, or anyone directly or indirectly employed by any of them, or by anyone for whose acts they may be liable and for which the Contractor is responsible under Sections 10.2.1.2 and 10.2.1.3, except damage or loss attributable to acts or omissions of the Owner or Architect or anyone directly or indirectly employed by either of them, or by anyone for whose acts either of them may be liable, and not attributable to the fault or negligence of the Contractor. The foregoing obligations of the Contractor are in addition to the Contractor's obligations under Section 3.18.
- § 10.2.6 The Contractor shall designate a responsible member of the Contractor's organization at the site whose duty shall be the prevention of accidents. This person shall be the Contractor's superintendent unless otherwise designated by the Contractor in writing to the Owner and Architect.
- § 10.2.7 The Contractor shall not permit any part of the construction or site to be loaded so as to cause damage or create an unsafe condition.

§ 10.2.8 Injury or Damage to Person or Property

Contractor shall promptly report in writing to the Architect and Owner all accidents whatsoever arising out of, or in connection with, the performance of the Work, whether on or off the Site, which caused death, personal injury or property damage, giving full details and statements of witnesses, if Contractor reasonably believes there may be a claim made against Owner. In addition, if death or serious injury or damages are caused, the accident shall be reported immediately to Owner by telephone or messenger. If either party suffers injury or damage to person or property because of an act or omission of the other party, or of others for whose acts such party is legally responsible, written

notice of such injury or damage, whether or not insured, shall be given to the other party within a reasonable time not exceeding 21 days after discovery. The notice shall provide sufficient detail to enable the other party to investigate the matter.

§ 10.3 Hazardous Materials and Substances

- § 10.3.1 The Contractor is responsible for compliance with any requirements included in the Contract Documents regarding hazardous materials or substances. If the Contractor encounters a hazardous material or substance not addressed in the Contract Documents and if reasonable precautions will be inadequate to prevent foreseeable bodily injury or death to persons resulting from a material or substance, including but not limited to asbestos or polychlorinated biphenyl (PCB), encountered on the site by the Contractor, the Contractor shall, upon recognizing the condition, immediately stop Work in the affected area and report the condition to the Owner and Architect in writing.
- § 10.3.2 Upon receipt of the Contractor's written notice, the Owner shall obtain the services of a licensed laboratory to verify the presence or absence of the material or substance reported by the Contractor and, in the event such material or substance is found to be present, to cause it to be rendered harmless. Unless otherwise required by the Contract Documents, the Owner shall furnish in writing to the Contractor and Architect the names and qualifications of persons or entities who are to perform tests verifying the presence or absence of such material or substance or who are to perform the task of removal or safe containment of such material or substance. The Contractor and the Architect will promptly reply to the Owner in writing stating whether or not either has reasonable objection to the persons or entities proposed by the Owner. If either the Contractor or Architect has an objection to a person or entity proposed by the Owner, the Owner shall propose another to whom the Contractor and the Architect have no reasonable objection. When the material or substance has been rendered harmless, Work in the affected area shall resume upon written agreement of the Owner and Contractor. A Change Order shall be issued to the Contractor as provided in Article 7, providing for an equitable adjustment in the Contract Sum or Contract Time for an increase or delay in the Work or the cost thereof in the event the Contractor encounters hazardous substances.
- § 10.3.3 The Owner shall indemnify and hold harmless the Contractor, Subcontractors, Architect, Architect's consultants, and agents and employees of any of them from and against claims, damages, losses, and expenses, including but not limited to attorneys' fees, arising out of or resulting from performance of the Work in the affected area if in fact the material or substance presents the risk of bodily injury or death as described in Section 10.3.1 and has not been rendered harmless, despite Owner's representations to the contrary, provided that such claim, damage, loss, or expense is attributable to bodily injury, sickness, disease or death, or to injury to or destruction of tangible property (other than the Work itself), except to the extent that such damage, loss, or expense is due to the fault or negligence of the party seeking indemnity.
- § 10.3.4 The Owner shall not be responsible under this Section 10.3 for materials or substances the Contractor brings to the site unless such materials or substances are required by the Contract Documents and are used, stored, transported, handled and installed properly by Contractor. The Owner shall be responsible for hazardous materials or substances required by the Contract Documents, except to the extent of the Contractor's fault or negligence in the use, handling, storage, transportation or installation of such materials or substances.
- § 10.3.5 The Contractor shall indemnify, defend and hold the Owner harmless against any and all liabilities, damages, losses, costs, penalties, expenses or responsibilities (1) for remediation of materials or substances the Contractor brings to the site and negligently handles, (2) where the Contractor fails to perform its obligations under Section 10.3.1, except to the extent that the cost and expense are due to the Owner's fault or negligence, and (3) materials and substances which are Contractor's responsibility under Section 10.3.4.
- § 10.3.6 Owner assumes the risk of, and shall indemnify, defend and hold Contractor harmless against any and all liabilities, damages, losses, costs, penalties, expenses or responsibilities, including reasonable attorneys' fees and disbursements, sustained in connection with any request or order arising directly or indirectly out of the release, threatened release, or handling of hazardous substances or materials at the Project site or at any site to which a hazardous substance is moved or alleged to have been moved provided that Contractor provides written notice to Owner of any such claim, demand, cause of action, suit, judgment, subpoena or administrative proceeding within 5 days of receipt of same. The duties and responsibilities of the Owner as set forth in this section shall not apply to the extent that the liability, damage, loss, economic loss, cost, penalty, expense or responsibility results from hazardous substances or material brought onto the Project site by Contractor or arises out of negligence of Contractor after the

hazardous substance or materials is discovered on the Project site and, if applicable, written notice of the discovery along with documents identifying the materials provided to the Contractor.

§ 10.4 Emergencies

In an emergency affecting safety of persons or property, the Contractor shall act, at the Contractor's discretion, to prevent threatened damage, injury, or loss. Additional compensation or extension of time claimed by the Contractor on account of an emergency shall be determined as provided in Article 15 and Article 7.

ARTICLE 11 INSURANCE AND BONDS

§ 11.1 Contractor's Insurance and Bonds

§ 11.1.1 The Contractor shall purchase and maintain such insurance as will protect the Contractor from claims set forth below which may arise out of or result from the Contractor's operations and completed operations under the Contract and for which the Contractor may be legally liable, whether such operations be by the Contractor or by a Subcontractor or by anyone directly or indirectly employed by any of them, or by anyone for whose acts any of them may be liable:

- .1 Claims under workers' compensation, disability benefit and other similar employee benefit acts that are applicable to the Work to be performed;
- .2 Claims for damages because of bodily injury, occupational sickness or disease, or death of the Contractor's employees;
- .3 Claims for damages because of bodily injury, sickness or disease, or death of any person other than the Contractor's employees;
- .4 Claims for damages insured by usual personal injury liability coverage;
- .5 Claims for damages, other than to the Work itself, because of injury to or destruction of tangible property, including loss of use resulting therefrom;
- .6 Claims for damages because of bodily injury, death of a person or property damage arising out of ownership, maintenance or use of a motor vehicle;
- .7 Claims for bodily injury or property damage arising out of completed operations; and
- .8 Claims involving contractual liability insurance applicable to the Contractor's obligations under Section 3.18;
- .9 claims arising out of or relating to the use of Contractor's autos, trucks and other vehicles in connection with the Project; and
- .10 claims arising from Contractor's professional services, if any.

The Contractor shall purchase and maintain the required insurance from an insurance company or companies rated A or better by A.M. Best and lawfully authorized to do business in the jurisdiction in which the Project is located.

§ 11.1.2 The Contractor shall to furnish bonds covering faithful performance of the Contract and payment of obligations arising thereunder (including the Guaranteed Maximum Price, as may be amended by Change Order(s)). The Contractor shall purchase and maintain the required bonds from a company or companies lawfully authorized to issue surety bonds in the jurisdiction where the Project is located. Upon the request of any person or entity appearing to be a potential beneficiary of bonds covering payment of obligations arising under the Contract, the Contractor shall promptly furnish a copy of the bonds or shall authorize a copy to be furnished. The Contractor may require its Subcontractors to furnish bonds in its discretion.

§ 11.1.3 The Owner, its board members, directors, officers, agents (including the Architect) and employees, and PNC Bank National Association, its successors and/or assigns, and Columbia Housing SLP Corporation, its successors and/or assigns, shall be named as additional insureds, on a primary and non-contributory basis, on all insurance policies providing coverage for the claims identified in Sections 11.1.1.3 through 11.1.1.9 above, including the completed operations coverage therein and the endorsements thereto. (See Exhibit to Agreement for the Certificate of Insurance). Any insurance maintained by Owner will be in excess of and not contribute with additional insured coverage provided by Contractor regardless of any "other insurance" clause. "Vertical exhaustion" shall apply when determining responsibility for coverage.

§ 11.1.4 The insurance required by Section 11.1.1 shall be written for not less than limits of liability specified in the Contract Documents or required by law, whichever coverage is greater. Coverages, whether written on an occurrence or claims-made basis, shall be maintained without interruption from the date of commencement of the Work until the date of final payment and termination of any coverage required to be maintained after final payment and, with respect

to the Contractor's completed operations coverage, for the period identified in 11.1.8.1 below. If any of the insurance required by Section 11.1.1 is maintained on a claims-made basis, then prior to final payment, Contractor shall purchase tail insurance affording the same coverage and having the same limits, for a period of not less than three years from the date of Final Payment.

§ 11.1.5 Certificates of insurance acceptable to the Owner shall be filed with the Owner prior to commencement of the Work and thereafter upon renewal or replacement of each required policy of insurance.

§ 11.1.6 Notice of Cancellation or Expiration of Contractor's Required Insurance. Within three (3) business days of the date the Contractor becomes aware of an impending or actual cancellation or expiration of any insurance required by the Contract Documents, the Contractor shall provide notice to the Owner of such impending or actual cancellation or expiration. Upon receipt of notice from the Contractor, the Owner shall, unless the lapse in coverage arises from an act or omission of the Owner, have the right to stop the Work until the lapse in coverage has been cured by the procurement of replacement coverage by the Contractor. The furnishing of notice by the Contractor shall not relieve the Contractor of any contractual obligation to provide any required coverage.

§ 11.1.7 Workers' Compensation and Employers' Liability: Contractor agrees that it and all its Subcontractors will comply with all statutes and law with regard to Workers' Compensation/Occupational Disease applying to employees or their beneficiaries. Minimum limits shall be:

.1 Workers' Compensation-Statutory Limits

.2 Employers' Liability Not Less Than:

Bodily Injury by Accident \$100,000 Each Accident Bodily Injury by Disease \$100,000 Each Employee \$500,000 Policy Limit

This insurance shall contain provisions waiving each underwriter's rights of subrogation against Owner.

§ 11.1.8 Other insurance required is as follows:

.1 Commercial General Liability

Limits not less than:

Bodily Injury and Property \$1,000,000 Each Occurrence

Damage Combined Single Limit: \$2,000,000 Annual General Aggregate – Per Policy

\$2,000,000 Products Liability/Completed Operations

Fire Damage \$50,000

Medical Payment \$5,000

Personal Injury and Advertising Injury \$1,000,000

Commercial General Liability coverage must provide:

General Aggregate Limit must apply per project

Premises and Operations

Products Liability/Completed Operations Liability

Contractual Liability

XCU Coverage (explosion, collapse and underground property damage)

Note: Products Liability/Completed Operations Coverage shall continue in force for two (2) years from the earlier of 90 days following Substantial Completion or final payment by the Owner.

.2 Auto

Limits not less than:

Bodily injury and Property Damage

Combined Single Limit Policy must be intended to cover all owned, Non-owned and hired vehicles: \$1,000,000 Each Accident

.3 Umbrella Liability

Not less than:

Bodily Injury and Property Damage \$10,000,000 Each Occurrence

Combined Single Limit \$10,000,000 Annual

§ 11.1.9 Insurance required of Subcontractors:

- All MEP Subcontractors and any other Subcontractor who will perform Work on the Project Site where the aggregate amount to be paid to the Subcontractor totals \$1,000,000.00 or more of the construction budget shall:
 - (a) Maintain the same limits as the Contractor for insurance required under Sections 11.1.7, 11.1.8.1 and 11.1.8.2 herein; and
 - (b) Umbrella coverage of \$5,000,000.
- Any other Subcontractor who will perform Work on Site shall maintain the same insurance as the Contractor for insurance required under Sections 11.1.7, and 11.1.8, excluding pollution liability, but for per occurrence amounts of not less than \$250,000.00 up to the amount of Subcontractor's contract, whichever is more; and maintain umbrella coverage of not less than \$1,000,000.
- .3 All other Subcontractors (not performing Work on site) shall have adequate insurance coverage up to the minimum amount required by law. If a Subcontractor does not have the insurance required above, then Contractor shall advise Owner and Owner may, in its sole discretion, waive and/or modify the above insurance requirements for Subcontractors.
- § 11.1.10 The Owner and Architect assume no responsibility for Contractor's or Subcontractors' liabilities in the event that the limits set above are not adequate or coverage is not properly obtained by either of them.
- § 11.1.11 Contractor's insurance policies, terms and conditions shall govern unless otherwise modified, in writing, by the parties.
- § 11.1.12 Unless otherwise agreed by Owner, Contractor shall ensure that the insurance required of Subcontractors in Section 11.1.9 be maintained in an insurance company or companies rated A or better by A.M. Best and lawfully authorized to do business in the jurisdiction in which the Project is located.

§ 11.2 Owner's Insurance

§ 11.2.1 Owner's Liability Insurance. The Owner shall purchase and maintain insurance of the types and limits of liability, containing the endorsements, and subject to the terms and conditions, as described in the Agreement or elsewhere in the Contract Documents. The Owner shall purchase and maintain the required insurance from an insurance company or companies rated A or better by A.M. Best and lawfully authorized to do business in the jurisdiction in which the Project is located.

(Paragraph deleted)

User Notes:

§ 11.2.2 Owner's Property Insurance. The Owner shall purchase and maintain property insurance written on a builder's risk "all-risk" or equivalent policy form in the amount of the initial Contract Sum, plus value of subsequent Contract

Init.

Modifications and cost of materials supplied or installed by others, comprising total value for the entire Project at the site on a replacement cost basis without optional deductibles. Such property insurance shall be maintained, unless otherwise provided in the Contract Documents or otherwise agreed in writing by all persons and entities who are beneficiaries of such insurance, until final payment has been made as provided in Section 9.10 or until no person or entity other than the Owner has an insurable interest in the property required by this Section 11.2 to be covered, whichever is later. This insurance shall include interests as named insureds, but not as "first-named insureds" of the Owner, the Contractor, Subcontractors and Sub-subcontractors in the Project. Such insurance shall be purchased and maintained from a company or companies lawfully authorized to do business in the jurisdiction in which the Project is located.

- § 11.2.3 Property insurance shall be on an "all-risk" or equivalent policy form and shall include, without limitation, insurance against the perils of fire (with extended coverage) and physical loss or damage including, without duplication of coverage, theft, vandalism, malicious mischief, collapse, earthquake, flood, windstorm, falsework, testing and startup, temporary buildings and debris removal including demolition occasioned by enforcement of any applicable legal requirements, and shall cover reasonable compensation for Architect's and Contractor's services and expenses required as a result of such insured loss.
- § 11.2.4 Failure to Purchase Required Property Insurance. If the Owner fails to purchase and maintain the required property insurance, with all of the coverages and in the amounts described in the Agreement or elsewhere in the Contract Documents, the Owner shall inform the Contractor in writing prior to commencement of the Work. Upon receipt of notice from the Owner, the Contractor may delay commencement of the Work and may obtain insurance that will protect the interests of the Contractor, Subcontractors, and Sub-Subcontractors in the Work. The cost of the insurance shall be charged to the Owner by a Change Order. If the Owner does not provide written notice, and the Contractor is damaged by the failure or neglect of the Owner to purchase or maintain the required insurance, the Owner shall reimburse the Contractor for all reasonable costs and damages attributable thereto.
- § 11.2.5 Notice of Cancellation or Expiration of Owner's Required Property Insurance. Within three (3) business days of the date the Owner becomes aware of an impending or actual cancellation or expiration of any property insurance required by the Contract Documents, the Owner shall provide notice to the Contractor of such impending or actual cancellation or expiration. Unless the lapse in coverage arises from an act or omission of the Contractor: (1) the Contractor, upon receipt of notice from the Owner, shall have the right to stop the Work until the lapse in coverage has been cured by the procurement of replacement coverage by either the Owner or the Contractor; and (2) the Contract Time and Contract Sum shall be equitably adjusted. If the Contractor purchases replacement coverage, the cost of the insurance shall be charged to the Owner by an appropriate Change Order. The furnishing of notice by the Owner shall not relieve the Owner of any contractual obligation to provide required insurance.
- § 11.2.6 This property insurance shall cover portions of the Work stored off the site, and also portions of the Work in transit.
- § 11.2.7 Except to the extent covered by insurance, the Owner shall not be liable for loss or damage to any tools owned by mechanics, any other tools, equipment, staging, towers, and forms owned or rented by the Contractor, the capital value of which is not included in the Cost of the Work, or any shanties, houses or other temporary structures.
- § 11.2.8 The Owner shall provide the Contractor with a certificate of insurance for each policy that includes insurance coverages required by this Section 11.2.
- § 11.2.9 Partial occupancy or use in accordance with Section 9.9 shall not commence until the insurance company or companies providing property insurance have consented to such partial occupancy or use by endorsement or otherwise. The Owner and the Contractor shall take reasonable steps to obtain consent of the insurance company or companies and shall, without mutual written consent, take no action with respect to partial occupancy or use that would cause cancellation, lapse, or reduction of insurance.

§ 11.3 Waivers of Subrogation

§ 11.3.1 The Owner and Contractor waive all rights against (1) each other and any of their subcontractors, sub-subcontractors, agents, and employees, each of the other; (2) the Architect and Architect's consultants; and (3) Separate Contractors, if any, and any of their subcontractors, sub-subcontractors, agents, and employees, for damages caused by fire, or other causes of loss, to the extent those losses are covered by property insurance required by the

Agreement or other property insurance applicable to the Project, except such rights as they have to proceeds of such insurance. The Owner or Contractor, as appropriate, shall require similar written waivers in favor of the individuals and entities identified above from the Architect, Architect's consultants, Separate Contractors, subcontractors, and sub-subcontractors. The policies of insurance purchased and maintained by each person or entity agreeing to waive claims pursuant to this section 11.3.1 shall not prohibit this waiver of subrogation. This waiver of subrogation shall be effective as to a person or entity (1) even though that person or entity would otherwise have a duty of indemnification, contractual or otherwise, (2) even though that person or entity did not pay the insurance premium directly or indirectly, or (3) whether or not the person or entity had an insurable interest in the damaged property.

§ 11.3.2 If during the Project construction period the Owner insures properties, real or personal or both, at or adjacent to the site by property insurance under policies separate from those insuring the Project, or if after final payment property insurance is to be provided on the completed Project through a policy or policies other than those insuring the Project during the construction period, to the extent permissible by such policies, the Owner waives all rights in accordance with the terms of Section 11.3.1 for damages caused by fire or other causes of loss covered by this separate property insurance.

§ 11.4 Loss of Use, Business Interruption, and Delay in Completion Insurance

The Owner, at the Owner's option, may purchase and maintain insurance that will protect the Owner against loss of use of the Owner's property, or the inability to conduct normal operations, due to fire or other causes of loss.

§11.5 Adjustment and Settlement of Insured Loss

§ 11.5.1 A loss insured under the Owner's property insurance shall be adjusted by the Owner and made payable to the Owner for the insureds, as their interests may appear, subject to requirements of any applicable mortgagee clause and of Section 11.5.2. The Owner shall pay the Architect and Contractor their just shares of insurance proceeds received by the Owner, and by appropriate agreements the Architect and Contractor shall make payments to their consultants and Subcontractors in similar manner.

§ 11.5.2 Prior to settlement of an insured loss, the Owner shall notify the Contractor of the terms of the proposed settlement as well as the proposed allocation of the insurance proceeds. The Contractor shall have 14 days from receipt of notice to object to the proposed settlement or allocation of the proceeds. If the Contractor does not object, the Owner shall settle the loss and the Contractor shall be bound by the settlement and allocation. Upon receipt, the Owner shall deposit the insurance proceeds in a separate account and make the appropriate distributions. Thereafter, if no other agreement is made or the Owner does not terminate the Contract for convenience, the Owner and Contractor shall execute a Change Order for reconstruction of the damaged or destroyed Work in the amount allocated for that purpose. If the Contractor timely objects to either the terms of the proposed settlement or the allocation of the proceeds, the Owner may proceed to settle the insured loss, and any dispute between the Owner and Contractor arising out of the settlement or allocation of the proceeds shall be resolved pursuant to Article 15. Pending resolution of any dispute, the Owner may issue a Construction Change Directive for the reconstruction of the damaged or destroyed Work.

ARTICLE 12 UNCOVERING AND CORRECTION OF WORK

§ 12.1 Uncovering of Work

§ 12.1.1 If a portion of the Work is covered contrary to the Architect's request or to requirements specifically expressed in the Contract Documents, it must, if requested in writing by the Architect, be uncovered for the Architect's examination and be replaced at the Contractor's expense without change in the Contract Time.

§ 12.1.2 If a portion of the Work has been covered that the Architect has not specifically requested to examine prior to its being covered, the Architect may request to see such Work and it shall be uncovered by the Contractor. If such Work is in accordance with the Contract Documents, costs of uncovering and replacement shall, by appropriate Change Order, be at the Owner's expense. If such Work is not in accordance with the Contract Documents, such costs and the cost of correction, shall be at the Contractor's expense, unless the condition was caused by the Owner or a separate contractor in which event the Owner shall be responsible for payment of such costs.

§ 12.2 Correction of Work

§ 12.2.1 Before Substantial Completion

The Contractor shall promptly correct Work rejected by the Architect or failing to conform to the requirements of the Contract Documents, whether discovered before or after Substantial Completion and whether or not fabricated,

Init.

AlA Document A201 – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017. All rights reserved. "The American Institute of Architects," "American Institute of Architects," "AlA," the AIA Logo, and "AIA Contract Documents" are trademarks of The American Institute of Architects. This document was produced at 15:38:55 ET on 10/30/2023 under Order No.2114399858 which expires on 02/01/2024, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail docinfo@aiacontracts.com.

User Notes: (1162898282)

installed or completed. Costs of correcting such rejected Work, including additional testing and inspections, the cost of uncovering and replacement, and compensation for the Architect's services and expenses made necessary thereby, shall be at the Contractor's expense.

§ 12.2.2 After Substantial Completion

- § 12.2.2.1 In addition to the Contractor's obligations under Section 3.5 and the requirements under Clause 23 of the HUD 5370 General Conditions of the Contract for Construction, if, within one year after the date of Substantial Completion of the Work or designated portion thereof or after the date for commencement of warranties established under Section 9.9.1, or by terms of an applicable special warranty required by the Contract Documents, any of the Work is found to be not in accordance with the requirements of the Contract Documents, the Contractor shall correct it promptly after receipt of written notice from the Owner to do so, unless the Owner has previously given the Contractor a written acceptance of such condition. The Owner shall give such notice promptly after discovery of the condition. If the Contractor fails to correct nonconforming Work within a reasonable time during that period after receipt of notice from the Owner or Architect, the Owner may correct it in accordance with Section 2.5.
- § 12.2.2.2 The one-year period for correction of Work shall be extended with respect to portions of Work first performed after Substantial Completion by the period of time between Substantial Completion and the actual completion of that portion of the Work.
- § 12.2.2.3 The one-year period for correction of Work shall not be extended by corrective Work performed by the Contractor pursuant to this Section 12.2, except for Work not properly completed in the first instance.
- § 12.2.3 The Contractor shall remove from the site portions of the Work that are not in accordance with the requirements of the Contract Documents and are neither corrected by the Contractor nor accepted by the Owner.
- § 12.2.4 The Contractor shall bear the cost of correcting destroyed or damaged construction of the Owner or Separate Contractors, whether completed or partially completed, caused by the Contractor's correction or removal of Work that is not in accordance with the requirements of the Contract Documents.
- § 12.2.5 Nothing contained in this Section 12.2 shall be construed to establish a period of limitation with respect to other obligations the Contractor has under the Contract Documents. Establishment of the one-year period for correction of Work as described in Section 12.2.2 relates only to the specific obligation of the Contractor to correct the Work, and has no relationship to the time within which the obligation to comply with the Contract Documents may be sought to be enforced, nor to the time within which proceedings may be commenced to establish the Contractor's liability with respect to the Contractor's obligations other than specifically to correct the Work.

§ 12.3 Acceptance of Nonconforming Work

If the Owner prefers to accept Work that is not in accordance with the requirements of the Contract Documents, the Owner may do so instead of requiring its removal and correction, in which case the Contract Sum will be reduced as appropriate and equitable. Such adjustment shall be effected whether or not final payment has been made.

ARTICLE 13 MISCELLANEOUS PROVISIONS

§ 13.1 Governing Law

The Contract shall be governed by the law of the State of Wisconsin.

§ 13.2 Successors and Assigns

- § 13.2.1 The Owner and Contractor respectively bind themselves, their partners, successors, assigns, and legal representatives to covenants, agreements, and obligations contained in the Contract Documents. Except as provided in Section 13.2.2, neither party to the Contract shall assign the Contract as a whole without written consent of the other. If either party attempts to make such an assignment without such consent, that party shall nevertheless remain legally responsible for all obligations under the Contract.
- § 13.2.2 The Owner may, without consent of the Contractor, assign the Contract to a lender providing construction financing for the Project, if the lender assumes the Owner's rights and obligations under the Contract Documents. The Contractor shall execute all consents reasonably required to facilitate such assignment.

§ 13.3 Rights and Remedies

- § 13.3.1 Duties and obligations imposed by the Contract Documents and rights and remedies available thereunder shall be in addition to and not a limitation of duties, obligations, rights, and remedies otherwise imposed or available by law.
- § 13.3.2 No action or failure to act by the Owner, Architect, or Contractor shall constitute a waiver of a right or duty afforded them under the Contract, nor shall such action or failure to act constitute approval of or acquiescence in a breach thereunder, except as may be specifically agreed in writing or as set forth herein.

§ 13.4 Tests and Inspections

- § 13.4.1 Tests, inspections, and approvals of portions of the Work shall be made as required by the Contract Documents and by applicable laws, statutes, ordinances, codes, rules, and regulations or lawful orders of public authorities. Unless otherwise provided, the Contractor shall make arrangements for such tests, inspections, and approvals with an independent testing laboratory or entity acceptable to the Owner, or with the appropriate public authority, and shall bear all related costs of tests, inspections, and approvals. The Contractor shall give the Architect timely notice of when and where tests and inspections are to be made so that the Architect may be present for such procedures. The Owner shall bear costs of tests, inspections, or approvals that do not become requirements until after bids are received or negotiations concluded.
- § 13.4.2 If the Architect, Owner, or public authorities having jurisdiction determine that portions of the Work require additional testing, inspection, or approval not included under Section 13.4.1, the Architect will, upon written authorization from the Owner, instruct the Contractor to make arrangements for such additional testing, inspection, or approval, by an entity acceptable to the Owner, and the Contractor shall give timely notice to the Architect of when and where tests and inspections are to be made so that the Architect may be present for such procedures. Such costs, except as provided in Section 13.4.3, shall be at the Owner's expense.
- § 13.4.3 If procedures for testing, inspection, or approval under Sections 13.4.1 and 13.4.2 reveal failure of the portions of the Work to comply with requirements established by the Contract Documents, all costs made necessary by such failure, including those of repeated testing, inspection, or approval procedures and compensation for the Architect's services and expenses, shall be at the Contractor's expense.
- § 13.4.4 Required certificates of testing, inspection, or approval shall, unless otherwise required by the Contract Documents, be secured by the Contractor and promptly delivered to the Architect.
- § 13.4.5 If the Architect is to observe tests, inspections, or approvals required by the Contract Documents, the Architect will do so promptly and, where practicable, at the normal place of testing.
- § 13.4.6 Tests or inspections conducted pursuant to the Contract Documents shall be made promptly to avoid unreasonable delay in the Work.

§ 13.5 Interest

Payments due and unpaid under the Contract Documents shall bear interest from the date payment is due at such rate as the parties may agree upon in writing or, in the absence thereof, at the legal rate prevailing from time to time at the place where the Project is located.

ARTICLE 14 TERMINATION OR SUSPENSION OF THE CONTRACT

§ 14.1 Termination by the Contractor

- § 14.1.1 The Contractor may terminate the Contract if the Work is stopped for a period of 60 consecutive days through no act or fault of the Contractor, a Sub-subcontractor, their agents or employees, or any other persons or entities performing portions of the Work under direct or indirect contract with the Contractor, for any of the following reasons:
 - .1 Issuance of an order of a court or other public authority having jurisdiction that requires all Work to be stopped:
 - .2 An act of government, such as a declaration of national emergency, that requires all Work to be stopped;
 - **.3** Because the Architect has not issued a Certificate for Payment and has not notified the Contractor of the reason for withholding certification as provided in Section 9.4.1, or because the Owner has not made

- payment on a Certificate for Payment within the time stated in the Contract Documents and none of the circumstances prescribed in Sections 9.5.1.1 through 9.5.1.9 exist; or
- .4 The Owner has failed to furnish to the Contractor reasonable evidence as required by Section 2.2.
- § 14.1.2 The Contractor may terminate the Contract if, through no act or fault of the Contractor, a Subcontractor, a Sub-subcontractor, their agents or employees, or any other persons or entities performing portions of the Work under direct or indirect contract with the Contractor, repeated suspensions, delays, or interruptions of the entire Work by the Owner as described in Section 14.3, constitute in the aggregate more than 100 percent of the total number of days scheduled for completion, or 120 days in any 365-day period, whichever is less.
- § 14.1.3 If one of the reasons described in Section 14.1.1 or 14.1.2 exists, the Contractor may, upon seven days' written notice to the Owner and Architect, terminate the Contract and recover from the Owner payment for Work executed, including reasonable overhead and profit, actual and direct costs incurred by reason of such termination, and damages.
- § 14.1.4 If the Work is stopped for a period of 60 consecutive days through no act or fault of the Contractor, a Sub-subcontractor, or their agents or employees or any other persons or entities performing portions of the Work under contract with the Contractor because the Owner has repeatedly failed to fulfill the Owner's obligations under the Contract Documents with respect to matters important to the progress of the Work, the Contractor may, upon seven additional days' written notice to the Owner and the Architect, terminate the Contract and recover from the Owner as provided in Section 14.1.3.

§ 14.2 Termination by the Owner for Cause

- § 14.2.1 The Owner may terminate the Contract if the Contractor
 - refuses or fails to supply enough properly skilled workers or proper materials in accordance with Contractor's schedule for attaining Substantial Completion;
 - fails to make payment to Subcontractors for materials or labor in accordance with the respective agreements between the Contractor and the Subcontractors if and only if Contractor fails to provide Owner with reasonable evidence that funds have been placed by Contractor in a commercial escrow account sufficient to pay Subcontractor for the disputed items, fails to post a bond for the amount due Subcontractors, fails to obtain an endorsement to a title policy with respect to liens of Subcontractor, or if the portion of the Contract Sum owing to the Contractor held in retainage by Owner pursuant to Article 9 is less than the amount allegedly owed by Contractor to Subcontractor;
 - .3 disregards applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of a public authority;
 - .4 otherwise is guilty of substantial breach of a provision of the Contract Documents; or
 - .5 becomes insolvent, is in bankruptcy, or is in receivership.
- § 14.2.2 When any of the above reasons exist, the Owner, upon certification by the Architect that sufficient cause exists to justify such action, may, without prejudice to any other rights or remedies of the Owner and after giving the Contractor and the Contractor's surety, if any, seven days' written notice and provided the Contractor, within such seven (7) day period, has not cured such cause or breach, terminate employment of the Contractor and may, subject to any prior rights of the surety:
 - .1 Exclude the Contractor from the site and take possession of all materials, equipment, tools, and construction equipment and machinery thereon owned by the Contractor;
 - .2 Accept assignment of subcontracts pursuant to Section 5.4; and
 - .3 Finish the Work by whatever reasonable method the Owner may deem expedient. Upon written request of the Contractor, the Owner shall furnish to the Contractor a detailed accounting of the costs incurred by the Owner in finishing the Work.
- § 14.2.3 When the Owner terminates the Contract for one of the reasons stated in Section 14.2.1, the Contractor shall not be entitled to receive further payment until the Work is finished.
- § 14.2.4 If the unpaid balance of the Contract Sum exceeds costs of finishing the Work, including compensation for the Architect's services and expenses made necessary thereby, and other damages incurred by the Owner and not expressly waived, such excess shall be paid to the Contractor. If such costs and damages exceed the unpaid balance, the Contractor shall pay the difference to the Owner. The amount to be paid to the Contractor or Owner, as the case

may be, shall be certified by the Architect, upon application, and this obligation for payment shall survive termination of the Contract.

§ 14.2.5 Upon termination of the Contract, Contractor shall immediately stop the Work hereunder except such as shall be essential to preserve and safeguard existing Work and immediately advise Owner of the status of all outstanding subcontracts and purchase orders. Such outstanding purchase orders and subcontracts shall be canceled or assigned to Owner as Owner may direct. In the event of any such termination, Contractor shall deliver all of the Work completed and in process to the date thereof to Owner and shall deliver to Owner all documents of title, reports, estimates, schedules, and other documents and data as Owner shall require.

§ 14.3 Suspension by the Owner for Convenience

§ 14.3.1 The Owner may, without cause, order the Contractor in writing to suspend, delay or interrupt the Work, in whole or in part for such period of time as the Owner may determine.

§ 14.3.2 The Contract Sum and Contract Time shall be adjusted for increases in the cost and time caused by suspension, delay, or interruption as described in Section 14.3.1. Adjustment of the Contract Sum shall include profit. No adjustment shall be made to the extent

- .1 that performance is, was, or would have been, so suspended, delayed, or interrupted, by another cause for which the Contractor is responsible; or
- .2 that an equitable adjustment is made or denied under another provision of the Contract.

§ 14.4 Termination by the Owner for Convenience

§ 14.4.1 The Owner may, at any time, terminate the Contract for the Owner's convenience and without cause.

§ 14.4.2 Upon receipt of written notice from the Owner of such termination for the Owner's convenience, the Contractor shall

- .1 cease operations as directed by the Owner in the notice;
- .2 take actions necessary, or that the Owner may direct, for the protection and preservation of the Work; and
- .3 except for Work directed to be performed prior to the effective date of termination stated in the notice, terminate all existing subcontracts and purchase orders and enter into no further subcontracts and purchase orders.

§ 14.4.3 In case of such termination for the Owner's convenience, the Contractor shall be entitled to receive payment for Work properly executed (including Contractor's fee for the Work completed), and costs incurred by reason of such termination.

ARTICLE 15 CLAIMS AND DISPUTES

§ 15.1 Claims

§ 15.1.1 Definition

A Claim is a demand or assertion by one of the parties seeking, as a matter of right, payment of money, adjustment or interpretation of the terms of this Agreement, or other relief with respect to the terms of the Contract. The term "Claim" also includes other disputes and matters in question between the Owner and Contractor arising out of or relating to the Contract. The responsibility to substantiate Claims shall rest with the party making the Claim. This Section 15.1.1 does not require the Owner to file a Claim in order to impose liquidated damages in accordance with the Contract Documents.

§ 15.1.2 Time Limits on Claims INTENTIONALLY OMITTED

§ 15.1.3 Notice of Claims

§ 15.1.3.1 Claims by either the Owner or Contractor must be initiated by written notice to the other party and to the Initial Decision Maker with a copy sent to the Architect, if the Architect is not serving as the Initial Decision Maker. Claims by either party must be initiated within 21 days after occurrence of the event giving rise to such Claim or within 21 days after the claimant first recognizes the condition giving rise to the Claim, whichever is later.

(Paragraph deleted)

Init.

AIA Document A201 – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017. All rights reserved. "The American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are trademarks of The American Institute of Architects. This document was produced at 15:38:55 ET on 10/30/2023 under Order No.2114399858 which expires on 02/01/2024, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail docinfo@aiacontracts.com.

§ 15.1.4 Continuing Contract Performance

§ 15.1.4.1 Pending final resolution of a Claim, except as otherwise agreed in writing or as provided in Section 9.7 and Article 14, the Contractor shall proceed diligently with performance of the Contract and the Owner shall continue to make payments in accordance with the Contract Documents.

(Paragraph deleted)

§ 15.1.5 Claims for Additional Cost

Subject to all other applicable provisions of the Contract Documents, if the Contractor wishes to make a Claim for an increase in the Contract Sum, notice as provided herein shall be given before proceeding to execute the portion of the Work that is the subject of the Claim. Prior notice is not required for Claims relating to an emergency endangering life or property arising under Section 10.4.

§ 15.1.6 Claims for Additional Time

§ 15.1.6.1 If the Contractor wishes to make a Claim for an increase in the Contract Time, written notice as provided herein shall be given. The Contractor's Claim shall include an estimate of cost and of probable effect of delay on progress of the Work. In the case of a continuing delay, only one Claim is necessary.

§ 15.1.6.2 If adverse weather conditions are the basis for a Claim for additional time, such Claim shall be documented by data substantiating that weather conditions were abnormal for the period of time for the geographical region where the Project is located, could not have been reasonably anticipated, and had an adverse effect on the scheduled construction.

§ 15.1.7 Waiver of Claims for Consequential Damages

Except to the extent that it is a covered loss under insurance required for this Project, the Contractor and Owner waive Claims against each other for consequential damages arising out of or relating to this Contract. This mutual waiver includes

- .1 damages incurred by the Owner for rental expenses, for losses of use, income, profit, financing, business and reputation, and for loss of management or employee productivity or of the services of such persons; and
- .2 damages incurred by the Contractor for principal office expenses including the compensation of personnel stationed there, for losses of financing, business and reputation, and for loss of profit, except anticipated profit arising directly from the Work.

This mutual waiver is applicable, without limitation, to all consequential damages due to either party's termination in accordance with Article 14. Nothing contained in this Section 15.1.7 shall be deemed to preclude assessment of liquidated damages, when applicable, in accordance with the requirements of the Contract Documents.

§ 15.2 Resolution of Claims and Disputes

§ 15.2.1 The Owner and Contractor shall endeavor to resolve Claims through good faith direct discussions between the parties' representatives. If, at any time, the Owner's and Contractor's representatives are unable to resolve their differences, a senior executive from either party may refer such dispute to a senior executive (of the requesting party's choice) of the other party. Within 21 calendar days of such referral, the executives shall meet to resolve the dispute.

§ 15.2.2 In the event of a Claim against the Contractor, the Owner may, but is not obligated to, notify the surety, if any, of the nature and amount of the Claim. If the Claim relates to a possibility of a Contractor's default, the Owner may, but is not obligated to, notify the surety and request the surety's assistance in resolving the controversy.

§ 15.2.3 If a Claim relates to or is the subject of a mechanic's lien, the party asserting such Claim may proceed in accordance with applicable law to comply with the lien notice or filing deadlines.

(Paragraphs deleted)

§ 15.3 Mediation/Litigation

§ 15.3.1 Claims, disputes, or other matters in controversy unresolved per Section 15.2 and arising out of or related to the Contract, except those waived as provided for in Sections 9.10.4, 9.10.5, and 15.1.7, shall be subject to mediation as a condition precedent to binding dispute resolution.

§ 15.3.2 The parties shall endeavor to resolve their Claims unresolved per Section 15.2 by mediation which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Mediation Procedures in effect on the date of the Agreement. A request for mediation shall be made in writing, delivered to the other party to the Contract, and filed with the person or entity administering the mediation. The request may be made concurrently with the filing of binding dispute resolution proceedings but, in such event, mediation shall proceed in advance of binding dispute resolution proceedings, which shall be stayed pending mediation for a period of 60 days from the date of filing, unless stayed for a longer period by agreement of the parties or court order.

§ 15.3.3 INTENTIONALLY OMITTED

§ 15.3.4 The parties shall share the mediator's fee and any filing fees equally. The mediation shall be held in the place where the Project is located, unless another location is mutually agreed upon. Agreements reached in mediation shall be enforceable as settlement agreements in any court having jurisdiction thereof.

(Paragraphs deleted)

§ 15.3.5 If the parties do not resolve a dispute through mediation pursuant to this Section 15.3, the method of binding dispute resolution shall be the following litigation in a court of competent jurisdiction located in Milwaukee, Wisconsin. The parties hereby agree and consent to personal jurisdiction in state or federal court located in Milwaukee, Wisconsin.

§ 15.3.6 In the event of litigation arising out of this Agreement, the prevailing party's attorneys' fees and costs shall be paid by the non-prevailing party. The "prevailing party" means the party in whose favor a judgment, decree, or final order is rendered.

Additions and Deletions Report for

AIA® Document A201® - 2017

This Additions and Deletions Report, as defined on page 1 of the associated document, reproduces below all text the author has added to the standard form AIA document in order to complete it, as well as any text the author may have added to or deleted from the original AIA text. Added text is shown underlined. Deleted text is indicated with a horizontal line through the original AIA text.

Note: This Additions and Deletions Report is provided for information purposes only and is not incorporated into or constitute any part of the associated AIA document. This Additions and Deletions Report and its associated document were generated simultaneously by AIA software at 15:38:55 ET on 10/30/2023.

PAGE 1

CONTRACT 23-006

Highland Gardens

Located at 1818 West Juneau Avenue, Milwaukee, WI 53233 Rehabilitation of 114-unit apartment building and associated site improvements

(Name, legal status and address)

Highland Gardens LLC c/o Housing Authority of the City of Milwaukee 809 North Broadway, 3rd Floor Milwaukee, WI 53202

Galbraith Carnahan Architects LLC 6404 West North Avenue Wauwatosa, WI 53213

PAGE 3

3.2, 3.3.2, 3.12.8, 3.18, 4.2.3, 8.3.1, 9.5.1, 10.2.5, 10.2.8, 13.3.2, 14.1, 15.1.2, 1, 15.2

PAGE 10

The Contract Documents are enumerated in the Agreement between the Owner and Contractor (hereinafter the Agreement) and consist of the Agreement, Conditions of the Contract (General, Supplementary and other Conditions), Drawings, Specifications, Addenda issued prior to execution of the Contract, other documents listed in the Agreement, and Modifications issued after execution of the Contract. A Modification is (1) a written amendment to the Contract signed by both parties, (2) a Change Order, (3) a Construction Change Directive, or (4) a written order for a minor change in the Work issued by the Architect. Architect, or (5) Field Change (as defined herein). Unless specifically enumerated in the Agreement, the Contract Documents do not include the advertisement or invitation to bid, Instructions to Bidders, sample forms, other information furnished by the Owner in anticipation of receiving bids or proposals, the Contractor's bid or proposal, or portions of Addenda relating to bidding or proposal requirements.

The Contract Documents form the Contract for Construction. The Contract represents the entire and integrated agreement between the parties hereto and supersedes prior negotiations, representations, or agreements, either written or oral. The Contract may be amended or modified only by a Modification. The Contract Documents shall not be construed to create a contractual relationship of any kind (1) between the Contractor and the Architect or the Architect's consultants, (2) between the Owner and a Subcontractor or a Sub-subcontractor, (3) between the Owner

Additions and Deletions Report for AIA Document A201 – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017. All rights reserved. "The American Institute of Architects," "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are trademarks of The American Institute of Architects. This document was produced at 15:38:55 ET on 10/30/2023 under Order No.2114399858 which expires on 02/01/2024, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail docinfo@aiacontracts.com. (1162898282)

and the Architect or the Architect's consultants, or (4) between any persons or entities other than the Owner and the Contractor. The Architect shall, however, be entitled to performance and enforcement of obligations under the Contract intended to facilitate performance of the Architect's duties and the Owner shall be entitled to performance and enforcement of obligations under the agreement between Contractor and Subcontractors intended to complete the Work.

...

The Drawings are the graphic and pictorial portions of the Contract Documents showing the design, location and dimensions of the Work, generally including plans, elevations, sections, details, schedules, and diagrams. diagrams, and any building information modeling (BIM).

• • •

The Specifications are that portion of the Contract Documents consisting of the written requirements for materials, equipment, systems, standards and workmanship for the Work, and performance of related services. The Specifications shall also include performance specifications necessary for Contractor to complete any design-assist portion of the Work.

..

Instruments of Service are representations, in any medium of expression now known or later developed, of the tangible and intangible creative work performed by the Architect and the Architect's consultants under their respective professional services agreements. agreements, in addition to any tangible and intangible creative work performed by Contractor related to any design-assist portion of the Work. Instruments of Service may include, without limitation, studies, surveys, models, sketches, drawings, specifications, and other similar materials.

§ 1.1.8 Initial Decision Maker

The Initial Decision Maker is the person identified in the Agreement to render initial decisions on Claims in accordance with Section 15.2. The Initial Decision Maker shall not show partiality to the Owner or Contractor and shall not be liable for results of interpretations or decisions rendered in good faith.

§ 1.1.8 I INTENTIONALLY OMITTED

§ 1.2.1 The intent of the Contract Documents is to Contract Documents shall include all items necessary for the proper execution and completion of the Work by the Contractor. The Contract Documents are complementary, and what is required by one shall be as binding as if required by all; performance by the Contractor shall be required only to the extent consistent with the Contract Documents and reasonably inferable from them as being necessary to produce the indicated results. In the event of any inconsistency among the Contract Documents, at Owner's discretion, the better quality or more stringent requirement shall apply.

PAGE 11

- § 1.2.4 The general character and scope of Work is indicated by the Drawings. Where a portion of the Work is fully drawn and the remainder is merely indicated, the portion fully drawn shall apply to all similar parts of the Work to the extent it can be reasonably inferred.
- § 1.2.5 Lists of "work included" and "work excluded" are not intended to enumerate each and every item of work or appurtenants required. All work indicated shall be supplied except items specially noted as "by others", "by Owner", "not in contract", "existing", or similarly noted. Words and abbreviations which have well-known technical or trade meanings are used in the Contract Documents in accordance with such recognized meanings and consistent with the Stipulated Sum and the detail provided therein.
- § 1.2.6 Where items of material, equipment and labor are referred to in the singular, such item or items shall be provided in the number necessary for the proper completion of the Work.
- § 1.2.7 Unless otherwise specifically indicated, all references to "days" refers to calendar days The term "calendar day"

shall mean the period from midnight of one day to the following midnight of the next day. The term "working day" shall mean the calendar days Mondays through Fridays, excluding holidays recognized by the City of Milwaukee, Wisconsin."

- § 1.5.1 The Architect and the Architect's consultants shall be deemed the authors and owners of their respective Owner shall be deemed the owner of the Instruments of Service, including the Drawings and Specifications, and retain all common law, statutory, and other reserved rights in their Instruments of Service, including copyrights. The Contractor, Sub-contractors, Sub-subcontractors, and Specifications. The Contractor, Subcontractors, Sub-subcontractors, and material or equipment suppliers shall not own or claim a copyright in the Instruments of Service. Submittal or distribution to meet official regulatory requirements or for other purposes in connection with the this Project is not to be construed as publication in derogation of the Architect's or Architect's consultants' reserved rights.
- § 1.5.2 The Contractor, Subcontractors, Sub-subcontractors, and material or equipment suppliers are authorized to use and reproduce the Instruments of Service provided to them, subject to any protocols established pursuant to Sections 1.7 and 1.8, solely and exclusively for execution of the Work. All copies made under this authorization shall bear the copyright notice, if any, shown on the Instruments of Service. The Contractor, Subcontractors, Sub-subcontractors, and material or equipment suppliers may not use the Instruments of Service on other projects or for additions to the this Project outside the scope of the Work without the specific written consent of the Owner, Architect, and the Architect's consultants.and, where applicable, Architect and the Architect's consultants.
- § 1.5.3 Contractor shall indemnify, hold harmless and defend Owner, Architect, Architect's Consultants, against any claim, loss, damage or injury resulting from Contractor's or Subcontractor's: (1) unauthorized use of the Instruments of Service; or (2) use of the Instruments of Service on any other project. This duty shall include the reasonable attorneys' fees incurred by Owner in enforcing the terms set forth in this section.

PAGE 12

The parties shall agree upon protocols governing the transmission and use of Instruments of Service or any other information or documentation in digital form. The Unless otherwise indicated in the Agreement or herein, the parties will use AIA Document E203TM-2013, Building Information Modeling and Digital Data Exhibit, to establish the protocols for the development, use, transmission, and exchange of digital data.

§ 1.8 Building Information Models Use and Reliance

Any use of, or reliance on, all or a portion of a building information model without agreement to protocols governing the use of, and reliance on, the information contained in the model and without having those protocols set forth in AIA Document E203TM 2013, Building Information Modeling and Digital Data Exhibit, and the requisite AIA Document G202TM 2013, Project Building Information Modeling Protocol Form, shall be at the using or relying party's sole risk and without liability to the other party and its contractors or consultants, the authors of, or contributors to, the building information model, and each of their agents and employees.

§ 1.8 Building Information Models Use

Contractor shall provide Drawings and BIM as necessary to meet permitting and design coordination requirements for the Work.

§ 2.1.1 The Owner is the person or entity identified as such in the Agreement and is referred to throughout the Contract Documents as if singular in number. The Owner shall designate in writing a representative those representatives who shall have express authority to bind the Owner with respect to all matters requiring the Owner's approval or authorization. Except as otherwise provided in Section 4.2.1, the Architect does not have such authority. The term "Owner" means the Owner or the Owner's authorized representative.

§ 2.1.2 The Owner shall furnish to the Contractor, within fifteen (15) days after receipt of a written request, information necessary and relevant for the Contractor to evaluate, give notice of, or enforce mechanic's lien rights. Such information shall include a correct statement of the record legal title to the property on which the Project is located, usually referred to as the site, and the Owner's interest therein.

- § 2.2.1 Prior to commencement of the Work and upon written request by the Contractor, the Owner shall furnish to the Contractor Work, the Contractor may request in writing that the Owner provide reasonable evidence that the Owner has made financial arrangements to fulfill the Owner's obligations under the Contract. The Contractor shall have no obligation to commence the Work until the Owner provides such evidence. If commencement of the Work is delayed under this Section 2.2.1, the Contract Time shall be extended appropriately.
- § 2.2.2 Following commencement of the Work and upon written request by the Contractor, the Owner shall furnish to the Contractor reasonable evidence that the Owner has made financial arrangements to fulfill the Owner's obligations under the Contract only Thereafter, the Contractor may only request such reasonable evidence if (1) the Owner fails to make payments to the Contractor as the Contract Documents require; (2) the Contractor identifies in writing a reasonable concern regarding the Owner's ability to make payment when due; or (3) a change in the Work materially changes the Contract Sum. If the Owner fails to provide such evidence, as required, within fourteen (14) days of the Contractor's request, the Contractor may immediately stop the Work and, in that event, shall notify the Owner that the Work has stopped. However, if the request is made because a change in the Work materially changes the Contract Sum under (3) above, the Contractor may immediately stop only that portion of the Work affected by the change until reasonable evidence is provided. If the Work is stopped under this Section 2.2.2, the Contract Time shall be extended appropriately and the Contract Sum shall be increased by the amount of the Contractor's reasonable costs of shutdown, delay and start-up, plus interest as provided in the Contract Documents.
- § 2.2.3 After the Owner furnishes evidence of financial arrangements under this Section 2.2, the Owner shall not materially vary such financial arrangements without prior notice to the Contractor.INTENTIONALLY OMITTED **PAGE 13**
- § 2.3.1 Except for permits and fees that are the responsibility of the Contractor under the Contract Documents, including those required under Section 3.7.1, the Owner shall secure and pay for necessary approvals, easements, assessments and charges required for construction, use or occupancy of permanent structures or for permanent changes in existing facilities. Owner shall furnish and pay for water and utilities utilized by the Contractor, Subcontractors, Sub-subcontractors and Architect in connection with their performance of the Work.

- § 2.3.4 The Except as set forth in the Contract Documents, the Owner shall furnish surveys describing physical characteristics, legal limitations and utility locations for the site of the Project, and a legal description of the site. The Contractor shall be entitled to rely on the accuracy of information furnished by the Owner but shall exercise proper precautions relating to the safe performance of the Work. Contractor shall notify Owner of any discrepancies or information which is lacking in the information provided by the Owner of which Contractor is aware or reasonably should be aware.
- § 2.3.5 The Owner shall furnish information or services required of the Owner by the Contract Documents with reasonable promptness. The Owner shall also furnish any other information or services under the Owner's control and relevant to the Contractor's performance that are necessary for completion of the Work with reasonable promptness after receiving the Contractor's written request for such information or services.

If the Contractor-Contractor: (a) fails to correct Work that is not in accordance with the requirements of the Contract Documents as required by Section 12.2 after fifteen (15) days prior written notice; or (b) repeatedly fails to carry out Work in accordance with the Contract Documents, Documents and, after fifteen (15) days prior written notice, and in the case under either (a) or (b) the Contractor does not undertake efforts to cure the non-conforming Work,, the Owner may issue a written order to the Contractor to stop the Work, or any portion thereof, until the cause for such order has

been eliminated; however, the right of the Owner to stop the Work shall not give rise to a duty on the part of the Owner to exercise this right for the benefit of the Contractor or any other person or entity, except to the extent required by Section 6.1.3.

If the Contractor defaults or neglects to carry out the Work in accordance with the Contract Documents and fails within a ten-day fifteen-day period after receipt of written notice from the Owner to commence and continue correction of such default or neglect with diligence and promptness, the Owner may, without prejudice to other remedies the Owner may have, correct such default or neglect. Such action by the Owner and amounts charged to the Contractor are both subject to prior approval of the Architect and the Architect may, pursuant to Section 9.5.1, withhold or nullify a Certificate for Payment in whole or in part, to the extent reasonably necessary to reimburse the Owner for the such deficiencies. In such case an appropriate Change Order shall be issued deducting from payments then or thereafter due the Contractor the reasonable cost of correcting such deficiencies, including Owner's expenses and compensation for the Architect's additional services and expenses made necessary by such default, neglect, or failure. If eurrent and future payments payments then or thereafter due the Contractor are not sufficient to cover such amounts, the Contractor shall pay the difference to the Owner. If the Contractor disagrees with the actions of the Owner or the Architect, or the amounts claimed as costs to the Owner, the Contractor may file a Claim pursuant to Article 15.

PAGE 14

- § 3.1.1 The Contractor is the person or entity identified as such in the Agreement and is referred to throughout the Contract Documents as if singular in number. The Contractor shall be lawfully licensed, and all applicable Subcontractors and suppliers shall be lawfully licensed and registered, if required in the jurisdiction where the Project is located. The Contractor shall designate in writing a representative who shall have express authority to bind the Contractor with respect to all matters under this Contract. The term "Contractor" means the Contractor or the Contractor's authorized representative.
- § 3.1.2 The Contractor shall perform the Work in accordance with the Contract Documents. Documents and Submittals approved pursuant to Section 3.12 herein.

- § 3.2.1 Execution of the Contract by the Contractor is a representation that the Contractor has visited the site, become generally familiar with local conditions under which the Work is to be performed, and correlated personal observations with requirements of the Contract Documents. Notwithstanding such observations, the Contractor shall nevertheless be entitled to rely upon any Owner provided information about the Project, Work or site, including, but not limited to, surveys, subsurface tests and data, the identification and location of hazardous substances and the condition and characteristics of structures on the site, unless Contractor has specific knowledge that the Owner provided information contrary to existing conditions. If that occurs, then Contractor shall ask for clarification from Owner or Architect. Whenever possible, the Contractor and all of its necessary Subcontractors, as appropriate, shall endeavor to verify existing field conditions that are visible (i.e., accessible through normal and customary methods of observation and investigation). Where existing field conditions are not visible (e.g., below grade) or concealed within construction assemblies, and existing items are indicated on the drawings, such items shall be considered as being shown schematically only, if so represented or designated as such by Architect.
- § 3.2.2 Because the Contract Documents are complementary, the Contractor shall, before starting each portion of the Work, carefully study and compare the various Contract Documents relative to that portion of the Work, as well as the information furnished by the Owner pursuant to Section 2.3.4, shall take field measurements of any existing conditions related to that portion of the Work, and shall observe any conditions at the site affecting it. These obligations are for the purpose of facilitating coordination and construction by the Contractor and are not for the purpose of discovering errors, omissions, or inconsistencies in the Contract Documents; however, the Documents. The Contractor shall promptly report to the Architect and Owner any errors, inconsistencies or omissions discovered discovered, including design errors or omissions, by or made known to the Contractor as a request for information in such form as the Architect may require. The Contractor shall halt all affected work until such discrepancies have been corrected or until directed otherwise by Owner or Architect. The direct out-of-pocket costs related to this Work stoppage shall be reimbursed to the Contractor and, if the Work stoppage results in a delay in the Project Schedule, the

Completion Date shall be extended by Change Order. It is recognized that the Contractor's review is made in the Contractor's capacity as a contractor and not as a licensed design professional, unless otherwise specifically provided in the Contract Documents.

- § 3.2.3 The Except as specifically set forth in the Contract Documents, the Contractor is not required to ascertain that the Contract Documents are in accordance with applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, but the Contractor shall promptly report to the Architect and Owner any nonconformity discovered by or made known to the Contractor as a request for information in such form as the Architect may require.
- § 3.2.4 If the Contractor believes that additional cost or time is involved because of clarifications or instructions the Architect issues in response to the Contractor's notices or requests for information pursuant to Sections 3.2.2 or 3.2.3, the Contractor shall submit or the Architect's failure to respond to such requests for information or notices, the Contractor shall make Claims as provided in Article 15. If the Contractor fails to perform the obligations of Sections 3.2.2 or 3.2.3, the Contractor shall pay such costs and damages to the Owner, subject to Section 15.1.7, as would have been avoided if the Contractor had performed such obligations. If the Contractor performs those obligations, the Contractor shall not be liable to the Owner or Architect for damages resulting from errors, inconsistencies or omissions in the Contract Documents, for differences between field measurements or conditions and the Contract Documents, or for nonconformities of the Contract Documents to applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities, authorities, unless the Contractor recognized such error, inconsistency, omission or difference and failed to report it to the Architect. If the Contractor performs any construction activity involving an error, inconsistency, omission or difference in the Contract Documents that Contractor knowingly recognized or should have reasonably recognized without such notice to the Architect and Owner, the costs for correction will be allocated among the responsible parties.
- § 3.2.5 The Contractor shall provide reasonably necessary estimating and coordination services to evaluate the value engineering and cost savings options. Contractor shall not be entitled to an increase in the Contract Price for such services. For those options the Owner approves as a lump sum adjustment, the Contract Price shall be reduced by the full amount of direct costs associated with the change through the Change Order process.

PAGE 15

§ 3.3.1 The Contractor shall supervise and direct the Work, using the Contractor's best skill and attention. The Contractor shall be solely responsible for, and have control over, construction means, methods, techniques, sequences, and procedures, and for coordinating all portions of the Work under the Contract, unless the Contract Documents give other specific instructions concerning these matters. If the Contract Documents give specific instructions concerning construction means, methods, techniques, sequences, or procedures, the Contractor shall evaluate the jobsite safety thereof and shall be and, except as stated below, shall be fully and solely responsible for the jobsite safety of such means, methods, techniques, sequences, or procedures. If the Contractor determines that such means, methods, techniques, sequences or procedures may not be safe, the Contractor shall give timely written notice to the Owner and Architect, and shall propose alternative means, methods, techniques, sequences, or procedures. The Architect shall evaluate the proposed alternative solely for conformance with the design intent for the completed construction. Unless the Architect objects to the Contractor's proposed alternative, the Contractor shall perform the Work using its alternative means, methods, techniques, sequences, or procedures, not proceed with that portion of the Work without further written instructions from the Architect. If the Contractor is then instructed to proceed with the required means, methods, techniques, sequences, or procedures without acceptance of changes proposed by the Contractor, the Owner shall be solely responsible for any loss or damage arising from those Owner-required means, methods, techniques, sequences, or procedures, unless Contractor is negligent.

§ 3.4.1 Unless otherwise provided in the Contract Documents, the Contractor shall provide and pay for labor, materials, equipment, tools, construction equipment and machinery, water, heat, utilities, transportation, and other facilities and services necessary for proper execution and completion of the Work, whether temporary or permanent and whether or not incorporated or to be incorporated in the Work. The Owner will not be liable to assume any responsibility for the damage or wear to the Contractor's tools, materials and/or equipment except to the extent covered by any applicable insurance or to the extent that the Owner and its agents, employees or contractors are responsible for the damage or wear.

- § 3.4.2 Except in the case of minor changes in the Work approved authorized by the Architect in accordance with Section 3.12.8 or ordered by the Architect in accordance with Section Sections 3.12.8 or 7.4, the Contractor may make substitutions of specified materials or equipment only with the consent and approval of the Owner, after evaluation and recommendation by the Architect and in accordance with a Change Order or Construction Change Directive.
- § 3.4.2.1 Where a definite material or method is specified, it is not the intention to discriminate against any "approved equal" product by another manufacturer. Rather, it is the intention to set a definite standard. Open competition is expected. The Contractor shall submit to the Architect the substitutions it proposes together with samples, complete evidence of quality and any credits that may accrue to the Owner for allowing the various substitutions. The Architect shall investigate all proposed substitutions, consult with the Owner for its review and approval, and render final decisions as is necessary to avoid any delay in the Work.
- § 3.4.2.2 The Contractor shall ascertain that items offered as equals to specified items will fit the physical limits of space shown on the Drawings, and leave ample clearance for proper installation, operation and servicing of the item and all adjacent items.
- § 3.4.2.3 Materials and equipment proposed as substitutes for specified items may be rejected by the Architect and Owner.
- § 3.4.2.4 If during the performance of Work any materials or equipment specified in the Contract Documents become unavailable because of government restrictions or because of other market conditions (which are not the result of Contractor's delay in order or purchasing), the Owner, Architect and Contractor shall collectively suggest and consider alternatives and substitutes. The Architect shall then issue a final recommendation for consideration and approval by Owner. Contractor will be entitled to an increase in the Contract Sum for actual increased costs incurred directly as a result of an alternative or substitution required because of government restrictions or other market conditions to the extent such cost could not be mitigated or otherwise avoided. Contractor will provide Owner with invoices or such other documents as Owner may reasonably require to verify Contractor's increased costs.
- § 3.4.3 The Contractor shall enforce strict discipline and good order among the Contractor's employees and other persons carrying out the Work. The Contractor shall not permit employment of unfit persons or persons not properly skilled in tasks assigned to them. The Contractor shall promptly remove any persons of whom Owner has a reasonable objection to their fitness for work.
- § 3.4.4 Contractor shall provide to Architect the required information for all materials and equipment which, pursuant to the Contract Documents must be approved by Architect. The Contractor shall provide such information to the Architect according to the submittal schedule included in the Construction Schedule, or, if not specifically included in the Construction Schedule, then timely enough to allow for Architect's review and to otherwise not cause a delay in the Project Schedule.

PAGE 16

- § 3.5.1 The Contractor warrants to the Owner and Architect that materials and equipment furnished under the Contract will be of good quality and new unless the Contract Documents require or permit otherwise. The Contractor further warrants that the Work will conform to the requirements of the Contract Documents and will be free from defects, except for those inherent in the quality of the Work the Contract Documents require or permit. Work, materials, or equipment not conforming to these requirements requirements, including substitutions not properly approved and authorized, may be considered defective. The Contractor's warranty excludes remedy for damage or defect caused by abuse, alterations to the Work not executed by the Contractor, improper or insufficient maintenance, improper operation, or normal wear and tear and normal usage. If required by the Architect, the Contractor shall furnish satisfactory evidence as to the kind and quality of materials and equipment.
- § 3.5.2 All material, equipment, or other special warranties required by the Contract Documents shall be issued in the name of the Owner, or shall be transferable to the Owner, and shall commence in accordance with Section 9.8.4. The Contractor's warranty under this Section is not exclusive, and any other express warranties stated elsewhere, may also be exercised by Owner at its option. In addition, the Contractor shall assign to the Owner all manufacturers' and suppliers' warranties, express or implied, respecting any part of the Work which Contractor or Subcontractors receive not later than at the time Final Payment is made. The assignments, copies of all warranties and all product operation

manuals for proper use and maintenance of equipment shall be conveyed to the Owner prior to Final Payment for the Work (i.e., final retainage). Owner may, at its option, release final retainage for that portion of the Work for which all contract close-out requirements have been satisfied.

§ 3.5.3 The Contractor and any applicable Subcontractors agree, as part of its post completion Work and its warranty obligations, to participate in warranty walk-throughs at the Project Site with the Owner and its Architect eleven (11) months after Substantial Completion of each phase of the Work to identify warranty work and any other items which must be corrected to conform with the requirements of the Contract Documents.

PAGE 17

§ 3.7.3 If the Contractor performs Work knowing it to be contrary to applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, the Contractor shall assume appropriate responsibility for such Work and shall bear the costs attributable to correction. However, if the Contractor observes that portions of the Contract Documents are at variance therewith and promptly notifies the Architect and Owner in writing, necessary changes shall be accomplished by appropriate Modification. Any Modification shall be in accordance with Article 7. Notwithstanding the above, for those trades requiring licensed tradesmen, Work depicted for those trades shall be installed in a code conforming manner irrespective of references or representations in the documents to the contrary, at no additional cost to the Owner.

...

If the Contractor encounters conditions at the site that are (1) subsurface or otherwise concealed physical conditions that differ materially from those indicated in the Contract Documents or (2) unknown physical conditions of an unusual nature that differ materially from those ordinarily found to exist and generally recognized as inherent in construction activities of the character provided for in the Contract Documents, the Contractor shall promptly provide notice to the Owner and the Architect before conditions are disturbed and in no event later than 14-21 days after first observance of the conditions. The Architect will promptly investigate such conditions and, if the Architect determines that they differ materially and cause an increase or decrease in the Contractor's cost of, or time required for, performance of any part of the Work, will recommend that an equitable adjustment be made in the Contract Sum or Contract Time, or both, recommend, for Owner's review and approval or disapproval, an equitable adjustment in the Contract Time. Any adjustment to the Contract Sum recommended by the Architect will be based on the factors set forth in Sections 7.3.4.1 to 7.3.4.5. If the Architect determines that the conditions at the site are not materially different from those indicated in the Contract Documents and that no change in the terms of the Contract is justified, the Architect shall promptly notify the Owner and Contractor, Contractor of its recommendation in writing, stating the reasons. If either party disputes the Architect's determination or recommendation, that party may submit a Claim recommendation, that party proceed as provided in Article 15.

§ 3.7.5 If, in the course of the Work, the Contractor encounters human remains or recognizes the existence of burial markers, archaeological sites or wetlands not indicated in the Contract Documents, the Contractor shall immediately suspend any operations that would affect them and shall notify the Owner and Architect. Upon receipt of such notice, the Owner shall promptly take any action necessary to obtain governmental authorization required to resume the operations. The Contractor shall continue to suspend such operations until otherwise instructed by the Owner but shall continue with all other operations that do not affect those remains or features. Requests for adjustments in the Contract Sum and Contract Time arising from the existence of such remains or features may be made as provided in Article 15. If requested by Contractor, any adjustment to the Contract Sum or Contract Time will be based on the factors set forth in Sections 7.3.4.1 to 7.3.4.5.

PAGE 18

- § 3.9.1 The Contractor shall employ a competent superintendent and necessary assistants who shall be in attendance at the Project site during performance of the Work. at the Project site as necessary for the performance of the Work, and who shall not be replaced without the consent of the Owner. If the superintendent or an assistant leaves the employment of the Contractor, the replacement shall be subject to Owner's approval. The superintendent shall represent the Contractor, and communications given to the superintendent shall be as binding as if given to the Contractor. All communications shall be confirmed in writing.
- § 3.9.2 The Contractor, as soon as practicable after award of the Contract, shall notify furnish in writing to the Owner and Architect of the name and qualifications of a proposed superintendent. Within 14 days of receipt of the

information, the Architect may notify the Contractor, stating whether the Owner or the Architect The Owner may reply within fourteen (14) days to the Contractor in writing, stating whether the Owner (1) has reasonable objection to the proposed superintendent or (2) requires additional time for review. Failure of the Architect to provide notice to review. Failure to reply within the 14-day period shall constitute notice of no reasonable objection.

- § 3.9.3 The Contractor shall not employ a proposed superintendent to whom the Owner or Architect-has made reasonable and timely objection. The Contractor shall not change the superintendent without the Owner's consent, which shall not unreasonably be withheld or delayed.
- § 3.9.4 The Contractor shall have the Superintendent and other competent representatives familiar with the Work attend meetings which may be held at the job site or at such place as Owner designates.

- § 3.10.1 The If the Owner requires, the Contractor, promptly after being awarded the Contract, shall prepare and submit for the Owner's review and Architect's information-approval a Contractor's construction schedule for the Work. The schedule shall contain detail appropriate for the Project, including (1) the date of commencement of the Work, interim schedule milestone dates, and the date of Substantial Completion; (2) an apportionment of the Work by construction activity; and (3) the time required for completion of each portion of the Work. The schedule shall provide for the orderly progression of the Work to completion and shall not exceed time limits current under the Contract Documents. The schedule shall be revised at appropriate intervals as required by the conditions of the Work and Project. Project, shall be related to the entire Project to the extent required by the Contract Documents, and shall provide for expeditious and practicable execution of the Work.
- § 3.10.2 The Contractor, Contractor shall prepare a submittal schedule promptly after being awarded the Contract and thereafter as necessary to maintain a current submittal schedule, and shall submit a submittal the schedule for the Architect's approval. The Architect's and Owner's review and approval, which approval shall not be unreasonably delayed or withheld. The submittal schedule shall (1) be coordinated with the Contractor's construction schedule, and (2) allow the Architect and Owner reasonable time to review submittals. If the Contractor fails to submit a submittal schedule, or fails to provide submittals in accordance with the approved submittal schedule, the Contractor shall not be entitled to any increase in Contract Sum or extension of Contract Time based on the time required for review of submittals.

The Contractor shall make available, at the Project site, the Contract Documents, including Change Orders, Construction Change Directives, and other Modifications, in good order and marked currently to indicate field changes and selections made during construction, and the approved Shop Drawings, Product Data, Samples, and similar required submittals. These shall be in electronic form or paper copy, available to the Architect and Owner, and delivered to the Architect for submittal to the Owner upon completion of the Work as a record of the Work as constructed. § 3.11.1 The Contractor shall maintain at the Project site, one (1) copy of the Contract Documents, including Drawings, Specifications, Addenda, Change Orders, Construction Change Directives, and other Modifications, in good order and marked currently to indicate Field Changes and selections made during construction, and the approved Shop Drawings, Product Data, Samples, and similar required submittals. These shall be in electronic form or paper copy, available to the Architect and Owner, and delivered to the Architect for submittal to the Owner upon completion of the Work as a record of the Work as constructed and prior to final payment (i.e. final retainage). Owner may, at its option, release final retainage for that portion of the Work for which all contract close-out requirements have been satisfied.

§ 3.11.2 At completion of each phase of the Work, record drawings of all mechanical and electrical systems, if any, shall be submitted to the Architect, which indicate all Field Changes. If not submitted, Applications for Payment will not be processed.

PAGE 19

§ 3.12.4 Shop Drawings, Product Data, Samples, and similar submittals are not Contract Documents. Their purpose is to demonstrate how the way by which the Contractor proposes to conform to the information given and the design concept expressed in the Contract Documents for those portions of the Work for which the Contract Documents

require submittals. Review by the Architect is subject to the limitations of Section 4.2.7. Informational submittals upon which the Architect is not expected to take responsive action may be so identified in the Contract Documents. Submittals that are not required by the Contract Documents may be returned by the Architect without action.

§ 3.12.5 The Contractor shall review for compliance with the Contract Documents, approve, and submit to the Architect, Shop Drawings, Product Data, Samples, and similar submittals required by the Contract Documents, in accordance with the submittal schedule approved by the Architect and Owner or, in the absence of an approved submittal schedule, with reasonable promptness and in such sequence as to cause no delay in the Work or in the activities of the Owner or of Separate Contractors.

- § 3.12.7 The Contractor shall perform no portion of the Work for which the Contract Documents require submittal and review of Shop Drawings, Product Data, Samples, or similar submittals, submittals until the respective submittal has been approved by the Architect.
- § 3.12.8 The Work shall be in accordance with approved submittals except that the Contractor shall not be relieved of responsibility for deviations from the requirements of the Contract Documents by the Architect's approval of Shop Drawings, Product Data, Samples, or similar submittals, unless the Contractor has specifically notified informed the Architect in writing of such deviation at the time of submittal and (1) the Architect has given written approval to the specific deviation as a minor change in the Work, or (2) a Change Order or Construction Change Directive has been issued authorizing the deviation. The Contractor shall not be relieved of responsibility for errors or omissions in Shop Drawings, Product Data, Samples, or similar submittals, by the Architect's approval thereof.
- § 3.12.9 The Contractor shall direct specific attention, in writing or on resubmitted Shop Drawings, Product Data, Samples, or similar submittals, to revisions other than those requested by the Architect on previous submittals. In the absence of such written notice, the Architect's approval of a resubmission shall not apply to such revisions.

§ 3.12.10.1 If professional design services or certifications by a design professional related to systems, materials, or equipment are specifically required of the Contractor by the Contract Documents, the Owner and the Architect will specify all performance and design criteria that such services must satisfy. The Contractor shall be entitled to rely upon the adequacy and accuracy of the performance and design criteria provided in the Contract Documents. The Contractor shall cause such services or certifications to be provided by an appropriately a properly licensed design professional, whose signature and seal shall appear on all drawings, calculations, specifications, certifications, Shop Drawings, and other submittals prepared by such professional. Shop Drawings, and other submittals related to the Work, designed or certified by such professional, if prepared by others, shall bear such professional's written approval when submitted to the Architect. The Owner and the Architect shall be entitled to rely upon the adequacy and accuracy adequacy, accuracy, and completeness of the services, certifications, and approvals performed or provided by such design professionals, provided the Owner and Architect have specified to the Contractor the appropriate performance and design criteria that such services must satisfy. Pursuant to this Section 3.12.10, the The Architect will review and approve approve, reject, or take other appropriate action on submittals only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents.(i.e., "revise and resubmit" or "proceed as revised") in accordance with Section 4.2.7 herein. Except to the extent Contractor is provided Design-Build services or has retained Subcontractors that are performing design-build services (e.g., MEP Subcontractors), the Contractor shall not be responsible for the adequacy of the performance and design criteria specified in the Contract Documents, but shall be responsible to promptly report to the Architect and Owner any errors, inconsistencies or omissions discovered in the performance and design criteria.

PAGE 20

§ 3.12.11 Contractor shall furnish three (3) complete sets of manuals containing the manufacturers' warranties and instructions necessary for maintenance and operation of each item of equipment and apparatus it furnishes under the Contract Documents, warranty information and any additional data specifically requested under the various sections of the Specifications for each division of the Work to the Architect for subsequent conveyance with all Contractors' manuals to the Owner at or before Final Payment. Each manual shall be arranged in logical order, indexed and suitably bound. Alternatively, the Contractor may provide this information in the as-built BIM for the Work.

The Contractor shall confine operations at the site to areas permitted by applicable laws, statutes, ordinances, codes, rules and regulations, lawful orders of public authorities, and the Contract Documents and shall not unreasonably encumber the site with materials or equipment. § 3.13.1 The Contractor shall confine operations at the site to areas permitted by applicable laws, statutes, ordinances, codes, rules and regulations, lawful orders of public authorities, and the Contract Documents and shall not unreasonably encumber the site with materials or equipment.

- § 3.13.2 The Contractor shall store materials on the site where directed and in such a manner as will not damage the area in which they are stored. Material deliveries shall be scheduled so that they are not stored longer than necessary. All items furnished to the site by the Owner shall be stored as directed.
- § 3.13.3 Where the Contractor's operations will affect the Owner's operation and use of existing facilities, the Contractor will arrange its work to minimize its effect on the existing facility, and will coordinate the timing, sequencing and duration of the disruptions with the Owner's representative and proceed with such work only after receiving authority to do so.

- § 3.14.1 The Contractor shall be responsible for cutting, fitting, or patching required to complete the Work or to make its parts fit together properly. All areas requiring cutting, fitting, or and patching shall be restored to the condition existing prior to the cutting, fitting, or and patching, unless otherwise required by the Contract Documents.
- § 3.14.2 The Contractor shall not damage or endanger a portion of the Work or fully or partially completed construction of the Owner or Separate Contractors by cutting, patching, or otherwise altering such construction, or by excavation. The Contractor shall not cut or otherwise alter <u>such</u> construction by the Owner or a Separate Contractor except with written consent of the Owner and of the-such Separate Contractor. Consent shall not be unreasonably withheld. The Contractor shall not unreasonably withhold, from the Owner or a Separate Contractor, its consent to cutting or otherwise altering the Work.

§ 3.15.1 The Contractor shall keep the premises and surrounding area free from accumulation of waste materials and or rubbish caused by operations under the Contract. At completion of the Work, the Contractor shall remove waste materials, rubbish, the Contractor's tools, construction equipment, machinery, and surplus materials from and about the Project.

PAGE 21

The Contractor shall pay all royalties and license fees. The Contractor shall defend suits or claims for infringement of copyrights and patent rights and shall hold the Owner and Architect harmless from loss on account thereof, but shall not be responsible for such defense or loss when a particular design, process, or product of a particular manufacturer or manufacturers is required by the Contract Documents, or where the copyright violations are contained in Drawings, Specifications, or other documents prepared by the Owner or Architect. However, if the Contractor has reason to believe that the required design, process or produce is an infringement of a copyright or patent is discovered by, or made known to, the Contractor, the Contractor shall be responsible for the such loss unless the such information is promptly furnished to the Architect.

§ 3.18.1 To the fullest extent permitted by law, the Contractor shall indemnify and hold harmless the Owner, Architect, Architect's consultants, and agents and employees of any of them-indemnify, defend, and hold harmless the Owner and its board members, officers, directors, agents and employees from and against claims, damages, losses, and expenses, including including, but not limited to attorneys' fees, arising out of or resulting from performance of the Work, provided that such claim, damage, loss, or expense is attributable to bodily injury, sickness, disease or death, or to injury to or destruction of tangible property (other than the Work itself), property, but only to the extent caused by the negligent acts or omissions of the Contractor, a Subcontractor, anyone directly or indirectly employed by them, or anyone for whose acts they may be liable, regardless of whether or not such claim, damage, loss, or expense is caused in part by a party indemnified hereunder. Such obligation shall not be construed to negate, abridge, or reduce other rights or obligations of indemnity that would otherwise exist as to a party or person described in this Section 3.18.

§ 4.1.2 Duties, responsibilities, and limitations of authority of the Architect as set forth in the Contract Documents shall not be restricted, modified, or extended without written consent of the Owner, Contractor, and Architect. Consent shall not be unreasonably withheld. Owner and written notice of such change is provided to the Contractor.

- § 4.2.1 The Architect will provide administration of the Contract as described in the Contract Documents and will be an Owner's representative during construction (1) during construction, (2) until the date the Architect issues the final Certificate for Payment. Payment, and (3) with the Owner's concurrence, from time to time during the one-year period of correction of Work described in Section 12.2.2. The Architect will have authority to act on behalf of the Owner only to the extent provided in the Contract Documents.
- § 4.2.2 The Architect will visit the site at intervals appropriate to the stage of construction, or as otherwise agreed with the Owner, (1) to become generally familiar with the progress and quality of the portion of the Work completed, and (2) to inspect the Work, (3) to determine in general if the Work observed is being performed in a manner indicating that the Work, when fully completed, will be in accordance with the Contract Documents. Documents, and (4) when applicable, to determine whether to issue a Certificate for Payment. However, the Architect will not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. The Architect will not have control over, charge of, or responsibility for the construction means, methods, techniques, sequences or procedures, or for the safety precautions and programs in connection with the Work, since these are solely the Contractor's rights and responsibilities under the Contract Documents, except as provided in Section 3.3.1.
- § 4.2.3 On the basis of the site visits, the Architect will keep the Owner reasonably informed about the progress and quality of the portion of the Work completed, and promptly report to the Owner (1) known deviations from the Contract Documents, (2) known deviations Documents and from the most recent construction schedule submitted by the Contractor, and (3)(2) defects and deficiencies observed in the Work. The Architect will not be responsible for the Contractor's failure to perform the Work in accordance with the requirements of the Contract Documents. Documents, except as set forth herein. The Architect will not have control over or charge of, and will not be responsible for acts or omissions of, the Contractor, Subcontractors, or their agents or employees, or any other persons or entities performing portions of the Work.

PAGE 22

The Except as otherwise provided in the Contract Documents, the Owner and Contractor shall include the Architect in all communications that relate to or affect the Architect's services or professional responsibilities. The Owner shall promptly notify the Architect of the substance of any direct communications between the Owner and the Contractor otherwise relating to the Project. Communications by and with the Architect's consultants shall be through the Architect. Communications by and with Subcontractors and suppliers shall be through the Contractor. Communications by and with Separate Contractors shall be through the Owner. The Contract Documents may specify other communication protocols.

§ 4.2.6 The Architect has authority to reject Work that does not conform to the Contract Documents. Whenever the Architect considers it necessary or advisable, the Architect will have authority to require inspection or testing of the Work in accordance with Sections 13.4.2 and 13.4.3, whether or not the such Work is fabricated, installed or completed. However, neither this authority of the Architect nor a decision made in good faith either to exercise or not to exercise such authority shall give rise to a duty or responsibility of the Architect to the Contractor, Subcontractors, material and equipment suppliers, their agents or employees, or other persons or entities performing portions of the Work.

- § 4.2.7 The Architect will review and approve, or take other appropriate action upon, the Contractor's submittals such as Shop Drawings, Product Data, and Samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Architect's action will be taken in accordance with the submittal schedule approved by the Architect or, in the absence of an approved submittal schedule, with reasonable promptness while allowing sufficient time in the Architect's professional judgment to permit adequate review. Review of such submittals is not conducted for the purpose of determining the accuracy and completeness of other details such as dimensions and quantities, or for substantiating instructions for installation or performance of equipment or systems, all of which remain the responsibility of the Contractor as required by the Contract Documents. The Architect's review of the Contractor's submittals shall not relieve the Contractor of the obligations under Sections 3.3, 3.5, and 3.12. The Architect's review shall not constitute approval of safety precautions or or procedures. The Architect's approval of a specific item shall not indicate approval of an assembly of which the item is a component.
- § 4.2.8 The Architect will prepare Change Orders and Construction Change Directives, and may order authorize minor changes in the Work as provided in Section 7.4. The Architect will investigate and make determinations and recommendations regarding concealed and unknown conditions as provided in Section 3.7.4.

§ 4.2.10 If the Owner and Architect agree, the Architect will provide one or more Project representatives to assist in carrying out the Architect's responsibilities at the site. The Owner shall notify the Contractor of any change in the duties, responsibilities and limitations of authority of the Project representatives. such Project representatives shall be

as set forth in an exhibit to be incorporated in the Contract Documents.

PAGE 23

§ 4.2.12 Interpretations and decisions of the Architect will be consistent with the intent of, and reasonably inferable from, the Contract Documents and will be in writing or in the form of drawings. When making such interpretations and decisions, the Architect will endeavor to secure faithful performance by both Owner and Contractor, will not show partiality to either, either and will not be liable for results of interpretations or decisions rendered in good faith.

. .

- § 5.2.1 Unless otherwise stated in the Contract Documents, Documents or the bidding requirements, the Contractor, as soon as practicable after award of the Contract, shall notify the Owner and Architect of the furnish in writing to the Owner through the Architect the names of persons or entities proposed for each principal portion of the Work, including those who are to furnish materials or equipment fabricated to a special design. Within 14-10 days of receipt of the information, the Owner or Architect may notify the Contractor in writing whether the Owner or the Architect (1) has reasonable objection to any such proposed person or entity or (2)-that the Owner or the Architect requires additional time for review. Failure of the Owner or Architect to provide notice within the 14-day 10-day period shall constitute notice of no reasonable objection.
- § 5.2.2 The Contractor shall not contract with a proposed person or entity to whom the Owner or Architect has made reasonable and timely objection. The Contractor shall not be required to contract with anyone to whom the Contractor has made reasonable objection. It shall be considered reasonable for the Contractor to reject a Subcontractor who fails or refuses to sign a form of subcontract reasonably acceptable to the Contractor.

• • •

§ 5.2.4 The Contractor shall not substitute a Subcontractor, person, or entity for one previously selected if the Owner or Architect makes reasonable objection to such substitution.

...

User Notes:

By appropriate written agreement, the Contractor shall require each Subcontractor, to the extent of the Work to be performed by the Subcontractor, to be bound to the Contractor by terms of the Contract Documents, and to assume toward the Contractor all the obligations and responsibilities, including the responsibility for safety of the

Subcontractor's Work that the Contractor, by these Contract Documents, assumes toward the Owner and Architect. Each subcontract agreement shall preserve and protect the rights of the Owner and Architect under the Contract Documents with respect to the Work to be performed by the Subcontractor so that subcontracting thereof will not prejudice such rights, and shall allow to the Subcontractor, unless specifically provided otherwise in the subcontract agreement, the benefit of all rights, remedies, and redress against the Contractor that the Contractor, by the Contract Documents, has against the Owner. Where appropriate, the Contractor shall require each Subcontractor to enter into similar agreements with Sub-subcontractors. The Contractor shall make available to each proposed Subcontractor, prior to the execution of the subcontract agreement, copies of the Contract Documents to which the Subcontractor will be bound, and, upon written request of the Subcontractor, identify to the Subcontractor terms and conditions of the proposed subcontract agreement that may be at variance with the Contract Documents. Subcontractors will similarly make copies of applicable portions of such documents available to their respective proposed Sub-subcontractors.§ **5.3.1** By appropriate written agreement, the Contractor shall require each Subcontractor, to the extent of the Work to be performed by the Subcontractor, to be bound to the Contractor by terms of the Contract Documents, and to assume toward the Contractor all the obligations and responsibilities, including the responsibility for safety of the Subcontractor's Work that the Contractor, by these Contract Documents, assumes toward the Owner and Architect. Each subcontract agreement shall preserve and protect the rights of the Owner and Architect under the Contract Documents with respect to the Work to be performed by the Subcontractor so that subcontracting thereof will not prejudice such rights. Where appropriate, the Contractor shall require each Subcontractor to enter into similar agreements with Sub-subcontractors. The Contractor shall make available to each proposed Subcontractor, prior to the execution of the subcontract agreement, copies of the Contract Documents to which the Subcontractor will be bound. Subcontractors will similarly make copies of applicable portions of such documents available to their respective proposed Sub-subcontractors.

§ 5.3.2 Owner is an intended third-party beneficiary of all subcontracts and material supply contracts of whatever tier, with the right to directly enforce, both during and after the construction period, subcontractor and material supplier obligations to meet prevailing standards or workmanship and to comply with the contract documents including but not limited to all applicable express and implied warranties. During the construction period, that right shall only be exercised in cooperation with Contractor, unless Contractor is in default under the Contract and fails to cure the same.

PAGE 24

.1 assignment is effective only after termination of the Contract by the Owner for cause pursuant to Section 14.2 and only for those subcontract agreements that the Owner accepts by notifying the Subcontractor and Contractor; Contractor in writing; and

When the Owner accepts the assignment of a subcontract agreement, the Owner assumes the Contractor's rights and obligations under the subcontract.

§ 5.4.2 Upon such assignment, if the Work has been suspended for more than 30 days, the Subcontractor's compensation shall be equitably adjusted for actual increases in eost-the direct costs resulting from the suspension.

§ 5.4.3 Upon such assignment to the Owner under this Section 5.4, the Owner may further assign the subcontract to a successor contractor or other entity. If the Owner assigns the subcontract to a successor contractor or other entity, the Owner shall nevertheless remain legally responsible for all of the successor contractor's obligations under the subcontract.

§ 6.1.1 The term "Separate Contractor(s)" shall mean other contractors retained by the Owner under separate agreements. The Owner reserves the right to perform construction or operations related to the Project with the Owner's own forces, and with Separate Contractors retained in connection with other portions of the Project or other construction or operations on the Project site under Conditions of the Contract identical or substantially similar to those of this Contract, including those provisions of the Conditions of the Contract related to insurance and waiver of subrogation.

- § 6.1.3 The Owner shall provide for coordination of the activities of the Owner's own forces and of each Separate Contractor with the Work of the Contractor, who shall cooperate with them. The Contractor shall participate with any other Separate Contractors and the Owner in reviewing their construction schedules. The Contractor shall make any revisions to its-the construction schedule deemed necessary after a joint review and mutual agreement. The construction schedules shall then constitute the schedules to be used by the Contractor, Separate Contractors, and the Owner until subsequently revised.
- § 6.1.4 Unless otherwise provided in the Contract Documents, when the Owner performs construction or operations related to the Project with the Owner's own forces or with Separate Contractors, the Owner or its Separate Contractors shall have the same obligations and rights that be deemed to be subject to the same obligations and to have the same rights that apply to the Contractor has under the Conditions of the Contract, including, without excluding others, those as stated in Article 3, this Article 6, and Articles 10, 11, and 12. PAGE 25
- § 6.2.2 If part of the Contractor's Work depends for proper execution or results upon construction or operations by the Owner or a Separate Contractor, the Contractor shall, prior to proceeding with that portion of the Work, promptly notify the Architect of apparent discrepancies or defects in the construction or operations by the Owner or Separate Contractor report to the Architect apparent discrepancies or defects in such other construction or operations that would render it unsuitable for proper execution and results of the Contractor's Work. Failure of the Contractor to notify the Architect of apparent discrepancies or defects prior to proceeding with the Work such proper execution. Failure of the Contractor to so report shall constitute an acknowledgment that the Owner's or Separate Contractor's completed or partially completed construction is fit and proper to receive the Contractor's Work. The Contractor shall not be responsible for discrepancies or defects in the construction or operations by the Owner or Separate Contractor that are not apparent. Work, except as to defects not then reasonably discoverable.
- § 6.2.3 The Contractor shall reimburse the Owner for costs the Owner incurs that are payable to a Separate Contractor because of the Contractor's delays, improperly timed activities or defective construction. The Owner shall be responsible to the Contractor for actual and direct costs the Contractor incurs because of a Separate Contractor's delays, improperly timed activities, damage to the Work or defective construction.
- § 6.2.4 The Contractor shall promptly remedy damage that the Contractor wrongfully causes to completed or partially completed construction or to property of the Owner or Separate Contractor as provided in Section 10.2.5. Section 10.2.5, but only to the extent caused by Contractor.

If a dispute arises among the Contractor, Separate Contractors, and the Owner as to the responsibility under their respective contracts for maintaining the premises and surrounding area free from waste materials and rubbish, the Owner may clean up and the Architect will allocate the cost among those responsible.

§ 7.1.4 Except for reasons of emergency or for minor changes in the Work (see Sections 7.4 herein), there will be no Changes in the Work unless first authorized in writing by Owner and Architect pursuant to the terms of this Article 7.

- § 7.2.2 Methods used in determining adjustments to the Contract Sum may include those listed in Section 7.3.3.
- § 7.3.1 A Construction Change Directive is a written order prepared by the Architect Owner and signed by the Owner and Architect, directing the form of which is attached hereto as "Exhibit 3," directing a change in the Work prior to agreement on adjustment, if any, in the Contract Sum or Contract Time, or both. The Owner may by Construction Change Directive, without invalidating the Contract, order changes in the Work within the general scope of the Contract consisting of additions, deletions, or other revisions, the Contract Sum and Contract Time being adjusted accordingly.

Additions and Deletions Report for AIA Document A201 – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017>. All rights reserved. "The American Institute of Architects," "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are trademarks of The American Institute of Architects. This draft was produced at 15:38:55 ET on 10/30/2023 under Order No. 2114399858 which expires on 02/01/2024, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail docinfo@aiacontracts.com. **User Notes:**

(1162898282)

- § 7.3.4 If the Contractor does not respond promptly or disagrees with the method for adjustment in the Contract Sum, the Architect shall determine the adjustment on the basis of reasonable expenditures and savings of those performing the Work attributable to the change, including, in case of an increase in the Contract Sum, an amount for overhead and profit as set forth in the Agreement, or if no such amount is set forth in the Agreement, a reasonable amount, amount, all on a not to exceed basis unless otherwise agreed. In such case, and also under Section 7.3.3.3, the Contractor shall keep and present, in such form as the Architect may prescribe, an itemized accounting together with appropriate supporting data. Unless otherwise provided in the Contract Documents, costs for the purposes of this Section 7.3.4 shall be limited to the following:
 - Costs of labor, including applicable payroll taxes, Actual costs of labor, including social security, old .1 age and unemployment insurance, fringe benefits required by agreement or custom, workers' compensation insurance, and other employee costs approved by the Architect; and workers' compensation insurance,;

- Costs of premiums for all bonds and insurance, insurance (itemized separately and subject to the provisions of the Agreement), permit fees, and sales, use, or similar taxes, directly related to the change; and
- Costs Additional costs of supervision and field office personnel directly attributable to the change.

- § 7.3.9 Pending final determination of the total cost of a Construction Change Directive to the Owner, the Contractor may request payment for Work completed under the Construction Change Directive in Applications for Payment. The The Contractor may request payment for changes in the Work not in dispute in Applications for Payment accompanied by a Change Order indicating the parties' agreement with part or all of such costs. For any portion of such cost that remains in dispute, the Architect will make an interim determination for purposes of monthly certification for payment for those costs and certify for payment the amount that the Architect determines, in the Architect's professional judgment, to be reasonably justified. The Architect's interim determination of cost shall adjust the Contract Sum on the same basis as a Change Order, subject to the right of either party to disagree and assert a Claim in accordance with Article 15.
- § 7.3.10 When the Owner and Contractor agree with a determination made by the Architect reach an agreement concerning the adjustments in the Contract Sum and Contract Time, or otherwise reach agreement upon the adjustments, such agreement shall be effective immediately and the Architect will prepare a Change Order. Change Orders may be issued for all or any part of a Construction Change Directive. shall be recorded by preparation and execution of an appropriate Change Order. The Contractor shall be responsible for preparation of the Change Order. The Contractor's agreement with a Change Order constitutes a waiver and release by the Contractor and its Subcontractors and suppliers for any claim for delay, cumulative impact, cost of extended general conditions, and any other indirect cost associated with the Owner-directed change in Work.

PAGE 27

The Architect may have the authority to order minor changes in the Work that are consistent with the intent of the Contract Documents and do not involve an adjustment in the Contract Sum or an extension of the Contract Time. The Architect's order for minor changes shall be in writing. If the Contractor believes that the proposed minor change in the Work will affect the Contract Sum or Contract Time, the Contractor shall notify the Architect and shall not proceed to implement the change in the Work. If the Contractor performs the Work set forth in the Architect's order for a minor change without prior notice to the Architect that such change will affect the Contract Sum or Contract Time, the Contractor waives any adjustment to the Contract Sum or extension of the Contract Time. writing signed by the Architect and shall be binding on the Owner and Contractor. The Contractor shall carry out such written orders promptly.

§ 7.5 FIELD CHANGES

§ 7.5.1 Upon the written request of Contractor, Owner may authorize minor changes in the Work which occur in the field ("Field Change(s)") without prior presentation to the Architect, so long as each of the following criteria is

satisfied	

- .1 the proposed Field Change is consistent with the intent of the Contract Documents;
- .2 the proposed Field Change will not result in an extension of the Contract Time;
- .3 the proposed Field Change will not result in a cumulative increase in the GMP or the Contract Sum by more than \$5,000.00.
- § 7.5.2 By presenting the proposed Field Change, the Contractor is verifying that each of the criteria listed above is satisfied with respect to the proposed Field Change.
- § 7.5.3 If a Field Change may result in a cumulative increase in the Contract Sum of more than \$5,000.00, the Field Change set forth in Section 7.5.1 above does not apply. Rather, the Contractor must follow the written change order provisions set forth in Sections 7.1 and 7.2 above.
- § 7.5.4 The issuance of a Field Change and the provisions of this Section 7.5 shall not limit the Owner's right of review and approval of any Field Change in the progress payment process or as part of final payment.
- § 8.1.2 The date of commencement of the Work is the date established in the Agreement. Notice to Proceed. If there is no Notice to Proceed, it shall be the date of the Owner-Contractor Agreement or such other date as may be established therein.
- § 8.1.3 The date of Substantial Completion of the Work or designated portion thereof is the date certified by the Architect in accordance with Section 9.8.

§ 8.2.1 Time limits stated in the Contract Documents are of the essence of the Contract. By executing the Agreement, the Contractor confirms that the Contract Time is a reasonable period for performing the Work. Work and the dates established for Substantial Completion and Final Completion are reasonable.

§ 8.2.2 The Contractor shall not knowingly, except by agreement or instruction of the Owner in writing, commence the Work prematurely commence operations on the Project site or elsewhere prior to the effective date of insurance required by Article 11 to be furnished by the Contractor and Owner. The date of commencement of the Work shall not be changed by the effective date of such insurance. Unless the date of commencement is established by the Contract Documents or a notice to proceed given by the Owner, the Contractor shall notify the Owner in writing not less than five (5) days or other agreed period before commencing the Work to permit the timely filing of mortgages, mechanic's liens and other security interests.

PAGE 28

§ 8.3.1 If the Contractor is delayed at any time in the commencement or progress of the Work by (1) an act or neglect of negligence of the Owner or Architect, or of an employee of either, or of a Separate Contractor; (2) by changes ordered in the Work; (3) by labor disputes, fire, unusual delay in deliveries, unavoidable easualties, adverse weather conditions documented in accordance with Section 15.1.6.2, casualties or other causes beyond the Contractor's control; (4) by delay authorized by the Owner pending mediation and binding dispute resolution; or (5) by other causes that the Contractor asserts, and the Architect determines, and/or Owner determines may justify delay, then the Contract Time shall be extended by Change Order for such reasonable time as the Architect may determine. Owner may allow.

§ 8.3.3 This Section 8.3 does not preclude recovery of damages for delay by either party under other provisions of the Contract Documents. Contractor's remedy for delays caused by the Owner or Architect shall be limited solely to (a) extra time; and (b) any applicable "general conditions" expenses allowed by Section 9.7 and Section 14.3 of the

Additions and Deletions Report for AIA Document A201 – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017>. All rights reserved. "The American Institute of Architects," "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are trademarks of The American Institute of Architects. This draft was produced at 15:38:55 ET on 10/30/2023 under Order No.2114399858 which expires on 02/01/2024, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail docinfo@aiacontracts.com. **User Notes:** (1162898282)

General Conditions. With respect to any delays caused by Owner or Architect, if Owner contends that acceleration of the Work is necessary, Contractor shall accelerate the Work as directed by Owner. Owner shall pay Contractor only the Contractor's extra labor cost over the amount for regular time during the period of such overtime, including additional insurance and taxes incurred by the Contractor with respect thereto. Time slips covering said overtime must be submitted to Owner's designated representative for checking and approval.

Contractor's sole remedy in the event of (i) any delay not caused by or attributable to the Owner; or (ii) a delay excusable per Section 8.3.1 above; shall be an equitable extension of the Contract Time and Contractor shall not obtain any monetary damages for the delay or any addition to the Contract Sum or Guaranteed Maximum Price; however, Contractor shall be entitled to use Contractor's construction contingency funds (if any) to cover additional costs incurred as a result of any such delay.

Notwithstanding any implication to the contrary in Section 8.3.1 or elsewhere herein, Contractor shall not be entitled to any time extension for any delay that is the fault of Contractor or any of its Subcontractors or suppliers, or their Subcontractors or suppliers at any level. There shall be no adjustment of compensation for acceleration efforts undertaken by Contractor to meet the Substantial Completion date, as the same may be modified as provided for herein, as a result of any delay that is the fault of Contractor or any of its Subcontractors or suppliers, or their Subcontractors or supplier at any level. However, Contractor shall be entitled to use Contractor's construction contingency funds (if any) to cover additional costs incurred as a result of any such delay.

§ 8.3.4 If there is a delay or anticipated delay of the Construction Schedule because of the actions or omissions of the Contractor or any Subcontractor and the Contractor is unable to produce within fifteen (15) working days after notice by the Owner a recovery schedule for its Work to address such delay or anticipated delay acceptable to the Owner, then the Owner shall have the right to order the Contractor to take such actions as may be necessary, consistent with the same performance of the Work affected thereby, to recapture the time lost by any such delay. Such action shall include increasing staff; increase in shifts or hours worked per day, or performance of work on Saturdays, Sundays or national holidays; use of any available work float in the Project schedule; and changing the sequence of work activities. Owner shall pay Contractor only the Contractor's extra labor cost over the amount for regular time during the period of such overtime, including additional insurance and taxes incurred by the Contractor with respect thereto. Time slips covering said overtime must be submitted to Owner's designated representative for checking and approval.

The Owner, at its option, shall also be entitled to accelerate performance of the Work where there is no delay or anticipated delay. The costs of such acceleration where there is no delay or anticipated delay shall be paid for by the Owner.

§ 8.3.5 If Contractor intends to claim an extension of time to perform as a result of a delay not caused by Contractors or any of its Subcontractors or suppliers, Contractor must give Owner written notice within seven (7) days after the event giving rise to the claim and follow the procedures in Article 7 for securing a Change Order; otherwise such a claim shall be deemed waived.

PAGE 29

§ 9.1.2 If unit prices are stated in the Contract Documents or subsequently agreed upon, and if quantities originally contemplated are materially changed so that application of such unit prices to the actual quantities causes substantial inequity to the Owner or Contractor, the applicable unit prices shall be equitably adjusted.

Where the Contract is based on a stipulated sum or Guaranteed Maximum Price, the Contractor shall submit a schedule of values to the Owner and. Architect before the first Application for Payment, allocating the entire Contract Sum to the various portions of the Work. The schedule of values shall be prepared in the form, and supported by the data to substantiate its accuracy, required by the Architect. This schedule, unless objected to by the Architect, shall be used as a basis for reviewing the Contractor's Applications for Payment. Any changes to the schedule of values shall be submitted to the Architect and supported by such data to substantiate its accuracy as the Architect may require, and unless objected to by the Owner or Architect, shall be used as a basis for reviewing the Contractor's subsequent Applications for Payment.

...

User Notes:

§ 9.3.1 At least ten days before the date established for each progress payment, the Contractor shall submit to the Owner and Architect an itemized Application for Payment prepared in accordance with the schedule of values, if required under Section 9.2, for completed portions of the Work. The Such application shall be notarized, if required, and supported by all-such data substantiating the Contractor's right to payment that as the Owner or Architect may require, such as copies of requisitions, and releases and waivers of liens from Subcontractors and material suppliers, and shall reflect retainage if provided for in the Contract Documents.

- § 9.3.1.2 Applications for Payment shall not include requests for payment for portions of the Work for which the Contractor does not intend to pay a Subcontractor or material supplier, unless such Work has been performed by others whom the Contractor intends to pay.
- § 9.3.2 Unless otherwise provided in the Contract Documents, payments shall be made on account of materials and equipment delivered and suitably stored at the site for subsequent incorporation in the Work. If approved in advance by the Owner, payment may similarly be made for materials and equipment suitably stored off the site at a location agreed upon in writing. Payment for materials and equipment stored on or off the site shall be conditioned upon compliance by the Contractor with procedures satisfactory to the Owner to establish the Owner's title to such materials and equipment or otherwise protect the Owner's interest, and shall include the costs of applicable insurance, storage, and transportation to the site, site for such materials and equipment stored off the site.
- § 9.3.3 The Contractor warrants that title to all Work covered by an Application for Payment will pass to the Owner no later than the time of payment. The Contractor further warrants that upon submittal of an Application for Payment all Work for which Certificates for Payment have been previously issued and payments received from the Owner shall, to the best of the Contractor's knowledge, information, and belief, be free and clear of liens, claims, security interests, or encumbrances, in favor of the Contractor, Subcontractors, material suppliers, or other persons or entities that making a claim by reason of having provided labor, materials, and equipment relating to the Work.
- § 9.3.4 With each Application for Payment, the Contractor shall provide to Owner; (a) a partial lien waiver and release from each subcontractor or supplier who was to receive funds from the previous Application for Payment, and (b) the Contractor's own conditional lien waiver for all funds to be received for that Application.
- § 9.3.5 Provided that Owner shall make payment of amounts due to Contractor, Contractor shall keep the Project site free from any liens by Contractor, its Subcontractors, Architect or engineers retained by Contractor, or their respective assigns. No Uniform Commercial Code (UCC) or similar state statutory filings shall be made by Contractor, Subcontractors, or such Architect or engineers, except upon express prior written consent of Owner. If, because of any act or omission of Contractor or anyone claiming through or under Contractor, any mechanic's or other lien, encumbrance or order is filed against Owner or the Project site, Contractor shall, at Contractor's own cost and expense, cause the same to be canceled and discharged of record or bonded over within thirty (30) days after receiving notice thereof, and shall indemnify and hold Owner harmless from and against all costs, expenses, claims, losses or damages, including reasonable attorneys' fees, resulting therefrom or by reason thereof. PAGE 30
- § 9.4.1 The Architect will, within seven days after receipt of the Contractor's Application for Payment, either (1) issue to the Owner a Certificate for Payment in the full amount of the Application for Payment, with a copy to the Contractor; or (2) issue to the Owner a Certificate for Payment Contractor for such amount as the Architect determines is properly due, and or notify the Contractor and Owner in writing of the Architect's reasons for withholding certification in part as provided in Section 9.5.1; or (3) withhold certification of the entire Application for Payment, and notify the Contractor and Owner of the Architect's reason for withholding certification in whole as provided in Section 9.5.1. Section 9.5.1.
- § 9.4.2 The issuance of a Certificate for Payment will constitute a representation by the Architect to the Owner, based on the Architect's evaluation of the Work and the data in-comprising the Application for Payment, that, to the best of the Architect's knowledge, information, and belief, the Work has progressed to the point indicated, and that the quality of the Work is in accordance with the Contract Documents, and that the Contractor is entitled to payment in the amount certified. Documents. The foregoing representations are subject to an evaluation of the Work for conformance with the Contract Documents upon Substantial Completion, to results of subsequent tests and inspections, to

correction of minor deviations from the Contract Documents prior to completion, and to specific qualifications expressed by the Architect. The issuance of a Certificate for Payment will further constitute a representation that the Contractor is entitled to payment in the amount certified. However, the issuance of a Certificate for Payment will not be a representation that the Architect has (1) made exhaustive or continuous on-site inspections to check the quality or quantity of the Work; (2) reviewed construction means, methods, techniques, sequences, or procedures; (3) reviewed copies of requisitions received from Subcontractors and material suppliers and other data requested by the Owner to substantiate the Contractor's right to payment; or (4) made examination to ascertain how or for what purpose the Contractor has used money previously paid on account of the Contract Sum.

§ 9.5.1 The Architect and/or Owner may withhold a Certificate for Payment in whole or in part, to the extent reasonably necessary to protect the Owner, if in the Owner's or Architect's opinion the representations to the Owner required by Section 9.4.2 cannot be made. If the Architect is unable to certify payment in the amount of the Application, the Architect will notify the Contractor and Owner as provided in Section 9.4.1. If the Contractor and Architect cannot agree on a revised amount, the Architect will promptly issue a Certificate for Payment for the amount for which the Architect is able to make such representations to the Owner. The Architect Owner and the Owner shall pay Contractor in accordance with Section 9.6.1. The Architect and/or Owner may also withhold a Certificate for Payment or, because of subsequently discovered evidence, may nullify the whole or a part of a Certificate for Payment previously issued, to such extent as may be necessary in the Owner's or Architect's opinion to protect the Owner from loss for which the Contractor is responsible, including loss resulting from acts and omissions described in Section 3.3.2, because of

- third party claims not resulting from the Owner's failure or refusal to pay Contractor filed or reasonable evidence indicating probable filing of such claims, unless security acceptable to the Owner is provided by the Contractor:
- .3 failure of the Contractor to make payments properly to Subcontractors or suppliers-for labor, materials or equipment;

- .5 damage to the Owner or a Separate Contractor; Contractor and not covered by insurance;
- reasonable evidence that the Work will not be completed within the Contract Time, Time as a result of the Contractor's fault, and that the unpaid balance would not be adequate to cover actual or liquidated damages for the anticipated delay; or
- .7 repeated failure to carry out the Work in accordance with the Contract Documents. Documents;
- failure of the Contractor to comply with the most current approved Project construction schedule; or
- the existence of any event of material default under the Contract Documents
- § 9.5.2 When either party disputes the Architect's decision regarding a Certificate for Payment under Section 9.5.1, in whole or in part, that party may submit a Claim in accordance with Article 15.INTENTIONALLY OMITTED
- § 9.5.3 When the above reasons for withholding certification are removed, certification will be made for amounts previously withheld.
- § 9.5.4 If the Owner or Architect withholds certification for payment under Section 9.5.1.3, the Owner may, at its sole option, issue joint checks to the Contractor and to any Subcontractor or supplier or material or equipment suppliers to whom the Contractor failed to make payment for Work properly performed or material or equipment suitably delivered. If the Owner makes payments by joint check, the Owner shall notify the Architect and the Contractor Architect shall reflect such payment on its next Application the next Certificate for Payment. PAGE 31
- § 9.6.2 The Contractor shall pay each Subcontractor, no later than seven days after receipt of payment from the Owner, the amount to which the Subcontractor is entitled, either by agreement or by reflecting percentages actually retained from payments to the Contractor on account of the Subcontractor's portion of the Work. The Contractor shall, by

20

appropriate agreement with each Subcontractor, require each Subcontractor to make payments to Sub-subcontractors in a similar manner.

- § 9.6.3 The Architect will, on request, furnish to a Subcontractor, if practicable, information regarding percentages of completion or amounts applied for by the Contractor and action taken thereon by the Architect and Owner on account of portions of the Work done by such Subcontractor.INTENTIONALLY OMITTED
- § 9.6.4 The Owner has the right to request written evidence from the Contractor that the Contractor has properly paid Subcontractors and material and equipment suppliers amounts paid by the Owner to the Contractor for subcontracted Work. If the Contractor fails to furnish such evidence within seven days, the Owner shall have the right to contact Subcontractors and material and equipment suppliers to ascertain whether they have been properly paid. Neither the Owner nor Architect shall have an obligation to pay, or to see to the payment of money to, a Subcontractor or material or equipment supplier, except as may otherwise be required by law.
- § 9.6.5 The Contractor's payments to material and equipment suppliers shall be treated in a manner similar to that provided in Sections 9.6.2, 9.6.3 and 9.6.4.

§ 9.6.7 Unless the Contractor provides the Owner with a payment bond in the full penal sum of the Contract Sum, payments received by the Contractor for Work properly performed by Subcontractors or provided by material and equipment suppliers shall be held by the Contractor for those Subcontractors or suppliers who performed Work or furnished materials, or both, under contract with the Contractor for which payment was made by the Owner. Nothing contained herein shall require money to be placed in a separate account and not commingled with money of the Contractor, shall create any fiduciary liability or tort liability on the part of the Contractor for breach of trust, or shall entitle any person or entity to an award of punitive damages against the Contractor for breach of the requirements of this provision.

If the Architect does not issue a Certificate for Payment, through no fault of the Contractor, within seven days after receipt of the Contractor's Application for Payment, or if the Owner does not pay the Contractor within seven days after the date established in the Contract Documents, the amount certified by the Architect or awarded by binding dispute resolution, then the Contractor may, upon seven additional days' written notice to the Owner and Architect, stop the Work until payment of the amount owing has been received. The Contract Time shall be extended appropriately and the Contract Sum shall be increased by the amount of the Contractor's reasonable costs of shutdown, delay and start-up, plus interest as provided for in the Contract Documents.

PAGE 32

- § 9.8.1 Substantial Completion is the stage in the progress of the Work when (i) the Work or designated portion thereof is sufficiently complete in accordance with the Contract Documents so that the Owner can occupy or utilize the Work for its intended use, for its intended use, subject only to the completion of Punch list items or other Work that cannot be completed because of weather conditions or because of unavailability of products or materials, none of which may materially affect Owner's intended use and enjoyment of the Project or cause unreasonable disruption of interference of Owner's intended use or otherwise affect Owner's ability to fully utilize the Project for its intended use; (ii) all required permits and approvals, including a certificate of occupancy, have been issued and obtained, unless any such permit, approval or certificate has not been issued through no fault of the Contractor; and (iii) all utilities required for Owner's operation of the Project are in place and operational, including gas, electricity, water, sanitary, sewer, and telephone; (iv) a Certificate of Substantial Completion is issued and signed-off on by Owner; and (v) Contractor has complied with all terms and conditions of the Agreement.
- § 9.8.2 When the Contractor considers that the Work, or a portion thereof which the Owner agrees to accept separately, is substantially complete, the Contractor shall prepare and submit to the Architect a comprehensive list of items to be completed or corrected prior to final payment. The Contractor shall proceed promptly to complete and correct items on the list. Failure to include an item on such list does not alter the responsibility of the Contractor to complete all Work in accordance with the Contract Documents. Observation by the Architect and/or the Owner during the construction shall not be considered as acceptance of any part of the Work, and the conditions found when final

inspection is made shall be taken as governing the fitness of the Work and whether or not it conforms to the provisions of the Contract.

- § 9.8.3 Upon receipt of the Contractor's list, the Architect will make an inspection to determine whether the Work or designated portion thereof is substantially complete. If the Architect's inspection discloses any item, whether or not included on the Contractor's list, which is list not sufficiently complete in accordance with the Contract Documents so that the Owner can occupy or utilize the Work or designated portion thereof for its intended use, the Contractor shall, before issuance of the Certificate of Substantial Completion, complete or correct such item upon notification by the Architect. In such case, the Contractor shall then submit a request for another inspection by the Architect to determine Substantial Completion.
- § 9.8.4 When the Work or designated portion thereof is substantially complete, the Architect will prepare a Certificate of Substantial Completion that shall establish the date of Substantial Completion; shall establish responsibilities of the Owner and Contractor for security, maintenance, heat, utilities, damage to the Work and insurance; and shall fix the time within which the Contractor shall finish all items on the list accompanying the Certificate. Warranties required by the Contract Documents shall commence on the date of Substantial Completion of the Work or designated portion thereof unless otherwise provided in the Certificate of Substantial Completion.
- § 9.8.5 The Certificate of Substantial Completion shall be submitted to the Owner and Contractor for their written acceptance of responsibilities assigned to them in the such Certificate. Upon such acceptance, and consent of surety if any, the Owner shall make payment of retainage applying to the such Work or designated portion thereof. Such payment shall be adjusted for Work that is incomplete or not in accordance with the requirements of the Contract Documents.
- § 9.8.6 Contractor shall reach Final Completion of the Work within 60 days of the date of Substantial Completion except for those items which cannot be completed because of weather conditions or because of unavailability of products or materials or that Owner agrees do not have to be completed within the 60-day period.

§ 9.9.1 The Owner may occupy or use any completed or partially completed portion of the Work at any stage when such portion is designated by separate agreement with the Contractor, provided such occupancy or use is consented to by the insurer as required under Section 11.2.9 and authorized by public authorities having jurisdiction over the Project. Such partial occupancy or use may commence whether or not the portion is substantially complete, provided the Owner and Contractor have accepted in writing the responsibilities assigned to each of them for payments, retainage, if any, security, maintenance, heat, utilities, damage to the Work and insurance, and have agreed in writing concerning the period for correction of the Work and commencement of warranties required by the Contract Documents. When the Contractor considers a portion substantially complete, the Contractor shall prepare and submit a list to the Architect as provided under Section 9.8.2. Consent of the Contractor to partial occupancy or use shall not be unreasonably withheld. The stage of the progress of the Work shall be determined by written agreement between the Owner and Contractor or, if no agreement is reached, by decision of the Architect.

PAGE 33

§ 9.9.3 Unless otherwise agreed upon, upon in writing and signed by Owner, partial occupancy or use of a portion or portions of the Work shall not constitute acceptance of Work not complying with the requirements of the Contract Documents.

§ 9.10.1 Upon receipt of the Contractor's written notice that the Work is ready for final inspection and acceptance and upon receipt of a final Application for Payment, the Architect will promptly make such inspection. When inspection, and when the Architect finds the Work acceptable under the Contract Documents and the Contract fully performed, the Architect will promptly issue a final Certificate for Payment stating that to the best of the Architect's knowledge, information and belief, and on the basis of the Architect's on-site visits and inspections, the Work has been completed in accordance with terms and conditions of the Contract Documents and that the entire balance found to be due the Contractor and noted in the final Certificate is due and payable. The Architect's final Certificate for Payment will

constitute a further representation that conditions listed in Section 9.10.2 as precedent to the Contractor's being entitled to final payment have been fulfilled.

§ 9.10.2 Neither final payment nor any remaining retained percentage shall become due until the Contractor submits to the Architect (1) an affidavit that payrolls, bills for materials and equipment, and other indebtedness connected with the Work for which the Owner or the Owner's property might be responsible or encumbered (less amounts withheld by Owner) have been paid or otherwise satisfied, (2) a certificate evidencing that insurance required by the Contract Documents to remain in force after final payment is currently in effect, effect and will not be cancelled or allowed to expire until at least 30 days' prior written notice has been given to the Owner, (3) a written statement that the Contractor knows of no substantial reason that the insurance will not be renewable to cover the period required by the Contract Documents, (4) consent of surety, if any, to final payment, (5) documentation of any special warranties, such as manufacturers' warranties or specific Subcontractor warranties, and (6) (6) a marked-up version of all Construction Documents showing the record condition of the Work; (7) a Final Lien Waiver and Release from Contractor and from all subcontractors and suppliers who provided any labor, materials or Work for the Project; and (8) if required by the Owner, other data establishing payment or satisfaction of obligations, such as receipts and releases and waivers of liens, claims, security interests, or encumbrances arising out of the Contract, to the extent and in such form as may be designated by the Owner. If a Subcontractor refuses to furnish a release or waiver required by the Owner, the Contractor may furnish a bond satisfactory to the Owner to indemnify the Owner against such lien, claim, security interest, or encumbrance. If a-such lien, claim, security interest, or encumbrance remains unsatisfied after payments are made, the Contractor shall refund to the Owner all money that the Owner may be compelled to pay in discharging the such lien, claim, security interest, or encumbrance, including all costs and reasonable attorneys' fees, fees, unless Contractor is proceeding diligently and in good faith to contest such lien, claim, security interest, or encumbrance and has furnished to Owner reasonable security, such as a bond, title insurance, or letter of credit to reasonably protect Owner against such lien, claim, security interest, or encumbrance.

§ 9.10.3 If, after Substantial Completion of the Work, final completion thereof is materially delayed through no fault of the Contractor or by issuance of Change Orders affecting final completion, and the Architect so confirms, the Owner shall, upon application by the Contractor and certification by the Architect, and without terminating the Contract, make payment of the balance due for that portion of the Work fully completed, corrected, completed and accepted. If the remaining balance for Work not fully completed or corrected is less than retainage stipulated in the Contract Documents, and if bonds have been furnished, the written consent of the surety to payment of the balance due for that portion of the Work fully completed and accepted shall be submitted by the Contractor to the Architect prior to certification of such payment. Such payment shall be made under terms and conditions governing final payment, except that it shall not constitute a waiver of Claims.

- .2 failure of the Work to comply with the requirements of the Contract Documents; or
- .3 terms of special warranties required by the Contract Documents; or
- audits performed by the Owner, if permitted by the Contract Documents, after final payment. Documents.

§ 9.10.5 Acceptance of final payment by the Contractor, a Subcontractor, or a material or equipment supplier, shall constitute a waiver of claims by that payee except those previously made in writing and identified by that payee as unsettled at the time of final Application for Payment.

PAGE 34

§ 10.2.5 The Contractor shall promptly remedy damage and loss (other than damage or loss insured under property insurance required by the Contract Documents) to property referred to in Sections 10.2.1.2 and 10.2.1.3 eaused in whole or in part to the extent caused by the Contractor, a Sub-subcontractor, or anyone directly or indirectly employed by any of them, or by anyone for whose acts they may be liable and for which the Contractor is responsible under Sections 10.2.1.2 and 10.2.1.3. The Contractor may make a Claim for the cost to remedy the damage or loss to the extent such damage or loss is 10.2.1.3, except damage or loss attributable to acts or omissions of the Owner or Architect or anyone directly or indirectly employed by either of them, or by anyone for whose acts either of them may be liable, and not attributable to the fault or negligence of the Contractor. The foregoing obligations of the Contractor are in addition to the Contractor's obligations under Section 3.18.

Contractor shall promptly report in writing to the Architect and Owner all accidents whatsoever arising out of, or in connection with, the performance of the Work, whether on or off the Site, which caused death, personal injury or property damage, giving full details and statements of witnesses, if Contractor reasonably believes there may be a claim made against Owner. In addition, if death or serious injury or damages are caused, the accident shall be reported immediately to Owner by telephone or messenger. If either party suffers injury or damage to person or property because of an act or omission of the other party, or of others for whose acts such party is legally responsible, written notice of the such injury or damage, whether or not insured, shall be given to the other party within a reasonable time not exceeding 21 days after discovery. The notice shall provide sufficient detail to enable the other party to investigate the matter.

PAGE 35

- § 10.3.1 The Contractor is responsible for compliance with any requirements included in the Contract Documents regarding hazardous materials or substances. If the Contractor encounters a hazardous material or substance not addressed in the Contract Documents and if reasonable precautions will be inadequate to prevent foreseeable bodily injury or death to persons resulting from a material or substance, including but not limited to asbestos or polychlorinated biphenyl (PCB), encountered on the site by the Contractor, the Contractor shall, upon recognizing the condition, immediately stop Work in the affected area and notify report the condition to the Owner and Architect of the condition.in writing.
- § 10.3.2 Upon receipt of the Contractor's written notice, the Owner shall obtain the services of a licensed laboratory to verify the presence or absence of the material or substance reported by the Contractor and, in the event such material or substance is found to be present, to cause it to be rendered harmless. Unless otherwise required by the Contract Documents, the Owner shall furnish in writing to the Contractor and Architect the names and qualifications of persons or entities who are to perform tests verifying the presence or absence of the such material or substance or who are to perform the task of removal or safe containment of the such material or substance. The Contractor and the Architect will promptly reply to the Owner in writing stating whether or not either has reasonable objection to the persons or entities proposed by the Owner. If either the Contractor or Architect has an objection to a person or entity proposed by the Owner, the Owner shall propose another to whom the Contractor and the Architect have no reasonable objection. When the material or substance has been rendered harmless, Work in the affected area shall resume upon written agreement of the Owner and Contractor. By Change Order, the Contract Time shall be extended appropriately and the Contract Sum shall be increased by the amount of the Contractor's reasonable additional costs of shutdown, delay, and start up.A Change Order shall be issued to the Contractor as provided in Article 7, providing for an equitable adjustment in the Contract Sum or Contract Time for an increase or delay in the Work or the cost thereof in the event the Contractor encounters hazardous substances.
- § 10.3.3 To the fullest extent permitted by law, the The Owner shall indemnify and hold harmless the Contractor, Subcontractors, Architect, Architect's consultants, and agents and employees of any of them from and against claims, damages, losses, and expenses, including but not limited to attorneys' fees, arising out of or resulting from performance of the Work in the affected area if in fact the material or substance presents the risk of bodily injury or death as described in Section 10.3.1 and has not been rendered harmless, despite Owner's representations to the contrary, provided that such claim, damage, loss, or expense is attributable to bodily injury, sickness, disease or death, or to injury to or destruction of tangible property (other than the Work itself), except to the extent that such damage, loss, or expense is due to the fault or negligence of the party seeking indemnity.
- § 10.3.4 The Owner shall not be responsible under this Section 10.3 for hazardous materials or substances the Contractor brings to the site unless such materials or substances are required by the Contract Documents. Documents and are used, stored, transported, handled and installed properly by Contractor. The Owner shall be responsible for hazardous materials or substances required by the Contract Documents, except to the extent of the Contractor's fault or negligence in the use and handling use, handling, storage, transportation or installation of such materials or substances.
- § 10.3.5 The Contractor shall reimburse the Owner for the cost and expense the Owner incurs (1) for remediation of hazardous-indemnify, defend and hold the Owner harmless against any and all liabilities, damages, losses, costs, penalties, expenses or responsibilities (1) for remediation of materials or substances the Contractor brings to the site and negligently handles, or (2) where the Contractor fails to perform its obligations under Section 10.3.1, except to the

extent that the cost and expense are due to the Owner's fault or negligence.negligence, and (3) materials and substances which are Contractor's responsibility under Section 10.3.4.

§ 10.3.6 If, without negligence on the part of the Contractor, the Contractor is held liable by a government agency for the cost of remediation of a hazardous material or substance solely by reason of performing Work as required by the Contract Documents, the Owner shall reimburse the Contractor for all cost and expense thereby incurred. Owner assumes the risk of, and shall indemnify, defend and hold Contractor harmless against any and all liabilities, damages, losses, costs, penalties, expenses or responsibilities, including reasonable attorneys' fees and disbursements, sustained in connection with any request or order arising directly or indirectly out of the release, threatened release, or handling of hazardous substances or materials at the Project site or at any site to which a hazardous substance is moved or alleged to have been moved provided that Contractor provides written notice to Owner of any such claim, demand, cause of action, suit, judgment, subpoena or administrative proceeding within 5 days of receipt of same. The duties and responsibilities of the Owner as set forth in this section shall not apply to the extent that the liability, damage, loss, economic loss, cost, penalty, expense or responsibility results from hazardous substances or material brought onto the Project site by Contractor or arises out of negligence of Contractor after the hazardous substance or materials is discovered on the Project site and, if applicable, written notice of the discovery along with documents identifying the materials provided to the Contractor.

PAGE 36

§ 11.1.1 The Contractor shall purchase and maintain insurance of the types and limits of liability, containing the endorsements, and subject to the terms and conditions, as described in the Agreement or elsewhere in the Contract Documents. such insurance as will protect the Contractor from claims set forth below which may arise out of or result from the Contractor's operations and completed operations under the Contract and for which the Contractor may be legally liable, whether such operations be by the Contractor or by a Subcontractor or by anyone directly or indirectly employed by any of them, or by anyone for whose acts any of them may be liable:

- Claims under workers' compensation, disability benefit and other similar employee benefit acts that are applicable to the Work to be performed;
- Claims for damages because of bodily injury, occupational sickness or disease, or death of the Contractor's employees:
- Claims for damages because of bodily injury, sickness or disease, or death of any person other than the .3 Contractor's employees;
- Claims for damages insured by usual personal injury liability coverage;
- Claims for damages, other than to the Work itself, because of injury to or destruction of tangible property, including loss of use resulting therefrom;
- Claims for damages because of bodily injury, death of a person or property damage arising out of ownership, maintenance or use of a motor vehicle;
- Claims for bodily injury or property damage arising out of completed operations; and
- Claims involving contractual liability insurance applicable to the Contractor's obligations under Section 3.18;
- claims arising out of or relating to the use of Contractor's autos, trucks and other vehicles in connection with the Project; and
- claims arising from Contractor's professional services, if any.

The Contractor shall purchase and maintain the required insurance from an insurance company or insurance companies lawfully authorized to issue insurance in the jurisdiction where the Project is located. The Owner, Architect, and Architect's consultants shall be named as additional insureds under the Contractor's commercial general liability policy or as otherwise described in the Contract Documents.companies rated A or better by A.M. Best and lawfully authorized to do business in the jurisdiction in which the Project is located.

§ 11.1.2 The Contractor shall provide surety bonds of the types, for such penal sums, and subject to such terms and conditions as required by the Contract Documents. to furnish bonds covering faithful performance of the Contract and payment of obligations arising thereunder (including the Guaranteed Maximum Price, as may be amended by Change Order(s)). The Contractor shall purchase and maintain the required bonds from a company or companies lawfully authorized to issue surety bonds in the jurisdiction where the Project is located. Upon the request of any person or entity appearing to be a potential beneficiary of bonds covering payment of obligations arising under the Contract, the

25

Contractor shall promptly furnish a copy of the bonds or shall authorize a copy to be furnished. The Contractor may require its Subcontractors to furnish bonds in its discretion.

- § 11.1.3 Upon the request of any person or entity appearing to be a potential beneficiary of bonds covering payment of obligations arising under the Contract, the Contractor shall promptly furnish a copy of the bonds or shall authorize a copy to be furnished. The Owner, its board members, directors, officers, agents (including the Architect) and employees, and PNC Bank National Association, its successors and/or assigns, and Columbia Housing SLP Corporation, its successors and/or assigns, shall be named as additional insureds, on a primary and non-contributory basis, on all insurance policies providing coverage for the claims identified in Sections 11.1.1.3 through 11.1.1.9 above, including the completed operations coverage therein and the endorsements thereto. (See Exhibit to Agreement for the Certificate of Insurance). Any insurance maintained by Owner will be in excess of and not contribute with additional insured coverage provided by Contractor regardless of any "other insurance" clause. "Vertical exhaustion" shall apply when determining responsibility for coverage.
- § 11.1.4 Notice of Cancellation or Expiration of Contractor's Required Insurance. Within three (3) business days of the date the Contractor becomes aware of an impending or actual cancellation or expiration of any insurance required by the Contract Documents, the Contractor shall provide notice to the Owner of such impending or actual cancellation or expiration. Upon receipt of notice from the Contractor, the Owner shall, unless the lapse in coverage arises from an act or omission of the Owner, have the right to stop the Work until the lapse in coverage has been cured by the procurement of replacement coverage by the Contractor. The furnishing of notice by the Contractor shall not relieve the Contractor of any contractual obligation to provide any required coverage. The insurance required by Section 11.1.1 shall be written for not less than limits of liability specified in the Contract Documents or required by law, whichever coverage is greater. Coverages, whether written on an occurrence or claims-made basis, shall be maintained without interruption from the date of commencement of the Work until the date of final payment and termination of any coverage required to be maintained after final payment and, with respect to the Contractor's completed operations coverage, for the period identified in 11.1.8.1 below. If any of the insurance required by Section 11.1.1 is maintained on a claims-made basis, then prior to final payment, Contractor shall purchase tail insurance affording the same coverage and having the same limits, for a period of not less than three years from the date of Final Payment.
- § 11.1.5 Certificates of insurance acceptable to the Owner shall be filed with the Owner prior to commencement of the Work and thereafter upon renewal or replacement of each required policy of insurance.
- § 11.1.6 Notice of Cancellation or Expiration of Contractor's Required Insurance. Within three (3) business days of the date the Contractor becomes aware of an impending or actual cancellation or expiration of any insurance required by the Contract Documents, the Contractor shall provide notice to the Owner of such impending or actual cancellation or expiration. Upon receipt of notice from the Contractor, the Owner shall, unless the lapse in coverage arises from an act or omission of the Owner, have the right to stop the Work until the lapse in coverage has been cured by the procurement of replacement coverage by the Contractor. The furnishing of notice by the Contractor shall not relieve the Contractor of any contractual obligation to provide any required coverage.
- § 11.1.7 Workers' Compensation and Employers' Liability: Contractor agrees that it and all its Subcontractors will comply with all statutes and law with regard to Workers' Compensation/Occupational Disease applying to employees or their beneficiaries. Minimum limits shall be:
 - Workers' Compensation-Statutory Limits

.2 Employers' Liability	Not Less Than:
Bodily Injury by Accident	\$100,000 Each Accident
Bodily Injury by Disease	\$100,000 Each Employee
	\$500,000 Policy Limit

This insurance shall contain provisions waiving each underwriter's rights of subrogation against Owner.

§ 11.1.8 Other insurance required is as follows:

.1 Commercial General Liability

_	Limits not less than:						
_	Bodily Injury and Property	\$1,000,000 Each Occurrence					
	Damage Combined Single Limit:	\$2,000,000 Annual General Aggregate – Per Policy \$2,000,000 Products Liability/Completed Operations					
_	Fire Damage	\$50,000					
	Medical Payment	\$5,000					
Personal Injury and Advertising Injury \$1,000,000							
_	Commercial General Liability coverage must provide: General Aggregate Limit must apply per project Premises and Operations Products Liability/Completed Operations Liability Contractual Liability						
	XCU Coverage (explosion, collapse and underground property damage)						
_		Operations Coverage shall continue in force for two (2) years from tantial Completion or final payment by the Owner.					
<u>.2</u>	Auto						
	Limits not less than:						
_	Bodily injury and Property Damage						
_	Combined Single Limit Policy must be intended to cover all owned, Non-owned and hired vehicles: \$1,000,000 Each Accident						
<u>.3</u>	Umbrella Liability						
_	Not less than:						
_	Bodily Injury and Property Damage	\$10,000,000 Each Occurrence					
_	Combined Single Limit	\$10,000,000 Annual					
.9 Ins	urance required of Subcontractors:						
.1		other Subcontractor who will perform Work on the Project Site paid to the Subcontractor totals \$1,000,000.00 or more of the					
		s as the Contractor for insurance required under and 11.1.8.2 herein; and					
_	(b) Umbrella coverage of \$5	5,000,000.					
<u>.2</u>	Contractor for insurance required	l perform Work on Site shall maintain the same insurance as the under Sections 11.1.7, and 11.1.8, excluding pollution liability, f not less than \$250,000.00 up to the amount of Subcontractor's					

Additions and Deletions Report for AIA Document A201 – 2017. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997, 2007 and 2017>. All rights reserved. "The American Institute of Architects," "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are trademarks of The American Institute of Architects. This draft was produced at 15:38:55 ET on 10/30/2023 under Order No.2114399858 which expires on 02/01/2024, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail docinfo@aiacontracts.com.

User Notes:

contract, whichever is more; and maintain umbrella coverage of not less than \$1,000,000.

All other Subcontractors (not performing Work on site) shall have adequate insurance coverage up

to the minimum amount required by law. If a Subcontractor does not have the insurance required above, then Contractor shall advise Owner and Owner may, in its sole discretion, waive and/or modify the above insurance requirements for Subcontractors.

- § 11.1.10 The Owner and Architect assume no responsibility for Contractor's or Subcontractors' liabilities in the event that the limits set above are not adequate or coverage is not properly obtained by either of them.
- § 11.1.11 Contractor's insurance policies, terms and conditions shall govern unless otherwise modified, in writing, by the parties.
- § 11.1.12 Unless otherwise agreed by Owner, Contractor shall ensure that the insurance required of Subcontractors in Section 11.1.9 be maintained in an insurance company or companies rated A or better by A.M. Best and lawfully authorized to do business in the jurisdiction in which the Project is located.

 PAGE 38
- § 11.2.1 Owner's Liability Insurance. The Owner shall purchase and maintain insurance of the types and limits of liability, containing the endorsements, and subject to the terms and conditions, as described in the Agreement or elsewhere in the Contract Documents. The Owner shall purchase and maintain the required insurance from an insurance company or insurance companies lawfully authorized to issue insurance in the jurisdiction where companies rated A or better by A.M. Best and lawfully authorized to do business in the jurisdiction in which the Project is located.
- § 11.2.2 Failure to Purchase Required Property Insurance. If the Owner fails to purchase and maintain the required property insurance, with all of the coverages and in the amounts described in the Agreement or elsewhere in the Contract Documents, the Owner shall inform the Contractor in writing prior to commencement of the Work. Upon receipt of notice from the Owner, the Contractor may delay commencement of the Work and may obtain insurance that will protect the interests of the Contractor, Subcontractors, and Sub-Subcontractors in the Work. When the failure to provide coverage has been cured or resolved, the Contract Sum and Contract Time shall be equitably adjusted. In the event the Owner fails to procure coverage, the Owner waives all rights against the Contractor, Subcontractors, and Sub-subcontractors to the extent the loss to the Owner would have been covered by the insurance to have been procured by the Owner. The cost of the insurance shall be charged to the Owner by a Change Order. If the Owner does not provide written notice, and the Contractor is damaged by the failure or neglect of the Owner to purchase or maintain the required insurance, the Owner shall reimburse the Contractor for all reasonable costs and damages attributable thereto.
- § 11.2.2 Owner's Property Insurance. The Owner shall purchase and maintain property insurance written on a builder's risk "all-risk" or equivalent policy form in the amount of the initial Contract Sum, plus value of subsequent Contract Modifications and cost of materials supplied or installed by others, comprising total value for the entire Project at the site on a replacement cost basis without optional deductibles. Such property insurance shall be maintained, unless otherwise provided in the Contract Documents or otherwise agreed in writing by all persons and entities who are beneficiaries of such insurance, until final payment has been made as provided in Section 9.10 or until no person or entity other than the Owner has an insurable interest in the property required by this Section 11.2 to be covered, whichever is later. This insurance shall include interests as named insureds, but not as "first-named insureds" of the Owner, the Contractor, Subcontractors and Sub-subcontractors in the Project. Such insurance shall be purchased and maintained from a company or companies lawfully authorized to do business in the jurisdiction in which the Project is located.
- § 11.2.3 Notice of Cancellation or Expiration of Owner's Required Property Insurance. Within three (3) business days of the date the Owner becomes aware of an impending or actual cancellation or expiration of any property insurance required by the Contract Documents, the Owner shall provide notice to the Contractor of such impending or actual cancellation or expiration. Unless the lapse in coverage arises from an act or omission of the Contractor: (1) the Contractor, upon receipt of notice from the Owner, shall have the right to stop the Work until the lapse in coverage has been cured by the procurement of replacement coverage by either the Owner or the Contractor; (2) the Contract Time and Contract Sum shall be equitably adjusted; and (3) the Owner waives all rights against the Contractor, Subcontractors, and Sub-subcontractors to the extent any loss to the Owner would have been covered by the insurance had it not expired or been cancelled. If the Contractor purchases replacement coverage, the cost of the insurance shall be charged to the Owner by an appropriate Change Order. The furnishing of notice by the Owner shall not relieve the Owner of any contractual obligation to provide required insurance. Property insurance shall be on an "all-risk" or

equivalent policy form and shall include, without limitation, insurance against the perils of fire (with extended coverage) and physical loss or damage including, without duplication of coverage, theft, vandalism, malicious mischief, collapse, earthquake, flood, windstorm, falsework, testing and startup, temporary buildings and debris removal including demolition occasioned by enforcement of any applicable legal requirements, and shall cover reasonable compensation for Architect's and Contractor's services and expenses required as a result of such insured

- § 11.2.4 Failure to Purchase Required Property Insurance. If the Owner fails to purchase and maintain the required property insurance, with all of the coverages and in the amounts described in the Agreement or elsewhere in the Contract Documents, the Owner shall inform the Contractor in writing prior to commencement of the Work. Upon receipt of notice from the Owner, the Contractor may delay commencement of the Work and may obtain insurance that will protect the interests of the Contractor, Subcontractors, and Sub-Subcontractors in the Work. The cost of the insurance shall be charged to the Owner by a Change Order. If the Owner does not provide written notice, and the Contractor is damaged by the failure or neglect of the Owner to purchase or maintain the required insurance, the Owner shall reimburse the Contractor for all reasonable costs and damages attributable thereto.
- § 11.2.5 Notice of Cancellation or Expiration of Owner's Required Property Insurance. Within three (3) business days of the date the Owner becomes aware of an impending or actual cancellation or expiration of any property insurance required by the Contract Documents, the Owner shall provide notice to the Contractor of such impending or actual cancellation or expiration. Unless the lapse in coverage arises from an act or omission of the Contractor: (1) the Contractor, upon receipt of notice from the Owner, shall have the right to stop the Work until the lapse in coverage has been cured by the procurement of replacement coverage by either the Owner or the Contractor; and (2) the Contract Time and Contract Sum shall be equitably adjusted. If the Contractor purchases replacement coverage, the cost of the insurance shall be charged to the Owner by an appropriate Change Order. The furnishing of notice by the Owner shall not relieve the Owner of any contractual obligation to provide required insurance.
- § 11.2.6 This property insurance shall cover portions of the Work stored off the site, and also portions of the Work in transit.
- § 11.2.7 Except to the extent covered by insurance, the Owner shall not be liable for loss or damage to any tools owned by mechanics, any other tools, equipment, staging, towers, and forms owned or rented by the Contractor, the capital value of which is not included in the Cost of the Work, or any shanties, houses or other temporary structures.
- § 11.2.8 The Owner shall provide the Contractor with a certificate of insurance for each policy that includes insurance coverages required by this Section 11.2.
- § 11.2.9 Partial occupancy or use in accordance with Section 9.9 shall not commence until the insurance company or companies providing property insurance have consented to such partial occupancy or use by endorsement or otherwise. The Owner and the Contractor shall take reasonable steps to obtain consent of the insurance company or companies and shall, without mutual written consent, take no action with respect to partial occupancy or use that would cause cancellation, lapse, or reduction of insurance.

PAGE 40

The Owner, at the Owner's option, may purchase and maintain insurance that will protect the Owner against loss of use of the Owner's property, or the inability to conduct normal operations, due to fire or other causes of loss. The Owner waives all rights of action against the Contractor and Architect for loss of use of the Owner's property, due to fire or other hazards however caused.

§ 11.5.1 A loss insured under the Owner's property insurance required by the Agreement shall be adjusted by the Owner as fiduciary and made payable to the Owner as fiduciary for the insureds, as their interests may appear, subject to requirements of any applicable mortgagee clause and of Section 11.5.2. The Owner shall pay the Architect and Contractor their just shares of insurance proceeds received by the Owner, and by appropriate agreements the Architect and Contractor shall make payments to their consultants and Subcontractors in similar manner.

§ 12.1.2 If a portion of the Work has been covered that the Architect has not specifically requested to examine prior to its being covered, the Architect may request to see such Work and it shall be uncovered by the Contractor. If such Work is in accordance with the Contract Documents, the Contractor shall be entitled to an equitable adjustment to the Contract Sum and Contract Time as may be appropriate. costs of uncovering and replacement shall, by appropriate Change Order, be at the Owner's expense. If such Work is not in accordance with the Contract Documents, the costs of uncovering the Work, such costs and the cost of correction, shall be at the Contractor's expense, unless the condition was caused by the Owner or a separate contractor in which event the Owner shall be responsible for payment of such costs.

The Contractor shall promptly correct Work rejected by the Architect or failing to conform to the requirements of the Contract Documents, whether discovered before or after Substantial Completion and whether or not fabricated, installed or completed. Costs of correcting such rejected Work, including additional testing and inspections, the cost of uncovering and replacement, and compensation for the Architect's services and expenses made necessary thereby, shall be at the Contractor's expense.

PAGE 41

§ 12.2.2.1 In addition to the Contractor's obligations under Section 3.5, Section 3.5 and the requirements under Clause 23 of the HUD 5370 General Conditions of the Contract for Construction, if, within one year after the date of Substantial Completion of the Work or designated portion thereof or after the date for commencement of warranties established under Section 9.9.1, or by terms of any an applicable special warranty required by the Contract Documents, any of the Work is found to be not in accordance with the requirements of the Contract Documents, the Contractor shall correct it promptly after receipt of written notice from the Owner to do so, unless the Owner has previously given the Contractor a written acceptance of such condition. The Owner shall give such notice promptly after discovery of the condition. During the one-year period for correction of Work, if the Owner fails to notify the Contractor and give the Contractor an opportunity to make the correction, the Owner waives the rights to require correction by the Contractor and to make a claim for breach of warranty. If the Contractor fails to correct nonconforming Work within a reasonable time during that period after receipt of notice from the Owner or Architect, the Owner may correct it in accordance with Section 2.5.

§ 12.2.2.3 The one-year period for correction of Work shall not be extended by corrective Work performed by the Contractor pursuant to this Section 12.2. Section 12.2, except for Work not properly completed in the first instance.

The Contract shall be governed by the law of the place where the Project is located, excluding that jurisdiction's choice of law rules. If the parties have selected arbitration as the method of binding dispute resolution, the Federal Arbitration Act shall govern Section 15.4. State of Wisconsin.

§ 13.2.1 The Owner and Contractor respectively bind themselves, their partners, successors, assigns, and legal representatives to covenants, agreements, and obligations contained in the Contract Documents. Except as provided in Section 13.2.2, neither party to the Contract shall assign the Contract as a whole without written consent of the other. If either party attempts to make such an assignment without such consent, that party shall nevertheless remain legally responsible for all obligations under the Contract.

§ 13.2.2 The Owner may, without consent of the Contractor, assign the Contract to a lender providing construction financing for the Project, if the lender assumes the Owner's rights and obligations under the Contract Documents. The Contractor shall execute all consents reasonably required to facilitate the such assignment.

PAGE 42

§ 13.3.2 No action or failure to act by the Owner, Architect, or Contractor shall constitute a waiver of a right or duty afforded them under the Contract, nor shall such action or failure to act constitute approval of or acquiescence in a breach thereunder, except as may be specifically agreed upon in writing in writing or as set forth herein.

...

§ 13.4.1 Tests, inspections, and approvals of portions of the Work shall be made as required by the Contract Documents and by applicable laws, statutes, ordinances, codes, rules, and regulations or lawful orders of public authorities. Unless otherwise provided, the Contractor shall make arrangements for such tests, inspections, and approvals with an independent testing laboratory or entity acceptable to the Owner, or with the appropriate public authority, and shall bear all related costs of tests, inspections, and approvals. The Contractor shall give the Architect timely notice of when and where tests and inspections are to be made so that the Architect may be present for such procedures. The Owner shall bear costs of tests, inspections, or approvals that do not become requirements until after bids are received or negotiations concluded. The Owner shall directly arrange and pay for tests, inspections, or approvals where building codes or applicable laws or regulations so require.

...

§ 13.4.3 If procedures for testing, inspection, or approval under Sections 13.4.1 and 13.4.2 reveal failure of the portions of the Work to comply with requirements established by the Contract Documents, all costs made necessary by such failure, including those of repeated <u>testing</u>, <u>inspection</u>, <u>or approval</u> procedures and compensation for the Architect's services and expenses, shall be at the Contractor's expense.

...

Payments due and unpaid under the Contract Documents shall bear interest from the date payment is due at the rate such rate as the parties <u>may</u> agree upon in writing or, in the absence thereof, at the legal rate prevailing from time to time at the place where the Project is located.

...

§ 14.1.1 The Contractor may terminate the Contract if the Work is stopped for a period of 30-60 consecutive days through no act or fault of the Contractor, a Subcontractor, a Sub-subcontractor, their agents or employees, or any other persons or entities performing portions of the Work, Work under direct or indirect contract with the Contractor, for any of the following reasons:

...

Because the Architect has not issued a Certificate for Payment and has not notified the Contractor of the reason for withholding certification as provided in Section 9.4.1, or because the Owner has not made payment on a Certificate for Payment within the time stated in the Contract Documents; Documents and none of the circumstances prescribed in Sections 9.5.1.1 through 9.5.1.9 exist; or

PAGE 43

- § 14.1.2 The Contractor may terminate the Contract if, through no act or fault of the Contractor, a Subcontractor, a Sub-subcontractor, their agents or employees, or any other persons or entities performing portions of the Work, Work under direct or indirect contract with the Contractor, repeated suspensions, delays, or interruptions of the entire Work by the Owner as described in Section 14.3, constitute in the aggregate more than 100 percent of the total number of days scheduled for completion, or 120 days in any 365-day period, whichever is less.
- § 14.1.3 If one of the reasons described in Section 14.1.1 or 14.1.2 exists, the Contractor may, upon seven days' written notice to the Owner and Architect, terminate the Contract and recover from the Owner payment for Work executed, as well as reasonable overhead and profit on Work not executed, and including reasonable overhead and profit, actual and direct costs incurred by reason of such termination. termination, and damages.
- § 14.1.4 If the Work is stopped for a period of 60 consecutive days through no act or fault of the Contractor, a Sub-subcontractor, or their agents or employees or any other persons or entities performing portions

of the Work under contract with the Contractor because the Owner has repeatedly failed to fulfill the Owner's obligations under the Contract Documents with respect to matters important to the progress of the Work, the Contractor may, upon seven additional days' written notice to the Owner and the Architect, terminate the Contract and recover from the Owner as provided in Section 14.1.3.

- repeatedly refuses or fails to supply enough properly skilled workers or proper materials; materials in accordance with Contractor's schedule for attaining Substantial Completion;
- .2 fails to make payment to Subcontractors or suppliers-for materials or labor in accordance with the respective agreements between the Contractor and the Subcontractors or suppliers; if and only if Contractor fails to provide Owner with reasonable evidence that funds have been placed by Contractor in a commercial escrow account sufficient to pay Subcontractor for the disputed items, fails to post a bond for the amount due Subcontractors, fails to obtain an endorsement to a title policy with respect to liens of Subcontractor, or if the portion of the Contract Sum owing to the Contractor held in retainage by Owner pursuant to Article 9 is less than the amount allegedly owed by Contractor to Subcontractor;
- repeatedly disregards applicable laws, statutes, ordinances, codes, rules and regulations, or lawful .3 orders of a public authority; or
- .4 otherwise is guilty of substantial breach of a provision of the Contract Documents; or
- .5 becomes insolvent, is in bankruptcy, or is in receivership.

§ 14.2.2 When any of the reasons described in Section 14.2.1 exist, and above reasons exist, the Owner, upon certification by the Architect that sufficient cause exists to justify such action, the Owner-may, without prejudice to any other rights or remedies of the Owner and after giving the Contractor and the Contractor's surety, if any, seven days' notice, written notice and provided the Contractor, within such seven (7) day period, has not cured such cause or breach, terminate employment of the Contractor and may, subject to any prior rights of the surety:

- § 14.2.4 If the unpaid balance of the Contract Sum exceeds costs of finishing the Work, including compensation for the Architect's services and expenses made necessary thereby, and other damages incurred by the Owner and not expressly waived, such excess shall be paid to the Contractor. If such costs and damages exceed the unpaid balance, the Contractor shall pay the difference to the Owner. The amount to be paid to the Contractor or Owner, as the case may be, shall be certified by the Initial Decision Maker, Architect, upon application, and this obligation for payment shall survive termination of the Contract.
- § 14.2.5 Upon termination of the Contract, Contractor shall immediately stop the Work hereunder except such as shall be essential to preserve and safeguard existing Work and immediately advise Owner of the status of all outstanding subcontracts and purchase orders. Such outstanding purchase orders and subcontracts shall be canceled or assigned to Owner as Owner may direct. In the event of any such termination, Contractor shall deliver all of the Work completed and in process to the date thereof to Owner and shall deliver to Owner all documents of title, reports, estimates, schedules, and other documents and data as Owner shall require.

PAGE 44

§ 14.3.2 The Contract Sum and Contract Time shall be adjusted for increases in the cost and time caused by suspension, delay, or interruption under as described in Section 14.3.1. Adjustment of the Contract Sum shall include profit. No adjustment shall be made to the extent

§ 14.4.2 Upon receipt of written notice from the Owner of such termination for the Owner's convenience, the Contractor shall

§ 14.4.3 In case of such termination for the Owner's convenience, the Owner shall pay the Contractor for Work properly executed; costs incurred by reason of the termination, including costs attributable to termination of

Subcontracts; and the termination fee, if any, set forth in the Agreement. Contractor shall be entitled to receive payment for Work properly executed (including Contractor's fee for the Work completed), and costs incurred by reason of such termination.

A Claim is a demand or assertion by one of the parties seeking, as a matter of right, payment of money, a change in the Contract Time, adjustment or interpretation of the terms of this Agreement, or other relief with respect to the terms of the Contract. The term "Claim" also includes other disputes and matters in question between the Owner and Contractor arising out of or relating to the Contract. The responsibility to substantiate Claims shall rest with the party making the Claim. This Section 15.1.1 does not require the Owner to file a Claim in order to impose liquidated damages in accordance with the Contract Documents.

The Owner and Contractor shall commence all Claims and causes of action against the other and arising out of or related to the Contract, whether in contract, tort, breach of warranty or otherwise, in accordance with the requirements of the binding dispute resolution method selected in the Agreement and within the period specified by applicable law, but in any case not more than 10 years after the date of Substantial Completion of the Work. The Owner and Contractor waive all Claims and causes of action not commenced in accordance with this Section 15.1.2. INTENTIONALLY OMITTED

§ 15.1.3.1 Claims by either the Owner or Contractor, where the condition giving rise to the Claim is first discovered prior to expiration of the period for correction of the Work set forth in Section 12.2.2, shall be initiated by Contractor must be initiated by written notice to the other party and to the Initial Decision Maker with a copy sent to the Architect, if the Architect is not serving as the Initial Decision Maker. Claims by either party under this Section 15.1.3.1 shall must be initiated within 21 days after occurrence of the event giving rise to such Claim or within 21 days after the claimant first recognizes the condition giving rise to the Claim, whichever is later.

§ 15.1.3.2 Claims by either the Owner or Contractor, where the condition giving rise to the Claim is first discovered after expiration of the period for correction of the Work set forth in Section 12.2.2, shall be initiated by notice to the other party. In such event, no decision by the Initial Decision Maker is required.

PAGE 45

§ 15.1.4.2 The Contract Sum and Contract Time shall be adjusted in accordance with the Initial Decision Maker's decision, subject to the right of either party to proceed in accordance with this Article 15. The Architect will issue Certificates for Payment in accordance with the decision of the Initial Decision Maker.

HSubject to all other applicable provisions of the Contract Documents, if the Contractor wishes to make a Claim for an increase in the Contract Sum, notice as provided in Section 15.1.3 herein shall be given before proceeding to execute the portion of the Work that is the subject of the Claim. Prior notice is not required for Claims relating to an emergency endangering life or property arising under Section 10.4.

§ 15.1.6.1 If the Contractor wishes to make a Claim for an increase in the Contract Time, written notice as provided in Section 15.1.3-herein shall be given. The Contractor's Claim shall include an estimate of cost and of probable effect of delay on progress of the Work. In the case of a continuing delay, only one Claim is necessary.

§ 15.1.6.2 If adverse weather conditions are the basis for a Claim for additional time, such Claim shall be documented by data substantiating that weather conditions were abnormal for the period of time, time for the geographical region where the Project is located, could not have been reasonably anticipated, and had an adverse effect on the scheduled construction.

...

The Except to the extent that it is a covered loss under insurance required for this Project, the Contractor and Owner waive Claims against each other for consequential damages arising out of or relating to this Contract. This mutual waiver includes

..

User Notes:

§ 15.2 Initial Decision Resolution of Claims and Disputes

§ 15.2.1 Claims, excluding those where the condition giving rise to the Claim is first discovered after expiration of the period for correction of the Work set forth in Section 12.2.2 or arising under Sections 10.3, 10.4, and 11.5, shall be referred to the Initial Decision Maker for initial decision. The Architect will serve as the Initial Decision Maker, unless otherwise indicated in the Agreement. Except for those Claims excluded by this Section 15.2.1, an initial decision shall be required as a condition precedent to mediation of any Claim. If an initial decision has not been rendered within 30 days after the Claim has been referred to the Initial Decision Maker, the party asserting the Claim may demand mediation and binding dispute resolution without a decision having been rendered. Unless the Initial Decision Maker and all affected parties agree, the Initial Decision Maker will not decide disputes between the Contractor and persons or entities other than the Owner. The Owner and Contractor shall endeavor to resolve Claims through good faith direct discussions between the parties' representatives. If, at any time, the Owner's and Contractor's representatives are unable to resolve their differences, a senior executive from either party may refer such dispute to a senior executive (of the requesting party's choice) of the other party. Within 21 calendar days of such referral, the executives shall meet to resolve the dispute.

§ 15.2.2 The Initial Decision Maker will review Claims and within ten days of the receipt of a Claim take one or more of the following actions: (1) request additional supporting data from the claimant or a response with supporting data from the other party, (2) reject the Claim in whole or in part, (3) approve the Claim, (4) suggest a compromise, or (5) advise the parties that the Initial Decision Maker is unable to resolve the Claim if the Initial Decision Maker lacks sufficient information to evaluate the merits of the Claim or if the Initial Decision Maker concludes that, in the Initial Decision Maker's sole discretion, it would be inappropriate for the Initial Decision Maker to resolve the Claim. In the event of a Claim against the Contractor, the Owner may, but is not obligated to, notify the surety, if any, of the nature and amount of the Claim. If the Claim relates to a possibility of a Contractor's default, the Owner may, but is not obligated to, notify the surety and request the surety's assistance in resolving the controversy.

§ 15.2.3 In evaluating Claims, the Initial Decision Maker may, but shall not be obligated to, consult with or seek information from either party or from persons with special knowledge or expertise who may assist the Initial Decision Maker in rendering a decision. The Initial Decision Maker may request the Owner to authorize retention of such persons at the Owner's expense. If a Claim relates to or is the subject of a mechanic's lien, the party asserting such Claim may proceed in accordance with applicable law to comply with the lien notice or filing deadlines.

§ 15.2.4 If the Initial Decision Maker requests a party to provide a response to a Claim or to furnish additional supporting data, such party shall respond, within ten days after receipt of the request, and shall either (1) provide a response on the requested supporting data, (2) advise the Initial Decision Maker when the response or supporting data will be furnished, or (3) advise the Initial Decision Maker that no supporting data will be furnished. Upon receipt of the response or supporting data, if any, the Initial Decision Maker will either reject or approve the Claim in whole or in part.

§ 15.2.5 The Initial Decision Maker will render an initial decision approving or rejecting the Claim, or indicating that the Initial Decision Maker is unable to resolve the Claim. This initial decision shall (1) be in writing; (2) state the reasons therefor; and (3) notify the parties and the Architect, if the Architect is not serving as the Initial Decision Maker, of any change in the Contract Sum or Contract Time or both. The initial decision shall be final and binding on the parties but subject to mediation and, if the parties fail to resolve their dispute through mediation, to binding dispute resolution.

§ 15.2.6 Either party may file for mediation of an initial decision at any time, subject to the terms of Section 15.2.6.1.

(1162898282)

- § 15.2.6.1 Either party may, within 30 days from the date of receipt of an initial decision, demand in writing that the other party file for mediation. If such a demand is made and the party receiving the demand fails to file for mediation within 30 days after receipt thereof, then both parties waive their rights to mediate or pursue binding dispute resolution proceedings with respect to the initial decision.
- § 15.2.7 In the event of a Claim against the Contractor, the Owner may, but is not obligated to, notify the surety, if any, of the nature and amount of the Claim. If the Claim relates to a possibility of a Contractor's default, the Owner may, but is not obligated to, notify the surety and request the surety's assistance in resolving the controversy.
- § 15.2.8 If a Claim relates to or is the subject of a mechanic's lien, the party asserting such Claim may proceed in accordance with applicable law to comply with the lien notice or filing deadlines.

§ 15.3 Mediation Mediation/Litigation

- § 15.3.1 Claims, disputes, or other matters in controversy unresolved per Section 15.2 and arising out of or related to the Contract, except those waived as provided for in Sections 9.10.4, 9.10.5, and 15.1.7, shall be subject to mediation as a condition precedent to binding dispute resolution.
- § 15.3.2 The parties shall endeavor to resolve their Claims <u>unresolved per Section 15.2</u> by mediation which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Mediation Procedures in effect on the date of the Agreement. A request for mediation shall be made in writing, delivered to the other party to the Contract, and filed with the person or entity administering the mediation. The request may be made concurrently with the filing of binding dispute resolution proceedings but, in such event, mediation shall proceed in advance of binding dispute resolution proceedings, which shall be stayed pending mediation for a period of 60 days from the date of filing, unless stayed for a longer period by agreement of the parties or court order. If an arbitration is stayed pursuant to this Section 15.3.2, the parties may nonetheless proceed to the selection of the arbitrator(s) and agree upon a schedule for later proceedings.
- § 15.3.3 Either party may, within 30 days from the date that mediation has been concluded without resolution of the dispute or 60 days after mediation has been demanded without resolution of the dispute, demand in writing that the other party file for binding dispute resolution. If such a demand is made and the party receiving the demand fails to file for binding dispute resolution within 60 days after receipt thereof, then both parties waive their rights to binding dispute resolution proceedings with respect to the initial decision. INTENTIONALLY OMITTED PAGE 46

§ 15.4 Arbitration

- § 15.4.1 If the parties have selected arbitration as the method for binding dispute resolution in the Agreement, any Claim subject to, but not resolved by, mediation shall be subject to arbitration which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Arbitration Rules in effect on the date of the Agreement. The Arbitration shall be conducted in the place where the Project is located, unless another location is mutually agreed upon. A demand for arbitration shall be made in writing, delivered to the other party to the Contract, and filed with the person or entity administering the arbitration. The party filing a notice of demand for arbitration must assert in the demand all Claims then known to that party on which arbitration is permitted to be demanded.
- § 15.4.1.1 A demand for arbitration shall be made no earlier than concurrently with the filing of a request for mediation, but in no event shall it be made after the date when the institution of legal or equitable proceedings based on the Claim would be barred by the applicable statute of limitations. For statute of limitations purposes, receipt of a written demand for arbitration by the person or entity administering the arbitration shall constitute the institution of legal or equitable proceedings based on the Claim.
- § 15.4.2 The award rendered by the arbitrator or arbitrators shall be final, and judgment may be entered upon it in accordance with applicable law in any court having jurisdiction thereof.
- § 15.4.3 The foregoing agreement to arbitrate and other agreements to arbitrate with an additional person or entity duly consented to by parties to the Agreement, shall be specifically enforceable under applicable law in any court having jurisdiction thereof.

§ 15.4.4 Consolidation or Joinder

§ 15.4.4.1 Subject to the rules of the American Arbitration Association or other applicable arbitration rules, either party may consolidate an arbitration conducted under this Agreement with any other arbitration to which it is a party provided that (1) the arbitration agreement governing the other arbitration permits consolidation, (2) the arbitrations to be consolidated substantially involve common questions of law or fact, and (3) the arbitrations employ materially similar procedural rules and methods for selecting arbitrator(s).

§ 15.4.4.2 Subject to the rules of the American Arbitration Association or other applicable arbitration rules, either party may include by joinder persons or entities substantially involved in a common question of law or fact whose presence is required if complete relief is to be accorded in arbitration, provided that the party sought to be joined consents in writing to such joinder. Consent to arbitration involving an additional person or entity shall not constitute consent to arbitration of any claim, dispute or other matter in question not described in the written consent.

§ 15.3.5 If the parties do not resolve a dispute through mediation pursuant to this Section 15.3, the method of binding dispute resolution shall be the following litigation in a court of competent jurisdiction located in Milwaukee, Wisconsin. The parties hereby agree and consent to personal jurisdiction in state or federal court located in Milwaukee, Wisconsin.

§ 15.3.6 In the event of litigation arising out of this Agreement, the prevailing party's attorneys' fees and costs shall be paid by the non-prevailing party. The "prevailing party" means the party in whose favor a judgment, decree, or final order is rendered.

§ 15.4.4.3 The Owner and Contractor grant to any person or entity made a party to an arbitration conducted under this Section 15.4, whether by joinder or consolidation, the same rights of joinder and consolidation as those of the Owner and Contractor under this Agreement.

36

Certification of Document's Authenticity

AIA® Document D401 ™ - 2003

I, , hereby certify, to the best of my knowledge, information and belief, the simultaneously with its associated Additions and Deletions Report and this cunder Order No. 2114399858 from AIA Contract Documents software and document I made no changes to the original text of AIA® Document A201 ^{TI} Contract for Construction, other than those additions and deletions shown in Report.	ertification at 15:38:55 ET on $10/30/2023$ that in preparing the attached final $^{M}-2017$, General Conditions of the
(Signed)	-
(Title)	-
(Dated)	

General Conditions for Construction Contracts - Public Housing Programs

U.S. Department of Housing and Urban Development

Office of Public and Indian Housing OMB Approval No. 2577-0157 (exp. 11/30/2023)

Applicability. This form is applicable to any construction/development contract greater than \$250,000.

Public reporting burden for this collection of information is estimated to average 1.0 hours per response, including the time for reviewing instructions, searching existing data sources, gathering and maintaining the data needed, and completing and reviewing the collection of information. Comments regarding the accuracy of this burden estimate and any suggestions for reducing this burden can be sent to the Reports Management Officer, Office of Policy Development and Research, REE, Department of Housing and Urban Development, 451 7th St SW, Room 4176, Washington, DC 20410-5000. When providing comments, please refer to OMB Approval No. 2577-0157. This form includes those clauses required by OMB's common rule on grantee procurement, implemented at HUD in 2 CFR 200, and those requirements set forth in Section 3 of the Housing and Urban Development Act of 1968 and its amendment by the Housing and Community Development Act of 1992, implemented by HUD at 24 CFR Part 75. The form is required for construction contracts awarded by Public Housing Agencies (PHAs). The form is used by Housing Authorities in solicitations to provide necessary contract clauses. If the form were not used, PHAs would be unable to enforce their contracts. Responses to the collection of information are required to obtain a benefit or to retain a benefit. The information requested does not lend itself to confidentiality. HUD may not conduct or sponsor, and a person is not required to respond to a collection of information unless it displays a currently valid OMB number.

	lT .				
	Clause	Page		Clause	Page
1.	Definitions	2		Administrative Requirements	
2.	Contractor's Responsibility for Work	2	25.	Contract Period	9
3.	Architect's Duties, Responsibilities and Authority	2	26.	Order of Precedence	9
4.	Other Contracts	3	27.	Payments	9
	Construction Requirements		28.	Contract Modifications	10
5.	Preconstruction Conference and Notice to Proceed	3	29.	Changes	10
6.	Construction Progress Schedule	3	30.	Suspension of Work	11
7.	Site Investigation and Conditions Affecting the Work	3	31.	Disputes	11
8.	Differing Site Conditions	4	32.	Default	11
9.	Specifications and Drawings for Construction	4	33.	Liquidated	12
10.	As-Built Drawings	5	34.	Termination of Convenience	12
11.	Material and Workmanship	5	35.	Assignment of Contract	12
12.	Permits and Codes	5	36.	Insurance	12
13.	Health, Safety, and Accident Prevention	6	37.	Subcontracts	13
14.	Temporary Buildings and Transportation Materials	6	38.	Subcontracting with Small and Minority Firms, Women's Business Enterprise, and Labor Surplus Area Firms	13
15.	Availability and Use of Utility Services	6	39.	Equal Employment Opportunity	13
16.	Protection of Existing Vegetation, Structures, Equipment, Utilities, and Improvements	6	40.	Employment, Training, and Contracting Opportunities for Low-Income Persons, Section 3 of the Housing and Urban Development Act of 1968	14
17.	Temporary Buildings and Transportation Materials	7	41.	Interest of Members of Congress	15
18.	Clean Air and Water	7	42.	Interest of Members, Officers, or Employees and Former Members, Officers, or Employees	15
19.	Energy Efficiency	7	43.	Limitations on Payments Made to Influence	15
20.	Inspection and Acceptance of Construction	7	44.	Royalties and Patents	15
		8	45.	Examination and Retention of Contractor's Records	15
	Warranty of Title	8	46.	Labor Standards-Davis-Bacon and Related Acts	15
23.	Warranty of	8	47.	Non-Federal Prevailing Wage Rates	19
24.	Prohibition Against	9	48.	Procurement of Recovered	19

Liens Materials

1. Definitions

- (a) "Architect" means the person or other entity engaged by the PHA to perform architectural, engineering, design, and other services related to the work as provided for in the contract. When a PHA uses an engineer to act in this capacity, the terms "architect" and "engineer" shall be synonymous. The Architect shall serve as a technical representative of the Contracting Officer. The Architect's authority is as set forth elsewhere in this contract.
- (b) "Contract" means the contract entered into between the PHA and the Contractor. It includes the forms of Bid, the Bid Bond, the Performance and Payment Bond or Bonds or other assurance of completion, the Certifications, Representations, and Other Statements of Bidders (form HUD-5370), these General Conditions of the Contract for Construction (form HUD-5370), the applicable wage rate determinations from the U.S. Department of Labor, any special conditions included elsewhere in the contract, the specifications, and drawings. It includes all formal changes to any of those documents by addendum, change order, or other modification.
- (c) "Contracting Officer" means the person delegated the authority by the PHA to enter into, administer, and/or terminate this contract and designated as such in writing to the Contractor. The term includes any successor Contracting Officer and any duly authorized representative of the Contracting Officer also designated in writing. The Contracting Officer shall be deemed the authorized agent of the PHA in all dealings with the Contractor.
- (d) "Contractor" means the person or other entity entering into the contract with the PHA to perform all of the work required under the contract.
- (e) "Drawings" means the drawings enumerated in the schedule of drawings contained in the Specifications and as described in the contract clause entitled Specifications and Drawings for Construction herein.
- (f) "HUD" means the United States of America acting through the Department of Housing and Urban Development including the Secretary, or any other person designated to act on its behalf. HUD has agreed, subject to the provision Annual Contributions Terms and Conditions (ACC), to provide financial assistance to the PHA, which includes assistance in financing the work to be performed under this contract. As defined elsewhere in these General Conditions or the contract documents, the determination of HUD may be required to authorize changes in the work or for release of funds to the PHA for payment to the Contractor. Notwithstanding HUD's role, nothing in this contract shall be construed to create any contractual relationship between the Contractor and HUD.
- (g) "Project" means the entire project, whether construction or rehabilitation, the work for which is provided for in whole or in part under this contract.
- (h) "PHA" means the Public Housing Agency organized under applicable state laws which is a party to this contract.
- (j) "Specifications" means the written description of the technical requirements for construction and includes the criteria and tests for determining whether the requirements are met.
- (I) "Work" means materials, workmanship, and manufacture and fabrication of components.

2. Contractor's Responsibility for Work

- (a) The Contractor shall furnish all necessary labor, materials, tools, equipment, and transportation necessary for performance of the work. The Contractor shall also furnish all necessary water, heat, light, and power not made available to the Contractor by the PHA pursuant to the clause entitled Availability and Use of Utility Services herein
- (b) The Contractor shall perform on the site, and with its own organization, work equivalent to at least [] (12 percent unless otherwise indicated) of the total amount of work to be performed under the order. This percentage may be reduced by a supplemental agreement to this order if, during performing the work, the Contractor requests a reduction and the Contracting Officer determines that the reduction would be to the advantage of the PHA.
- (c) At all times during performance of this contract and until the work is completed and accepted, the Contractor shall directly superintend the work or assign and have on the work site a competent superintendent who is satisfactory to the Contracting Officer and has authority to act for the Contractor.
- (d) The Contractor shall be responsible for all damages to persons or property that occur as a result of the Contractor's fault or negligence, and shall take proper safety and health precautions to protect the work, the workers, the public, and the property of others. The Contractor shall hold and save the PHA, its officers and agents, free and harmless from liability of any nature occasioned by the Contractor's performance. The Contractor shall also be responsible for all materials delivered and work performed until completion and acceptance of the entire work, except for any completed unit of work which may have been accepted under the contract.
- (e) The Contractor shall lay out the work from base lines and bench marks indicated on the drawings and be responsible for all lines, levels, and measurements of all work executed under the contract. The Contractor shall verify the figures before laying out the work and will be held responsible for any error resulting from its failure to do so.
- act on its behalf. HUD has agreed, subject to the provisions of an (f) The Contractor shall confine all operations (including Annual Contributions Terms and Conditions (ACC), to storage of materials) on PHA premises to areas provide financial assistance to the PHA, which includes authorized or approved by the Contracting Officer.
 - (g) The Contractor shall at all times keep the work area, including storage areas, free from accumulations of waste materials. After completing the work and before final inspection, the Contractor shall (1) remove from the premises all scaffolding, equipment, tools, and materials (including rejected materials) that are not the property of the PHA and all rubbish caused by its work; (2) leave the work area in a clean, neat, and orderly condition satisfactory to the Contracting Officer; (3) perform all specified tests; and, (4) deliver the installation in complete and operating condition.
 - (h) The Contractor's responsibility will terminate when all work has been completed, the final inspection made, and the work accepted by the Contracting Officer. The Contractor will then be released from further obligation except as required by the warranties specified elsewhere in the contract.

3. Architect's Duties, Responsibilities, and Authority

(a) The Architect for this contract, and any successor, shall be designated in writing by the Contracting Officer.

- (b) The Architect shall serve as the Contracting Officer's technical representative with respect to architectural, Schedule engineering, and design matters related to the work performed under the contract. The Architect may provide direction on contract performance. Such direction shall be within the scope of the contract and may not be of a nature which: (1) institutes additional work outside the scope of the contract; (2) constitutes a change as defined in the Changes clause herein; (3) causes an increase or decrease in the cost of the contract; (4) alters the Construction Progress Schedule; or (5) changes any of the other express terms and represcribilities may include but
- (c) The Architect's duties and responsibilities may include but shall not be limited to:
- (1) Making periodic visits to the work site, and on the basis of his/her on-site inspections, issuing written reports to the PHA which shall include all observed deficiencies. The Architect shall file a copy of the report with the Contractor's designated representative at the site:
- (2) Making modifications in drawings and technical specifications and assisting the Contracting Officer in the preparation of change orders and other contract modifications for issuance by the Contracting Officer;
- (3) Reviewing and making recommendations with respect to - (i) the Contractor's construction progress schedules; (ii) the Contractor's shop and detailed drawings; (iii) the machinery, mechanical and other equipment and materials or other articles proposed for use by the Contractor; and, (iv) the Contractor's price breakdown and progress payment estimates; and.
- (4) Assisting in inspections, signing Certificates of Completion, and making recommendations with respect to acceptance of work completed under the contract.

4. Other Contracts

The PHA may undertake or award other contracts for additional work at or near the site of the work under this contract. The Contractor shall fully cooperate with the other contractors and with PHA employees and shall carefully adapt scheduling and performing the work under this contract to accommodate the additional work, heeding any direction that may be provided by the Contracting Officer. The Contractor shall not commit or permit any act that will interfere with the performance of work by any other contractor or by PHA employees

Construction Requirements

5. Pre-construction Conference and Notice to Proceed of the work, and that it has investigated and satisfied itself

- (a) Within ten calendar days of contract execution, and prior to the commencement of work, the Contractor shall attend a preconstruction conference with representatives of the PHA, its Architect, and other interested parties convened by the PHA. The conference will serve to acquaint the participants with the general plan of the construction operation and all other requirements of the contract. The PHA will provide the Contractor with the date, time, and place of the conference.
- (b) The contractor shall begin work upon receipt of a written Notice to Proceed from the Contracting Officer or designee. The Contractor shall not begin work prior to receiving such notice.

6. Construction Progress

- (a) The Contractor shall, within five days after the work commences on the contract or another period of time determined by the Contracting Officer, prepare and submit to the Contracting Officer for approval three copies of a practicable schedule showing the order in which the Contractor proposes to perform the work, and the dates on which the Contractor contemplates starting and completing the several salient features of the work (including acquiring labor, materials, and equipment). The schedule shall be in the form of a progress chart of suitable scale to indicate appropriately the percentage of work scheduled for completion by any given date during the period. If the Contractor fails to submit a schedule within the time prescribed, the Contracting Officer may withhold approval of progress payments or take other remedies under the contract until the Contractor submits the required schedule.
- (b) The Contractor shall enter the actual progress on the chart as required by the Contracting Officer, and immediately deliver three copies of the annotated schedule to the Contracting Officer. If the Contracting Officer determines, upon the basis of inspection conducted pursuant to the clause entitled Inspection and Acceptance of Construction, herein that the Contractor is not meeting the approved schedule, the Contractor shall take steps necessary to improve its progress, including those that may be required by the Contracting Officer, without additional cost to the PHA. In this circumstance, the Contracting Officer may require the Contractor to increase the number of shifts, overtime operations, days of work, and/or the amount of construction plant, and to submit for approval any supplementary schedule or schedules in chart form as the Contracting Officer deems necessary to demonstrate how the approved rate of progress will be regained.
- (c) Failure of the Contractor to comply with the requirements of the Contracting Officer under this clause shall be grounds for a determination by the Contracting Officer that the Contractor is not prosecuting the work with sufficient diligence to ensure completion within the time specified in the Contract. Upon making this determination, the Contracting Officer may terminate the Contractor's right to proceed with the work, or any separable part of it, in accordance with the Default clause of this contract.

7. Site Investigation and Conditions Affecting the Work

(a) The Contractor acknowledges that it has taken steps reasonably necessary to ascertain the nature and location

as to the general and local conditions which can affect the work or its cost, including but not limited to, (1) conditions bearing upon transportation, disposal, handling, and storage of materials; (2) the availability of labor, water, electric power, and roads; (3) uncertainties of weather, river stages, tides, or similar physical conditions at the site; (4) the conformation and conditions of the ground; and (5) the character of equipment and facilities needed preliminary to and during work performance. The Contractor also acknowledges that it has satisfied itself as to the character, quality, and quantity of surface and subsurface materials or obstacles to be encountered insofar as this information is

- reasonably ascertainable from an inspection of the site,
- including all exploratory work done by the PHA, as well as from the drawings and specifications made a part of this contract. Any failure of the Contractor to take the actions described and acknowledged in this paragraph will not relieve the Contractor from responsibility for estimating properly the difficulty and cost of successfully performing the work, or for proceeding to successfully perform the work without additional expense to the PHA.
- (b) The PHA assumes no responsibility for any conclusions or interpretations made by the Contractor based on the information made available by the PHA. Nor does the PHA assume responsibility for any understanding reached or representation made concerning conditions which can affect the work by any of its officers or agents before the execution of this contract, unless that understanding or representation is expressly stated in this contract.

8. Differing Site Conditions

- (a) The Contractor shall promptly, and before the conditions are disturbed, give a written notice to the Contracting Officer of (1) subsurface or latent physical conditions at the site which differ materially from those indicated in this contract, or (2) unknown physical conditions at the site(s), of an unusual nature, which differ materially from those ordinarily encountered and generally recognized as inhering in work of the character provided for in the contract.
- (b) The Contracting Officer shall investigate the site conditions promptly after receiving the notice. Work shall not proceed at the affected site, except at the
 - Contractor's risk, until the Contracting Officer has provided written instructions to the Contractor. If the conditions do materially so differ and cause an increase or decrease in the Contractor's cost of, or the time required for, performing any part of the work under this contract, whether or not changed as a result of the conditions, the Contractor shall file a claim in writing to the PHA within ten days after receipt of such instructions and, in any event, before proceeding with the work. An equitable adjustment in the contract price, the delivery schedule, or both shall be made under this clause and the contract modified in writing accordingly.
- (c) No request by the Contractor for an equitable adjustment to the contract under this clause shall be allowed, unless the Contractor has given the written notice required; provided, that the time prescribed in (a) above for giving written notice may be extended by the Contracting Officer.
- (d) No request by the Contractor for an equitable adjustment to the contract for differing site conditions shall be allowed if made after final payment under this contract.

9. Specifications and Drawings for Construction

(a) The Contractor shall keep on the work site a copy of the drawings and specifications and shall at all times give the Contracting Officer access thereto. Anything mentioned in the specifications and not shown on the drawings, or shown on the drawings and not mentioned in the specifications, shall be of like effect as if shown or mentioned in both. In case of difference between drawings and specifications, the specifications shall govern. In case of discrepancy in the figures, in the drawings, or in the specifications, the matter shall be

- promptly submitted to the Contracting Officer, who shall promptly make a determination in writing. Any adjustment by the Contractor without such a determination shall be at its own risk and expense. The Contracting Officer shall furnish from time to time such detailed drawings and other information as considered necessary, unless otherwise provided.
- (b) Wherever in the specifications or upon the drawings the words "directed", "required", "ordered", "designated", "prescribed", or words of like import are used, it shall be understood that the "direction", "requirement", "order", "designation", or "prescription", of the Contracting Officer is intended and similarly the words "approved", "acceptable", "satisfactory", or words of like import shall mean "approved by", or "acceptable to", or "satisfactory to" the Contracting Officer, unless otherwise expressly stated.
- (c) Where "as shown" "as indicated", "as detailed", or words of similar import are used, it shall be understood that the reference is made to the drawings accompanying this contract unless stated otherwise. The word "provided" as used herein shall be understood to mean "provide complete in place" that is "furnished and installed".
- (d) "Shop drawings" means drawings, submitted to the PHA by the Contractor, subcontractor, or any lower tier subcontractor, showing in detail (1) the proposed fabrication and assembly of structural elements and (2) the installation (i.e., form, fit, and attachment details) of materials of equipment. It includes drawings, diagrams, layouts, schematics, descriptive literature, illustrations, schedules, performance and test data, and similar materials furnished by the Contractor to explain in detail specific portions of the work required by the contract. The PHA may duplicate, use, and disclose in any manner and for any purpose shop drawings delivered under this contract.
- (e) If this contract requires shop drawings, the Contractor shall coordinate all such drawings, and review them for accuracy, completeness, and compliance with other contract requirements and shall indicate its approval thereon as evidence of such coordination and review. Shop drawings submitted to the Contracting Officer without evidence of the Contractor's approval may be returned for resubmission. The Contracting Officer will indicate an approval or disapproval of the shop drawings and if not approved as submitted shall indicate the PHA's reasons therefore. Any work done before such approval shall be at the Contractor's risk. Approval by the Contracting Officer shall not relieve the Contractor from responsibility for any errors or omissions in such drawings, nor from responsibility for complying with the requirements of this contract, except with respect to variations described and approved in accordance with (f) below
- (f) If shop drawings show variations from the contract requirements, the Contractor shall describe such variations in writing, separate from the drawings, at the time of submission. If the Architect approves any such variation and the Contracting Officer concurs, the Contracting Officer shall issue an appropriate modification to the contract, except that, if the variation is minor or does not involve a change in price or in time of performance, a modification need not be issued.

 (g) It shall be the responsibility of the Contractor to make
- (g) It shall be the responsibility of the Contractor to make timely requests of the PHA for such large scale and full size drawings, color schemes, and other additional information, not already in his possession, which shall be

- required in the planning and production of the work. Such requests may be submitted as the need arises, but each such request shall be filed in ample time to permit appropriate action to be taken by all parties involved so as to avoid delay.
- (h) The Contractor shall submit to the Contracting Officer for approval four copies (unless otherwise indicated) of all shop drawings as called for under the various headings of these specifications. Three sets (unless otherwise indicated) of all shop drawings, will be retained by the PHA and one set will be returned to the Contractor. As required by the Contracting Officer, the Contractor, upon completing the work under this contract, shall furnish a complete set of all shop drawings as finally approved. These drawings shall show all changes and revisions made up to the time the work is completed and accepted.
- (i) This clause shall be included in all subcontracts at any tier. It shall be the responsibility of the Contractor to ensure that all shop drawings prepared by subcontractors are submitted to the Contracting Officer.

10. As-Built Drawings

- (a) "As-built drawings," as used in this clause, means drawings submitted by the Contractor or subcontractor at any tier to show the construction of a particular structure or work as actually completed under the contract. "As-built drawings" shall be synonymous with "Record drawings."
- (b) As required by the Contracting Officer, the Contractor shall provide the Contracting Officer accurate information to be used in the preparation of permanent as-built drawings. For this purpose, the Contractor shall record on one set of contract drawings all changes from the installations originally indicated, and record final locations of underground lines by depth from finish grade and by accurate horizontal offset distances to permanent surface improvements such as buildings, curbs, or edges of walks
- (c) This clause shall be included in all subcontracts at any tier. It shall be the responsibility of the Contractor to ensure that all as-built drawings prepared by subcontractors are submitted to the Contracting Officer.

11. Material and Workmanship

- (a) All equipment, material, and articles furnished under this contract shall be new and of the most suitable grade for the purpose intended, unless otherwise specifically provided in this contract. References in the contract to equipment, material, articles, or patented processes by trade name, make, or catalog number, shall be regarded as establishing a standard of quality and shall not be construed as limiting competition. The Contractor may, at its option, use any equipment, material, article, or process that, in the judgment of, and as approved by the Contracting Officer, is equal to that named in the specifications, unless otherwise specifically provided in this contract.
- (b) Approval of equipment and materials.
- (1) The Contractor shall obtain the Contracting Officer's approval of the machinery and mechanical and other equipment to be incorporated into the work. When requesting approval, the Contractor shall furnish to the Contracting Officer the name of the manufacturer, the model number, and other information concerning the performance, capacity, nature, and rating of the

- machinery and mechanical and other equipment. When required by this contract or by the Contracting Officer, the Contractor shall also obtain the Contracting Officer's approval of the material or articles which the Contractor contemplates incorporating into the work. When requesting approval, the Contractor shall provide full information concerning the material or articles. Machinery, equipment, material, and articles that do not have the required approval shall be installed or used at the risk of subsequent rejection.
- (2) When required by the specifications or the Contracting Officer, the Contractor shall submit appropriately marked samples (and certificates related to them) for approval at the Contractor's expense, with all shipping charges prepaid. The Contractor shall label, or otherwise properly mark on the container, the material or product represented, its place of origin, the name of the producer, the Contractor's name, and the identification of the construction project for which the material or product is intended to be used.
- (3) Certificates shall be submitted in triplicate, describing each sample submitted for approval and certifying that the material, equipment or accessory complies with contract requirements. The certificates shall include the name and brand of the product, name of manufacturer, and the location where produced.
- (4) Approval of a sample shall not constitute a waiver of the PHA right to demand full compliance with contract requirements. Materials, equipment and accessories may be rejected for cause even though samples have been approved.
- (5) Wherever materials are required to comply with recognized standards or specifications, such specifications shall be accepted as establishing the technical qualities and testing methods, but shall not govern the number of tests required to be made nor modify other contract requirements. The Contracting Officer may require laboratory test reports on items submitted for approval or may approve materials on the basis of data submitted in certificates with samples. Check tests will be made on materials delivered for use only as frequently as the Contracting Officer determines necessary to insure compliance of materials with the specifications. The Contractor will assume all costs of retesting materials which fail to meet contract requirements and/or testing materials offered in substitution for those found deficient.
- (6) After approval, samples will be kept in the Project office until completion of work. They may be built into the work after a substantial quantity of the materials they represent has been built in and accepted.
- (c) Requirements concerning lead-based paint. The Contractor shall comply with the requirements concerning lead-based paint contained in the Lead-Based Paint Poisoning Prevention Act (42 U.S.C. 4821-4846) as implemented by 24 CFR Part 35.

12. Permits and Codes

(a) The Contractor shall give all notices and comply with all applicable laws, ordinances, codes, rules and regulations. Notwithstanding the requirement of the Contractor to comply with the drawings and specifications in the contract, all work installed shall comply with all applicable codes and regulations as amended by any waivers. Before installing the work, the Contractor shall examine the drawings and the specifications for compliance with applicable codes and regulations bearing on the work and shall immediately report any discrepancy it may discover to the Contracting Officer. Where the requirements of the drawings and specifications fail to comply with the applicable code or regulation, the Contracting Officer shall modify the contract by change order pursuant to the clause entitled Changes herein to conform to the code or regulation.

- (b) The Contractor shall secure and pay for all permits, fees, and licenses necessary for the proper execution and completion of the work. Where the PHA can arrange for the issuance of all or part of these permits, fees and licenses, without cost to the Contractor, the contract amount shall be reduced accordingly.
- 13. Health, Safety, and Accident Prevention
- (a) In performing this contract, the Contractor shall:
- (1) Ensure that no laborer or mechanic shall be required to work in surroundings or under working conditions which are unsanitary, hazardous, or dangerous to his/her health and/or safety as determined under construction safety and health standards promulgated by the Secretary of Labor by regulation;
- (2) Protect the lives, health, and safety of other persons;
- (3) Prevent damage to property, materials, supplies, and equipment; and.
- (4) Avoid work interruptions.
- (b) For these purposes, the Contractor shall:
- (1) Comply with regulations and standards issued by the Secretary of Labor at 29 CFR Part 1926. Failure to comply may result in imposition of sanctions pursuant to the Contract Work Hours and Safety Standards Act (Public Law 91-54, 83 Stat. 96), 40 U.S.C. 3701 et seq.; and
- (2) Include the terms of this clause in every subcontract so that such terms will be binding on each subcontractor.
- (c) The Contractor shall maintain an accurate record of exposure data on all accidents incident to work performed under this contract resulting in death, traumatic injury, occupational disease, or damage to property, materials, supplies, or equipment, and shall report this data in the manner prescribed by 29 CFR Part 1904
- (d) The Contracting Officer shall notify the Contractor of any noncompliance with these requirements and of the corrective action required. This notice, when delivered to the Contractor or the Contractor's representative at the site of the work, shall be deemed sufficient notice of the noncompliance and corrective action required. After receiving the notice, the Contractor shall immediately take corrective action. If the Contractor fails or refuses to take corrective action promptly, the Contracting Officer may issue an order stopping all or part of the work until satisfactory corrective action has been taken. The Contractor shall not base any claim or request for equitable adjustment for additional time or money on any stop order issued under these circumstances.
- (e) The Contractor shall be responsible for its subcontractors' compliance with the provisions of this clause. The Contractor shall take such action with respect to any subcontract as the PHA, the Secretary of Housing and Urban Development, or the Secretary of Labor shall direct as a means of enforcing such provisions.

14. Temporary Heating

The Contractor shall provide and pay for temporary heating, covering, and enclosures necessary to properly protect all work and materials against damage by dampness and cold, to dry out the work, and to facilitate the completion of the work. Any permanent heating equipment used shall be turned over to the PHA in the condition and at the time required by the specifications.

15. Availability and Use of Utility Services

- (a) The PHA shall make all reasonably required amounts of utilities available to the Contractor from existing outlets and supplies, as specified in the contract. Unless otherwise provided in the contract, the amount of each utility service consumed shall be charged to or paid for by the Contractor at prevailing rates charged to the PHA or, where the utility is produced by the PHA, at reasonable rates determined by the Contracting Officer. The Contractor shall carefully conserve any utilities furnished without charge.
- (b) The Contractor, at its expense and in a manner satisfactory to the Contracting Officer, shall install and maintain all necessary temporary connections and distribution lines, and all meters required to measure the amount of each utility used for the purpose of determining charges. Before final acceptance of the work by the PHA, the Contractor shall remove all the temporary connections, distribution lines, meters, and associated paraphernalia.
- 16. Protection of Existing Vegetation, Structures, Equipment, Utilities, and Improvements
- (a) The Contractor shall preserve and protect all structures, equipment, and vegetation (such as trees, shrubs, and grass) on or adjacent to the work site, which are not to be removed under this contract, and which do not unreasonably interfere with the work required under this
- (b) The Contractor shall only remove trees when specifically authorized to do so, and shall avoid damaging vegetation that will remain in place. If any limbs or branches of trees are broken during performance of this contract, or by the careless operation of equipment, or by workmen, the Contractor shall trim those limbs or branches with a clean cut and paint the cut with a tree-pruning compound as directed by the Contracting Officer.
- (c) The Contractor shall protect from damage all existing improvements and utilities (1) at or near the work site and (2) on adjacent property of a third party, the locations of which are made known to or should be known by the Contractor. Prior to disturbing the ground at the construction site, the Contractor shall ensure that all underground utility lines are clearly marked.
- (d) The Contractor shall shore up, brace, underpin, secure, and protect as necessary all foundations and other parts of existing structures adjacent to, adjoining, and in the vicinity of the site, which may be affected by the excavations or other operations connected with the construction of the project.
- (e) Any equipment temporarily removed as a result of work under this contract shall be protected, cleaned, and replaced in the same condition as at the time of award of this contract.

- (f) New work which connects to existing work shall correspond in all respects with that to which it connects and/or be similar to existing work unless otherwise required by the specifications.
- (g) No structural members shall be altered or in any way weakened without the written authorization of the Contracting Officer, unless such work is clearly specified in the plans or specifications.
- (h) If the removal of the existing work exposes discolored or unfinished surfaces, or work out of alignment, such surfaces shall be refinished, or the material replaced as necessary to make the continuous work uniform and harmonious. This, however, shall not be construed to require the refinishing or reconstruction of dissimilar finishes previously exposed, or finished surfaces in good condition, but in different planes or on different levels Construction when brought together by the removal of intervening work, unless such refinishing or reconstruction is specified in the plans or specifications.
- The Contractor shall give all required notices to any adjoining or adjacent property owner or other party before the commencement of any work.
- (j) The Contractor shall indemnify and save harmless the PHA from any damages on account of settlement or the loss of lateral support of adjoining property, any damages from changes in topography affecting drainage, and from all loss or expense and all damages for which the PHA may become liable in consequence of such injury or damage to adjoining and adjacent structures and their premises.
- (k) The Contractor shall repair any damage to vegetation, structures, equipment, utilities, or improvements, including those that are the property of a third party, resulting from failure to comply with the requirements of this contract or failure to exercise reasonable care in performing the work. If the Contractor fails or refuses to repair the damage promptly, the Contracting Officer may have the necessary work performed and charge the cost to the Contractor.

17. Temporary Buildings and Transportation of Materials

- (a) Temporary buildings (e.g., storage sheds, shops, offices, sanitary facilities) and utilities may be erected by the Contractor only with the approval of the Contracting Officer and shall be built with labor and materials furnished by the Contractor without expense to the PHA. The temporary buildings and utilities shall remain the property of the Contractor and shall be removed by the Contractor at its expense upon completion of the work. With the written consent of the Contracting Officer, the buildings and utilities may be abandoned and need not be removed.
- (b) The Contractor shall, as directed by the Contracting Officer, use only established roadways, or use temporary roadways constructed by the Contractor when and as authorized by the Contracting Officer. When materials are transported in prosecuting the work, vehicles shall not be loaded beyond the loading capacity recommended by the manufacturer of the vehicle or prescribed by any federal, state, or local law or regulation. When it is necessary to cross curbs or sidewalks, the Contractor shall protect them from damage. The Contractor shall repair or pay for the repair of any damaged curbs, sidewalks, or roads.

18. Clean Air and Water

The contactor shall comply with the Clean Air Act, as amended, 42 USC 7401 et seq., the Federal Water Pollution Control Water Act, as amended, 33 U.S.C. 1251 et seq., and standards issued pursuant thereto in the facilities in which this contract is to be performed.

19. Energy Efficiency

The Contractor shall comply with mandatory standards and policies relating to energy efficiency which are contained in the energy conservation plan issued in compliance with the Energy Policy and Conservation Act (Pub.L. 94-163) for the State in which the work under the contract is performed.

20. Inspection and Acceptance of

- (a) Definitions. As used in this clause -
- (1) "Acceptance" means the act of an authorized representative of the PHA by which the PHA approves and assumes ownership of the work performed under this contract. Acceptance may be partial or complete.
- (2) "Inspection" means examining and testing the work performed under the contract (including, when appropriate, raw materials, equipment, components, and intermediate assemblies) to determine whether it conforms to contract requirements.
- (3) "Testing" means that element of inspection that determines the properties or elements, including functional operation of materials, equipment, or their components, by the application of established scientific principles and procedures.
- (b) The Contractor shall maintain an adequate inspection system and perform such inspections as will ensure that the work performed under the contract conforms to contract requirements. All work is subject to PHA inspection and test at all places and at all reasonable times before acceptance to ensure strict compliance with the terms of the contract.
- (c) PHA inspections and tests are for the sole benefit of the PHA and do not: (1) relieve the Contractor of responsibility for providing adequate quality control measures; (2) relieve the Contractor of responsibility for loss or damage of the material before acceptance; (3) constitute or imply acceptance; or, (4) affect the continuing rights of the PHA after acceptance of the completed work under paragraph (j) below.
- (d) The presence or absence of the PHA inspector does not relieve the Contractor from any contract requirement, nor is the inspector authorized to change any term or condition of the specifications without the Contracting Officer's written authorization. All instructions and approvals with respect to the work shall be given to the Contractor by the Contracting Officer.
- (e) The Contractor shall promptly furnish, without additional charge, all facilities, labor, and material reasonably needed for performing such safe and convenient inspections and tests as may be required by the Contracting Officer. The PHA may charge to the Contractor any additional cost of inspection or test when work is not ready at the time specified by the Contractor for inspection or test, or when prior rejection makes reinspection or retest necessary. The PHA shall perform all inspections and tests in a manner that will not unnecessarily delay the work. Special, full size, and performance tests shall be performed as described in the contract.

- (f) The PHA may conduct routine inspections of the construction site on a daily basis.
- (g) The Contractor shall, without charge, replace or correct work found by the PHA not to conform to contract requirements, unless the PHA decides that it is in its interest to accept the work with an appropriate adjustment in contract price. The Contractor shall promptly segregate and remove rejected material from the premises.
- (h) If the Contractor does not promptly replace or correct rejected work, the PHA may (1) by contract or otherwise, replace or correct the work and charge the cost to the Contractor, or (2) terminate for default the Contractor's right to proceed.
- (i) If any work requiring inspection is covered up without approval of the PHA, it must, if requested by the Contracting Officer, be uncovered at the expense of the Contractor. If at any time before final acceptance of the entire work, the **Construction** PHA considers it necessary or advisable, to examine work already completed by removing or tearing it out, the Contractor, shall on request, promptly furnish all necessary facilities, labor, and material. If such work is found to be defective or nonconforming in any material respect due to the fault of the Contractor or its subcontractors, the Contractor shall defray all the expenses of the examination and of satisfactory reconstruction. If, however, such work is found to meet the requirements of the contract, the Contracting Officer shall make an equitable adjustment to cover the cost of the examination and reconstruction, including, if completion of the work was thereby delayed, an extension of time
- (j) The Contractor shall notify the Contracting Officer, in writing, as to the date when in its opinion all or a designated portion of the work will be substantially completed and ready for inspection. If the Architect determines that the state of preparedness is as represented, the PHA will promptly arrange for the inspection. Unless otherwise specified in the contract, the PHA shall accept, as soon as practicable after completion and inspection, all work required by the contract or that portion of the work the Contracting Officer determines and designates can be accepted separately. Acceptance shall be final and conclusive except for latent defects, fraud, gross mistakes amounting to fraud, or the PHA's right under any warranty or guarantee.

21. Use and Possession Prior to Completion

- (a) The PHA shall have the right to take possession of or use any completed or partially completed part of the work. Before taking possession of or using any work, the Contracting Officer shall furnish the Contractor a list of items of work remaining to be performed or corrected on those portions of the work that the PHA intends to take possession of or use. However, failure of the Contracting Officer to list any item of work shall not relieve the Contractor of responsibility for complying with the terms of the contract. The PHA's possession or use shall not be deemed an acceptance of any work under the contract.
- (b) While the PHA has such possession or use, the Contractor shall be relieved of the responsibility for (1) the loss of or damage to the work resulting from the PHA's possession or use, notwithstanding the terms of the clause entitled Permits and Codes herein; (2) all maintenance costs on the areas occupied; and, (3) furnishing heat, light, power, and water used in the areas

occupied without proper remuneration therefore. If prior possession or use by the PHA delays the progress of the work or causes additional expense to the Contractor, an equitable adjustment shall be made in the contract price or the time of completion, and the contract shall be modified in writing accordingly.

22. Warranty of Title

The Contractor warrants good title to all materials, supplies, and equipment incorporated in the work and agrees to deliver the premises together with all improvements thereon free from any claims, liens or charges, and agrees further that neither it nor any other person, firm or corporation shall have any right to a lien upon the premises or anything appurtenant thereto.

23. Warranty of

- (a) In addition to any other warranties in this contract, the Contractor warrants, except as provided in paragraph (j) of this clause, that work performed under this contract conforms to the contract requirements and is free of any defect in equipment, material, or workmanship performed by the Contractor or any subcontractor or supplier at any tier. This warranty shall continue for a period of (one year unless otherwise indicated) from the date of final acceptance of the work. If the PHA takes possession of any part of the work before final acceptance, this warranty shall continue for a period of (one year unless otherwise indicated) from the date that the PHA takes possession.
- (b) The Contractor shall remedy, at the Contractor's expense, any failure to conform, or any defect. In addition, the Contractor shall remedy, at the Contractor's expense, any damage to PHA-owned or controlled real or personal property when the damage is the result of— (1) The Contractor's failure to conform to contract requiremonts; or
 - (2) Any defects of equipment, material, workmanship or design furnished by the Contractor.
- (c) The Contractor shall restore any work damaged in fulfilling the terms and conditions of this clause. The Contractor's warranty with respect to work repaired or replaced will run for (one year unless otherwise indicated) from the date of repair or replacement.
- (d) The Contracting Officer shall notify the Contractor, in writing, within a reasonable time after the discovery of any failure, defect or damage.
- (e) If the Contractor fails to remedy any failure, defect, or damage within a reasonable time after receipt of notice, the PHA shall have the right to replace, repair or otherwise remedy the failure, defect, or damage at the Contractor's expense.
- (f) With respect to all warranties, express or implied, from subcontractors, manufacturers, or suppliers for work performed and materials furnished under this contract, the Contractor shall:
 - Obtain all warranties that would be given in normal commercial practice;
 - (2) Require all warranties to be executed in writing, for the benefit of the PHA; and.
 - (3) Enforce all warranties for the benefit of the PHA.
- (g) In the event the Contractor's warranty under paragraph (a) of this clause has expired, the PHA may bring suit at its own expense to enforce a subcontractor's, manufacturer's or supplier's warranty.

- (h) Unless a defect is caused by the negligence of the Contractor or subcontractor or supplier at any tier, the Contractor shall not be liable for the repair of any defect of material or design furnished by the PHA nor for the repair of any damage that results from any defect in PHA furnished material or design.
- (i) Notwithstanding any provisions herein to the contrary, the establishment of the time periods in paragraphs (a) and (c) above relate only to the specific obligation of the Contractor to correct the work, and have no relationship to the time within which its obligation to comply with the contract may be sought to be enforced, nor to the time within which proceedings may be commenced to establish the Contractor's liability with respect to its obligation other than specifically to correct the work.
- (j) This warranty shall not limit the PHA's rights under the Inspection and Acceptance of Construction clause of this contract with respect to latent defects, gross mistakes or fraud.

24. Prohibition Against Liens

The Contractor is prohibited from placing a lien on the PHA's property. This prohibition shall apply to all subcontractors at any tier and all materials suppliers.

Administrative Requirements

25. Contract Period

this contract within calendar days of the effective date of the contract, or within the time schedule established in the notice to proceed issued by the Contracting Officer.

26. Order of Provisions

accordance with the terms and conditions of the
In the event of a conflict between these General
Conditions and the Specifications, the General
Conditions shall prevail. In the event of a conflict between
the contract and any applicable state or local law or
regulation, the state or local law or regulation shall
prevail; provided that such state or local law or regulation
does not conflict with, or is less restrictive than applicable
federal law, regulation, or Executive Order. In the event of
such a conflict, applicable federal law, regulation, and
Executive Order shall prevail.

27. Payments

- (a) The PHA shall pay the Contractor the price as provided in this contract
- (b) The PHA shall make progress payments approximately every 30 days as the work proceeds, on estimates of work accomplished which meets the standards of quality established under the contract, as approved by the Contracting Officer. The PHA may, subject to written determination and approval of the Contracting Officer, make more frequent payments to contractors which are qualified small businesses.
- (c) Before the first progress payment under this contract, the Contractor shall furnish, in such detail as requested by the Contracting Officer, a breakdown of the total contract price showing the amount included therein for each principal category of the work, which shall substantiate the payment amount requested in order to provide a

- basis for determining progress payments. The breakdown shall be approved by the Contracting Officer and must be acceptable to HUD. If the contract covers more than one project, the Contractor shall furnish a separate breakdown for each. The values and quantities employed in making up this breakdown are for determining the amount of progress payments and shall not be construed as a basis for additions to or deductions from the contract price. The Contractor shall prorate its overhead and profit over the construction period of the contract.
- (d) The Contractor shall submit, on forms provided by the PHA, periodic estimates showing the value of the work performed during each period based upon the approved

submitted not later than ______ days in advance of the date set for payment and are subject to correction and revision as required. The estimates must be approved by the Contracting Officer with the concurrence of the Architect prior to payment. If the contract covers more than one project, the Contractor shall furnish a separate progress payment estimate for each.

- (e) Along with each request for progress payments and the required estimates, the Contractor shall furnish the following certification, or payment shall not be made: I hereby certify, to the best of my knowledge and belief, that:
 - The amounts requested are only for performance in accordance with the specifications, terms, and conditions of the contract;
 - (2) Payments to subcontractors and suppliers have been made from previous payments received under the contract, and timely payments will be made from the proceeds of the payment covered by this certification, in accordance with subcontract agreements; and,
 - (3) This request for progress payments does not include any amounts which the prime contractor intends to withhold or retain from a subcontractor or supplier in

Name:		

subcontract.

Date:

Title:

- (f) Except as otherwise provided in State law, the PHA shall retain ten (10) percent of the amount of progress payments until completion and acceptance of all work under the contract; except, that if upon completion of 50 percent of the work, the Contracting Officer, after consulting with the Architect, determines that the Contractor's performance and progress are satisfactory, the PHA may make the remaining payments in full for the work subsequently completed. If the Contracting Officer subsequently determines that the Contractor's performance and progress are unsatisfactory, the PHA shall reinstate the ten (10) percent (or other percentage as provided in State law) retainage until such time as the Contracting Officer determines that performance and progress are satisfactory.
- (g) The Contracting Officer may authorize material delivered on the site and preparatory work done to be taken into consideration when computing progress payments.

- Material delivered to the Contractor at locations other than the site may also be taken into consideration if the Contractor furnishes satisfactory evidence that (1) it has acquired title to such material; (2) the material is properly stored in a bonded warehouse, storage yard, or similar suitable place as may be approved by the Contracting Officer; (3) the material is insured to cover its full value; and (4) the material will be used to perform this contract. Before any progress payment which includes delivered material is made, the Contractor shall furnish such documentation as the Contracting Officer may require to assure the protection of the PHA's interest in such materials. The Contractor shall remain responsible for such stored material notwithstanding the transfer of title to the PHA
- (h) All material and work covered by progress payments made shall, at the time of payment become the sole property of the PHA, but this shall not be construed as (1) relieving the Contractor from the sole responsibility for all material and work upon which payments have been made or the restoration of any damaged work; or, (2) waiving the right of the PHA to require the fulfillment of all of the terms of the contract. In the event the work of the Contractor has been damaged by other contractors or persons other than employees of the PHA in the course of their employment, the Contractor shall restore such damaged work without cost to the PHA and to seek redress for its damage only from those who directly caused it.
- (i) The PHA shall make the final payment due the Contractor under this contract after (1) completion and final acceptance of all work; and (2) presentation of release of all claims against the PHA arising by virtue of this contract, other than claims, in stated amounts, that the Contractor has specifically excepted from the operation of the release. Each such exception shall embrace no more than one claim, the basis and scope of which shall be clearly defined. The amounts for such excepted claims shall not be included in the request for final payment. A release may also be required of the assignee if the Contractor's claim to amounts payable under this contract has been assigned.
- (j) Prior to making any payment, the Contracting Officer may require the Contractor to furnish receipts or other evidence of payment from all persons performing work and supplying material to the Contractor, if the Contracting Officer determines such evidence is necessary to substantiate claimed costs.
- (k) The PHA shall not; (1) determine or adjust any claims for payment or disputes arising there under between the Contractor and its subcontractors or material suppliers; or, (2) withhold any moneys for the protection of the subcontractors or material suppliers. The failure or refusal of the PHA to withhold moneys from the Contractor shall in nowise impair the obligations of any surety or sureties under any bonds furnished under this contract.

28. Contract Modifications

- (a) Only the Contracting Officer has authority to modify any term or condition of this contract. Any contract modification shall be authorized in writing.
- (b) The Contracting Officer may modify the contract unilaterally (1) pursuant to a specific authorization stated in a contract clause (e.g., Changes); or (2) for administrative matters which do not change the rights or

- responsibilities of the parties (e.g., change in the PHA address). All other contract modifications shall be in the form of supplemental agreements signed by the Contractor and the Contracting Officer.
- (c) When a proposed modification requires the approval of HUD prior to its issuance (e.g., a change order that exceeds the PHA's approved threshold), such modification shall not be effective until the required approval is received by the PHA.

29. Changes

- (a) The Contracting Officer may, at any time, without notice to the sureties, by written order designated or indicated to be a change order, make changes in the work within the general scope of the contract including changes:
 - (1) In the specifications (including drawings and designs);
 - (2) In the method or manner of performance of the work;
 - (3) PHA-furnished facilities, equipment, materials, services or site: or
 - services, or site; or,
 (4) Directing the acceleration in the performance of the work.
- (b) Any other written order or oral order (which, as used in this paragraph (b), includes direction, instruction, interpretation, or determination) from the Contracting Officer that causes a change shall be treated as a change order under this clause; provided, that the Contractor gives the Contracting Officer written notice stating (1) the date, circumstances and source of the order and (2) that the Contractor regards the order as a change order.
- (c) Except as provided in this clause, no order, statement or conduct of the Contracting Officer shall be treated as a change under this clause or entitle the Contractor to an equitable adjustment.
- (d) If any change under this clause causes an increase or decrease in the Contractor's cost of, or the time required for the performance of any part of the work under this contract, whether or not changed by any such order, the Contracting Officer shall make an equitable adjustment and modify the contract in writing. However, except for a adjustment based on defective specifications, no proposal for any change under paragraph (b) above shall be allowed for any costs incurred more than 20 days (5 days for oral orders) before the Contractor gives written notice as required. In the case of defective specifications for which the PHA is responsible, the equitable adjustment shall include any increased cost reasonably incurred by the Contractor in attempting to comply with the defective specifications.
- (e) The Contractor must assert its right to an adjustment under this clause within 30 days after (1) receipt of a written change order under paragraph (a) of this clause, or (2) the furnishing of a written notice under paragraph (b) of this clause, by submitting a written statement describing the general nature and the amount of the proposal. If the facts justify it, the Contracting Officer may extend the period for submission. The proposal may be included in the notice required under paragraph (b) above. No proposal by the Contractor for an equitable adjustment shall be allowed if asserted after final payment under this contract.
- (f) The Contractor's written proposal for equitable adjustment shall be submitted in the form of a lump sum proposal supported with an itemized breakdown of all increases and decreases in the contract in at least the following details:

- (1) Direct Costs. Materials (list individual items, the quantity and unit cost of each, and the aggregate cost); Transportation and delivery costs associated with materials; Labor breakdowns by hours or unit costs (identified with specific work to be performed); Construction equipment exclusively necessary for the change; Costs of preparation and/ or revision to shop drawings resulting from the change; Worker's Compensation and Public Liability Insurance; Employment taxes under FICA and FUTA; and, Bond Costs when size of change warrants revision.
- (2) Indirect Costs. Indirect costs may include overhead, general and administrative expenses, and fringe benefits not normally treated as direct costs.
- (3) Profit. The amount of profit shall be negotiated and may vary according to the nature, extent, and complexity of the work required by the change. The allowability of the direct and indirect costs shall be determined in accordance with the Contract Cost Principles and Procedures for Commercial Firms in Part 31 of the Federal Acquisition Regulation (48 CFR 1-31), as implemented by HUD Handbook 2210.18, in effect on the date of this contract. The Contractor shall not be allowed a profit on the profit received by any subcontractor. Equitable adjustments for deleted work shall include a credit for profit and may include a credit for indirect costs. On proposals covering both increases and decreases in the amount of the contract, the application of indirect costs and profit shall be on the net-change in direct costs for the Contractor or subcontractor performing the work.
- (g) The Contractor shall include in the proposal its request for time extension (if any), and shall include sufficient information and dates to demonstrate whether and to what extent the change will delay the completion of the contract in its entirety.
- (h) The Contracting Officer shall act on proposals within 30 days after their receipt, or notify the Contractor of the date when such action will be taken.
- (i) Failure to reach an agreement on any proposal shall be a dispute under the clause entitled Disputes herein.
 Nothing in this clause, however, shall excuse the Contractor from proceeding with the contract as changed.
- (j) Except in an emergency endangering life or property, no change shall be made by the Contractor without a prior order from the Contracting Officer.

30. Suspension of Work

- (a) The Contracting Officer may order the Contractor in writing to suspend, delay, or interrupt all or any part of the work of this contract for the period of time that the
 - Contracting Officer determines appropriate for the convenience of the PHA.
- (b) If the performance of all or any part of the work is, for an unreasonable period of time, suspended, delayed, or interrupted (1) by an act of the Contracting Officer in the administration of this contract, or (2) by the Contracting Officer's failure to act within the time specified (or within a reasonable time if not specified) in this contract an adjustment shall be made for any increase in the cost of performance of the contract (excluding profit) necessarily caused by such unreasonable suspension, delay, or interruption and the contract modified in writing accordingly. However, no adjustment shall be made under this clause for any suspension, delay, or interruption to the extent that performance would have

- been so suspended, delayed, or interrupted by any other cause, including the fault or negligence of the Contractor or for which any equitable adjustment is provided for or
- excluded under any other provision of this contract.

 (c) A claim under this clause shall not be allowed (1) for any costs incurred more than 20 days before the Contractor shall have notified the Contracting Officer in writing of the act or failure to act involved (but this requirement shall not apply as to a claim resulting from a suspension order); and, (2) unless the claim, in an amount stated, is asserted in writing as soon as practicable after the termination of the suspension, delay, or interruption, but not later than the date of final payment under the contract.

31. Disputes

- (a) "Claim," as used in this clause, means a written demand or written assertion by one of the contracting parties seeking, as a matter of right, the payment of money in a sum certain, the adjustment or interpretation of contract terms, or other relief arising under or relating to the contract. A claim arising under the contract, unlike a claim relating to the contract, is a claim that can be resolved under a contract clause that provides for the relief sought by the claimant. A voucher, invoice, or other routine request for payment that is not in dispute when submitted is not a claim. The submission may be converted to a claim by complying with the requirements of this clause, if it is disputed either as to liability or amount or is not acted upon in a reasonable time.
- (b) Except for disputes arising under the clauses entitled Labor Standards - Davis Bacon and Related Acts, herein, all disputes arising under or relating to this contract, including any claims for damages for the alleged breach thereof which are not disposed of by agreement, shall be resolved under this clause.
- (c) All claims by the Contractor shall be made in writing and submitted to the Contracting Officer for a written decision. A claim by the PHA against the Contractor shall be subject to a written decision by the Contracting Officer.
- (d) The Contracting Officer shall, within 60 (unless otherwise indicated) days after receipt of the request, decide the claim or notify the Contractor of the date by which the decision will be made.
- (e) The Contracting Officer's decision shall be final unless the Contractor (1) appeals in writing to a higher level in the PHA in accordance with the PHA's policy and procedures, (2) refers the appeal to an independent mediator or arbitrator, or (3) files suit in a court of competent jurisdiction. Such appeal must be made within (30 unless otherwise indicated) days after receipt of the Contracting Officer's decision.
- (f) The Contractor shall proceed diligently with performance of this contract, pending final resolution of any request for relief, claim, appeal, or action arising under or relating to the contract, and comply with any decision of the Contracting Officer.

32. Default

(a) If the Contractor refuses or fails to prosecute the work, or any separable part thereof, with the diligence that will insure its completion within the time specified in this contract, or any extension thereof, or fails to complete said work within this time, the Contracting Officer may, by written notice to the Contractor, terminate the right to proceed with the work (or separable part of the work) that has been delayed. In this event, the PHA may take over the work and complete it, by contract or otherwise, and may take possession of and use any materials, equipment, and plant on the work site necessary for completing the work. The Contractor and its sureties shall be liable for any damage to the PHA resulting from the Convenience Contractor's refusal or failure to complete the work within the specified time, whether or not the Contractor's right to proceed with the work is terminated. This liability includes any increased costs incurred by the PHA in completing

- (b) The Contractor's right to proceed shall not be terminated or the Contractor charged with damages under this clause if—
- (1) The delay in completing the work arises from unforeseeable causes beyond the control and without the fault or negligence of the Contractor. Examples of such causes include (i) acts of God, or of the public enemy, (ii) acts of the PHA or other governmental entity in either its sovereign or contractual capacity, (iii) acts of another contractor in the performance of a contract with the PHA, (iv) fires, (v) floods, (vi) epidemics, (vii) quarantine restrictions, (viii) strikes, (ix) freight embargoes, (x) unusually severe weather, or (xi) delays of subcontractors or suppliers at any tier arising from unforeseeable causes beyond the control and without the fault or negligence of both the Contractor and the subcontractors or suppliers; and
- (2) The Contractor, within days (10 days unless otherwise indicated) from the beginning of such delay (unless extended by the Contracting Officer) notifies the Contracting Officer in writing of the causes of delay. The Contracting Officer shall ascertain the facts and the extent of the delay. If, in the judgment of the Contracting Officer, the findings of fact warrant such action, time for completing the work shall be extended by written modification to the contract. The findings of the Contracting Officer shall be reduced to a written decision which shall be subject to the provisions of the Disputes clause of this contract.
- (c) If, after termination of the Contractor's right to proceed, it is determined that the Contractor was not in default, or that the delay was excusable, the rights and obligations of the parties will be the same as if the termination had been for convenience of the PHA.

33. Liquidated Damages

- (a) If the Contractor fails to complete the work within the time specified in the contract, or any extension, as specified in the clause entitled Default of this contract, the Contractor shall pay to the PHA as liquidated damages, the sum of \$ _____ Contracting Officer insert amount] for each day of delay. If different completion dates are specified in the contract for separate parts or stages of the work, the amount of liquidated damages shall be assessed on those parts or stages which are delayed. To the extent that the Contractor's delay or nonperformance is excused under another clause in this contract, liquidated damages shall not be due the PHA. The Contractor remains liable for damages caused other than by delay.
- (b) If the PHA terminates the Contractor's right to proceed, the resulting damage will consist of liquidated damages until such reasonable time as may be required for final

- completion of the work together with any increased costs occasioned the PHA in completing the work.
- (c) If the PHA does not terminate the Contractor's right to proceed, the resulting damage will consist of liquidated damages until the work is completed or accepted.

34. Termination for

- (a) The Contracting Officer may terminate this contract in whole, or in part, whenever the Contracting Officer determines that such termination is in the best interest of the PHA. Any such termination shall be effected by delivery to the Contractor of a Notice of Termination specifying the extent to which the performance of the work under the contract is terminated, and the date upon which such termination becomes effective.
- (b) If the performance of the work is terminated, either in whole or in part, the PHA shall be liable to the Contractor for reasonable and proper costs resulting from such termination upon the receipt by the PHA of a properly presented claim setting out in detail: (1) the total cost of the work performed to date of termination less the total amount of contract payments made to the Contractor; (2) the cost (including reasonable profit) of settling and paying claims under subcontracts and material orders for work performed and materials and supplies delivered to the site, payment for which has not been made by the PHA to the Contractor or by the Contractor to the subcontractor or supplier; (3) the cost of preserving and protecting the work already performed until the PHA or assignee takes possession thereof or assumes responsibility therefore; (4) the actual or estimated cost of legal and accounting services reasonably necessary to prepare and present the termination claim to the PHA; and (5) an amount constituting a reasonable profit on the value of the work performed by the Contractor.
- (c) The Contracting Officer will act on the Contractor's claim within days (60 days unless otherwise indicated) of receipt of the Contractor's claim.
- (d) Any disputes with regard to this clause are expressly made subject to the provisions of the Disputes clause of this contract.

35. Assignment of Contract

The Contractor shall not assign or transfer any interest in this contract; except that claims for monies due or to become due from the PHA under the contract may be assigned to a bank, trust company, or other financial institution. Such assignments of claims shall only be made with the written concurrence of the Contracting Officer. If the Contractor is a partnership, this contract shall inure to the benefit of the surviving or remaining member(s) of such partnership as approved by the Contracting Officer.

36. Insurance

- (a) Before commencing work, the Contractor and each subcontractor shall furnish the PHA with certificates of insurance showing the following insurance is in force and will insure all operations under the Contract:
 - (1) Workers' Compensation, in accordance with state or Territorial Workers' Compensation laws.
 - (2) Commercial General Liability with a combined single limit for bodily injury and property damage of not less than \$ _____ [Contracting Officer insert amount]

- per occurrence to protect the Contractor and each subcontractor against claims for bodily injury or death and damage to the property of others. This shall cover the use of all equipment, hoists, and vehicles on the site(s) not covered by Automobile Liability under (3) below. If the Contractor has a "claims made" policy, then the following additional requirements apply: the policy must provide a "retroactive date" which must be on or before the execution date of the Contract; and the extended reporting period may not be less than five years following the completion date of the Contract.
- (3) Automobile Liability on owned and non -owned motor vehicles used on the site(s) or in connection therewith for a combined single limit for bodily injury and property damage of not less than \$
- [Contracting Officer insert amount] per occurrence.

 (b) Before commencing work, the Contractor shall furnish the PHA with a certificate of insurance evidencing that Builder's Risk (fire and extended coverage) Insurance on all work in place and/or materials stored at the building site(s), including foundations and building equipment, is in force. The Builder's Risk Insurance shall be for the benefit of the Contractor and the PHA as their interests may appear and each shall be named in the policy or policies as an insured. The Contractor in installing equipment supplied by the PHA shall carry insurance on such equipment from the time the Contract work is accepted by the PHA. The Builder's Risk Insurance need not be carried on excavations, piers, footings, or foundations until such time as work on the superstructure is started. It
 - need not be carried on landscape work. Policies shall furnish coverage at all times for the full cash value of all completed construction, as well as materials in place and/or stored at the site(s), whether or not partial payment has been made by the PHA. The Contractor may terminate this insurance on buildings as of the date taken over for occupancy by the PHA. The Contractor is not required to carry Builder's Risk Insurance for modernization work which does not involve structural alterations or additions and where the PHA's existing fire and extended coverage policy can be endorsed to include such work.
- (c) All insurance shall be carried with companies which are financially responsible and admitted to do business in the State in which the project is located. If any such insurance is due to expire during the construction period, the Contractor (including subcontractors, as applicable) shall not permit the coverage to lapse and shall furnish evidence of coverage to the Contracting Officer. All certificates of insurance, as evidence of coverage, shall provide that no coverage may be canceled or nonrenewed by the insurance company until at least 30 days prior written notice has been given to the Contracting Officer.

37. Subcontracts

- (a) Definitions. As used in this contract -
 - (1) "Subcontract" means any contract, purchase order, or other purchase agreement, including modifications and change orders to the foregoing, entered into by a subcontractor to furnish supplies, materials, equipment, and services for the performance of the prime contract or a subcontract.

- (2) "Subcontractor" means any supplier, vendor, or firm that furnishes supplies, materials, equipment, or services to or for the Contractor or another subcontractor.
- (b) The Contractor shall not enter into any subcontract with any subcontractor who has been temporarily denied participation in a HUD program or who has been suspended or debarred from participating in contracting programs by any agency of the United States Government or of the state in which the work under this contract is to be performed.
- (c) The Contractor shall be as fully responsible for the acts or omissions of its subcontractors, and of persons either directly or indirectly employed by them as for the acts or omissions of persons directly employed by the Contractor.
- (d) The Contractor shall insert appropriate clauses in all subcontracts to bind subcontractors to the terms and conditions of this contract insofar as they are applicable to the work of subcontractors.
- (e) Nothing contained in this contract shall create any contractual relationship between any subcontractor and the PHA or between the subcontractor and HUD.

38. Subcontracting with Small and Minority Firms, Women's Business Enterprise, and Labor Surplus Area Firms

The Contractor shall take the following steps to ensure that, whenever possible, subcontracts are awarded to small business firms, minority firms, women's business enterprises, and labor surplus area firms:

- (a) Placing qualified small and minority businesses and women's business enterprises on solicitation lists;
- (b) Ensuring that small and minority businesses and women's business enterprises are solicited whenever they are potential sources;
- (c) Dividing total requirements, when economically feasible, into smaller tasks or quantities to permit maximum participation by small and minority businesses and women's business enterprises:
- (d) Establishing delivery schedules, where the requirements of the contract permit, which encourage participation by small and minority businesses and women's business enterprises; and
- (e) Using the services and assistance of the U.S. Small Business Administration, the Minority Business Development Agency of the U.S. Department of Commerce, and State and local governmental small business agencies.

39. Equal Employment Opportunity

During the performance of this contract, the Contractor/ Seller agrees as follows:

- (a) The Contractor/Seller shall not discriminate against any employee or applicant for employment because of of race color, religion, sex, sexual orientation, gender identity, disability, or national origin.
- (b) The Contractor/Seller shall take affirmative action to ensure that applicants are employed, and that employees are treated during employment without regard to their race, color, religion, sex, sexual orientation, gender identity, disability, or national origin. Such action shall include, but not be limited to, (1) employment, (2) upgrading demotion, (4) transfer, (5) recruitment or recruitment advertising, (6) layoff or termination, (7) rates of pay or other forms of compensation, and (8) selection for training,including apprenticeship

- (c) The Contractor/Seller agrees to post in conspicuous places available to employees and applicants for employment the notices to be provided by the Contracting Officer setting forth the provisions of this nondiscrimination clause.
- (d) The Contractor/Seller shall, in all solicitations or advertisements for employees placed by or on behalf of the Contractor/Seller, state that all qualified applicants will receive consideration for employment without regard to race, color, religion, sex, or national origin.
- (e) The Contractor/Seller shall send, to each labor union or representative of workers with which it has a collective bargaining agreement or other contract or understanding, the notice to be provided by the Contracting Officer advising the labor union or workers' representative of the Contractor's commitments under this clause, and post copies of the notice in conspicuous places available to employees and applicants for employment.
- (f) The Contractor/Seller shall comply with Executive Order 11246, as amended, and the rules, regulations, and orders of the Secretary of Labor.
- (g) The Contractor/Seller shall furnish all information and reports required by Executive Order 11246, as amended, Section 503 of the Rehabilitation Act of 1973, as amended, and by rules, regulations, and orders of the Secretary of Labor, or pursuant thereto. The Contractor/Seller shall permit
 - access to its books, records, and accounts by the Secretary of Labor for purposes of investigation to ascertain compliance with such rules, regulations, and orders.
- (h) In the event of a that the Contractor/Seller is in noncompliance with the nondiscrimination clauses of this contract or with any of such rules, regulations, or orders, this contract may be canceled, terminated or suspended in whole or in part and the contractor/seller may be declared ineligible for further Government contracts in accordance with procedures authorized in Executive Order 11246 of September 24, 1965, and such other sanctions may be imposed and remedies invoked as provided in Executive Order 11246 of September 24, 1965, or by rule, regulation, or order of the Secretary of Labor, or as otherwise provided by law.
- (i)The contractor/seller will include the provisions of paragraphs (a) through (h) in every subcontract or purchase order unless exempted by rules, regulations, or orders of the Secretary of Labor issued pursuant to section 204 of Executive Order 11246 of September 24, 1965, so that such provisions will be binding upon each sub[contractor/seller] or vendor. The [contractor/seller] will take such action with respect to any subcontract or purchase order as may be directed by the Secretary of Labor as a means of enforcing such provisions in cluding sanctions for noncompliance: Provided, however, that in the event the [contractor/seller] becomes involved in, or is threatened with, litigation with a subcontractor or vendor as a result of such direction, the [contractor/seller] may request the United States to enter into such litigation to protect the interests of the United States.
- (j) Compliance with the requirements of this clause shall be to the maximum extent consistent with, but not in derogation of, compliance with section 7(b) of the Indian Self-Determination and Education Assistance Act and the Indian Preference clause of this contract.
- Employment, Training, and Contracting Opportunities for Low-Income Persons, Section 3 of the Housing and Urban Development Act of 1968.

- (a) The work to be performed under this contract is subject to the requirements of Section 3 of the Housing and Urban Development Act of 1968, as amended, 12 U.S.C. 1701u (section 3). The purpose of section 3 is to ensure that employment and other economic opportunities generated by HUD assistance or HUD-assisted projects covered by Section 3, shall, to the greatest extent feasible, be directed to low- and very low-income persons, particularly persons who are recipients of HUD assistance for housing.
- (b) The parties to this contract agree to comply with HUD's regulations in 24 CFR Part 75, which implement Section 3. As evidenced by their execution of this contract, the parties to this contract certify that they are under no contractual or other impediment that would prevent them from complying with the Part 75 regulations.
- (c) The contractor agrees to send to each labor organization or representative of workers with which the contractor has a collective bargaining agreement or other understanding, if any, a notice advising the labor organization or workers' representative of the contractor's commitments under this section 3 clause and will post copies of the notice in conspicuous places at the work site where both employees and applicants for training and employment positions can see the notice. The notice shall describe the Section 3 prioritization requirements and shall state the minimum percentages of labor hour requirements established in the Benchmark Notice (FR-6085-N-04).
- (d) The contractor agrees to include this section 3 clause in every subcontract subject to compliance with regulations in 24 CFR Part 75, and agrees to take appropriate action, as provided in an applicable provision of the subcontract or in this section 3 clause, upon a finding that the subcontractor is in violation of the regulations in 24 CFR Part 75. The contractor will not subcontract with any subcontractor where the contractor has notice or knowledge that the subcontractor has been found in violation of the regulations in 24 CFR Part 75.
- (e) Noncompliance with HUD's regulations in 24 CFR Part 75 may result in sanctions, termination of this contract for default, and debarment or suspension from future HUD assisted contracts.
- (f) Contracts, subcontracts, grants, or subgrants subject to Section 7(b) of the Indian Self-Determination and Education Assistance Act (25 U.S.C. 5307(b)) or subject to tribal preference requirements as authorized under 101(k) of the Native American Housing Assistance and Self-Determination Act (25 U.S.C. 4111(k)) must provide preferences in employment, training, and business opportunities to Indians and Indian organizations, and are therefore not subject to the requirements of 24 CFR Part 75.

41. Interest of Members of Congress

No member of or delegate to the Congress of the United States of America shall be admitted to any share or part of this contract or to any benefit that may arise therefrom.

42. Interest of Members, Officers, or Employees and Former Members, Officers, or Employees

No member, officer, or employee of the PHA, no member of the governing body of the locality in which the project is situated, no member of the governing body of the locality in which the PHA was activated, and no other public official of such locality or localities who exercises any functions or responsibilities with respect to the project, shall, during his or her tenure, or for one year thereafter, have any interest, direct or indirect, in this contract or the proceeds thereof.

43. Limitations on Payments made to Influence Certain Federal Financial Transactions

- (a) The Contractor agrees to comply with Section 1352 of Title 31, United States Code which prohibits the use of Acts Federal appropriated funds to pay any person for influencing or attempting to influence an officer or employee of any agency, a Member of Congress, and officer or employee of Congress, or an employee of a Member of Congress in connection with any of the following covered Federal actions: the awarding of any Federal contract; the making of any Federal grant; the making of any Federal loan; the entering into of any cooperative agreement; or the modification of any Federal contract, grant, loan, or cooperative agreement.
- (b) The Contractor further agrees to comply with the requirement of the Act to furnish a disclosure (OMB Standard Form LLL, Disclosure of Lobbying Activities) if any funds other than Federal appropriated funds (including profit or fee received under a covered Federal transaction) have been paid, or will be paid, to any person for influencing or attempting to influence an officer or employee of any agency, a Member of Congress, an officer or employee of Congress, or an employee of a Member of Congress in connection with a Federal contract, grant, loan, or cooperative agreement.

44. Royalties and Patents

The Contractor shall pay all royalties and license fees. It shall defend all suits or claims for infringement of any patent rights and shall save the PHA harmless from loss on account thereof; except that the PHA shall be responsible for all such loss when a particular design, process or the product of a particular manufacturer or manufacturers is specified and the Contractor has no reason to believe that the specified design, process, or product is an infringement. If, however, the Contractor has reason to believe that any design, process or product specified is an infringement of a patent, the Contractor shall promptly notify the Contracting Officer. Failure to give such notice shall make the Contractor responsible for resultant loss.

45. Examination and Retention of Contractor's Records

- (a) The PHA, HUD, or Comptroller General of the United States, or any of their duly authorized representatives shall, until 3 years after final payment under this contract, have access to and the right to examine any of the Contractor's directly pertinent books, documents, papers, or other records involving transactions related to this contract for the purpose of making audit, examination, excerpts, and transcriptions.
- (b) The Contractor agrees to include in first-tier subcontracts under this contract a clause substantially the same as paragraph (a) above. "Subcontract," as used in this clause, excludes purchase orders not exceeding \$10,000.
- (c) The periods of access and examination in paragraphs (a) and (b) above for records relating to (1) appeals under the Disputes clause of this contract, (2) litigation or settlement of claims arising from the performance of this contract, or (3) costs and expenses of this contract to which the PHA, HUD, or Comptroller General or any of their duly authorized representatives has taken exception shall continue until disposition of such appeals, litigation, claims, or exceptions.

46. Labor Standards - Davis-Bacon and Related

If the total amount of this contract exceeds \$2,000, the Federal labor standards set forth in the clause below shall apply to the development or construction work to be performed under the contract.

(a) Minimum Wages.

(1) All laborers and mechanics employed under this contract in the development or construction of the project(s) involved will be paid unconditionally and not less often than once a week, and without subsequent deduction or rebate on any account (except such payroll deductions as are permitted by regulations issued by the Secretary of Labor under the Copeland Act (29 CFR Part 3)), the full amount of wages and bona fide fringe benefits (or cash equivalents thereof) due at time of payment computed at rates not less than those contained in the wage determination of the Secretary of Labor which is attached hereto and made a part hereof, regardless of any contractual relationship which may be alleged to exist between the Contractor and such laborers and mechanics. Contributions made or costs reasonably anticipated for bona fide fringe benefits under Section 1(b)(2) of the Davis-Bacon Act on behalf of laborers or mechanics are considered wages paid to such laborers or mechanics, subject to the provisions of 29 CFR 5.5(a)(1)(iv); also, regular contributions made or costs incurred for more than a weekly period (but not less often than quarterly) under plans, funds, or programs which cover the regular weekly period, are deemed to be constructively made or incurred during such weekly period. Such laborers and mechanics shall be paid the appropriate wage rate and fringe benefits in the wage determination for the classification of work actually performed, without regard to skill, except as provided in 29 CFR 5.5(a)(4). Laborers or mechanics performing work in morè than one classification may be compensated at the rate specified for each classification for the time actually worked therein; provided, that the employer's payroll records accurately set forth the time spent in each classification in which work is performed. The wage determination (including any additional classification and wage rates conformed under 29 CFR 5.5(a)(1)(ii) and the Davis-Bacon poster (WH-1321) shall

be posted at all times by the Contractor and its subcontractors at the site of the work in a prominent and accessible place where it can be easily seen by the workers.

- (2) (i) Any class of laborers or mechanics, including helpers, which is not listed in the wage determination and which is to be employed under the contract shall be classified in conformance with the wage determination. HUD shall approve an additional classification and wage rate and fringe benefits therefor only when all the following criteria have been met: (A) The work to be performed by the classification requested is not performed by a classification in the wage determination; and (B) The classification is utilized in the area by the construction industry; and (C) The proposed wage rate, including any bona fide fringe benefits, bears a reasonable relationship to the wage rates contained in the wage determination.
- (ii) If the Contractor and the laborers and mechanics to be employed in the classification (if known), or their representatives, and HUD or its designee agree on the classification and wage rate (including the amount designated for fringe benefits where appropriate), a report of the action taken shall be sent by HUD or its designee to the Administrator of the Wage and Hour Division, Employee Standards Administration, U.S. Department of Labor, Washington, DC 20210. The Administrator, or an authorized representative, will approve, modify, or disapprove every additional classification action within 30 days of receipt and so advise HUD or its designee or will notify HUD or its designee within the 30-day period that additional time is necessary.
- In the event the Contractor, the laborers or (iii) mechanics to be employed in the classification or their representatives, and HUD or its designee do not agree on the proposed classification and wage rate (including the amount designated for fringe benefits, where appropriate), HUD or its designee shall refer the questions, including the views of all interested parties and the recommendation of HUD or its designee, to the Administrator of the Wage and Hour Division for determination. The Administrator, or an authorized representative, will issue a determination within 30 days of receipt and so advise HUD or its designee or will notify HUD or its designee within the 30-day period that additional time is necessary.
- (iv) The wage rate (including fringe benefits where appropriate) determined pursuant to subparagraphs (a)(2)(ii) or (iii) of this clause shall be paid to all workers performing work in the classification under this contract from the first day on which work is performed in classification.
 - (3) Whenever the minimum wage rate prescribed in the contract for a class of laborers or mechanics includes a fringe benefit which is not expressed as an hourly rate, the Contractor shall either pay the benefit as stated in the wage determination or shall pay another bona fide fringe benefit or an hourly cash equivalent thereof.
 - (4) If the Contractor does not make payments to a trustee or other third person, the Contractor may consider as part of the wages of any laborer or mechanic the

- amount of any costs reasonably anticipated in providing bona fide fringe benefits under a plan or program; provided, that the Secretary of Labor has found, upon the written request of the Contractor, that the applicable standards of the Davis-Bacon Act have been met. The Secretary of Labor may require the Contractor to set aside in a separate account assets for the meeting of obligations under the plan or program.
- (b) Withholding of funds. HUD or its designee shall, upon its own action or upon written request of an authorized representative of the Department of Labor, withhold or cause to be withheld from the Contractor under this contract or any other Federal contract with the same prime Contractor, or any other Federally-assisted contract subject to Davis-Bacon prevailing wage requirements, which is held by the same prime Contractor, so much of the accrued payments or advances as may be considered necessary to pay laborers and mechanics, including apprentices, trainees, and helpers, employed by the Contractor or any subcontractor the full amount of wages required by the contract. In the event of failure to pay any laborer or mechanic, including any apprentice, trainee, or helper, employed or working in the construction or development of the project, all or part of the wages required by the contract, HUD or its designee may, after written notice to the Contractor, take such action as may be necessary to cause the suspension of any further payment, advance, or quarantee of funds until such violations have ceased. HUD or its designee may, after written notice to the Contractor, disburse such amounts withheld for and on account of the Contractor or subcontractor to the respective employees to whom they are due.
- (c) Payrolls and basic records.
 - (1) Payrolls and basic records relating thereto shall be maintained by the Contractor during the course of the work and preserved for a period of three years thereafter for all laborers and mechanics working in the construction or development of the project. Such records shall contain the name, address, and social security number of each such worker, his or her correct classification, hourly rates of wages paid (including rates of contributions or costs anticipated for bona fide fringe benefits or cash equivalents thereof of the types described in section 1(b)(2)(B) of the Davis-Bacon Act), daily and weekly number of hours worked, deductions made, and actual wages paid. Whenever the Secretary of Labor has found, under 29 CFR 5.5(a)(1)(iv), that the wages of any laborer or mechanic include the amount of costs reasonably anticipated in providing benefits under a plan or program described in section 1(b)(2)(B) of the Davis-Bacon Act, the Contractor shall maintain records which show that the commitment to provide such benefits is enforceable, that the plan or program is financially responsible, and that the plan or program has been communicated in writing to the laborers or mechanics affected, and records which show the costs anticipated or the actual cost incurred in providing such benefits. Contractors employing apprentices or trainees under approved programs shall maintain written evidence of the registration of apprenticeship programs and certification of trainee programs, the registration of the apprentices and trainees, and the ratios and wage rates prescribed in the applicable programs.

- (2) (i) The Contractor shall submit weekly for each week in which any contract work is performed a copy of all payrolls to the Contracting Officer for transmission to HUD or its designee. The payrolls submitted shall set out accurately and completely all of the information required to be maintained under subparagraph (c)(1) of this clause. This information may be submitted in any form desired. Optional Form WH-347 (Federal Stock Number 029-005-00014-1) is available for this purpose and may be purchased from the Superintendent of Documents, U.S. Government Printing Office, Washington, D.C. 20402. The Contractor is responsible for the submission of copies of payrolls by all subcontractors. (Approved by the Office of Management and Budget under OMB Control Number 1214-0149.)
 - (ii) Each payroll submitted shall be accompanied by a "Statement of Compliance," signed by the Contractor or subcontractor or his or her agent who pays or supervises the payment of the persons employed under the contract and shall certify the following:
- certify the following:

 (A) That the payroll for the payroll period contains the information required to be maintained under paragraph (c) (1) of this clause and that such information is correct and complete;
- (B) That each laborer or mechanic (including each helper, apprentice, and trainee) employed on the contract during the payroll period has been paid the full weekly wages earned, without rebate, either directly or indirectly, and that no deductions have been made either directly or indirectly from the full wages earned, other than permissible deductions as set forth in 29 CFR Part 3; and
- (C) That each laborer or mechanic has been paid not less than the applicable wage rates and fringe benefits or cash equivalents for the classification of work performed, as specified in the applicable wage determination incorporated into the contract.
 - (iii) The weekly submission of a properly executed certification set forth on the reverse side of Optional Form WH-347 shall satisfy the requirements for submission of the "Statement of Compliance" required by subparagraph (c)(2)(ii) of this clause
 - (iv) The falsification of any of the above certifications may subject the Contractor or subcontractor to civil or criminal prosecution under Section 1001 of Title 18 and Section 3729 of Title 31 of the United States Code.
 - (3) The Contractor or subcontractor shall make the records required under subparagraph (c)(1) available for inspection, copying, or transcription by authorized representatives of HUD or its designee, the Contracting Officer, or the Department of Labor and shall permit such representatives to interview employees during working hours on the job. If the Contractor or subcontractor fails to submit the required records or to make them available, HUD or its designee may, after written notice to the Contractor, take such action as may be necessary to cause the suspension of any further payment, advance, or guarantee of funds. Furthermore, failure to submit the required records upon request or to

- make such records available may be grounds for debarment action pursuant to 29 CFR 5.12.
- (d) (1) Apprentices. Apprentices will be permitted to work at less than the predetermined rate for the work they performed when they are employed pursuant to and individually registered in a bona fide apprenticeship program registered with the U.S. Department of Labor, Employment and Training Administration, Office of Apprenticeship and Training, Employer and Labor Services (OATELS), or with a State Apprenticeship Agency recognized by OATELS, or if a person is employed in his or her first 90 days of probationary employment as an apprentice in such an apprenticeship program, who is not individually registered in the program, but who has been certified by OATELS or a State Apprenticeship Agency (where appropriate) to be eligible for probationary employment as an apprentice. The allowable ratio of apprentices to journeymen on the job site in any craft classification shall not be greater than the ratio permitted to the Contractor as to the entire work force under the registered program. Any worker listed on a payroll at an apprentice wage rate, who is not registered or otherwise employed as stated in this paragraph, shall be paid not less than the applicable wage rate on the wage determination for the classification of work actually performed. In addition, any apprentice performing work on the job site in excess of the ratio permitted under the registered program shall be paid not less than the applicable wage rate on the wage determination for the work actually performed. Where a contractor is performing construction on a project in a locality other than that in which its program is registered, the ratios and wage rates (expressed in percentages of the journeyman's hourly rate) specified in the Contractor's or subcontractor's registered program shall be observed. Every apprentice must be paid at not less than the rate specified in the registered program for the apprentice's level of progress, expressed as a percentage of the journeyman hourly rate specified in the applicable wage determination. Apprentices shall be paid fringe benefits in accordance with the provisions of the apprenticeship program. If the apprenticeship program does not specify fringe benefits, apprentices must be paid the full amount of fringe benefits listed on the wage determination for the applicable classification. If the Administrator of the Wage and Hour Division determines that a different practice prevails for the applicable apprentice classification, fringes shall be paid in accordance with that determination. In the event OATELS, or a State Apprenticeship Agency recognized by OATELS, withdraws approval of an apprenticeship program, the Contractor will no longer be permitted to utilize apprentices at less than the applicable predetermined rate for the work performed until an acceptable
 - (2) Trainees. Except as provided in 29 CFR 5.16, trainees will not be permitted to work at less than the predetermined rate for the work performed unless they are employed pursuant to and individually registered in a program which has received prior approval, evidenced by formal certification by the U.S. Department of Labor, Employment and Training Administration. The ratio of trainees to journeymen on the job site shall not be greater than permitted under

program is approved.

the plan approved by the Employment and Training Administration. Every trainee must be paid at not less than the rate specified in the approved program for the trainee's level of progress, expressed as a percentage of the journeyman hourly rate specified in the applicable wage determination. Trainees shall be paid fringe benefits in accordance with the provisions of the trainee program. If the trainee program does not mention fringe benefits, trainees shall be paid the full amount of fringe benefits listed in the wage determination unless the Administrator of the Wage and Hour Division determines that there is an apprenticeship program associated with the corresponding journeyman wage rate in the wage determination which provides for less than full fringe benefits for apprentices. Any employee listed on the payroll at a trainee rate who is not registered and participating in a training plan approved by the Employment and Training Administration shall be paid not less than the applicable wage rate in the wage determination for the classification of work actually performed. In addition, any trainee performing work on the job site in excess of the ratio permitted under the registered program shall be paid not less than the applicable wage rate in the wage determination for the work actually performed. In the event the Employment and Training Administration withdraws approval of a training program, the Contractor will no longer be permitted to utilize trainees at less than the applicable predetermined rate for the work performed until an acceptable program is approved.

- (3) Equal employment opportunity. The utilization of apprentices, trainees, and journeymen under this clause shall be in conformity with the equal employment opportunity requirements of Executive Order 11246, as amended, and 29 CFR Part 30.
- (e) Compliance with Copeland Act requirements. The Contractor shall comply with the requirements of 29 CFR Part 3, which are hereby incorporated by reference in this contract.
- (f) Contract termination; debarment. A breach of this contract clause may be grounds for termination of the contract and for debarment as a Contractor and a subcontractor as provided in 29 CFR 5.12.
- (g) Compliance with Davis-Bacon and related Act requirements. All rulings and interpretations of the Davis-Bacon and related Acts contained in 29 CFR Parts 1, 3, and 5 are herein incorporated by reference in this contract
- (h) Disputes concerning labor standards. Disputes arising out of the labor standards provisions of this clause shall not be subject to the general disputes clause of this contract. Such disputes shall be resolved in accordance with the procedures of the Department of Labor set forth in 29 CFR Parts 5, 6, and 7. Disputes within the meaning of this clause include disputes between the Contractor (or any of its subcontractors) and the PHA, HUD, the U.S. Department of Labor, or the employees or their representatives.
- (i) Certification of eligibility.
 - (1) By entering into this contract, the Contractor certifies that neither it (nor he or she) nor any person or firm who has an interest in the Contractor's firm is a person or firm ineligible to be awarded contracts by the United States Government by virtue of section 3(a) of the Davis-Bacon Act or 29 CFR 5.12(a)(1).

- (2) No part of this contract shall be subcontracted to any person or firm ineligible for award of a United States Government contract by virtue of section 3(a) of the Davis-Bacon Act or 29 CFR 5.12(a)(1).
- (3) The penalty for making false statements is prescribed in the U. S. Criminal Code, 18 U.S.C. 1001.
- (j) Contract Work Hours and Safety Standards Act. As used in this paragraph, the terms "laborers" and "mechanics" include watchmen and guards.
 - (1) Overtime requirements. No contractor or subcontractor contracting for any part of the contract work which may require or involve the employment of laborers or mechanics, including watchmen and guards, shall require or permit any such laborer or mechanic in any workweek in which the individual is employed on such work to work in excess of 40 hours in such workweek unless such laborer or mechanic receives compensation at a rate not less than one and one-half times the basic rate of pay for all hours worked in excess of 40 hours in such workweek.
 - (2) Violation; liability for unpaid wages; liquidated damages. In the event of any violation of the provisions set forth in subparagraph (j)(1) of this clause, the Contractor and any subcontractor responsible therefor shall be liable for the unpaid wages. In addition, such Contractor and subcontractor shall be liable to the United States (in the case of work done under contract for the District of Columbia or a territory, to such District or to such territory), for liquidated damages. Such liquidated damages shall be computed with respect to each individual laborer or mechanic (including watchmen and guards) employed in violation of the provisions set forth in subparagraph (i)(1) of this clause, in the sum of \$27 for each calendar day on which such individual was required or permitted to work in excess of the standard workweek of 40 hours without payment of the overtime wages required by provisions set forth in subparagraph (j)(1) of this clause. DOL posts current fines at: https://www.dol.gov/whd/ govcontracts/cwhssa.htm#cmp
 - (3) Withholding for unpaid wages and liquidated damages. HUD or its designee shall upon its own action or upon written request of an authorized representative of the Department of Labor withhold or cause to be withheld, from any moneys payable on account of work performed by the Contractor or subcontractor under any such contract or any Federal contract with the same prime Contractor, or any other Federally-assisted contract subject to the Contract Work Hours and Safety Standards Act, which is held by the same prime Contractor, such sums as may be determined to be necessary to satisfy any liabilities of such Contractor or subcontractor for unpaid wages and liquidated damages as provided in the provisions set forth in subparagraph (j)(2) of this clause.
- (k) Subcontracts. The Contractor or subcontractor shall insert in any subcontracts all the provisions contained in this clause, and such other clauses as HUD or its designee may by appropriate instructions require, and also a clause requiring the subcontractors to include these provisions in any lower tier subcontracts. The prime Contractor shall be responsible for the compliance by any subcontractor or lower tier subcontractor with all these provisions.

47. Non-Federal Prevailing Wage Rates

- (a) Any prevailing wage rate (including basic hourly rate and any fringe benefits), determined under State or tribal law to be prevailing, with respect to any employee in any trade or position employed under the contract, is inapplicable to the contract and shall not be enforced against the Contractor or any subcontractor, with respect to employees engaged under the contract whenever such non-Federal prevailing wage rate exceeds:
- (1) The applicable wage rate determined by the Secretary of Labor pursuant to the Davis-Bacon Act (40 U.S.C. 3141 et seq.) to be prevailing in the locality with respect to such trade;
- (b) An applicable apprentice wage rate based thereon specified in an apprenticeship program registered with the U.S. Department of Labor (DOL) or a DOL-recognized State Apprenticeship Agency; or
 (c) An applicable trainee wage rate based thereon specified
- 48. Procurement of Recovered Materials.

in a DOL-certified trainee program.

(a) In accordance with Section 6002 of the Solid Waste Disposal Act, as amended by the Resource Conservation and Recovery Act, the Contractor shall procure items designated in guidelines of the Environmental Protection Agency (EPA) at 40 CFR Part 247 that contain the highest percentage of recovered materials practicable, consistent with maintaining a satisfactory level of competition. The Contractor shall procure items designated in the EPA guidelines that contain the highest percentage of recovered materials practicable unless the Contractor determines that such items: (1) are not reasonably available in a reasonable period of time; (2) fail to meet reasonable performance standards, which shall be determined on the basis of the guidelines of the National Institute of Standards and Technology, if applicable to the item; or (3) are only available at an

unreasonable price.

() Paragraph (a) of this clause shall apply to items purchased under this contract where: (1) the

Contractor purchases in excess of \$10,000 of the item under this contract; or (2) during the preceding Federal fiscal year, the Contractor: (i) purchased any amount of the items for use under a contract that was funded with Federal appropriations and was with a Federal agency or a State agency or agency of a political subdivision of a State; and (ii) purchased a total of in excess of \$10,000 of the item both under and outside that contract.

-2023 GALBRAITH CARNAHAN ARCHITECTS LLC

Z Ш G A R GHLAND

က က \sim က 2 <u>-</u> ≥ ш Ш MILWAUK Ш \supset Z Ш > ⋖ \supset ⋖ Ш Z \supset \neg ഗ Ш ≥ ∞ $\overline{}$ ∞



PUTERIOR ELEMENTON TAG LE, ELEMENTON ON SHEET AN DETALLS 1, 2, 3, 4 DETALTAG LE. DETAL 1 ON SHE **\(\)** ⋖ 38.6 SYMBOL KEY • 1

District to the property of th

OWNER:
HIGHLAND PARK COMMUNITY LLC
809 N. BROADWAY
MILWAUKEE, WI 53202 CONTACT INFORMATION

Torgue And Groove Telephore Top Of Top Of Concorde Top Of Concorde Top Of Steel Vryf Bass Vryf Cenposto Verty J. Daki Vrsen Pand Vryf Well Cover NOT FOR CONSTRUCTION

SYMBOL KEY

OWNER:
HIGHLAND PARK COMMUNITY LLC
809 N. BROADWAY
MILWAUKEE, WI 53202 CONTACT INFORMATION

o 2023 GALBRAITH CARNAHAN ARCHITECTS LLC

S Z Ш G A R GHLAND

က က $^{\circ}$ က 2 <u>-</u> ≥ . Ш Ш MILWAUK Ш \supset Z Ш > ⋖ \supset ⋖ Ш N - \vdash ഗ Ш ≥ ∞ $\overline{}$

 ∞



<u>_</u>	910	8100	3101	20102	M010	A103	Attal	A102	A111	7933	100	V333						
	DOOR NUMBER - SEE DOOR SCHEDULE.		MANDOW I THE SAM WANDOW SUBSTILLS.				NEW DOCK - SEE DOOR SCHEDULE		a magazza abod ansi- abod bettera			LE LEWING BLEWTON TAG		del library			T.D. FOUND SPOT ELEVATION MARKER	
	COLUMN GRID LINE: SEE STRUCTURAL DROWNINGS.	(9	ROOM NAME & NUMBER	_		ELEVATION TAG		No. of Contractions		-∢	DRAWING TILE 44400)2	>	4	WALL-THO)	reterring	•
((A))	ROOM	NAME		•						ATIO DOWNING NAME)		n		284	

1	1	1				0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	14 Process Version
1	1	1	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		0 4	O O O O O O O O O O O O O O O O O O O	inter- O COMP O COMP Interest to the comp
Comparison	Column	Compared	33555555555555555555555555555555555555			00 00 00 00 00 00 00 00 00 00 00 00 00	Variation Variat
Company Comp	Comparison Com		Thorne 0000			O SO	Of Carry Of
The control of the	Commonweight Comm	1	11 to 2 to			O S S S S S S S S S S S S S S S S S S S	Of Control
		The control of the co	There is a second secon			PERMIT PROPERTY AND STATE OF S	Of Comba
100 100	1996 1996	Comparison Com	100 100			P C C C C C C C C C C C C C C C C C C C	of scenario Opera
100 100	Comparison Com		136 pt 2 pt			ON- ON- ON- ON- ON- ON- ON- ON- ON- ON-	Operate into 3 Operate into 3 Operate into 4 Operate into 4 Operat
Column	Comparison	1	1500MW 15			OPAG PECT PECT PECT PECT PECT PECT PECT PECT	Operation Operat
1		1	I Baselor 0000 000 000 000 000 000 000 000 000			P CC PCC PCUMS PCU	Concession of the Concession o
The control of the	Compared	1	18 seeker 00 00 00 00 00 00 00 00 00 00 00 00 00			PCC PCUAS PCUAS PCUAS PCUAS PCCAS PC	Fire-fast Controls Fire-fast Control Fire-fast Controls Fire-fast Control Fire-fast Co
The control of the	10 10 10 10 10 10 10 10	1	100 100	1		PCC PCUMB PCUMB PCCMM PCCMM PCCMM PCCMM	Pre-Clast Concess Harting Harting Harting Harting Harting Planca
1	Compared by Comp		Best tapmer Bayler DVI	1		PLUMS PLUMS PLUMS PRESH	Particle Controls Particle Controls Particle Controls Particle Controls Particle Controls Particle Controls Particle Control
10 10 10 10 10 10 10 10	10 10 10 10 10 10 10 10					PLUMB PLUMB PLUMB PMETRY PMETRY	Horstein Place Commande Place Commande Place Sind on 1 Shelf Profession Travior Place of Place of Place of Place of
1	10 10 10 10 10 10 10 10					PLUMB PLUMB PREZH PREZH	Floor Carnavia Plantesy Plantesy Plantesy Pratesian Pratesian Pratesian Plantesy Pla
100 100	100 100	The state of the s		l speed		PLUMB PLUMB PRESH PT T	Electricity Fluctors Fluctors Professors Professor
1	1	1	Statement EFS			PRESH PRESH	Phenosa Phenosa Preference Pressure Province Previous Province Previous Province
	10 10 10 10 10 10 10 10		September ETS	1	-	PRESH N	Therease Publishers Shelves Shelf Publishers Present Processe Phant or Pannishe
100 100	10 10 10 10 10 10 10 10		South Electron South Electron South Electron Control	1	· [.	PHETR	Professional State Professional Process Proces
Column	1	The state of the s	Education (1987)		ļ	PT T	Prefessival Pressure Freehol Particle Particle Public of Particle Public of Particle
Column		The control of the co	Subsect Connect CE E E E E E E E E E E E E E E E E E E			E	Pressure Treated Plant or Plantsid Pstydrad Chlassia
Company	Company	The state of the s	Betham Clarker Betham				Paint or Painted Polyotini Chladde
Col. State Col.	Compared		Bothom ELLIV EPOM CONTRACT ETR			2	Potveryl Chloride
Compared	1	The state of the s	EP0M ED ETR		Ī	PAC	
10 Complete Comp	Compared	The state of the s	EB EB		Ī		
1		The state of the s	88		•	٥	
Control Cont		The state of the s			_		
Column C	Comparison Com	Control of the contro	FXRT	che	VI. ALL	2	Appeal page
Complete	Col.	The state of the s	000	The second secon		2000	Collected Collect Ches
1		1 200 0 10 10 10 10 10 10 10 10 10 10 10 10	The Factor of Street			20	Company Company
Company	1	Comment	13.0		TOTAL	2	ross uran
Company Comp	Column C	1	The same of the sa			ų.	HORIZON TO THE PROPERTY OF THE
1	10 10 10 10 10 10 10 10	900 100 100 100 100 100 100 100				400	Marfacod Margaret
1	1.0 1.0					OCID	Coquired
14	Library Company Compan	THOUSE STATE	Canstructor Nanager FD FI	or Drain or Fire Department	Maximum	Row	Room
1.5 1.5	1.5 1.5	1 6 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	Carenda Mascrry Unit				
10 10 10 10 10 10 10 10	10 10 10 10 10 10 10 10	A M	201	or Cabbrell		œ	
15 15 15 15 15 15 15 15	1.5 1.5	2	FINER			SHIPPE.	Sparties
15 15 15 15 15 15 15 15	15 15 15 15 15 15 15 15	25 H O H E	1001			35	Share Elvotor
150 Heaving #1 150	LONG STATE OF STATE O	648.47 64	83			-10	Study Study
12 Chicago de Caracterio de Ca	100 100	29± E	5690			Control of	Provided City Passell and as
1,000 1,00	1.1.1.25(1) 1.1.1.25 1.1.1.	918	88	_		armin and a	Charles on Constant
100 100	17 1900 19	110	S				change or opposed
Opening Table Fill Fooding LEGING Moderate Measure Opposite Male Communication Communication Communication Fooding Communication Communication Fooding Communication Communication File Moderate Communication Communication File Communication Communication File Communication File Communication Communication File Commun	Fig. 5-config. Fig. 5-config. Historic Modine-Newton St. 5-config. Historic Modine-Newto	Oavand Tile FTG		_		0	CALL PRINCIPAL CARRIE
Commercial and Pictor Proceedings IITR Model STC Company Of Company IITR Model STC Company	Common Man STG Countries No. 11 Man STG Countries Countries No. 11 Man STG Countries Countries STG C	011	Camping Tale		ě		vine.
Company IVP MAN ST. Conjunt Conjunt Conjunt	Country Market STL STREET STREET STREET	Comment of the same	Common Table			2000	Sound Intramination Coefficient
With Manual SHOCS	WINC. Material S 190051	Comment	Comment			1	2000
				MIN		SIMOS	attractive of attractive

UITE 1 RUWATOSA, WI 53213 114)-212-5006

ROOM NAME RANNER	SEOUTEN 740	2004/WO TTE	
		TO COMMO NAME	
of Common State of Common Stat	Understelle Ving Base Ving County Vinder II sale Vinder Vinder Vinder Vinder Vinder Vinder		
Top Of Concerts Top Of Steel To		William Willia	
28882 ± 38	8 > 1000 4 4 5 M	MOM MAN	
O C Ch Control CDC Ch Ch Control CDC Ch Ch Control CDC Ch		The state of the s	ST Stah STC Sound Transmission Coefficient STL Steel STRUCT Stupping or Structural
COTE Openar VM Board The Copenar VM Board The Make Cope The Make Make Make Make Make Make Make Mak	Help Photi Hosti Hosti Hosti Photi Hosti Photi Photi Construit Hosti Photi Charles Hosti Charles Hos	3201 3201 3201 3201 3201 3201 3201 3201	Modern Reistan Oppsan Well Board Model Moter
HE HOW	HATE COOR WELL	M M M M M M M M M M M M M M M M M M M	MRGM8 MTR MTR
Discussion Chromotode Discussion Discussion Chrosen Discussion Discussion Chrosen Chrosen Chrosen Chrosen Chrosen Chrosen Chromotode			Flooring Foundation
488 48888888			200
1 1	1		

NOT FOR CONSTRUCTION

HIGHLAND GARDENS





Proprietary and Confidential - Not to be issued, reproduced, or distributed without the express written permission from Travaux Inc.





Travaux Inc.

401 E. Ogden Ave

Milwaukee, WI 53202

414.209.1800

The below information is Travaux Inc.'s interpretation of the scope and is based solely on the existing documents. It is Travaux Inc's intent to include everything the D3G scope has identified. This schematic budget will be used as a guide to maintaining the project budget going forward.

- Site
 - Concrete Paver Replacement
 - The pavers are to removed and salvaged for reuse, the sand bed is to leveled, and the existing pavers are to be reinstalled
 - No sealing is included in cost
- Building Exterior
 - Exterior Double HM Doors
 - Frames are to be reused
 - Frames are not to be repainted
 - Exterior Single HM Doors
 - Frames are to be reused
 - Frames are not to be repainted
 - All windows to be replaced with a Quaker or equal
 - No exterior architectural metal work is included
 - o Roof overlay with built up system at areas without green roof.
- Equipment
 - o (1) New Fuel Tank for Generator
 - o (2) New Trash Compactors
 - o (2) 3/4HP Water Circulating Pumps
 - o (2) 1HP DHW Circulating Pumps
 - o (3) DHW Boilers
 - o (3) 4-Ton Roof-Top Units
 - o (1) 3-Ton Roof-Top Unit
 - o (2) 5-Ton Roof-Top Units
 - o (1) 10-Ton Roof-Top Unit
 - o (1) 16-Ton Roof-Top Unit
 - o Fire Control Panel & Associated Devices
 - All devices throughout the building are to be replaced along with the fire alarm control panels.
 - Key Card Panel
 - Only panel to be replaced, not card readers.
- Common Areas
 - Existing Flooring Replacement
 - No flooring replacement at ceramic tile, wood floors, or workout room flooring.
 - The assumption is that all LVT is to be overlaid on existing flooring with the exception of the carpet which will be removed.
 - Carpeting Replacement
 - All carpeting to be removed and replaced with LVT.
 - Lighting Replacement
 - All fixtures through the building to be replace or retrofit with LED fixtures
 - Exit Signs not included

- Repair of Cracking Mortar Joints in Common Bathrooms
- Remodel/Upgrade Neighborhood Network
 - Computers, furniture and equipment not included
- o Paint walls and drywall ceilings in area shown on map attached
 - Common Ceilings
 - Assuming 1 coat of primer and 2 coats of paint (sprayed and back rolled different color from existing walls).
 - Common Walls
 - Assuming 1 coat of primer and 2 coats of paint (sprayed and back rolled different color from existing walls).
 - No epoxy paints
- Units
 - Under-Counter Scald and Abrasion Protection in ADA Units
 - o VCT Replacement & Carpet Replacement with LVT
 - Some units will not have flooring replacement as it was replaced recently
 - o Bathroom Vanity Replacement with Countertop
 - Refrigerator, Range & Range Hood Replacement
 - All appliances to be black or white (not stainless)
 - Range hoods assumed to be recirculating
 - Window Replacement
 - Replacement from exterior, no interior work is assumed
 - No interior sills or drywall work assumed
 - Kitchen Counter Replacement
 - Counters to be replaced
 - New sinks are to be installed
 - Installation of Non-Skid Surface to Showers
 - Replacement of Showerheads and Faucets with Low Flow
 - Window Covering Replacement
 - Aluminum blinds are assumed
 - Replacement of T-Stats
 - Standard wired t-stat not "Nest" brand or similar
 - Lighting Replacement
 - All fixtures through the building to be replace or retrofit with LED fixtures
 - Exhaust Fan Replacement
 - Includes the supply and installation of new fan and new ducting
 - Emergency call system replacement with like product
- New Elevator Towers
 - Schindler 3300 Elevator is basis of estimate
 - o 3500 lb capacity, general purpose class A
 - o 150 FPM Speed
 - o 32 Ft. travel distance
 - o 4 stops
 - o 208 volt 60 Hz 3 phase power supply
 - Non-sprinklered Hostway
 - Two speed side opening doors
 - O Cab interior elevations 6' 9-5/16" Wide x 5' 6-7/8" Deep

- Elevator shaft overall size is assumed to be 10 by 9 outside dimension, 44 feet in height construction with 8" CMU and 8" split face on the outside.
- o 4th floor extension is assumed to use the existing precast deck as the floor of the infill
- O We have included 6 HEILCO PIERS at each tower 30' deep located under elevator pit
- o Electrical / HVAC / Plumbing cost is within the line item budget.

Infill Areas

We have included 8 eight infill areas that connect the main building to the elevator towers.

- Included is masonry demolition of exterior split face only on each side of the infill area on 8 locations.
- Wall demolition on new wall opening in the existing building so to allow access to the infills
- o Included in finish budget funds to finish out the infill areas.
- Included are eight (8) containment doors for new elevators. Did not include any of the existing elevators.

EXCLUSIONS

- Any alterations due to blower door testing results
- Replacement or painting of any doors
- Exit signs
- Any parking lot work
- New monument sign
- Green Roof replacement or modification
- Existing elevator revitalization
- Refinish Kitchen Cabinets
- Painting of residential units
- No hollow metal entrance door painting or refinishing included
- No work in existing elevators or shaft
- No hazardous materials assumed

CLARIFICATIONS

- Travaux Inc. will work with the mechanical and electrical engineers to maintain the plumbing, mechanical and electrical budgets as stands. We have established the MEPs as a not-to-exceed value.
- No Allowances at this time

D - ITEMIZED STATEMENT OF THE GUARANTEED MAXIMUM PRICE 10/26/2023

PAGE 1 OF 1 PAGES

AIA DOCUMENT G703

APPLICATION NO: APPLICATION DATE: PERIOD TO:

CONTINUATION SHEET

ALA Document G702, APPLICATION AND CERTIFICATION FOR PAYMENT containing Contractor's signed certification is attached. In tabulations below, amounts are stated of the nearest dollar. The Contract of the contract whose variety contracts whose variety benefits of the interest of the contract of the contract of the contract whose variety benefits of the contract whose variety contracts whose variety contracts.

×
7
d
ya
\geq
na
-
S
=
<u>≘</u>
•
ĭ
≔
ö
Ŧ
0
50
2
·=
#
~
<u>_e</u>
9
<u>च</u>
Ħ
5
ė
ë
-
≥
S
ಶ
g
븀
5
\circ
ö
Ι
Ξ
Ξ
2
0
\circ
60

Use Column 1	Use Column I on Contracts where variable retainage for line items may apply.	may apply.		•			•	-	ARCHI	ARCHITECT'S PROJECT NO:	
V	В	C			Q	E	F	Ü	H	I	ſ
Item Number	Description of Work	Scheduled Value	Adjustment	Adjusted Schedule Value	Work C	Work Completed	Materials Presently Stored (Not in D or E)	Total Completed and Stored to Date (D+E+F)	% (G/C)	Balance to Finish (C - G)	Retainage (if Variable Rate)
	Highland Gardens Remodel				From Previous Application	This Period					
-	Demolition	46,320.00		46,320.00						46,320.00	
2	Footings, Foundations, SOG	24,000.00		24,000.00					0.0%	24,000.00	
3	Site Utilities	00.00		0.00					0.0%		
4 (Asphalt Paving	0.00		0.00					0.0%	. 00 000	•
6	Site Favers	30,000.00		0.000					0.0%	30,000.00	
0 1	Curo and Guuers	00.00		0.00					0.0%		
× ×	Building Concrete & SOG	50.368.00		50.368.00						50.368.00	
6	Thermal Insulation and Vapor Barrier	0.00		0.00					0.0%	00:00:00	
10	Masonry	412,008.00		412,008.00					0.0%	412,008.00	
11	Expansion Joints	7,680.00		7,680.00					0.0%	7,680.00	
12	Structural Steel	00.000,09		00:000:09					%0.0	00.000,09	
13	Handrail	0.00		0.00					0.0%		
14	Rough Carpentry	0.00		00:00					%0.0		
15	Finish Carpentry-Labor	0.00		0.00					0.0%		
16	Millwork/Cabinets/Sills	239,400.00		239,400.00					0.0%	239,400.00	
17	Roofing and Flashing	974,000.00		974,000.00					0.0%	974,000.00	
18	Sealant	0.00		0.00					0.0%		
19	Metal Fabrications	2,400.00		2,400.00					0.0%	2,400.00	•
20	Apartment Doors and Frames	0.00		0.00					0.0%	•	•
21	Solid Doors, Frames and Hardware	94,000.00		94,000.00					0.0%	94,000.00	•
22	Windows	1,071,635.00		1,071,635.00					0.0%	1,071,635.00	•
23	Install Wood Window and Patior Doors	0.00		0.00					0.0%		•
24	Glass and Glazing	0.00		0.00					0.0%		
25	Drywall and Framing	9,300.00		9,300.00					0.0%	9,300.00	
26	Acoustic Ceiling	5,676.00		5,676.00					0.0%	5,676.00	
27	Carpet and Flooring	436,092.00		436,092.00					0.0%	436,092.00	•
28	Painting and Staining	154,719.00		154,719.00					0.0%	154,719.00	
29	Specialties/Wall Protection/Loilet Partitions	0.00		0.00					0.0%	00 000 110	
31	Trash Compactors	35 000 00		35,000,00					0.0%	35,000,00	
32	Window Blinds and Shades	20,800.00		20,800.00					0.0%	20,800.00	
33	Elevators	606,000.00		00.000,909						00.000,909	
34	Fire Protection System	0.00		00:0					0:0%		
35	Plumbing	430,004.00		430,004.00					0.0%	430,004.00	
36	HVAC- Mechanical	00.000,609		00:000,609					0:0%	00.000,009	
37	Electrical	1,730,000.00		1,730,000.00					0.0%	1,730,000.00	
38	Fire Protection System	0.00		0.00					0.0%	-	-
39	Travaux Contingency	423,561.00		423,561.00						423,561.00	-
	Bond/Builders Risk	95,000.00		95,000.00						95,000.00	
40	General Conditions / PPBonds	441,929.00		441,929.00					0.0%	441,929.00	•
41	Overhead	176,772.00		176,772.00					0.0%	176,772.00	•
42	CM Fees	441,929.00		441,929.00				•	0.0%	441,929.00	•
	A T WOW	00 000 000 0	000	00 503 050 0					7000	00 002 000 0	
	TOTAL	8,838,593.00	0.00	8,838,593.00	'	-	-	-	0.0%	8,838,593.00	-

8,838,593.00	176,771.86	441,929.65
\$8,838,593.00	176,771.86	441,929.65
Fotal before GC/OH/CM Fes		8
Total be	HO HO	CMFees

